# **UNITED STATES SECURITIES AND EXCHANGE COMMISSION**

WASHINGTON, DC 20549

FO	ORM 10-Q
(Mark One)	
☑ QUARTERLY REPORT PURSUANT TO SECTION 1934	ON 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
For the quarterl	y period ended June 30, 2016
	or
☐ TRANSITION REPORT PURSUANT TO SECTION 1934	ON 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
Commission	n file number: 001-36182
	encor, Inc. gistrant as Specified in its Charter)
Delaware	20-1622502
(State or Other Jurisdiction of Incorporation or Organization)	(I.R.S. Employer Identification No.)
111 West Lemon Avenue, Monrovia, CA	91016
(Address of Principal Executive Offices)	(Zip Code)
,	<b>626) 305-5900</b> one Number, Including Area Code)
	orts required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 193 rant was required to file such reports), and (2) has been subject to such filing
·	onically and posted on its corporate Web site, if any, every Interactive Data File 3-T ( $\S232.405$ of this chapter) during the preceding 12 months (or for such shorter $\boxtimes$ No $\square$
Indicate by check mark whether the registrant is a large accelerated definitions of "large accelerated filer", "accelerated filer" and "smaller rep	d filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. Se porting company" in Rule 12b-2 of the Exchange Act.
Large accelerated filer $\square$ Accelerated filer $\boxtimes$ Non-accelerated fi	iler $\square$ Smaller reporting company $\square$
Indicate by check mark whether the registrant is a shell company (a	as defined in Rule 12b-2 of the Securities Exchange Act of 1934). Yes $\Box$ No $\boxtimes$
Indicate the number of shares of each of the issuer's classes of common states.	tock, as of the latest practicable date:  Outstanding at July 28, 2016
Common stock \$0.01 par value	40 949 637

# Xencor, Inc.

# Quarterly Report on FORM 10-Q for the quarter ended June 30, 2016

# **Table of Contents**

		Page
	SPECIAL NOTE REGARDING FORWARD LOOKING STATEMENTS	3
PART I.	FINANCIAL INFORMATION	4
Item 1.	Financial Statements	4
	Balance Sheets as of June 30, 2016 (unaudited) and December 31, 2015	4
	Statements of Comprehensive Income (Loss) for the Three Months and Six Months ended June 30,	
	2016 and 2015 (unaudited)	5
	Statement of Stockholders' Equity as of June 30, 2016 (unaudited)	$\epsilon$
	Statements of Cash Flows for the Six Months Ended June 30, 2016 and 2015 (unaudited)	7
	Notes to Financial Statements (unaudited)	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	29
<u>Item 4.</u>	Controls and Procedures	29
PART II.	OTHER INFORMATION	29
Item 1.	Legal Proceedings	29
Item 1A.	Risk Factors	30
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	31
Item 6.	Exhibits	32
Signatures		33

In this report, unless otherwise stated or the context otherwise indicates, references to "Xencor," "the Company," "we," "us," "our" and similar references refer to Xencor, Inc. The Xencor logo is a registered trademark of Xencor, Inc. This report also contains registered marks, trademarks and trade names of other companies. All other trademarks, registered marks and trade names appearing in this report are the property of their respective holders.

# SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of federal securities laws. Forward-looking statements include statements that may relate to our plans, objectives, goals, strategies, future events, future revenues or performance, capital expenditures, financing needs and other information that is not historical information. Many of these statements appear, in particular, under the headings "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations". Forward-looking statements can often be identified by the use of terminology such as "subject to", "believe", "anticipate", "plan", "expect", "intend", "estimate", "project", "may", "will", "should", "could", "can", the negatives thereof, variations thereon and similar expressions, or by discussions of strategy.

All forward-looking statements, including, without limitation, our examination of historical operating trends, are based upon our current expectations and various assumptions. We believe there is a reasonable basis for our expectations and beliefs, but they are inherently uncertain. We may not realize our expectations, and our beliefs may not prove correct. Actual results could differ materially from those described or implied by such forward-looking statements. The following uncertainties and factors, among others (including those set forth under "Risk Factors"), could affect future performance and cause actual results to differ materially from those matters expressed in or implied by forward-looking statements:

- · our plans to develop and commercialize our product candidates;
- · our ongoing and planned clinical trials;
- · the timing of and our ability to obtain and maintain regulatory approvals for our product candidates;
- · our estimates regarding expenses, future revenue, capital requirements and needs for additional financing;
- · our ability to identify additional products or product candidates with significant commercial potential that are consistent with our business objectives;
- · the rate and degree of market acceptance and clinical utility of our products;
- · the capabilities and strategy of our suppliers and vendors including key manufacturers of our clinical drug supplies;
- · significant competition in our industry;
- · costs of litigation and the failure to successfully defend lawsuits and other claims against us;
- · our partners' ability to advance drug candidates into, and successfully complete, clinical trials;
- · our ability to receive research funding and achieve anticipated milestones under our collaborations;
- · our intellectual property position;
- · loss or retirement of key members of management;
- · costs of compliance and our failure to comply with new and existing governmental regulations;
- $\cdot \ failure \ to \ successfully \ execute \ our \ growth \ strategy, \ including \ any \ delays \ in \ our \ planned \ future \ growth; \ and$
- · our failure to maintain effective internal controls.

The factors, risks and uncertainties referred to above and others are more fully described under the heading "Risk Factors" in our Annual Report on Form 10-K for the fiscal year ended December, 31, 2015. Forward-looking statements should be regarded solely as our current plans, estimates and beliefs. You should not place undue reliance on forward-looking statements. We cannot guarantee future results, events, levels of activity, performance or achievements. We do not undertake and specifically decline any obligation to update, republish or revise forward-looking statements to reflect future events or circumstances or to reflect the occurrences of unanticipated events.

# PART I — FINANCIAL INFORMATION

Item1. Financial Statements

# Xencor, Inc. Balance Sheets (In thousands, except share amounts)

		June 30, 2016	D	ecember 31, 2015
		unaudited)		2015
Assets	,	unauuncu)		
Current assets				
Cash and cash equivalents	\$	7,877	\$	12,590
Marketable securities		83,228		83,840
Accounts receivable		150,354		44
Prepaid expenses and other current assets		2,192		1,201
Total current assets		243,651		97,675
Property and equipment, net		2,508		2,310
Patents, licenses, and other intangible assets, net		10,353		9,971
Marketable securities - long term		77,666		96,891
Other assets		103		63
Total assets	\$	334,281	\$	206,910
Liabilities and stockholders' equity				
Current liabilities				
Accounts payable	\$	7,694	\$	6,400
Accrued expenses		3,278		3,634
Current portion of deferred rent		137		108
Current portion of deferred revenue		103,063		33,287
Income taxes		1,781		<u> </u>
Total current liabilities		115,953		43,429
Deferred rent, less current portion		424		507
Deferred revenue, less current portion		9,307		542
Total liabilities		125,684		44,478
Commitments and contingencies				
Stockholders' equity				
Preferred stock, \$0.01 par value: 10,000,000 authorized shares; -0- issued and				
outstanding shares at June 30, 2016 and December 31, 2015		_		
Common stock, \$0.01 par value: 200,000,000 authorized shares at June 30, 2016 and December 31, 2015; 40,944,080 issued and outstanding at June 30, 2016 and				
40,551,039 issued and outstanding at December 31, 2015		409		405
Additional paid-in capital		428,790		424,128
Accumulated other comprehensive income (loss)		216		(516)
Accumulated deficit		(220,818)		(261,585)
Total stockholders' equity		208,597	_	162,432
Total liabilities and stockholders' equity	\$	334,281	\$	206,910
Total natifices and stockholders equity	Ψ	55 1,251	Ψ.	200,510

 $See\ accompanying\ notes.$ 

# Xencor, Inc. Statements of Comprehensive Income (Loss) (unaudited) (In thousands, except share and per share data)

	Three Months Ended				Six Months Ended				
		June	30,			Jun	e 30,		
		2016	_	2015		2016	_	2015	
Revenue									
Collaborations, licenses and milestones	\$	66,007	\$	1,014	\$	73,259	\$	2,505	
Operating expenses									
Research and development		14,408		7,476		24,443		12,681	
General and administrative		3,043		2,524		6,993		5,288	
Total operating expenses		17,451		10,000		31,436		17,969	
Income (loss) from operations		48,556		(8,986)		41,823		(15,464)	
Other income (expenses)									
Interest income		368		290		726		391	
Interest expense		(10)		(4)		(37)		(8)	
Other income				(168)		4		(231)	
Total other income, net		358		118		693		152	
Income (loss) before income tax		48,914		(8,868)	\$	42,516	\$	(15,312)	
Provision for income tax		1,749				1,749		_	
Net income (loss)		47,165		(8,868)		40,767		(15,312)	
Other comprehensive income (loss), net of tax									
Net unrealized gain (loss) on marketable securities		113		(55)		732		(90)	
Comprehensive income (loss)	\$	47,278	\$	(8,923)	\$	41,499	\$	(15,402)	
Basic net income (loss) per common share	\$	1.16	\$	(0.22)	\$\$	1.00	\$	(0.41)	
Diluted net income (loss) per common share	\$	1.13	\$	(0.22)	\$\$	0.98	\$	(0.41)	
Basic weighted average common shares outstanding	4	10,800,586		10,389,648	40,703,688		3	37,518,271	
Diluted weighted average common shares outstanding	4	1,738,460	40,389,648		41	1,701,262	3	37,518,271	

See accompanying notes.

# Xencor, Inc. Statement of Stockholders' Equity (in thousands, except share data)

	Accumulated										
				A	dditional	O	ther				Total
	Common Stock		Paid C		Comprehensive		Accumulated		Stockholders'		
Stockholders' Equity	Shares	An	nount	ir	ı-Capital	Incon	ne (Loss)		Deficit		Equity
Balance, December 31, 2015	40,551,039	\$	405	\$	424,128	\$	(516)	\$	(261,585)	\$	162,432
Issuance of common stock upon exercise and vesting of stock awards	374,275		4		443		_		_		447
Issuance of common stock under the Employee Stock Purchase Plan	18,766		_		226		_		_		226
Comprehensive income	_		_		_		732		40,767		41,499
Stock-based compensation					3,993			_	<u> </u>		3,993
Balance, June 30, 2016 (unaudited)	40,944,080	\$	409	\$	428,790	\$	216	\$	(220,818)	\$	208,597

See accompanying notes.

# Xencor, Inc. Statements of Cash Flows (unaudited) (in thousands)

Six Months Ended

June 30,			
	2016		2015
\$	40,767	\$	(15,312)
	628		494
	876		240
	3,993		2,300
	45		54
	_		(9)
	(3)		_
	(150,310)		2,513
	160		(670)
	(1,031)		(538)
	1,294		1,387
	(357)		328
	1,781		_
	(53)		617
	78,541		(413
	(23,669)		(9,009
	(7,123)		(148,328)
	(761)		(915
	(493)		(1,115
	26,660		_
	_		9
	18,283		(150,349
	_		122,906
	447		429
	226		247
	_		(7,702
	673		115,880
	(4,713)		(43,478
	12,590		54,649
\$	7,877	\$	11,171
\$	732	\$	(90)
	\$	\$ 40,767  628 876 3,993 45 — (3)  (150,310) 160 (1,031) 1,294 (357) 1,781 (53) 78,541 (23,669)  (7,123) (761) (493) 26,660 — 18,283  — 447 226 — 673 (4,713) 12,590 \$ 7,877	\$ 40,767 \$  628 876 3,993 45 — (3)  (150,310) 160 (1,031) 1,294 (357) 1,781 (53) 78,541 (23,669)  (7,123) (761) (493) 26,660 — 18,283  — 447 226 — 673 (4,713) 12,590 \$ 7,877 \$

See accompanying notes.

# Xencor, Inc.

# Notes to Financial Statements (unaudited)

June 30, 2016

# 1. Summary of Significant Accounting Policies

# **Basis of Presentation**

The accompanying unaudited interim financial statements for Xencor, Inc. (the Company) have been prepared in accordance with U.S. generally accepted accounting principles (GAAP) for interim financial information. The financial statements include all adjustments (consisting only of normal recurring adjustments) that the management of the Company believes are necessary for a fair presentation of the periods presented. The preparation of interim financial statements requires the use of management's estimates and assumptions that affect reported amounts of assets and liabilities at the date of the interim financial statements and the reported revenues and expenditures during the reported periods. These interim financial results are not necessarily indicative of the results expected for the full fiscal year or for any subsequent interim period.

The accompanying unaudited interim financial statements and related notes should be read in conjunction with the audited financial statements and notes thereto included in the Company's 2015 Annual Report on Form 10-K filed with the Securities and Exchange Commission (SEC) on March 8, 2016.

# **Marketable Securities**

The Company has an investment policy that includes guidelines on acceptable investment securities, minimum credit quality, maturity parameters and concentration and diversification. The Company invests its excess cash primarily in marketable securities issued by investment grade institutions.

The Company considers its marketable securities to be available-for-sale. These assets are carried at fair value and the unrealized gains and losses are included in accumulated other comprehensive income (loss). Accrued interest on marketable securities is included in marketable securities. If a decline in the value of a marketable security in the Company's investment portfolio is deemed to be other-than-temporary, the Company writes down the security to its current fair value and recognizes a loss as a charge against income. The Company reviews its portfolio of marketable securities, using both quantitative and qualitative factors, to determine if declines in fair value below cost are other-than-temporary.

# **Recent Accounting Pronouncements**

In March 2016, the FASB issued ASU No. 2016-09, "Compensation - Stock Compensation (Topic 718): Improvements to Employee Share-Based Payment Accounting," which amends the current stock compensation guidance. The amendments simplify the accounting for the taxes related to stock based compensation, including adjustments to how excess tax benefits and a company's payments for tax withholdings should be classified. The standard is effective for fiscal periods beginning after December 15, 2016, with early adoption permitted.

In April 2016, the FASB issued ASU No. 2016-10 "Revenue from Contracts with Customers (Topic 606): Identifying Performance Obligations and Licensing" which amends and clarifies the revenue recognition guidance on accounting for licenses of intellectual property (IP) and identifying performance obligations. The amendment clarifies how an entity should evaluate the nature of its promise in granting a license of IP which will determine whether it recognizes revenue over time or at a point in time and also clarifies when a promised good or service is separately identifiable, which is an important step in determining whether goods or services should be accounted for as separate performance obligations. The amendment has the same effective date as the new revenue recognition standard which is for fiscal periods after December 15, 2017.

The Company is currently evaluating the impact of the adoption of the new accounting pronouncements on its financial statements and related disclosures.

There have been no other material changes to the significant accounting policies previously disclosed in the Company's 2015 Annual Report on Form 10-K.

# 2. Fair Value of Financial Instruments

Financial instruments included in the financial statements include cash equivalents, marketable securities, trade accounts receivable, accounts payable and accrued expenses. Marketable securities and cash equivalents are carried at fair value. The fair value of the other financial instruments closely approximates their fair value due to their short maturities.

The Company accounts for recurring and non-recurring fair value measurements in accordance with FASB Accounting Standards Codification (ASC) 820, *Fair Value Measurements and Disclosures* (ASC 820). ASC 820 defines fair value, establishes a fair value hierarchy for assets and liabilities measured at fair value, and requires expanded disclosure about fair value measurements. The ASC 820 hierarchy ranks the quality of reliable inputs, or assumptions, used in the determination of fair value and requires assets and liabilities carried at fair value to be classified and disclosed in one of the following three categories:

- Level 1—Fair Value is determined by using unadjusted quoted prices that are available in active markets for identical assets or liabilities.
- Level 2—Fair Value is determined by using inputs other than Level 1 quoted prices that are directly or indirectly observable. Inputs can include quoted prices for similar assets or liabilities in active markets or quoted prices for identical assets or liabilities in markets that are not active. Related inputs can also include those used in valuation or other pricing models, such as interest rates and yield curves that can be corroborated by observable market data.
- Level 3—Fair value is determined by inputs that are unobservable and not corroborated by market data. Use of these inputs involves significant and subjective judgments to be made by the reporting entity e.g. determining an appropriate discount factor for illiquidity associated with a given security.

The Company measures the fair value of financial assets using the highest level of inputs that are reasonably available as of the measurement date. The assets recorded at fair value are classified within the hierarchy as follows for the periods reported (in thousands):

		June 30, 2016						December 31, 2015						
	F	Total air Value	Level 1		Level 1		Level 1 Level 2		Total Fair Value		Level 1			Level 2
Money Market Funds	\$	6,969	\$	6,969	\$	_	\$	9,453	\$	9,453	\$	_		
Corporate Securities		103,405		_		103,405		114,846		_		114,846		
Government Securities		57,489		_		57,489		65,885		_		65,885		
	\$	167,863	\$	6,969	\$	160,894	\$	190,184	\$	9,453	\$	180,731		

Our policy is to record transfers of assets between Level 1 and Level 2 at their fair values as of the end of each reporting period, consistent with the date of the determination of fair value. During the three and six months ended June 30, 2016, there were no transfers between Level 1 and Level 2. The Company does not have any Level 3 assets or liabilities.

# 3. Net Income (Loss) Per Share

We compute net income (loss) per common share by dividing the net income (loss) attributable to common stockholders by the weighted-average number of common shares outstanding during the period without consideration of common stock equivalents. Diluted income (loss) per share is computed by dividing the net income (loss) attributable to common stockholders by the weighted-average number of common stock equivalents outstanding for the period. The treasury stock method is used to determine the dilutive effect of the Company's stock option grants. Potentially dilutive securities consisting of stock issuable under options and our 2013 Employee Stock Purchase Plan (ESPP) are not included in the diluted net loss per common share calculation where the inclusion of such shares would have had an antidilutive effect.

Basic and diluted income (loss) per common share is computed as follows (in thousands except share and per share data):

	Three Months Ended June 30,					Six Mon	ths Ended		
		2016		2015		2016		2015	
				(in thousa	ıds,	except			
				per sha	re d	ata)			
Numerator:									
Net income (loss) used for calculation of basic and diluted EPS	\$	47,165	\$	(8,868)	\$	40,767	\$	(15,312)	
Denominator:									
Weighted-average shares outstanding, basic		40,800,586		40,389,648		40,703,688		37,518,271	
Dilutive effect of stock options		937,874		_		997,574		_	
Weighted average shares outstanding, diluted		41,738,460		40,389,648		41,701,262		37,518,271	
Net income (loss) per share, basic	\$	1.16	\$	(0.22)	\$	1.00	\$	(0.41)	
Net income (loss) per share, diluted	\$	1.13	\$	(0.22)	\$	0.98	\$	(0.41)	

For the three and six months ended June 30, 2016 there were no shares from the Company's employee stock purchase plan that had a dilutive effect on shares outstanding. For the three and six months ended June 30, 2015, all outstanding potentially dilutive securities have been excluded from the calculation of diluted net loss per common share as the effect of including such securities would have been antidilutive.

# 4. Comprehensive income (loss)

Comprehensive income (loss) is comprised of net income (loss) and other comprehensive income (loss). For the three and six months ended June 30, 2016, the only component of other comprehensive income is net unrealized gains on marketable securities. For the three months and six months ended June 30, 2015, the only component of other comprehensive loss is net unrealized losses on marketable securities. There were no material reclassifications out of accumulated other comprehensive income (loss) during the three and six months ended June 30, 2016 and 2015.

# 5. Marketable Securities

The Company's marketable securities held as of June 30, 2016 and December 31, 2015 are summarized below:

		Amortized	ı	Gross Unrealized		Gross realized		
<u>June 30, 2016</u>		Cost		Gains		Losses		air Value
(in thousands)	_							
Money Market Funds	\$	6,96	9 \$	_	\$	_	\$	6,969
Corporate Securities		103,21	6	201		(12)		103,405
Government Securities		57,43	0	65		(6)		57,489
	\$	167,61	5 \$	266	\$	(18)	\$	167,863
	_							
Reported as								
Cash and cash equivalents							\$	6,969
Marketable securities								160,894
Total investments							\$	167,863

<u>December 31, 2015</u> (in thousands)	<b>A</b> 1	mortized Cost	Uı	Gross nrealized Gains	_	Gross nrealized Losses	Fair Value
Money Market Funds	\$	9,453	\$	_	\$	_	\$ 9,453
Corporate Securities		115,148		6		(308)	114,846
Government Securities		66,099		_		(214)	65,885
	\$	190,700	\$	6	\$	(522)	\$ 190,184
Reported as							
Cash and cash equivalents							\$ 9,453
Marketable securities							180,731
Total investments							\$ 190,184

The maturities of the Company's marketable securities are as follows:

	Amortized			Estimated	
<u>June 30, 2016</u>		Cost	Fair Value		
(in thousands)					
Mature in one year or less	\$	83,202	\$	83,228	
Mature after one year through five years		77,444		77,666	
	\$	160,646	\$	160,894	

December 31, 2015 (in thousands)	 Amortized Cost	 Estimated Fair Value
Mature in one year or less	\$ 83,963	\$ 83,840
Mature after one year through five years	97,284	96,891
	\$ 181,247	\$ 180,731

# 6. Stock Based Compensation

Our Board of Directors and the requisite stockholders previously approved the 2010 Equity Incentive Plan (the 2010 Plan). In October 2013, our Board of Directors approved the 2013 Equity Incentive Plan (the 2013 Plan) and in November 2013 our stockholders approved the 2013 Plan. The 2013 Plan became effective as of December 3, 2013, the date of the Company's initial public offering (IPO). As of December 2, 2013, we suspended the 2010 Plan and no additional awards may be granted under the 2010 Plan. Any shares of common stock covered by awards granted under the 2010 Plan that terminate after December 2, 2013 by expiration, forfeiture, cancellation or other means without the issuance of such shares will be added to the 2013 Plan reserve.

As of June 30, 2016 the total number of shares of common stock available for issuance under the 2013 Plan is 7,352,140, which includes 2,684,456 of common stock that were available for issuance under the 2010 Plan as of the effective date of the 2013 Plan. Unless otherwise determined by the Board, beginning January 1, 2014, and continuing until the expiration of the 2013 Plan, the total number of shares of common stock available for issuance under the 2013 Plan will automatically increase annually on January 1 of each year by 4% of the total number of issued and outstanding shares of common stock as of December 31 of the immediate preceding year. Pursuant to approval by our board on January 1, 2016, the total number of shares of common stock available for issuance under the 2013 Plan was increased by 1,400,000 shares. As of June 30, 2016 a total of 3,142,750 options had been issued under the 2013 Plan.

In November 2013, our Board of Directors and stockholders approved the 2013 Employee Stock Purchase Plan (ESPP), which became effective as of December 5, 2013. We have reserved a total of 581,286 shares of common stock for issuance under the ESPP. Unless otherwise determined by our Board, beginning on January 1, 2014, and continuing until the expiration of the ESPP, the total number of shares of common stock available for issuance under the ESPP will automatically increase annually on January 1 by the lesser of (i) 1% of the total number of issued and outstanding shares of common stock as of December 31 of the immediately preceding year, or (ii) 621,814 shares of common stock. Pursuant to approval by our board, there was no increase in the number of authorized shares in the ESPP in 2016. As of June 30, 2016, we have issued a total of 195,129 shares of common stock under the ESPP.

Total employee, director and non-employee stock-based compensation expense recognized for the three and six months ended June 30, 2016 are as follows (in thousands):

	Three Months Ended				Six Months Ended					
	 June 30,				June	30,				
	2016 2015				2016	2015				
General and administrative	\$ 874	\$	520	\$	1,826	\$	1,035			
Research and development	1,159		673		2,167		1,265			
	\$ 2,033	\$	1,193	\$	3,993	\$	2,300			

The following table summarizes option activity under our stock plans and related information:

		Weighted	Weighted		
		Average	Average		
	Number of	Exercise	Remaining	Ag	gregate
	Shares subject	Price	Contractual	Ir	ntrinsic
	to outstanding	(Per	Term	,	Value
	options	Share)	(in years)	(in t	nousands)
Balances at December 31, 2015	3,370,901	\$ 8.50	6.98		
Options granted	1,138,000	\$ 12.71			
Options forfeited	(43,486)	\$ 12.03			
Options cancelled	(1,031)	\$ 15.69			
Options exercised	(374,275)	\$ 1.20			
Balance at June 30, 2016	4,090,109	\$ 10.30	7.80	\$	35,630
Exercisable	1,728,711	\$ 7.24	6.34	\$	20,347

We calculate the intrinsic value as the difference between the exercise price of the options and the closing price of common stock of \$18.99 per share as of June 30, 2016.

Weighted average fair value of options granted during the six-month period ended June 30, 2016 and 2015 was \$8.45 and \$10.58 per share, respectively. There were 909,750 options granted during the period ended June 30, 2015. We estimated the fair value of each stock option using the Black-Scholes option-pricing model based on the date of grant of such stock option with the following weighted average assumptions for the three and six months ended June 30, 2016 and 2015:

	Options		Options	<u> </u>	
	Three Months E	Ended			
	June 30, June 30,				
	2016	2015	2016	2015	
Expected term (years)	5.9	5.9	6.1	6.0	
Expected volatility	76.1 %	75.6 %	75.8 %	76.4 %	
Risk-free interest rate	1.36 %	1.60 %	1.53 %	1.61 %	
Expected dividend yield	— %	— %	— %	— %	

	ESPP	<del></del>	ESPP				
	Three Months	Three Months Ended Six Months Ended					
	June 30	,					
	2016	2015	2016	2015			
Expected term (years)	0.5 - 2.0	0.5 - 2.0	0.5 - 2.0	0.5 - 2.0			
Expected volatility	67.8 - 79.6 %	70.6 - 82.9 %	67.8 - 79.6 %	70.6 - 82.9 %			
Risk-free interest rate	.47%93 %	.06%46 %	.47%93 %	.06%46 %			
Expected dividend yield	— %	— %	— %	— %			

As of June 30, 2016, the unamortized compensation expense related to unvested stock options was \$16.8 million, net of estimated forfeitures. The remaining unamortized compensation expense will be recognized over the next three years.

# 7. Commitments and Contingencies

# **Operating Leases**

The Company leases office and laboratory space in Monrovia, CA through June 2020 with an option to renew for an additional five years.

The Company also leases office space in San Diego, CA through April 2018 and includes an option to renew for a period of one year. In March 2016, the Company signed a lease for additional space contiguous with its existing office space. The combined lease expires in June 2020.

The leases are accounted for as non-cancellable operating leases and future minimum payments are as follows (in thousands):

# Years ending December 31,

For the remainder of the fiscal year	\$ 375
2017	807
2018	833
2019	859
2020	466
Thereafter	_

Rent expense for the six months ended June 30, 2016 and 2015 was \$298,000 and \$272,000 respectively.

# **Contingencies**

From time to time, the Company may be subject to various litigation and related matters arising in the ordinary course of business. The Company does not believe it is currently subject to any material matters where there is at least a reasonable possibility that a material loss may be incurred.

On March 3, 2015, a verified class action complaint, captioned DePinto v. John S. Stafford, et al., C.A. No. 10742, was filed in the Court of Chancery of the State of Delaware against certain of the Company's current and former directors alleging cause of action for Breach of Fiduciary Duty and Invalidity of Director and Stockholder Consents. In general, the complaint alleged that the plaintiff and the class he seeks to represent were shareholders of the Company during the recapitalization and certain related transactions that the Company underwent in 2013 and that the defendants breached their fiduciary duties in the course of approving that series of transactions. It also challenged as invalid certain corporate acts taken in the 2013 time period.

On June 10, 2015, the Company filed a Verified Petition for Relief under Del. C. Section 205 (the 205 Petition) related to the corporate acts challenged in the complaint. The defendants filed an answer to the class action complaint on June 22, 2015. On July 9, 2015, the Court consolidated the 205 Petition with the class action, joined the Company as a defendant and ordered it to file the claims in the 205 Petition as counter-claims in the class action, which the Company has done.

On August 11, 2015, the Company filed a Motion for Leave to File an Amended Counter-Claim, along with the proposed Amended Counter-Claim and related documents. On October 5, the parties filed a Stipulation of Partial Settlement and related documents disclosing a settlement of the invalidity claims addressed in the complaint, the counter-claim and the proposed amended counter-claim including a request by plaintiff's counsel for reimbursement of legal fees up to \$950,000. On October 7, 2015, Xencor filed the Amended Counter-Claim and related documents. On December 14, 2015, the Court entered an Order and Partial Final Judgment approving the settlement of the invalidity claims, validating each corporate act challenged in the complaint, dismissing with prejudice Count II of the complaint (the invalidity claims) and granting plaintiff's counsel a fee award. We have paid the plaintiff's legal award of \$950,000 net of insurance proceeds of \$187,500 which has been reflected as a charge in our 2015 operations. We continue to recognize legal costs as incurred and offset any insurance proceeds when approved and issued.

Based on the nature of the remaining claim, the Company believes that it is not possible to estimate the likelihood of loss or a range of potential loss related to the claim; accordingly, no amount for any loss has been accrued at June 30, 2016.

# 8. Collaboration and Licensing Agreements

Following is a summary description of the arrangements that generated revenue in the six months ended June 30, 2016 and 2015.

### Novartis

In June 2016, the Company entered into a Collaboration and License Agreement (the Agreement) with Novartis Institutes for BioMedical Research, Inc., or Novartis, to develop and commercialize bispecific and other Fc modulated antibody drug candidates using the Company's proprietary XmAb® technologies and drug candidates. Pursuant to the Agreement:

- · The Company granted Novartis certain exclusive rights to research, develop and commercialize XmAb14045 and XmAb13676, two development stage products that incorporate the Company's bispecific Fc technology,
- · The Company will apply its bispecific technology in up to four target pair antibodies identified by Novartis (each a Global Discovery Program) and,
- · The Company will provide Novartis with a non-exclusive license to certain of its Fc technologies to apply against up to ten targets identified by Novartis.

The Company received a non-refundable upfront payment under the Agreement of \$150 million in July 2016 and is eligible to receive up to \$2.4 billion in future development, regulatory and sales milestones in total for all programs that could be developed under the Agreement.

Under the Agreement, the Company granted Novartis a worldwide co-exclusive license with Xencor to research, develop and manufacture XmAb14045 and XmAb13676. The Company also granted Novartis an exclusive license to commercialize XmAb14045 and XmAb13676 in all worldwide territories outside the United States (U.S.). XmAb14045 is a clinical candidate that binds the CD123 antigen and the cytotoxic T-cell binding domain CD3 (the XmAb14045 Program) and targets acute myeloid leukemia (AML). XmAb13676 is a clinical candidate that binds the CD20 antigen and the cytotoxic T-cell binding domain CD3 (the XmAb13676 Program) and targets B-cell malignancies. Assuming successful development and commercialization of a product, the Company could receive up to \$325 million in milestone payments for each of XmAb14045 and XmAb13676. The total potential milestones for each product include \$90 million in development milestones, \$110 million in regulatory milestones and, \$125 million in sales milestones. If commercialized, the Company is eligible to receive tiered low double-digit royalties on global net sales of approved products outside the US.

The Company and Novartis will co-develop XmAb14045 and XmAb13676 worldwide and share development costs equally. The Company may elect to opt-out of the development of either program by providing notice to Novartis. If the Company elects to opt-out with respect to a program, Novartis will receive the Company's U.S. rights to the program and the Company will receive low double-digit royalties on U.S. net sales in addition to the royalties on net sales outside the US.

Pursuant to the Novartis Agreement, the Company will apply its bispecific technology to up to four target pair antibodies selected, available for exclusive license to Novartis and not subject to a Xencor internal program. The Company will apply its bispecific technology to generate bispecific antibody candidates from starting target pair antibodies provided by Novartis for each of the four Global Discovery Programs and return the bispecific product candidate to Novartis for further testing, development and commercialization. Novartis has the right to substitute up to four of the original selected target pair antibodies during the research term provided that Novartis has not filed and received acceptance for an Investigational New Drug Application (IND) with the Xencor provided bispecific candidate. The research term is five years from the date of the Agreement.

Novartis will assume full responsibility for development and commercialization of each product candidate under each of the Global Discovery Programs. Assuming successful development and commercialization of each Global Discovery Program compound, the Company could receive up to \$250.0 million in milestones for each Global Discovery Program which includes \$50.0 million in development milestones, \$100.0 million in regulatory milestones and \$100.0 million in sales milestones. If commercialized, the Company is eligible to receive mid-single digit royalties on global net sales of approved products.

Under the Novartis Agreement, the Company has the right to participate in the development and commercialization of one of the Global Discovery Programs prior to filing an IND for Global Discovery Program. If the Company elects to participate in development, it will assume responsibility for 25% of the worldwide development costs for the program and 50% of commercialization costs and will receive 50% of the US profits on net sales of the product.

Under the Novartis Agreement, the Company is also granting Novartis a non-exclusive research license to use certain of the Company's Fc technologies, specifically Cytotoxic, Xtend and Immune Inhibitor to research, develop, commercialize and manufacture antibodies against up to ten targets selected by Novartis, available for non-exclusive license and not subject to a Xencor internal program. Novartis will assume all research, development and commercialization costs for products that are developed from application of the Fc technologies. Assuming successful development and commercialization of a compound that incorporates an Fc technology, the Company could receive up to \$75.0 million in milestones for each target which includes \$15.0 million in development milestones, \$30.0 million in regulatory milestones and \$30 million in sales milestones. If commercialized, the Company is eligible to receive low single-digit royalties on global net sales of approved products.

The Company evaluated the Agreement and determined that it is a revenue arrangement with multiple deliverables or performance obligations. The Company's substantive performance obligations under the Agreement include:

- · delivery of an exclusive license to commercialize XmAb14045 in worldwide territories outside the U.S., with worldwide co-exclusive rights with Xencor to research, develop and manufacture XmAb14045
- · delivery of an exclusive license to commercialize XmAb13676 in worldwide territories outside the U.S., with worldwide co-exclusive rights with Xencor to research, develop and manufacture XmAb13676
- · application of its bispecific technology to four Novartis selected target pair antibodies and delivery of four bispecific product candidates and,
- · delivery of a non-exclusive license to its Fc technologies: Cytotoxic, Xtend and Immune Inhibitor

The Company determined that the \$150 million upfront payment represents the total initial consideration and was allocated to each of the deliverables using the relative selling price method. The Company determined that each of the development and regulatory milestones is substantive. Although sales milestones are not considered substantive, they are still recognized upon achievement of a milestone. After identifying each of the deliverables included in the arrangement, the Company determined the relative selling price using its best estimate of selling price for each of the deliverables.

The estimated selling price for the licensing rights to the XmAb13676 Program are the Company's best estimate of selling price and was determined based on market conditions, similar arrangements entered into by third parties including the Company's understanding of pricing terms offered for comparable transactions that involve licensing bispecific antibody development candidates. The Company reviewed recent published market transactions that are comparable to the license of the XmAb13676 Program in the Novartis Agreement. The Company adjusted the value of the published market information to reflect differences in stage of development and rights and potential markets to determine the estimated selling price for the license rights to the XmAb13676 program. This amount represents the value that a third party would be willing to pay for certain rights to the XmAb13676 Program including the exclusive right to commercialize XmAb13676 in all territories outside the U.S.

The Company determined the estimated selling price for the rights to the XmAb14045 Program using the income approach by calculating a risk-adjusted present value of the potential revenue that could be earned from the license reduced by the minimum development costs that the Company is obligated to fund under the Agreement. This amount represents the value that a third party would be willing to pay for certain rights to the XmAb14045 Program including the right to commercialize XmAb14045 in all territories outside the U.S.

The Company's estimated selling price for each Global Discovery Program is the Company's best estimate. The best estimate for each Global Discovery Programs was determined using the income approach by calculating a risk-adjusted net present value of the potential revenue that could be earned from each Global Program license reduced by the estimated cost of the Company's efforts to deliver the completed Global Program bispecific candidate to Novartis. These amounts represent the value that a third party would be willing to pay as an upfront for access to the Company's bispecific technology and capabilities.

The Company's estimated selling price for the Fc licenses is its best estimate and was determined by considering market and entity-specific factors. The Company has previously licensed its Fc technologies on a limited basis to third parties. The Company considered the term of the Novartis license, scope of the rights granted for each license, the type of technologies subject to the license, and the potential number of targets that may be applied in establishing its best estimate for the Fc license.

The total allocable consideration of \$150 million was allocated to the deliverables based on the relative selling price method as follows:

- \* \$27.1 million to the XmAb14045 Program,
- \* \$31.4 million to the XmAb13676 Program,
- \* \$20.05 million to each of the five Global Discovery Programs and,
- \* \$11.3 million to the Fc licenses

The Company recognized as license revenue the amount of the total allocable consideration allocated to the XmAb13676 and XmAb14045 Programs upon delivery of the exclusive license to Novartis both of which were transferred as of the effective date of the Agreement. At the time that each Global Discovery Program is accepted by Novartis, the Company will recognize as collaboration revenue of \$20.05 million for each program. Since Novartis has substitution rights for up to four target pair antibodies, revenue recognition may be delayed until the earlier that Novartis has an open IND for a delivered bispecific Discovery Program or the right to substitute the target pair lapses. The Company will recognize as licensing revenue the amount of the total consideration allocated to the Fc license over the five year research term beginning from the effective date of the Agreement.

During the three and six months ended June 30, 2016, we recognized \$58.6 million of revenue under this arrangement. As of June 30, 2016 there is \$91.4 million in deferred revenue related to the arrangement.

# Amgen, Inc.

In September 2015, the Company entered into a research and license agreement (the Amgen Agreement) with Amgen, Inc. (Amgen) to develop and commercialize bispecific antibody product candidates using the Company's proprietary XmAb® bispecific Fc technology. Under the Amgen Agreement, the Company granted an exclusive license to Amgen to develop and commercialize bispecific drug candidates from the Company's preclinical program that bind the CD38 antigen and the cytotoxic T-cell binding domain CD3, (the CD38 Program). The Company will also apply its bispecific technology to five previously identified Amgen provided targets (each a Discovery Program). The Company received a \$45.0 million upfront payment from Amgen and is eligible to receive up to \$1.7 billion in future development, regulatory and sales milestones in total for all six programs and is eligible to receive royalties on any global net sales of products.

In the fourth quarter ended December 31, 2015, the Company transferred the research material and data related to its CD38 Program to Amgen. Amgen will assume full responsibility for the further development and commercialization of product candidates under the CD38 Program. Assuming successful development and commercialization of a product, the Company could receive up to \$355 million in milestones payments which include \$55 million in development milestones, \$70 million in regulatory milestones and, \$230 million in sales milestones. If commercialized, the Company is eligible to receive from high single-digit up to low double-digit royalties on global net sales of approved products under the CD38 Program.

Pursuant to the Amgen Agreement, for each of the five Discovery Programs the Company will apply its bispecific technology to antibody molecules provided by Amgen that bind Discovery Program Targets and return the

bispecific product candidates to Amgen for further testing, development and commercialization. Subject to approval by Xencor, Amgen has the right to substitute up to three of the previously identified targets during the research term provided that Amgen has not initiated non-human primate studies with the Xencor provided bispecific candidate. The initial research term is three years from the date of the Amgen Agreement but Amgen, at its option, may request an extension of one year if Xencor has not completed delivery of all five Discovery Program bispecific candidates to Amgen.

Amgen will assume full responsibility for development and commercialization of product candidates under each of the Discovery Programs. Assuming successful development and commercialization of each Discovery Program compound, the Company could receive up to \$260.5 million in milestones for each compound which include \$35.5 million in development milestones, \$55.0 million in regulatory milestones and \$170.0 million in sales milestones. If commercialized, the Company is eligible to receive mid to high single-digit royalties on global net sales of approved products.

The Company evaluated the Amgen Agreement and determined that it is a revenue arrangement with multiple deliverables or performance obligations. The Company's substantive performance obligations under the Amgen Agreement include delivery of research material and data related to its CD38 Program and application of its bispecific technology to five Amgen provided targets and delivery of the five bispecific product candidates. The Company evaluated the Amgen Agreement and determined that the CD38 Program and each of the five Discovery Programs represent separate units of accounting.

The \$45 million upfront payment represents the total initial consideration and was allocated to each of the deliverables using the relative selling price method. After identifying each of the deliverables included in the arrangement, the Company determined its best estimate of selling price for each of the deliverables. In order to determine the best estimate of selling price for the CD38 Program, the Company determined the value of the CD38 Program by calculating a risk-adjusted present value of the potential revenue from the future development and regulatory milestones under the Amgen Agreement. This amount represents the value that a third party would be willing to pay as an upfront fee to license the Company's CD38 Program.

The Company determined the value of each of the Discovery Programs by calculating a risk-adjusted net present value of the potential revenue from future development and regulatory milestones reduced by the estimated cost of the Company's efforts to apply its bispecific technology to the Amgen targets and deliver the five bispecific product candidates. These amounts represent the value that a third party would be willing to pay as an upfront for access to the Company's bispecific technology and capabilities.

The total allocable consideration of \$45 million was allocated to the deliverables based on the relative selling price method as follows:

\$13.75 million to the CD38 Program and, \$6.25 million to each of the five Discovery Programs

The Company recognized as collaboration revenue the amount of consideration allocated to the CD38 Programs upon delivery of the CD38 research material and data to Amgen in the fourth quarter of 2015.

In the first quarter ended March 31, 2016, Amgen exercised its substitution rights with respect to one of the previously identified Discovery targets. In the first quarter ended March 31, 2016, the Company delivered bispecific product candidates for three Discovery Program targets to Amgen. In the second quarter ended June 30, 2016 the Company delivered an additional product candidate for a Discovery target. As of June 30, 2016, the Company has delivered four of the discovery programs with Amgen exercising its substitution rights to one.

At the time that each bispecific Discovery Program is accepted by Amgen, the Company will recognize as collaboration revenue \$6.25 million for each program. Since Amgen has substitution rights for up to three targets, revenue recognition may be delayed until the earlier that Amgen initiates non-human primate studies for a delivered bispecific Discovery Program or the right to substitute the target lapses.

During the three and six months ended June 30, 2016, we recognized \$6.2 million and \$12.5 million in revenue under this arrangement, respectively. As of June 30, 2016 there is \$18.7 million in deferred revenue related to the arrangement.

# Merck Sharp & Dohme Corporation

In July 2013, we entered into a License Agreement with Merck Sharp & Dohme Corp (Merck). Under the terms of the agreement, we provided Merck with a non-exclusive commercial license to certain patent rights to our Fc domains to apply to one of their compounds. We also provided Merck with contingent options to take additional non-exclusive commercial licenses. The contingent options provide Merck an opportunity to take non-exclusive commercial licenses at an amount less than the amount paid for the original license. The agreement provided for an upfront payment of \$1.0 million and annual maintenance fees totaling \$0.5 million. We are also eligible to receive future milestones and royalties as Merck advances the compound into clinical development.

We determined that the deliverables under this agreement were the non-exclusive commercial license and the options. The options are considered substantive and contingent and no amount of the upfront payment was allocated to these options. We also determined that the future milestones and related payments were substantive and contingent and did not allocate any of the upfront payment to the milestones.

During each of the three and six months ended June 30, 2016 and 2015 we recognized \$25,000 and \$50,000 of revenue respectively. As of June 30, 2016, there is \$100,000 of deferred revenue related to this arrangement.

# Alexion Pharmaceuticals, Inc.

In January 2013, we entered into an option and license agreement with Alexion Pharmaceuticals, Inc. (Alexion). Under the terms of the agreement, we granted to Alexion an exclusive research license, with limited sublicensing rights, to make and use our Xtend technology to evaluate and advance compounds against six different target programs during a five-year research term under the agreement, up to completion of the first multi-dose human clinical trial for each target compound. Alexion may extend the research term for an additional three years upon written notice to us and payment of an extension fee of \$2.0 million. Alexion is responsible for conducting all research and development activities under the agreement at its own expense.

In addition, we granted to Alexion an exclusive option, on a target-by-target basis, to obtain an exclusive commercial, worldwide, royalty-bearing license, with sublicensing rights, under our Xtend technology to develop and commercialize products that contain the target for which the option is exercised. In order to exercise this option, Alexion must pay a \$4.0 million option fee with respect to each target for which the option is exercised. Alexion may exercise this option at any time during the research term. An option must be exercised for any compound that is advanced into development after the first multi-ascending dose trial is initiated.

Under the agreement, we received an upfront payment of \$3.0 million. Alexion is also required to pay an annual maintenance fee of \$0.5 million during the research term of the agreement and \$1.0 million during any extension of the research term. We determined that \$2.5 million of the upfront fee was allocated to the license and is being recognized into income over the initial research term of five years.

In the third quarter of 2014, Alexion achieved a clinical development milestone with an undisclosed molecule to be used against an undisclosed target. We received a milestone related to this trial in March 2015 upon issuance of certain patents related to our Xtend technology. In the fourth quarter of 2015, Alexion exercised its option to take an exclusive commercial license and achieved a further clinical development milestone. As a result of Alexion's exercise to take a commercial option to an undisclosed compound, the Company is eligible to receive additional development, regulatory and sales milestones under the agreement. If commercialized, the Company is eligible to receive royalties on global net sales of approved products.

During the three and six months ended June 30, 2016 and 2015 we recognized \$250,000 and \$500,000 of revenue respectively. As of June 30, 2016, we have deferred revenue related to this arrangement of \$1.0 million.

# Novo Nordisk A/S

In December 2014, we entered into a collaboration and license agreement with Novo Nordisk A/S (Novo). Under the terms of the agreement we granted Novo a research license to use certain Xencor technologies including our bispecific, Fcy-IIb, Xtend and other technologies during a two-year research term.

We are recognizing the \$2.5 million upfront payment as income over the two-year research term. The research funding is being recognized into income over the period that the services are being provided. We determined that future milestone payments were substantive and contingent and we did not allocate any of the upfront consideration to these milestones.

During each of the three months ended June 30, 2016 and 2015, we recognized \$0.7 million of revenue. During each of the six months ended June 30, 2016 and 2015, we recognized \$1.4 million of revenue. As of June 30, 2016, we have \$1.0 million in deferred revenue related to this arrangement.

# 9. Income taxes

The provision for income taxes for the three and six months ended June 30, 2016 represents the interim period tax allocation of the federal and state alternative minimum tax based on the Company's projected year end effective income tax rates which cannot be offset by the Company's net operating loss carryforwards.

The Company has deferred tax assets consisting primarily of net operating loss and tax credit carryforwards that have been fully offset by a valuation allowance.

# Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion and analysis should be read in conjunction with our financial statements and accompanying notes included in this Quarterly Report on Form 10-Q and the financial statements and accompanying notes thereto for the fiscal year ended December 31, 2015 and the related Management's Discussion and Analysis of Financial Condition and Results of Operations, both of which are contained in our Annual Report on Form 10-K for the year ended December 31, 2015. This Ouarterly Report on Form 10-O may contain "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, or the Securities Act, and Section 21E of the Securities Exchange Act of 1934, as amended, or the Exchange Act. Such forward-looking statements, which represent our intent, belief, or current expectations, involve risks and uncertainties. We use words such as "may," "will," "expect," "anticipate," "estimate," "intend," "plan," "predict," "potential," "believe," "should" and similar expressions to identify forward-looking statements, although not all forward-looking statements contain these identifying words. Such statements may include, but are not limited to, statements concerning: (i) the initiation, cost, timing, progress and results of our research and development activities, preclinical studies and future clinical trials, including our expected timeline for nominating clinical development candidates under our strategic alliances and our expected timeline for filing applications with regulatory authorities;(ii) our ability to obtain and maintain regulatory approval of our future product candidates, and any related restrictions, limitations, and/or warnings in the label of an approved product candidate; (iii) our ability to obtain funding for our operations; (iv) our plans to research, develop and commercialize our future product candidates; (v) our ability to attract collaborators with development, regulatory and commercialization expertise; (vi) our ability to obtain and maintain intellectual property protection for our technology; (vii) the size and growth potential of the markets for our technology and future product candidates, and our ability to serve those markets; (viii) our ability to successfully commercialize our technology and our future product candidates; (ix) our ability to develop sales and marketing capabilities, whether alone or with potential future collaborators; (x) regulatory developments in the United States and foreign countries; and (xi) the performance of our collaboration partners, licensees, third-party suppliers and manufacturers. Although we believe the expectations reflected in these forward-looking statements are reasonable, such statements are inherently subject to risk and we can give no assurances that our expectations will prove to be correct. You should not place undue reliance on these forward-looking statements, which apply only as of the date of this Quarterly Report on Form 10-Q. As a result of many factors, including without limitation those set forth under "Risk Factors" under Item 1A of Part II below, and elsewhere in this Quarterly Report on Form 10-Q, our actual results may differ materially from those anticipated in these forward-looking statements. We

undertake no obligation to update these forward-looking statements to reflect events or circumstances after the date of this report or to reflect actual outcomes.

# **Company Overview**

We are a clinical-stage biopharmaceutical company focused on discovering and developing engineered monoclonal antibodies to treat severe and life-threatening diseases with unmet medical needs. We use our proprietary XmAb technology platform to create next-generation antibody product candidates designed to treat autoimmune and allergic diseases, cancer and other conditions. In contrast to conventional approaches to antibody design, which focus on the portion of antibodies that interact with target antigens, we focus on the portion of the antibody that interacts with multiple segments of the immune system. This portion, referred to as the Fc domain, is constant and interchangeable among antibodies. Our engineered Fc domains, the XmAb technology, can be readily substituted for natural Fc domains.

Our business strategy is based on the plug-and-play nature of the XmAb technology, allowing us to create new antibody drug candidates for our internal development or licensing, or to selectively license access to one or more of our XmAb technologies or product candidates to pharmaceutical or biotechnology companies to use in developing their own proprietary antibodies and drug candidates with improved properties. These licensing transactions provide us with multiple revenue streams that help fund development of our wholly owned product candidates and usually require limited resources or efforts from us. There are currently nine antibody product candidates in clinical trials that have been engineered with XmAb technology, including seven candidates being advanced by licensees and development partners.

Our protein engineering capabilities allow us to continue to expand the functionality of the XmAb technology platform to identify new protein enhancements and create new antibody drug candidates with improved properties. Our bispecific technology, heterodimer Fc domains, enables the creation of bispecific drug candidates, which are antibodies that are engineered to bind two targets simultaneously. The core of our bispecific programs is a novel Fc domain that is a robust and portable scaffold for two, or potentially more, different antigen binding domains. Our Fc domain technology is designed to maintain full-length antibody properties in a bispecific antibody, potentially enabling stable molecules with favorable *in vivo* half-life and allowing for the use of standard antibody production methods. The portability of the bispecific technology, including the ability of bispecific candidates generated from our technology to use standard production methods, allows us to license access to our technology as highlighted in our two bispecific licensing transactions that we entered into in the last year.

In June 2016 we entered into the Novartis Agreement which included a \$150 million upfront payment and up to \$2.4 billion in potential development, regulatory and sales milestones. As part of the Agreement, we will apply our bispecific technology to up to four target pair antibodies selected, available for exclusive license to Novartis and not subject to a Xencor internal program.

We will apply our bispecific technology to generate bispecific antibody candidates from starting target pair antibodies provided by Novartis for each of the four Global Discovery Programs and return the bispecific product candidate to Novartis for further testing, development and commercialization. Assuming successful development and commercialization of each bispecific compound, we could receive up to \$250 million in milestones for each compound which includes \$50 million in development milestones, \$100 million in regulatory milestones and \$100 million in sales milestones. If commercialized, the Company is eligible to receive mid-single digit royalties on global net sales of approved products.

In September 2015 we entered into the Amgen Agreement which included a \$45 million upfront payment and up to \$1.7 billion in future development, regulatory and sales milestones if all programs under the agreement advance into development. In connection with the Amgen Agreement, we are applying our bispecific technology to up to five previously identified molecules identified by Amgen and approved by us. We are applying our bispecific technology to each of the five identified programs and returning the bispecific product candidates to Amgen, who is assuming full responsibility for further testing, development and commercialization. Assuming successful development and commercialization of each bispecific compound, we could receive up to \$260.5 million in milestones which include \$35.5 million in development milestones, \$55 million in regulatory milestones and \$170 million in sales milestones. If commercialized, we are eligible to receive mid to high-single digit royalties on global net sales of approved products. Through June 30, 2016 we have delivered four bispecific product candidates to Amgen under the Agreement.

Since we commenced active operations in 1998, we have devoted substantially all of our resources to staffing our company, business planning, raising capital, developing our technology platforms, identifying potential product candidates, undertaking pre-clinical and IND enabling studies and conducting clinical trials. We have no products approved for commercial sale and have not generated any revenues from product sales, and we continue to incur significant research and development expenses and other expenses related to our ongoing operations. To date, we have funded our operations primarily through the sale of stock and convertible promissory notes and through payments generated from our product development partnership and licensing arrangements. We raised \$80.5 million (\$72.5 million net of expenses) in December 2013 through the sale of common stock in connection with our Initial Public Offering (IPO) and full exercise by the underwriters of their over-allotment. We raised an additional \$122.9 million (\$115.2 million net of expenses) through a follow-on public offering of our common stock and full exercise by the underwriters of their over-allotment in March 2015. In September 2015 we received a \$45 million upfront payment from Amgen in connection with the 2015Amgen Agreement. In July 2016 we received a \$150 million upfront payment from Novartis in connection with the Novartis Agreement. Although it is difficult to predict our funding requirements, based upon our current operating plan, we anticipate that our cash and cash equivalents and related marketable securities as of June 30, 2016 in addition to the \$150 million received from Novartis in July 2016, will enable us to fund operations at least through the end of 2019.

As of June 30, 2016, we had an accumulated deficit of \$221 million. Substantially all of our operating losses that we have incurred resulted from expenses incurred in connection with our product candidate development programs, our research activities and general and administrative costs associated with our operations.

# **Company Programs**

We are developing a pipeline of candidates for clinical development based on our Immune Inhibitor Domain and Bispecific Domain technologies.

# **Immune Inhibitor Pipeline**

*XmAb5871* uses our XmAb Immune Inhibitor Fc Domain and targets B cells, an important component of the immune system. We believe that XmAb5871 has the potential to address a key unmet need in autoimmune therapies due to its combination of potent B-cell inhibition without B-cell depletion.

In March 2016 we initiated enrollment for two Phase 2 trials for XmAb5871, one trial in IgG4-Related Disease (IgG4-RD) and a trial in Systemic Lupus Erythematosus (SLE or Lupus). In July 2016 we initiated a Phase 1 trial with a subcutaneous formulation of XmAb5871.

*IgG4-RD*: we are currently enrolling a Phase 2 open-label pilot study of XmAb5871 for IgG4-RD. The current trial design is to enroll approximately 15 patients with scheduled treatment up to 24 weeks. The primary objective of the study is to evaluate the effect of every other week IV administration of XmAb5871 using the recently reported IgG4-RD Responder Index in patients with active IgG4-RD. Secondary objectives are to determine the safety and tolerability profile and to characterize the pharmacokinetics (PK) and immunogenicity of every other week IV administration of XmAb5871. IgG4-RD is a rare fibro-inflammatory autoimmune disorder that we estimate impacts up to 40,000 patients in the United States. IgG4-RD affects multiple organ systems and is characterized by the distinct microscopic appearance of disease organs, including dense presence of IgG4-positive plasma cells that is required for diagnosis. This objective diagnostic criterion is atypical for autoimmune diseases and offers advantages for accurately identifying patients. There are currently no approved therapies for this newly recognized disorder and corticosteroids are the current standard of care.

*SLE*: we are also enrolling a Phase 2 randomized, double blinded, placebo-controlled study of XmAb5871in SLE. This trial is designed to assess the effect of XmAb5871 on SLE disease activity in a shorter timeframe and using fewer patients compared to standard SLE trials, and XmAb5871 is the first newly developed agent being assessed with this novel trial design. The trial design calls for treating patients with moderate to severe, non-organ threatening SLE with XmAb5871 (or placebo) after their lupus disease activity has improved with a short course of intra-muscular (IM) steroid therapy. Background, potentially confounding, immunosuppressant medications will be stopped. In this double-blinded placebo-controlled study, the ability of XmAb5871 to maintain the improvement in disease activity after IM steroid therapy and in the absence of immunosuppressant medication will be assessed. Historically, SLE trial designs

generally add new medications to the many already taken by the patient, and hence display a discernible treatment effect only when restricted to the sickest patients. The trial will enroll approximately 90 subjects, 1:1 randomized to XmAb5871 or placebo, for up to 24 weeks.

*XmAb7195* uses our Immune Inhibitor Fc Domain and is being developed for the treatment of severe asthma and allergic diseases. XmAb 7195 is designed to reduce blood serum levels of IgE, which mediates allergic responses and allergic disease. In January 2015, we reported top-line interim data from Part 1 of the Phase 1a trial of XmAb7195, in which healthy volunteers received a single intravenous (IV) dose. Data showed rapid reduction of free IgE levels to below the limit of detection in 90% of treated subjects, including those treated at the lowest dose evaluated of 0.3 mg/kg, with parallel reductions in total IgE. In 2015, we continued the Phase 1a trial of XmAb7195, treating subjects with high baseline IgE levels, and in June 2015, we announced an expansion of the trial, adding cohorts of subjects that receive two IVdoses of XmAb7195. We announced complete data from these studies in May 2016. The results of these trials indicate that Xmab7195 was generally well tolerated when administered as a single IV infusion with transient, asymptomatic thrombocytopenia occurring at doses greater or equal to 2.0 mg/kg. XmAb7195 induced rapid and extensive depletion of serum free, serum total IgE, and basophil surface IgE at all doses tested. Across all dose levels tested, 93% of healthy adults (Part 1) and 75% of atopic subjects with predose total Ige over 300 IU/ml had reduction of free IgE levels to below the level of quantification following a single IV dose of XmAb7195. We plan on initiating a multi-dose Phase 1 trial for XmAb7195 with a subcutaneous formulation in healthy volunteers in the second half of 2016.

# **XmAb Bispecific Pipeline**

*XmAb14045* uses our XmAb bispecific Fc technology that allows us to create dual-antigen targeting molecules. We have an open IND and are planning enrollment of patients in the Phase 1 clinical trial for XmAb14045, our first bispecific oncology candidate, for the treatment of acute myeloid leukemia (AML). XmAb14045 targets CD123, an antigen on AML cells and leukemic stem cells, and CD3, an activating receptor on T cells. The trial is a Phase 1, open-label, multiple-dose, dose escalation study to assess safety, tolerability and preliminary anti-tumor activity in AML.

**XmAb13676** is our second bispecific oncology candidate and is expected to enter clinical trials in the second half of this year. It is a tumor-targeted antibody that contains both a B-cell tumor antigen binding domain (CD20) and a cytotoxic T-cell binding domain (CD3). The trial will be a Phase 1, open-label, multiple-dose, dose escalation study to assess safety, tolerability and preliminary anti-tumor activity in B-cell malignancies.

In connection with the Novartis Agreement we granted Novartis exclusive licenses to commercialize XmAb14045 and XmAb13676 in all worldwide territories outside the U.S., with worldwide co-exclusive rights with us to research, develop and manufacture XmAb14045 and XmAb13676. We continue to retain U.S. rights to both drug candidates and will co-develop worldwide both candidates with Novartis and share development costs equally. Upon successful development of each of Xmab14045 and XmAb13676 we are eligible to receive up to \$325 million in milestones which includes \$90 million in development milestones, \$110 million in regulatory milestones and \$125 million in sales milestones. If commercialized, the Company is eligible to receive tiered low double-digit royalties on net global sales outside the U.S.

*XmAb18087* is our third CD3 bispecific oncology candidate and we are beginning its development in the second half of 2016. XmAb18087 targets the Somatostatin Receptor 2 (SSTR2) and the cytotoxic T-cell binding domain CD3 (CD3) for the treatment of neuroendocrine tumors.

*XmAb20717* is our initial checkpoint inhibitor candidate that is being developed using our bispecific technology platform. XmAb20717 targets PD-1 and CTLA-4 and is being developed for broad oncology indications including solid tumors.

# **Out-Licensed Compounds**

In addition to our wholly-owned compounds in clinical development, we have used our XmAb technology to create antibody compounds which have been licensed to other pharmaceutical and biotechnology companies for further development. These licensed compounds do not require additional development effort by us as they advance into development by our partners. If successful, these candidates will generate additional milestone payments and royalties to

support our internal development efforts. These include XmAb5574/MOR208 (now MOR208) licensed to MorphoSys AG (MorphoSys), and XmAb13551, a bispecific CD38 x CD3 preclinical candidate, which we developed and licensed to Amgen.

			Primar	-	
Program	Target	Fc Domain	Indication	Development	Partner
XmAb5574/MOR208	CD19	Cytotoxic	CLL/NHL/ALL	Phase 2	Morphosys
XmAb13551	CD38 x CD3	Bispecific	Myeloma	Preclinical	Amgen
XmAb14045	CD123 x CD3	Bispecific	AML	Clinical	Novartis*
XmAb13676	CD20 x CD3	Bispecific	B-cell malignancies	Preclinical	Novartis*

<sup>\*</sup> In connection with our Novartis Agreement, we licensed XmAb14045 and XmAb13676 to Novartis for commercialization in all territories outside the US. We will co-develop both these candidates with Novartis worldwide.

# **Our Out-Licensed Technology**

We selectively license our XmAb technology to other companies for use in their own internal development candidates and to potentially make next-generation improvements to their marketed products. These licenses generally require little or no development effort by us and provide us with cash to fund our own research and development programs. These agreements typically provide the licensee with specific rights to use one or more of our Fc technologies to be applied to their proprietary antibodies or targets. The licensee is generally responsible for all development, of any resulting product candidate. As part of these agreements, we are generally entitled to receive upfront fees, annual licensing fees, potential milestone payments and royalties on the sales of any resulting products. In connection with our collaboration with Novo Nordisk, we also received research and development funding.

There are currently eight programs in development with our partners. The most advanced programs are with Alexion and CSL-Janssen, which both entered into Phase 2 clinical development in 2015.

						Current
		Xencor				Development
Licensee	Year	Technology	Indication	Milestones	Royalties	Stage
Alexion	2013	Xtend	Undisclosed	Yes	Yes	Phase 2
CSL-Janssen	2009	Cytotoxic	Oncology	Yes	Yes	Phase 2
						Phase 1
Boehringer Ingelheim	2007	Cytotoxic	Oncology	Yes	Yes	(2 candidates)
Janssen	2009	Xtend	Autoimmune disease	Yes	Yes	Preclinical
NIH (not licensed)		Xtend	HIV	N/A	N/A	Phase 1
Merck	2013	Fc optimization	Autoimmune disease	Yes	Yes	Phase 1
		Various, including				
Novo Nordisk	2014	Bi-specifics	Undisclosed	Yes	Yes	Preclinical
						5 Preclinical
Amgen	2015	Bi-specific	Oncology/Autoimmune	Yes	Yes	candidates
		Various, including				
Novartis	2016	Bi-specifics	Undisclosed	Yes	Yes	

# **Results of Operations**

# Comparison of the Three Months Ended June 30, 2016 and 2015

The following table summarizes our results of operations for the three months ended June 30, 2016 and 2015 (in millions):

		Three Months Ended						
	June 30,							
		2016	2015			Change		
Revenues:								
Research collaboration	\$	7.2	\$	0.7	\$	6.5		
Licensing		58.8		0.3		58.5		
Total revenues	\$	66.0	\$	1.0	\$	65.0		
Operating expenses:								
Research and development		14.4		7.5		6.9		
General and administrative		3.0		2.5		0.5		
Total operating expenses		17.5		10.0		7.5		
Other income, net		0.4		0.1		0.3		
Income (loss) before taxes		48.9		(8.9)		57.8		
Income tax provision		1.7		_		1.7		
Net income (loss)	\$	47.2	\$	(8.9)	\$	56.2		

## Revenues

Research collaboration revenues increased by \$6.5 million in the three months ended June 30, 2016 over 2015 amounts primarily due to revenue recognized under our 2015 collaboration agreement with Amgen.

Licensing revenues were \$58.5 million higher during the three months ended June 30, 2016 over 2015 primarily due to revenue recognized under our Novartis Agreement.

# Research and Development Expenses

The following table summarizes our research and development expenses for the three months ended June 30, 2016 and 2015 (in millions):

Three Months Ended

		Till ee Months Ended							
		June 30,							
	2	016		2015	C	hange			
Product programs:									
XmAb5871	\$	4.5	\$	1.7	\$	2.8			
XmAb7195		2.2		1.1		1.1			
Bi-specific		7.0		4.2		2.8			
Early research and discovery		0.7		0.5		0.2			
Total research and development expenses	\$	14.4	\$	7.5	\$	6.9			

Research and development expenses increased by \$6.9 million for the three months ended June 30, 2016 over the same period in 2015. Spending on the XmAb5871and bispecific programs increased during the three months ended June 30, 2016 compared to the same period in 2015. The \$2.8 million increase in spending associated with the XmAb5871 program is primarily due to expenses related to the initiation of the clinical trials in IgG4-RD and SLE. There was increased spending of \$2.8 million in the three months ended June 30, 2016 on our bispecific programs as we advanced our initial bispecific candidates, XmAb14045 and XmAb13676, toward clinical development and conducted additional work on our bispecific platform and other preclinical programs. The increased spending of \$1.1 million in the three months ended June 30, 2016 on XmAb7195 is primarily due to expenses related to drug manufacturing.

# General and Administrative Expenses

The following table summarizes our general and administrative expenses for the three months ended June 30, 2016 and 2015 (in millions):

## **Three Months Ended**

	June 30,						
	2	016	2	2015	CI	nange	
General and administrative	\$	3.0	\$	2.5	\$	0.5	

General and administrative expenses increased by \$0.5 million for the three months ended June 30, 2016 over the same period in 2015. The increase is primarily due to an increase in stock-based compensation costs.

## Other Income, Net

Other income, net was \$358,000 for the three months ended June 30, 2016 compared to \$117,000 for the same period in 2015 reflecting interest income on our investment in marketable securities.

# Income Tax Provision

The provision for income taxes was \$1.7 million for the three months ended June 30, 2016, compared to zero in 2015. The provision for income taxes for the second quarter of 2016 is a result of the interim period tax allocation of the federal and state alternative minimum tax based on the projected year end effective income tax rates which cannot be offset by the Company's net operating loss carryforwards.

# Comparison of the Six Months Ended June 30, 2016 and 2015

The following table summarizes our results of operations for the six months ended June 30, 2016 and 2015 (in millions):

### Six Months Ended

	June 30,						
		2016		2015		Change	
Revenues:							
Research collaboration	\$	14.1	\$	1.4	\$	12.7	
Licensing		59.1		0.6		58.5	
Milestone		_		0.5		(0.5)	
Total revenues	\$	73.2	\$	2.5	\$	70.7	
Operating expenses:							
Research and development		24.4		12.7		11.7	
General and administrative		7.0		5.3		1.7	
Total operating expenses		31.4		18.0		13.4	
Other income, net		0.7		0.2		0.5	
Income (loss) before taxes		42.5		(15.3)		57.8	
Income tax provision		1.7		_		1.7	
Net income (loss)	\$	40.8	\$	(15.3)	\$	56.1	

# Revenues

Research collaboration revenues for the six months ended June 30, 2016 increased by \$12.7 million over the same period in 2015 primarily due to revenue recognized under our Amgen Agreement.

Licensing revenues for the six months ended June 30, 2016 increased by \$58.5 million over the same period in 2015 primarily due to revenue recognized from our Novartis Agreement.

There were no milestone revenues for the six months ended June 30, 2016. Milestone revenues for the same period in 2015 were from Alexion.

# Research and Development Expenses

The following table summarizes our research and development expenses for the six months ended June 30, 2016 and 2015 (in millions):

		Six Months Ended							
	June 30,								
		2016		2015		Change			
Product programs:									
XmAb5871	\$	8.0	\$	2.9	\$	5.1			
XmAb7195		3.2		2.6		0.6			
Bi-specific		11.9		6.2		5.7			
Early research and discovery		1.3		1.0		0.3			
Total research and development expenses	\$	24.4	\$	12.7	\$	11.7			

Research and development expenses increased \$11.7 million for the six months ended June 30, 2016 over the same period in 2015. Spending on XmAb5871 increased by \$5.1 million primarily due to expenses related to the initiation of the clinical trials in IgG4-RD and SLE. Spending on our bispecific programs increased by \$5.7 million primarily due to expenses incurred advancing our XmAb14045 and XmAb13676 candidates towards clinical development as well as conducting additional work on our bispecific platform and other preclinical programs.

# General and Administrative Expenses

The following table summarizes our general and administrative expenses for the six months ended June 30, 2016 and 2015 (in millions):

	Six Months Ended						
	June 30,						
	2016		2015			Change	
General and administrative	\$	7.0	\$	5.3	\$	1.7	

General and administrative expenses increased by \$1.7 million for the six months ended June 30, 2016 over the same period in 2015. The increase is primarily due to an increase in stock-based compensation costs.

# Other Income, Net

Other income, net was \$693,000 for the six months ended June 30, 2016 compared to \$152,000 for the same period in 2015 reflecting interest income on our investment in marketable securities.

# Income Tax Provision

The provision for income taxes was \$1.7 million for the three months ended June 30, 2016, compared to zero in 2015. The provision for income taxes for six months ended June 30, 2016 is a result of the interim period tax allocation of the federal and state alternative minimum tax based on the projected year end effective income tax rates which cannot be offset by the Company's net operating loss carryforwards.

# Cash Flows

The following table sets forth the primary sources and uses of cash for each of the periods presented below (in thousands):

		Six Months Ended June 30,					
		2016		2015		Change	
Net cash provided by (used in):							
Operating activities		\$ (23,669)	\$	(9,009)	\$	(14,660)	
Investing activities		18,283		(150,349)		168,632	
	27						

Financing activities	673	115,880	(115,207)
Net decrease in cash	\$ (4,713)	\$ (43,478)	\$ 38,765

# **Operating Activities**

Cash used in operating activities for the six months ended June 30, 2016 increased by \$14.7 million over the same period in 2015 reflecting the increase in spending for research and development activities for our 5871 clinical programs and spending on our bispecific programs.

# **Investing Activities**

Investing activities consist primarily of investments in marketable securities available-for-sale, purchases of intangible assets, capitalization of patent and licensing costs and, purchases of property and equipment. Net cash provided by investing activities increased by \$169 million for the six months ended June 30, 2016 over the same period in 2015 primarily from proceeds from the sale or maturities of our marketable securities.

# Financing Activities

Net cash provided by financing activities consist primarily of net proceeds from the sale of common stock and from the issuance of common stock upon exercise of stock awards. Net financing proceeds decreased by \$115.2 million during the six months ended June 30, 2016 compared to the same period in 2015 due to the \$115.2 million received from our follow-on financing in March 2015.

# **Liquidity and Capital Resources**

We have financed our operations primarily through private placements of our equity and convertible notes, the public offerings of our common stock, and payments received under our product development partnerships and licensing arrangements.

On March 3, 2015, we finalized the sale of 8,625,000 shares of common stock at an offering price of \$14.25 per share, resulting in net proceeds of approximately \$115.2 million, after deducting underwriting discounts, commissions and offering expenses. In September 2015 we received a \$45 million upfront payment in connection with our 2015 Amgen transaction.

At June 30, 2016, we had \$168.8 million of cash, cash equivalents and marketable securities. In July 2016 we received a \$150 million upfront payment from Novartis in connection with our Novartis Agreement. We expect to continue to receive additional payments from our collaborators for research and development services rendered, additional milestone, contingent payments, opt-in and annual license maintenance payments. Our ability to receive milestone payments and contingent payments from our partners is dependent upon either our ability or our partners' abilities to achieve certain levels of research and development activities and is therefore uncertain at this time.

# **Funding Requirements**

We have not generated any revenue from product sales to date and do not expect to do so until such time as we obtain regulatory approval of and commercialize one or more of our product candidates. As we are currently in clinical stage of development, it will be some time before we expect to achieve this and it is uncertain that we ever will commercialize one or more of our product candidates. We expect that we will continue to increase our operating expenses in connection with ongoing as well as additional clinical and pre-clinical development of product candidates in our pipeline.

Although it is difficult to predict our funding requirements, we expect that our existing cash, cash equivalents and marketable securities and certain potential milestone and contingent contractual payments will fund our operating expenses and capital expenditure requirements at least through the end of 2019. We have based these estimates on assumptions that may prove to be wrong, and we could use our capital resources sooner than we currently expect. Additionally, the process of testing product candidates in clinical trials is costly, and the timing of progress in these trials is uncertain. Because our product candidates are in various stages of development and the outcome of these efforts is

uncertain, we cannot estimate the actual amounts necessary to successfully complete the development and commercialization of our product candidates or whether, or when, we may achieve profitability.

# **Off-Balance Sheet Arrangements**

We did not have during the periods presented, and we do not currently have, any off-balance sheet arrangements.

# **Critical Accounting Policies**

For a discussion on our material changes in critical accounting policies, see "Recent Accounting Pronouncements" in the notes to the financial statements included in this quarterly report on form 10-Q.

# ITEM 3. Quantitative and Qualitative Disclosures about Market Risk

Our primary objective when considering our investment activities is to preserve capital in order to fund our operations. Our primary exposure to market risk is related to changes in interest rates. Our current investment policy is to invest principally in deposits and securities issued by the U.S. government and its agencies, government sponsored agency debt obligations, corporate debt obligations and money market instruments. As of June 30, 2016 we had cash and cash equivalents and marketable securities of \$168.8 million consisting of bank deposits, interest-bearing money market accounts, and US government and corporate securities. Our primary exposure to market risk is interest rate sensitivity, which is affected by changes in the general level of U.S. interest rates. Due to the short-term maturities of our cash equivalents and marketable securities and the conservative risk profile of our marketable securities, a substantial change in interest rates would not have a material effect on the fair market value of our cash equivalents and marketable securities. We have the ability to hold our marketable securities until maturity, and we therefore do not expect a change in interest rates to affect our operating results or cash flows to any significant degree.

# ITEM 4. Controls and Procedures

# **Disclosure Controls and Procedures**

Our management, including our principal executive and principal financial officers, has evaluated the effectiveness of our disclosure controls and procedures as of June 30, 2016. Our disclosure controls and procedures are designed to provide reasonable assurance that the information required to be disclosed in this Quarterly Report on Form 10-Q has been appropriately recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms, and that such information is accumulated and communicated to our management, including our principal executive and principal financial officers, to allow timely decisions regarding required disclosure. Based on that evaluation, our principal executive and principal financial officers have concluded that our disclosure controls and procedures are effective at the reasonable assurance level as of June 30, 2016.

# **Changes in Internal Control**

There have been no changes in our internal control over financial reporting during our most recent fiscal quarter that have materially affected or are reasonably likely to materially affect our internal control over financial reporting.

# PART II — OTHER INFORMATION

# Item 1. Legal Proceedings.

On March 3, 2015, a verified class action complaint, captioned DePinto v. John S. Stafford, et al., C.A. No. 10742, was filed in the Court of Chancery of the State of Delaware against certain of the Company's current and former directors alleging cause of action for Breach of Fiduciary Duty and Invalidity of Director and Stockholder Consents. In general, the complaint alleged that the plaintiff and the class he seeks to represent were shareholders of the Company during the recapitalization and certain related transactions that the Company underwent in 2013 and that the

defendants breached their fiduciary duties in the course of approving that series of transactions. It also challenged as invalid certain corporate acts taken in the 2013 time period. On June 10, 2015, the Company filed a Verified Petition for Relief under Del. C. Section 205 (the 205 Petition) related to the corporate acts challenged in the complaint. The defendants filed an answer to the class action complaint on June 22, 2015. On July 9, 2015, the Court consolidated the 205 Petition with the class action, joined the Company as a defendant and ordered it to file the claims in the 205 Petition as counter-claims in the class action, which the Company has done.

On August 11, 2015, the Company filed a Motion for leave to File an Amended Counter-Claim, along with the proposed Amended Counter-Claim and related documents. On October 5, 2015, the parties filed a Stipulation of Partial Settlement and related documents disclosing a settlement of the invalidity claims addressed in the complaint, the counter-claim and the proposed amended counter-claim including a request by plaintiff's counsel for reimbursement of legal fees up to \$950,000. On October 7, 2015, Xencor filed the Amended Counter-Claim and the related documents. On December 14, 2015, the Court entered an Order and Partial Final Judgment approving the settlement of the invalidity claims, validating each corporate act challenged in the complaint, dismissing with prejudice Count II of the complaint (the invalidity claims) and granting plaintiff's counsel a fee award. We have paid the plaintiff's legal award cost of \$950,000 net of insurance proceeds of \$187,500 which has been reflected as a charge in our 2015 operations.

Based on the nature of the claim, the Company believes that it is not possible to estimate the likelihood of loss or a range of potential loss related to the claim; accordingly, no amount for any loss has been accrued at June 30, 2016.

# Item 1A. Risk Factors

The following factors, which supplement or update the risk factors set forth in "Risk Factors" in Item 1A of our Annual Report on Form 10-K for the year ended December 31, 2015 may affect our future financial condition and results of operations. The risk described below is not the only risk we face. See also "Special Note Regarding Forward-Looking Statements" included in this Quarterly Report on Form 10-Q. In addition to the risks set forth in our Annual Report on Form 10-K for the year ended December 31, 2015, additional risks and uncertainties not currently known to us or that we currently deem to be immaterial may also materially and adversely affect our business.

We are obligated to develop and maintain proper and effective internal control over financial reporting. If these internal controls are determined not to be effective, investor confidence in our company may be adversely affected and, as a result, the value of our common stock.

We regularly review and update our internal controls, disclosure controls and procedures, and corporate governance policies. We are required under the Sarbanes-Oxley Act of 2002 to report annually on our internal control over financial reporting, but as an emerging growth company we have been exempt from the requirement to have our independent accountants attest to our internal control over financial reporting. As of December 31, 2016, we will no longer qualify as an emerging growth company. As a result, our independent registered public accounting firm will be required to issue an attestation report on the effectiveness of our internal control over financial reporting. We are in the process of determining whether our existing internal controls over financial reporting systems are compliant with Section 404. This process requires the investment of substantial time and resources, including by members of our senior management, and may divert internal resources and take a significant amount of time and effort to complete. In addition, even if our management concludes that our internal control over financial reporting is effective, our independent registered public accounting firm may conclude that there are material weaknesses or significant deficiencies with respect to our internal controls or the level at which our controls are documented, designed, implemented or reviewed.

If it were to be determined that our internal control over financial reporting is not effective, such a shortcoming could result in an adverse reaction in the marketplace due to a loss of investor confidence in the reliability of our financial statements, which ultimately could negatively affect the market price of our shares, increase the volatility of our stock price and adversely affect our ability to raise additional funding.

# Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Recent Sales of Unregistered Securities

# None

Use of Proceeds from Registered Securities

On December 3, 2013, we completed our IPO and issued 14,639,500 shares of our common stock at \$5.50 per share, which included shares we issued pursuant to our underwriters' exercise of their over-allotment option, and received net proceeds of \$72.5 million, after underwriting discounts, commissions and estimated offering expenses. None of the expenses associated with the IPO were paid to directors, officers, persons owning 10% or more of any class of equity securities, or to their associates, or to our affiliates.

Shares of our common stock began trading on the NASDAQ Global Market on December 3, 2013. The shares were registered under the Securities Act on registration statements on Form S-1 (Registration No. 333-191689) effective as of December 2, 2013.

We are using the proceeds from the IPO to fund research and development activities and for working capital and general corporate purposes. We described the planned use of proceeds from our IPO in our prospectus dated December 2, 2013, filed with the SEC pursuant to Rule 424(b)(4) under the Securities Act of 1933, as amended, including using a portion of such proceeds for a planned Phase 2b clinical trial with XmAb5871. In October 2014, we announced that we would not be pursuing a Phase 2b clinical trial of XmAb5871 in RA and would initiate clinical development of XmAb5871 in IgG4-RD and possibly other autoimmune diseases. In 2016 we initiated the Phase 1 clinical trials with XmAb5871 in IgG4-RD and Lupus. As of June 30, 2016, we have used all of the funds from the IPO.

Item 6.	Exhibits
3.1	Amended and Restated Certificate of Incorporation of the Company (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed with the SEC on December 11, 2013).
3.2	Amended and Restated Bylaws of the Company (incorporated by reference to Exhibit 3.2 to the Company's Current Report on Form 8-K, filed with the SEC on December 11, 2013).
4.1	Form of Common Stock Certificate of the Company (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement on Form S-1, as amended (File No. 333-191689), originally filed with the SEC on October 25, 2013).
4.2	Third Amended and Restated Investor Rights Agreement, dated September 26, 2013, among the Company and certain of its stockholders incorporated by reference to Exhibit 4.2 to the Company's Registration Statement on Form S-1, as amended (File No. 333-191689), originally filed with the SEC on October 11, 2013).
10.1*	Severance Agreement dated May 26, 2016 by and between the Company and Bassil Dahiyat.
10.2*	Severance Agreement dated May 26, 2016 by and between the Company and John Kuch.
10.3*	Severance Agreement dated May 26, 2016 by and between the Company and John Desjarlais.
10.4*	Severance Agreement dated May 26, 2016 by and between the Company and Lloyd Rowland.
10.5*	Severance Agreement dated July 29, 2016 by and between the Company and Edgardo Baracchini
10.6**	Collaboration and License Agreement by and between the Company and Novartis Institutes for BioMedical Research, Inc. dated June 26, 2016.
31.1	Rule 13a-14(a) Certification of Principal Executive Officer.
31.2	Rule 13a-14(a) Certification of Principal Financial Officer.
32.1	Section 1350 Certification of Principal Executive Officer and Principal Financial Officer.
101.INS	XBRL Instance Document
101.SCH	XBRL Schema Document
101.CAL	XBRL Calculation Linkbase Document
101.DEF	XBRL Definition Linkbase Document
101.LAB	XBRL Labels Linkbase Document
101.PRE	XBRL Presentation Linkbase Document

Indicates management contract or compensatory plan.
 \*\* Confidential treatment has been requested with respect to certain portions of this exhibit. Omitted portions have been filed separately with the Security and Exchange Commission.

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

XENCOR, INC.

BY: /s/ BASSIL I. DAHIYAT

Bassil I. Dahiyat, Ph.D.

President and Chief Executive Officer

(Principal Executive Officer)

BY: /s/ JOHN J. KUCH

John J. Kuch

Vice President, Finance (Principal Financial Officer)

Dated: August 2, 2016

# EXHIBIT INDEX

3.1		Amended and Restated Certificate of Incorporation of the Company (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed with the SEC on December 11, 2013).
3.2		Amended and Restated Bylaws of the Company (incorporated by reference to Exhibit 3.2 to the Company's Current Report on Form 8-K, filed with the SEC on December 11, 2013).
4.1		Form of Common Stock Certificate of the Company (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement on Form S-1, as amended (File No. 333-191689), originally filed with the SEC on October 25, 2013).
4.2*		Third Amended and Restated Investor Rights Agreement, dated June 26, 2013, among the Company and certain of its stockholders incorporated by reference to Exhibit 4.2 to the Company's Registration Statement on Form S-1, as amended (File No. 333-191689), originally filed with the SEC on October 11, 2013).
10.1*		Severance Agreement dated May 26, 2016 by and between the Company and Bassil Dahiyat.
10.2*		Severance Agreement dated May 26, 2016 by and between the Company and John Kuch.
10.3*		Severance Agreement dated May 26, 2016 between the Company and John Desjarlais.
10.4*	Coverage Agreement dated July 20, 2016 by and be	Severance Agreement dated May 26, 2016 by and between the Company and Lloyd Rowland.
10.5*	Severance Agreement dated July 29, 2016 by and be	tween the Company and Edgardo Baracchini
10.6**	Collaboration and License Agreement by and betwee Research, Inc. dated June 26, 2016.	en the Company and Novartis Institutes for BioMedical
31.1		Rule 13a-14(a) Certification of Principal Executive Officer.
31.2		Rule 13a-14(a) Certification of Principal Financial Officer.
32.1		Section 1350 Certification of Principal Executive Officer and Principal Financial Officer.
101.INS		XBRL Instance Document
101.SCH		XBRL Schema Document
101.CAL		XBRL Calculation Linkbase Document
101.DEF		XBRL Definition Linkbase Document
101.LAB		XBRL Labels Linkbase Document
101.PRE		XBRL Presentation Linkbase Document

Indicates management contract or compensatory plan.
 \*\* Confidential treatment has been requested with respect to certain portions of this exhibit. Omitted portions have been filed separately with the Security and Exchange Commission.

# XENCOR, INC.

# SEVERANCE AGREEMENT

This Severance Agreement (the "*Agreement*") is dated as of May 26, 2016 (the "*Effective Date*"), by and between Bassil I. Dahiyat, Ph.D. ("*Executive*") and Xencor, Inc., a Delaware corporation (the "*Company*"). This Agreement is intended to provide Executive with certain severance and change in control benefits as described herein.

# **RECITALS**

- A. The Company and Executive are parties that certain Third Amended and Restated Executive Employment Agreement, dated September 4, 2013, by and between the Company and Executive (the "Employment Agreement"), which provides for certain severance benefits upon specific termination events.
- C. The Company's Board of Directors (the "*Board*") believes it is in the best interests of the Company and its stockholders to provide Executive with certain benefits upon a termination of Executive's employment under certain circumstances, including in connection with a Change in Control (as defined below), which benefits are intended to provide Executive with financial security and provide sufficient income and encouragement to Executive to remain with the Company, notwithstanding the possibility of a termination of Executive's employment with the Company.
- E. To accomplish the foregoing objectives, the Board desires to provide the opportunity for severance and change in control benefits to Executive on the terms provided in this Agreement, which shall supersede and replace the terms of the Employment Agreement with respect to Executive's severance and change in control benefits, as described in more detail in Section 9(b) of this Agreement.

Now therefore, in consideration of the mutual promises, covenants and agreements contained herein, and in consideration of the continuing employment of Executive by the Company, the parties hereto agree as follows:

- **1. Effectiveness and Term of Agreement.** This Agreement shall become effective as of the date indicated above, subject to execution by each of Executive and the Company. This Agreement shall remain in effect until it is terminated pursuant to its terms or until the Company has met all of its obligations under this Agreement following a termination of Executive's employment with the Company.
- **2. Definitions Used in This Agreement**. The terms used in this Agreement shall have the following meanings.
- **(a) Annual Base Salary.** "*Annual Base Salary*" means Executive's annual base salary in effect immediately prior to Executive's Involuntary Termination, ignoring any decrease that forms the basis for Executive's termination for Good Reason, if applicable.

- **(b) Cause**. "Cause" means Executive's:
- (i) gross negligence or willful misconduct in the performance of Executive's duties to the Company as an employee of the Company (other than a failure resulting from Executive's complete or partial incapacity due to physical or mental illness or impairment);
- **(ii)** material and willful violation of any federal or state law or regulation applicable to the business of the Company;
- (iii) refusal or failure to act in accordance with any lawful specific direction or order of the Board;
  - **(iv)** commission of any act of fraud with respect to the Company;
- **(v)** breach of any material provision of Executive's PIIA, including without limitation, Executive's theft or other misappropriation of the Company's proprietary information or trade secrets; or
- **(vi)** conviction of, or entry of plea of *nolo contendere* to, a felony or a crime involving moral turpitude.

Whether or not the actions or omissions of Executive constitute "Cause" shall be decided by the Board based upon a reasonable good faith investigation and determination.

- **(c) Change in Control.** "*Change in Control*" has the meaning set forth in the Company's 2013 Equity Incentive Plan.
- **(d) Change in Control Period.** "Change in Control Period" means the period of time beginning one (1) month prior to the execution of a definitive written agreement that if consummated in accordance with its terms would result in a Change in Control and ending on the earlier of (i) the termination of such agreement or (ii) twelve (12) months following the consummation of a Change in Control pursuant to such agreement.
- **(e) Code**. "*Code*" means the Internal Revenue Code of 1986, as amended, and the treasury regulations and other guidance promulgated thereunder.
- (f) Good Reason. "Good Reason" for Executive to terminate Executive's employment hereunder means the occurrence of any of the following events without Executive's consent; provided, however, that any resignation by Executive due to any of the following conditions shall only be deemed for Good Reason if: (i) Executive gives the Company written notice of the intent to terminate for Good Reason within sixty (60) days following the first occurrence of the condition(s) that Executive believes constitutes Good Reason, which notice shall describe such condition(s); (ii) the Company fails to remedy, if remediable, such condition(s) within thirty (30) days following receipt of the written notice (the "Cure Period") of such condition(s) from Executive; and (iii) Executive actually resigns his employment within the first fifteen (15) days after expiration of the Cure Period:
- **(i)** a material reduction in Executive's authority or job responsibilities as an employee of the Company or successor to the Company;

- **(ii)** a material reduction in Executive's annual base salary other than pursuant to a Company-wide reduction of annual base salaries for employees of the Company generally; or
- **(iii)** the relocation of the Executive's offices by a distance of fifty (50) miles or more, which relocation requires an increase in Executive's one-way driving distance by more than twenty-five (25) miles.
- **(g) Involuntary Termination**. "*Involuntary Termination*" means Executive's employment with the Company is terminated either (i) by the Company without Cause (and other than as a result of Executive's death or disability) or (ii) by Executive's resignation for Good Reason, and provided in either case such termination constitutes a Separation from Service.
- **(h) PIIA**. "*PIIA*" means the Proprietary Information and Inventions Agreement (or similar type agreement) between Executive and the Company.
- (i) Section 409A. "Section 409A" means Section 409A of the Code and any state law of similar effect.
- **(j) Separation from Service**. "*Separation from Service*" means a "separation from service", as defined under Treasury Regulation Section 1.409A-1(h).

## 3. Benefits Upon Termination of Employment.

- (a) General. If Executive's employment is terminated for any reason or no reason, the Company shall pay to Executive or to Executive's legal representatives, if applicable: (i) any base salary earned, but unpaid and (ii) any unreimbursed business expenses payable to Executive and any accrued but unused personal time off or vacation benefits and any other payments or benefits required by applicable law (collectively "Accrued Amounts"), which amounts shall be promptly paid in a lump sum to Executive, or in the case of Executive's death to Executive's estate. Other than the Accrued Amounts, Executive or Executive's legal representatives shall not be entitled to any additional compensation or benefits if Executive's employment is terminated for any reason other than by reason of Executive's Involuntary Termination as more fully provided below. If Executive's employment terminates due to an Involuntary Termination, Executive will be eligible to receive the compensation and benefits described in Sections 3(b) and 3(c), as applicable.
- **(b) Involuntary Termination.** Upon Executive's Involuntary Termination at any time, in addition to the Accrued Amounts, and provided that Executive signs and allows to become effective a release and waiver of claims substantially in the form attached hereto as **Exhibit A** (the "**Release**") within the time period provided therein (as further described in Section 4(a) below), then the Company shall provide Executive with the following severance benefits (the "**Severance Benefits**"):
- equivalent to the sum of (A) Executive's then-current Annual Base Salary for twenty-four (24) months and (B) an amount equal to Executive's then-current target bonus percentage of Executive's then-current Annual Base Salary as most recently established for Executive by the Board (or its compensation committee), assuming achievement of 100% of corporate and, as applicable, Executive's personal goals established for such year (or if no such goals have been established for such year, then the most recently established goals for a prior year) and without regard to any "stretch" or similar goals; provided that the portion of such payment that is payable under this clause (B) shall be pro rated based on the ratio that the number of days from the beginning of the calendar year in which such Involuntary Termination occurs through the date of Involuntary Termination bears to 365. The lump sum cash payment described in this Section 3(b)(i) shall be paid to Executive on the first

regular payroll date of the Company following the effective date of the Release, and in any event no later than March 15 of the year following the year in which the Involuntary Termination occurs.

- **(ii)** Accelerated Vesting of Stock Awards. The vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) as if Executive had completed an additional twenty-four (24) months of service with the Company as of the date of Executive's Involuntary Termination.
- Payment of Continued Group Health Plan Benefits. If Executive is eligible for (iii) and timely elects continued group health plan coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985 ("COBRA") following Executive's termination, the Company will pay the COBRA group health insurance premiums for Executive and Executive's eligible dependents until the earliest of (A) the close of the twelve (12) month period following the termination of Executive's employment (the "COBRA Payment Period"), (B) the expiration of Executive's eligibility for the continuation coverage under COBRA, or (C) the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment. References to COBRA premiums shall not include any amounts payable by Executive under a health care reimbursement plan pursuant to Section 125 of the Code. Notwithstanding the foregoing, if at any time the Company determines, in its sole discretion, that it cannot pay the COBRA premiums without potentially incurring financial costs or penalties under applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then regardless of whether Executive elects continued health coverage under COBRA, in lieu of providing the COBRA premiums, the Company will instead pay to Executive, on the last day of each remaining month of the COBRA Payment Period, a fully taxable cash payment equal to the COBRA premiums for that month, subject to applicable tax withholdings (such amount, the "Special Severance Payment"), which payments shall continue until the earlier of expiration of the COBRA Payment Period or the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment.
- **(c)** Additional Benefits Upon Involuntary Termination During the Change in Control Period. If Executive's Involuntary Termination occurs during the Change in Control Period, in addition to the Accrued Amounts and the Severance Benefits described in Section 3(b) above, Executive shall also be eligible for the following additional vesting acceleration benefit (collectively, with the Severance Benefits, the "Covered Severance Benefits"):
- (i) Full Accelerated Vesting of Stock Awards. Effective as of the later of Executive's Involuntary Termination or the effective date of the Change in Control, the vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) in full. Executive's stock awards shall remain outstanding following Executive's Involuntary Termination if and to the extent necessary to give effect to this Section 3(c)(i). For the avoidance of doubt, this vesting acceleration is conditioned upon the actual consummation of a Change in Control and in the event such Change in Control is not consummated, Executive shall receive the vesting acceleration benefits set forth in Section 3(b)(ii) above instead of the vesting acceleration benefits set forth in this Section 3(c)(i).

For the avoidance of doubt, the benefits set forth in this Section 3(c) shall in all events be subject to Executive's execution of an effective Release as described in Section 3(b) above.

## 4. Limitations and Conditions on Covered Severance Benefits.

- (a) No Covered Severance Benefits shall be payable until the effectiveness of the Release. The Board, in its sole discretion, may modify the form of the required Release to comply with applicable law and individual circumstance, and may incorporate the Release into a termination agreement or other agreement with Executive. In any case, the Release shall specifically relate to all of Executive's rights and claims in existence at the time of such execution and shall confirm Executive's continuing obligations to the Company (including but not limited to obligations under the PIIA and any confidentiality and/or non-solicitation agreement, as applicable, with the Company) and in all cases the Release by its terms must be signed and become effective no later than sixty (60) days following the Executive's Separation from Service.
- **(b)** All payments under this Agreement will be subject to applicable withholding for federal, state and local taxes. If Executive is indebted to the Company as of Executive's Involuntary Termination, the Company reserves the right to offset any Covered Severance Benefits by the amount of such indebtedness.
- **(c)** As a condition to receipt of any Covered Severance Benefits, Executive acknowledges and agrees to resign following termination of employment with the Company from all Company positions, including membership on the Board, if applicable, unless otherwise requested by the Company.
- **5. Successors.** Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company described in the foregoing sentence or any other successor who becomes bound by this Agreement by operation of law. The terms of this Agreement and all of Executive's rights hereunder and thereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.
- **6. At-Will Employment; No Employment Rights.** Executive acknowledges, affirms and agrees that Executive's employment with the Company is "at will," and may be terminated at any time and for any reason whatsoever by Executive or the Company, with or without Cause and with or without advance notice. This "at-will" employment relationship cannot be changed except in a writing signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board.

7. Section 409A. It is intended that all of the benefits and other payments payable under this Agreement satisfy, to the greatest extent possible, an exemption from the application of Section 409A, and this Agreement will be construed to the greatest extent possible as consistent with those provisions, and to the extent no so exempt, this Agreement (and any definitions hereunder) will be construed in a manner that complies with Section 409A, and any ambiguities herein shall be interpreted accordingly. Specifically, the benefits under this Agreement are intended to satisfy the exemptions from application of Section 409A provided under Treasury Regulations Sections 1.409A-1(b)(4), 1.409A-1(b)(5) and 1.409A-1(b)(9) and each installment of severance benefits, if any, is a separate "payment" for purposes of Treasury Regulations Section 1.409A-2(b)(2) (i). However, if such exemptions are not available and Executive is, upon Separation from Service, a "specified employee" for purposes of Section 409A, then, solely to the extent necessary to avoid adverse personal tax consequences under Section 409A, the timing of the severance benefits payments shall be delayed until the earlier of (i) six (6) months and one day after Executive's Separation from Service, or (ii) Executive's death. Severance benefits shall not commence until Executive has a Separation from Service. If the severance benefits are not covered by one or more exemptions from the application of Section 409A and the Release could become effective in the calendar year following the calendar year in which Executive's Separation from Service occurs, the Release will not be deemed effective, for purposes of payment of severance, any earlier than the first day of the second calendar year. Except to the minimum extent that payments must be delayed because Executive is a "specified employee" or until the effectiveness of the Release, all severance amounts will be paid as soon as practicable in accordance with this Agreement and the Company's normal payroll practices.

# 8. Parachute Payments.

- (a) If any payment or benefit Executive will or may receive from the Company or otherwise (a "280G Payment") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then any such 280G Payment (a "Payment") shall be equal to the Reduced Amount. The "Reduced Amount" shall be either (x) the largest portion of the Payment that would result in no portion of the Payment (after reduction) being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount (i.e., the amount determined by clause (x) or by clause (y)), after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in Executive's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in a Payment is required pursuant to the preceding sentence and the Reduced Amount is determined pursuant to clause (x) of the preceding sentence, the reduction shall occur in the manner (the "Reduction Method") that results in the greatest economic benefit for Executive. If more than one method of reduction will result in the same economic benefit, the items so reduced will be reduced pro rata (the "Pro Rata Reduction Method").
- (b) Notwithstanding the foregoing, if the Reduction Method or the Pro Rata Reduction Method would result in any portion of the Payment being subject to taxes pursuant to Section 409A that would not otherwise be subject to taxes pursuant to Section 409A, then the Reduction Method and/or the Pro Rata Reduction Method, as the case may be, shall be modified so as to avoid the imposition of taxes pursuant to Section 409A as follows: (A) as a first priority, the modification shall preserve to the greatest extent possible, the greatest economic benefit for Executive as determined on an after-tax basis; (B) as a second priority, Payments that are contingent on future events (e.g., being terminated without cause), shall be reduced (or eliminated) before Payments that are not contingent on future events; and (C) as a third priority, Payments that are "deferred compensation" within the meaning of Section 409A shall be reduced (or eliminated) before Payments that are not deferred compensation within the meaning of Section 409A.

- (c) Unless Executive and the Company agree on an alternative accounting firm or law firm, the accounting firm engaged by the Company for general tax compliance purposes as of the day prior to the effective date of the Change in Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change in Control, the Company shall appoint a nationally recognized accounting or law firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting or law firm required to be made hereunder. The Company shall use commercially reasonable efforts to cause the accounting or law firm engaged to make the determinations hereunder to provide its calculations, together with detailed supporting documentation, to Executive and the Company within fifteen (15) calendar days after the date on which Executive's right to a 280G Payment becomes reasonably likely to occur (if requested at that time by Executive or the Company) or such other time as requested by Executive or the Company.
- (d) If Executive receives a Payment for which the Reduced Amount was determined pursuant to clause (x) of Section 8(a) and the Internal Revenue Service determines thereafter that some portion of the Payment is subject to the Excise Tax, Executive agrees to promptly return to the Company a sufficient amount of the Payment (after reduction pursuant to clause (x) of Section 8(a) so that no portion of the remaining Payment is subject to the Excise Tax. For the avoidance of doubt, if the Reduced Amount was determined pursuant to clause (y) of Section 8(a), Executive shall have no obligation to return any portion of the Payment pursuant to the preceding sentence.

### 9. Miscellaneous Provisions.

- **(a) Waiver.** No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- **(b) Whole Agreement.** No agreements, representations or understandings (whether oral or written and whether express or implied) which are not expressly set forth in this Agreement have been made or entered into by either party with respect to the subject matter hereof. This Agreement supersedes the Employment Agreement, with respect to Sections 5.4, 5.5, 5.6(b), 5.7, 5.8, 5.9 and 5.10 thereof, and any other agreement (or portion thereof), representation or understanding (whether oral or written and whether express or implied) with respect to Executive's severance and change in control benefits, dated or made prior to the date of this Agreement, and by execution of this Agreement both parties agree that any such predecessor agreement (or applicable portion thereof), representation or understanding shall be deemed null and void.
- **(c) Choice of Law**. The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California without reference to conflict of laws provisions, and the parties hereto submit to the exclusive jurisdiction of the state and federal courts of the State of California.

- **(d) Severability.** If any term or provision of this Agreement or the application thereof to any circumstance shall, in any jurisdiction and to any extent, be invalid or unenforceable, such term or provision shall be ineffective as to such jurisdiction to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining terms and provisions of this Agreement or the application of such terms and provisions to circumstances other than those as to which it is held invalid or unenforceable, and a suitable and equitable term or provision shall be substituted therefor to carry out, insofar as may be valid and enforceable, the intent and purpose of the invalid or unenforceable term or provision.
- **Dispute Resolution**. To ensure the rapid and economical resolution of disputes that may arise in connection with Executive's employment with the Company, Executive and the Company agree that any and all disputes, claims, or causes of action, in law or equity, including but not limited to statutory claims, arising from or relating to the enforcement, breach, performance, or interpretation of this Agreement, Executive's employment with the Company, or the termination of Executive's employment from the Company, shall be resolved, to the fullest extent permitted by law, by final, binding and confidential arbitration conducted before a single arbitrator by JAMS, Inc. ("JAMS") or its successor, under JAMS' then applicable rules and procedures for employment disputes (which can be found at http://www.jamsadr.com/rules-clauses/, and which will be provided to Executive on request). The arbitration shall take place in the county (or comparable governmental unit) in which Executive was last employed by the Company, as determined by the arbitrator; provided that if the arbitrator determines there will be an undue hardship to Executive to have the arbitration in such location, the arbitrator will choose an alternative appropriate location. Executive and the Company each acknowledge that by agreeing to this arbitration procedure, they waive the right to resolve any such dispute through a trial by jury or judge or administrative proceeding. Executive will have the right to be represented by legal counsel at any arbitration proceeding. The arbitrator shall: (a) have the authority to compel adequate discovery for the resolution of the dispute and to award such relief as would otherwise be available under applicable law in a court proceeding; and (b) issue a written statement signed by the arbitrator regarding the disposition of each claim and the relief, if any, awarded as to each claim, the reasons for the award, and the arbitrator's essential findings and conclusions on which the award is based. The arbitrator, and not a court, shall also be authorized to determine whether the provisions of this section apply to a dispute, controversy, or claim sought to be resolved in accordance with these arbitration procedures. The Company shall pay all arbitration fees and costs in excess of the administrative fees that Executive would be required to incur if the dispute were filed or decided in a court of law. Nothing in this Agreement is intended to prevent either Executive or the Company from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration.
- **(f) Legal Fees and Expenses.** The parties shall each bear their own expenses, legal fees and other fees incurred in connection with the execution of this Agreement.
- **(g) No Assignment.** The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, including (without limitation) bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 9(g) shall be void.
- **(h) Counterparts**. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

IN WITNESS WHEREOF, each of the parties has executed this Agreement as of the Effective Date.

EXECUTIVE:

/S/
Name: Bassil I. Dahiyat, Ph.D.

COMPANY:

XENCOR, INC.

By:
/S/
Name: Lloyd Rowland

Title: SVP and General Counsel

### Exhibit A

### RELEASE AND WAIVER OF CLAIMS

In consideration of the receipt of benefits set forth in the Severance Agreement dated \_\_\_\_\_\_\_\_, 2016 (the "Agreement") to which this form is attached, I, \_\_\_\_\_\_\_\_, hereby furnish Xencor, Inc. and any and all affiliated, subsidiary, related, or successor corporations (collectively, the "Company"), with the following release and waiver ("Release and Waiver"). I understand that if I timely sign, date and return this Release and Waiver, and I do not revoke it, I will receive certain benefits pursuant to the terms and conditions of the Agreement. I understand that I am not entitled to such benefits unless I timely sign this Release and Waiver and allow it to become effective.

General Release and Waiver. In exchange for the consideration to be provided to me under the Agreement that I am not otherwise entitled to receive, I hereby generally and completely Release and Waiver, acquit and forever discharge the Company and its parent, subsidiary, and affiliated entities, and investors, along with its and their predecessors and successors and their respective directors, officers, employees, shareholders, partners, agents, attorneys, insurers, affiliates and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities and obligations, both known and unknown, that arise from or are in any way related to events, acts, conduct, or omissions occurring at any time prior to and including the date that I sign this Release and Waiver (collectively, the "Released Claims"). The Released Claims include, but are not limited to: (1) all claims arising out of or in any way related to my employment with the Company, or the termination of that employment; (2) all claims related to my compensation or benefits from the Company, including salary, bonuses, commissions, other incentive compensation, vacation pay and the redemption thereof, expense reimbursements, fringe benefits, stock, stock options, or any other ownership or equity interests in the Company; (3) all claims for breach of contract, wrongful termination, and breach of the implied covenant of good faith and fair dealing; (4) all tort claims, including but not limited to claims for fraud, defamation, emotional distress, and discharge in violation of public policy; and (5) all federal, state, and local statutory claims, including but not limited to claims for discrimination, harassment, retaliation, attorneys' fees, penalties, or other claims arising under the federal Civil Rights Act of 1964 (as amended), the federal Americans with Disabilities Act of 1990 (as amended), the federal Age Discrimination in Employment Act of 1967 (as amended) (the "ADEA"), the federal Family and Medical Leave Act ("FMLA"), the California Labor Code (as amended), and the California Fair Employment and Housing Act (as amended).

**Excluded Claims**. Notwithstanding the foregoing, the following are not included in the Released Claims (the "*Excluded Claims*"): (1) any rights or claims for indemnification I may have pursuant to any written indemnification agreement with the Company to which I am a party, the Company's bylaws, or applicable law; and (2) any rights which are not waivable as a matter of law. In addition, nothing in this Release and Waiver prevents me from filing, cooperating with, or participating in any investigation or proceeding before the Equal Employment Opportunity Commission, the Department of Labor, the California Department of Fair Employment and Housing, or any other government agency, except that I hereby waive my right to any monetary benefits in connection with any such claim, charge, investigation or proceeding. I hereby represent and warrant that, other than the Excluded Claims, I am not aware of any claims I have or might have against any of the Released Parties that are not included in the Released Claims.

**ADEA Waiver**. I acknowledge that I am knowingly and voluntarily waiving and releasing any rights I may have under the ADEA ("*ADEA Waiver*"). I also acknowledge that the consideration given for the ADEA Waiver is in addition to anything of value to which I was already entitled. I further acknowledge that I have

been advised by this writing, as required by the ADEA, that: (a) my ADEA Waiver does not apply to any rights or claims that arise after the date I sign this Release and Waiver; (b) I should consult with an attorney prior to signing this Release and Waiver; (c) I have twenty-one (21) days to consider this Release and Waiver (although I may choose to voluntarily sign it sooner); (d) I have seven (7) days following the date I sign this Release and Waiver to revoke the ADEA Waiver; and (e) the ADEA Waiver will not be effective until the date upon which the revocation period has expired unexercised, which will be the eighth day after I sign this Release and Waiver.

Section 1542 Waiver. In giving the general release herein, which includes claims which may be unknown to me at present, I acknowledge that I have read and understand Section 1542 of the California Civil Code, which reads as follows: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." I hereby expressly waive and relinquish all rights and benefits under that section and any law of any other jurisdiction of similar effect with respect to my release of claims, including but not limited to any unknown or unsuspected claims herein.

Other Agreements and Representations. I further agree: (a) not to disparage the Company, its officers, directors, employees, shareholders, and agents, in any manner likely to be harmful to its or their business, business reputations, or personal reputations; (b) not to voluntarily (except in response to legal compulsion) assist any third party in bringing or pursuing any proposed or pending litigation, arbitration, administrative claim or other formal proceeding against the Company, its parent or subsidiary entities, investors, affiliates, officers, directors, employees or agents; (c) to cooperate fully with the Company, by voluntarily (without legal compulsion) providing accurate and complete information, in connection with the Company's actual or contemplated defense, prosecution, or investigation of any claims or demands by or against third parties, or other matters, arising from events, acts, or failures to act that occurred during the period of my employment by the Company; and (d) I hereby acknowledge and reaffirm my continuing obligations under the terms of my Proprietary Information and Inventions Agreement. In addition, I hereby represent that I have been paid all wages earned owed and for all hours worked, I have received all the leave and leave benefits and protections for which I am eligible, pursuant to FMLA, the California Family Rights Act, or any applicable law or Company policy, and I have not suffered any on-the-job injury for which I have not already filed a workers' compensation claim.

I acknowledge my continuing obligations under my employee Proprietary Information and Inventions Agreement with the Company (the "*PIIA*").

This Release and Waiver attached to the Agreement as Exhibit A, along with the PIIA, constitutes the complete, final and exclusive embodiment of the entire agreement between the Company and me with regard to the subject matter hereof. I am not relying on any promise or representation by the Company that is not expressly stated herein. This Release and Waiver may only be modified by a writing signed by both me and a duly authorized officer of the Company.

# **UNDERSTOOD AND AGREED:**

NAME:			
DATE:			

### XENCOR, INC.

## SEVERANCE AGREEMENT

This Severance Agreement (the "*Agreement*") is dated as of May 26, 2016 (the "*Effective Date*"), by and between John J. Kuch ("*Executive*") and Xencor, Inc., a Delaware corporation (the "*Company*"). This Agreement is intended to provide Executive with certain severance and change in control benefits as described herein.

## **RECITALS**

- A. The Company and Executive are parties that certain Amended and Restated Change in Control Agreement, dated September 5, 2013, by and between the Company and Executive (the "*Prior Agreement*"), which provides for certain severance benefits upon specific termination events.
- C. The Company's Board of Directors (the "*Board*") believes it is in the best interests of the Company and its stockholders to provide Executive with certain benefits upon a termination of Executive's employment under certain circumstances, including in connection with a Change in Control (as defined below), which benefits are intended to provide Executive with financial security and provide sufficient income and encouragement to Executive to remain with the Company, notwithstanding the possibility of a termination of Executive's employment with the Company.
- E. To accomplish the foregoing objectives, the Board desires to provide the opportunity for severance and change in control benefits to Executive on the terms provided in this Agreement, which shall supersede and replace the Prior Agreement.

Now therefore, in consideration of the mutual promises, covenants and agreements contained herein, and in consideration of the continuing employment of Executive by the Company, the parties hereto agree as follows:

- **1. Effectiveness and Term of Agreement.** This Agreement shall become effective as of the date indicated above, subject to execution by each of Executive and the Company. This Agreement shall remain in effect until it is terminated pursuant to its terms or until the Company has met all of its obligations under this Agreement following a termination of Executive's employment with the Company.
- **2. Definitions Used in This Agreement**. The terms used in this Agreement shall have the following meanings.
- **(a) Annual Base Salary.** "*Annual Base Salary*" means Executive's annual base salary in effect immediately prior to Executive's Involuntary Termination, ignoring any decrease that forms the basis for Executive's termination for Good Reason, if applicable.
  - **(b) Cause**. "*Cause*" means Executive's:
- (i) gross negligence or willful misconduct in the performance of Executive's duties to the Company as an employee of the Company (other than a failure resulting from Executive's complete or partial incapacity due to physical or mental illness or impairment);
- **(ii)** material and willful violation of any federal or state law or regulation applicable to the business of the Company;

- (iii) refusal or failure to act in accordance with any lawful specific direction or order of the Board;
  - **(iv)** commission of any act of fraud with respect to the Company;
- **(v)** breach of any material provision of Executive's PIIA, including without limitation, Executive's theft or other misappropriation of the Company's proprietary information or trade secrets; or
- **(vi)** conviction of, or entry of plea of *nolo contendere* to, a felony or a crime involving moral turpitude.

Whether or not the actions or omissions of Executive constitute "Cause" shall be decided by the Board based upon a reasonable good faith investigation and determination.

- **(c) Change in Control.** "*Change in Control*" has the meaning set forth in the Company's 2013 Equity Incentive Plan.
- **(d) Change in Control Period.** "*Change in Control Period*" means the period of time beginning one (1) month prior to the execution of a definitive written agreement that if consummated in accordance with its terms would result in a Change in Control and ending on the earlier of (i) the termination of such agreement or (ii) twelve (12) months following the consummation of a Change in Control pursuant to such agreement.
- **(e) Code.** "*Code*" means the Internal Revenue Code of 1986, as amended, and the treasury regulations and other guidance promulgated thereunder.
- (f) Good Reason. "Good Reason" for Executive to terminate Executive's employment hereunder means the occurrence of any of the following events without Executive's consent; provided, however, that any resignation by Executive due to any of the following conditions shall only be deemed for Good Reason if: (i) Executive gives the Company written notice of the intent to terminate for Good Reason within sixty (60) days following the first occurrence of the condition(s) that Executive believes constitutes Good Reason, which notice shall describe such condition(s); (ii) the Company fails to remedy, if remediable, such condition(s) within thirty (30) days following receipt of the written notice (the "Cure Period") of such condition(s) from Executive; and (iii) Executive actually resigns his employment within the first fifteen (15) days after expiration of the Cure Period:
- (i) a material reduction in Executive's authority or job responsibilities as an employee of the Company or successor to the Company;
- (ii) a material reduction in Executive's annual base salary other than pursuant to a Company-wide reduction of annual base salaries for employees of the Company generally; or
- **(iii)** the relocation of the Executive's offices by a distance of fifty (50) miles or more, which relocation requires an increase in Executive's one-way driving distance by more than twenty-five (25) miles.
- **(g) Involuntary Termination**. *"Involuntary Termination"* means Executive's employment with the Company is terminated either (i) by the Company without Cause (and other than as a result of Executive's death or disability) or (ii) by Executive's resignation for Good Reason, and provided in either case such termination constitutes a Separation from Service.

- **(h) PIIA.** "*PIIA*" means the Proprietary Information and Inventions Agreement (or similar type agreement) between Executive and the Company.
- **(i) Section 409A.** "*Section 409A*" means Section 409A of the Code and any state law of similar effect.
- **(j) Separation from Service**. "*Separation from Service*" means a "separation from service", as defined under Treasury Regulation Section 1.409A-1(h).

## 3. Benefits Upon Termination of Employment.

- (a) General. If Executive's employment is terminated for any reason or no reason, the Company shall pay to Executive or to Executive's legal representatives, if applicable: (i) any base salary earned, but unpaid and (ii) any unreimbursed business expenses payable to Executive and any accrued but unused personal time off or vacation benefits and any other payments or benefits required by applicable law (collectively "Accrued Amounts"), which amounts shall be promptly paid in a lump sum to Executive, or in the case of Executive's death to Executive's estate. Other than the Accrued Amounts, Executive or Executive's legal representatives shall not be entitled to any additional compensation or benefits if Executive's employment is terminated for any reason other than by reason of Executive's Involuntary Termination as more fully provided below. If Executive's employment terminates due to an Involuntary Termination, Executive will be eligible to receive the compensation and benefits described in Sections 3(b) and 3(c), as applicable.
- **(b) Involuntary Termination.** Upon Executive's Involuntary Termination at any time, in addition to the Accrued Amounts, and provided that Executive signs and allows to become effective a release and waiver of claims substantially in the form attached hereto as **Exhibit A** (the "*Release*") within the time period provided therein (as further described in Section 4(a) below), then the Company shall provide Executive with the following severance benefits (the "*Severance Benefits*"):
- equivalent to the sum of (A) Executive's then-current Annual Base Salary for twelve (12) months and (B) an amount equal to Executive's then-current target bonus percentage of Executive's then-current Annual Base Salary as most recently established for Executive by the Board (or its compensation committee), assuming achievement of 100% of corporate and, as applicable, Executive's personal goals established for such year (or if no such goals have been established for such year, then the most recently established goals for a prior year) and without regard to any "stretch" or similar goals; provided that the portion of such payment that is payable under this clause (B) shall be pro rated based on the ratio that the number of days from the beginning of the calendar year in which such Involuntary Termination occurs through the date of Involuntary Termination bears to 365. The lump sum cash payment described in this Section 3(b)(i) shall be paid to Executive on the first regular payroll date of the Company following the effective date of the Release, and in any event no later than March 15 of the year following the year in which the Involuntary Termination occurs.
- (ii) Accelerated Vesting of Stock Awards. The vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) as if Executive had completed an additional twelve (12) months of service with the Company as of the date of Executive's Involuntary Termination.
- (iii) Payment of Continued Group Health Plan Benefits. If Executive is eligible for and timely elects continued group health plan coverage under the Consolidated Omnibus Budget

Reconciliation Act of 1985 ("COBRA") following Executive's termination, the Company will pay the COBRA group health insurance premiums for Executive and Executive's eligible dependents until the earliest of (A) the close of the twelve (12) month period following the termination of Executive's employment (the "COBRA Payment Period"), (B) the expiration of Executive's eligibility for the continuation coverage under COBRA, or (C) the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment. References to COBRA premiums shall not include any amounts payable by Executive under a health care reimbursement plan pursuant to Section 125 of the Code. Notwithstanding the foregoing, if at any time the Company determines, in its sole discretion, that it cannot pay the COBRA premiums without potentially incurring financial costs or penalties under applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then regardless of whether Executive elects continued health coverage under COBRA, in lieu of providing the COBRA premiums, the Company will instead pay to Executive, on the last day of each remaining month of the COBRA Payment Period, a fully taxable cash payment equal to the COBRA premiums for that month, subject to applicable tax withholdings (such amount, the "Special Severance Payment"), which payments shall continue until the earlier of expiration of the COBRA Payment Period or the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment.

- **(c)** Additional Benefits Upon Involuntary Termination During the Change in Control Period. If Executive's Involuntary Termination occurs during the Change in Control Period, in addition to the Accrued Amounts and the Severance Benefits described in Section 3(b) above, Executive shall also be eligible for the following additional vesting acceleration benefit (collectively, with the Severance Benefits, the "Covered Severance Benefits"):
- Executive's Involuntary Termination or the effective date of the Change in Control, the vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) in full. Executive's stock awards shall remain outstanding following Executive's Involuntary Termination if and to the extent necessary to give effect to this Section 3(c)(i). For the avoidance of doubt, this vesting acceleration is conditioned upon the actual consummation of a Change in Control and in the event such Change in Control is not consummated, Executive shall receive the vesting acceleration benefits set forth in Section 3(b)(ii) above instead of the vesting acceleration benefits set forth in this Section 3(c)(i).

For the avoidance of doubt, the benefits set forth in this Section 3(c) shall in all events be subject to Executive's execution of an effective Release as described in Section 3(b) above.

## 4. Limitations and Conditions on Covered Severance Benefits.

(a) No Covered Severance Benefits shall be payable until the effectiveness of the Release. The Board, in its sole discretion, may modify the form of the required Release to comply with applicable law and individual circumstance, and may incorporate the Release into a termination agreement or other agreement with Executive. In any case, the Release shall specifically relate to all of Executive's rights and claims in existence at the time of such execution and shall confirm Executive's continuing obligations to the Company (including but not limited to obligations under the PIIA and any confidentiality and/or non-solicitation agreement, as applicable, with the Company) and in all cases the Release by its terms must be signed and become effective no later than sixty (60) days following the Executive's Separation from Service.

- **(b)** All payments under this Agreement will be subject to applicable withholding for federal, state and local taxes. If Executive is indebted to the Company as of Executive's Involuntary Termination, the Company reserves the right to offset any Covered Severance Benefits by the amount of such indebtedness.
- **(c)** As a condition to receipt of any Covered Severance Benefits, Executive acknowledges and agrees to resign following termination of employment with the Company from all Company positions, including membership on the Board, if applicable, unless otherwise requested by the Company.
- **5. Successors.** Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company described in the foregoing sentence or any other successor who becomes bound by this Agreement by operation of law. The terms of this Agreement and all of Executive's rights hereunder and thereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.
- **6. At-Will Employment; No Employment Rights.** Executive acknowledges, affirms and agrees that Executive's employment with the Company is "at will," and may be terminated at any time and for any reason whatsoever by Executive or the Company, with or without Cause and with or without advance notice. This "at-will" employment relationship cannot be changed except in a writing signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board.
- 7. **Section 409A**. It is intended that all of the benefits and other payments payable under this Agreement satisfy, to the greatest extent possible, an exemption from the application of Section 409A, and this Agreement will be construed to the greatest extent possible as consistent with those provisions, and to the extent no so exempt, this Agreement (and any definitions hereunder) will be construed in a manner that complies with Section 409A, and any ambiguities herein shall be interpreted accordingly. Specifically, the benefits under this Agreement are intended to satisfy the exemptions from application of Section 409A provided under Treasury Regulations Sections 1.409A-1(b)(4), 1.409A-1(b)(5) and 1.409A-1(b)(9) and each installment of severance benefits, if any, is a separate "payment" for purposes of Treasury Regulations Section 1.409A-2(b)(2)(i). However, if such exemptions are not available and Executive is, upon Separation from Service, a "specified employee" for purposes of Section 409A, then, solely to the extent necessary to avoid adverse personal tax consequences under Section 409A, the timing of the severance benefits payments shall be delayed until the earlier of (i) six (6) months and one day after Executive's Separation from Service, or (ii) Executive's death. Severance benefits shall not commence until Executive has a Separation from Service. If the severance benefits are not covered by one or more exemptions from the application of Section 409A and the Release could become effective in the calendar year following the calendar year in which Executive's Separation from Service occurs, the Release will not be deemed effective, for purposes of payment of severance, any earlier than the first day of the second calendar year. Except to the minimum extent that payments must be delayed because Executive is a "specified employee" or until the effectiveness of the Release, all severance amounts will be paid as soon as practicable in accordance with this Agreement and the Company's normal payroll practices.

# 8. Parachute Payments.

**(a)** If any payment or benefit Executive will or may receive from the Company or otherwise (a "280G Payment") would (i) constitute a "parachute payment" within the meaning of Section

280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then any such 280G Payment (a "Payment") shall be equal to the Reduced Amount. The "Reduced Amount" shall be either (x) the largest portion of the Payment that would result in no portion of the Payment (after reduction) being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount (i.e., the amount determined by clause (x) or by clause (y)), after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in Executive's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in a Payment is required pursuant to the preceding sentence and the Reduced Amount is determined pursuant to clause (x) of the preceding sentence, the reduction shall occur in the manner (the "Reduction Method") that results in the greatest economic benefit for Executive. If more than one method of reduction will result in the same economic benefit, the items so reduced will be reduced pro rata (the "Pro Rata Reduction Method").

- **(b)** Notwithstanding the foregoing, if the Reduction Method or the Pro Rata Reduction Method would result in any portion of the Payment being subject to taxes pursuant to Section 409A that would not otherwise be subject to taxes pursuant to Section 409A, then the Reduction Method and/or the Pro Rata Reduction Method, as the case may be, shall be modified so as to avoid the imposition of taxes pursuant to Section 409A as follows: (A) as a first priority, the modification shall preserve to the greatest extent possible, the greatest economic benefit for Executive as determined on an after-tax basis; (B) as a second priority, Payments that are contingent on future events (e.g., being terminated without cause), shall be reduced (or eliminated) before Payments that are not contingent on future events; and (C) as a third priority, Payments that are "deferred compensation" within the meaning of Section 409A shall be reduced (or eliminated) before Payments that are not deferred compensation within the meaning of Section 409A.
- (c) Unless Executive and the Company agree on an alternative accounting firm or law firm, the accounting firm engaged by the Company for general tax compliance purposes as of the day prior to the effective date of the Change in Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change in Control, the Company shall appoint a nationally recognized accounting or law firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting or law firm required to be made hereunder. The Company shall use commercially reasonable efforts to cause the accounting or law firm engaged to make the determinations hereunder to provide its calculations, together with detailed supporting documentation, to Executive and the Company within fifteen (15) calendar days after the date on which Executive's right to a 280G Payment becomes reasonably likely to occur (if requested at that time by Executive or the Company) or such other time as requested by Executive or the Company.
- (d) If Executive receives a Payment for which the Reduced Amount was determined pursuant to clause (x) of Section 8(a) and the Internal Revenue Service determines thereafter that some portion of the Payment is subject to the Excise Tax, Executive agrees to promptly return to the Company a sufficient amount of the Payment (after reduction pursuant to clause (x) of Section 8(a) so that no portion of the remaining Payment is subject to the Excise Tax. For the avoidance of doubt, if the Reduced Amount was determined pursuant to clause (y) of Section 8(a), Executive shall have no obligation to return any portion of the Payment pursuant to the preceding sentence.

## 9. Miscellaneous Provisions.

**(a) Waiver**. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other

party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.

- **(b) Whole Agreement.** No agreements, representations or understandings (whether oral or written and whether express or implied) which are not expressly set forth in this Agreement have been made or entered into by either party with respect to the subject matter hereof. This Agreement supersedes the Prior Agreement in its entirety and any other agreement (or portion thereof), representation or understanding (whether oral or written and whether express or implied) with respect to Executive's severance and change in control benefits, dated or made prior to the date of this Agreement, and by execution of this Agreement both parties agree that any such predecessor agreement (or applicable portion thereof), representation or understanding shall be deemed null and void.
- **(c) Choice of Law.** The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California without reference to conflict of laws provisions, and the parties hereto submit to the exclusive jurisdiction of the state and federal courts of the State of California.
- **(d) Severability.** If any term or provision of this Agreement or the application thereof to any circumstance shall, in any jurisdiction and to any extent, be invalid or unenforceable, such term or provision shall be ineffective as to such jurisdiction to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining terms and provisions of this Agreement or the application of such terms and provisions to circumstances other than those as to which it is held invalid or unenforceable, and a suitable and equitable term or provision shall be substituted therefor to carry out, insofar as may be valid and enforceable, the intent and purpose of the invalid or unenforceable term or provision.
- **Dispute Resolution**. To ensure the rapid and economical resolution of disputes that may arise in connection with Executive's employment with the Company, Executive and the Company agree that any and all disputes, claims, or causes of action, in law or equity, including but not limited to statutory claims, arising from or relating to the enforcement, breach, performance, or interpretation of this Agreement, Executive's employment with the Company, or the termination of Executive's employment from the Company, shall be resolved, to the fullest extent permitted by law, by final, binding and confidential arbitration conducted before a single arbitrator by JAMS, Inc. ("JAMS") or its successor, under JAMS' then procedures for employment rules and disputes (which can http://www.jamsadr.com/rules-clauses/, and which will be provided to Executive on request). The arbitration shall take place in the county (or comparable governmental unit) in which Executive was last employed by the Company, as determined by the arbitrator; provided that if the arbitrator determines there will be an undue hardship to Executive to have the arbitration in such location, the arbitrator will choose an alternative appropriate location. Executive and the Company each acknowledge that by agreeing to this arbitration procedure, they waive the right to resolve any such dispute through a trial by jury or judge or administrative proceeding. Executive will have the right to be represented by legal counsel at any arbitration proceeding. The arbitrator shall: (a) have the authority to compel adequate discovery for the resolution of the dispute and to award such relief as would otherwise be available under applicable law in a court proceeding; and (b) issue a written statement signed by the arbitrator regarding the disposition of each claim and the relief, if any, awarded as to each claim, the reasons for the award, and the arbitrator's essential findings and conclusions on which the award is based. The arbitrator, and not a court, shall also be authorized to determine whether the provisions of this section apply to a dispute, controversy, or claim sought to be resolved in accordance with these arbitration procedures. The Company shall pay all arbitration fees and costs in excess of the administrative fees that Executive would be required to incur if the dispute were filed or decided in a court of law. Nothing in this Agreement is intended to prevent either Executive or the Company from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration.

- **(f) Legal Fees and Expenses**. The parties shall each bear their own expenses, legal fees and other fees incurred in connection with the execution of this Agreement.
- **(g) No Assignment**. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, including (without limitation) bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 9(g) shall be void.
- **(h) Counterparts**. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

In Witness Whereof, each of the parties has executed this Agreement as of the Effective Date.

EXECUTIVE:

/S/
Name: John J. Kuch

**COMPANY:** 

XENCOR, INC.

By: /S/

Name: Bassil Dahiyat

Title: President and CEO

### Exhibit A

### RELEASE AND WAIVER OF CLAIMS

In consideration of the receipt of benefits set forth in the Severance Agreement dated \_\_\_\_\_\_\_\_, 2016 (the "Agreement") to which this form is attached, I, \_\_\_\_\_\_\_\_, hereby furnish Xencor, Inc. and any and all affiliated, subsidiary, related, or successor corporations (collectively, the "Company"), with the following release and waiver ("Release and Waiver"). I understand that if I timely sign, date and return this Release and Waiver, and I do not revoke it, I will receive certain benefits pursuant to the terms and conditions of the Agreement. I understand that I am not entitled to such benefits unless I timely sign this Release and Waiver and allow it to become effective.

General Release and Waiver. In exchange for the consideration to be provided to me under the Agreement that I am not otherwise entitled to receive, I hereby generally and completely Release and Waiver, acquit and forever discharge the Company and its parent, subsidiary, and affiliated entities, and investors, along with its and their predecessors and successors and their respective directors, officers, employees, shareholders, partners, agents, attorneys, insurers, affiliates and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities and obligations, both known and unknown, that arise from or are in any way related to events, acts, conduct, or omissions occurring at any time prior to and including the date that I sign this Release and Waiver (collectively, the "Released Claims"). The Released Claims include, but are not limited to: (1) all claims arising out of or in any way related to my employment with the Company, or the termination of that employment; (2) all claims related to my compensation or benefits from the Company, including salary, bonuses, commissions, other incentive compensation, vacation pay and the redemption thereof, expense reimbursements, fringe benefits, stock, stock options, or any other ownership or equity interests in the Company; (3) all claims for breach of contract, wrongful termination, and breach of the implied covenant of good faith and fair dealing; (4) all tort claims, including but not limited to claims for fraud, defamation, emotional distress, and discharge in violation of public policy; and (5) all federal, state, and local statutory claims, including but not limited to claims for discrimination, harassment, retaliation, attorneys' fees, penalties, or other claims arising under the federal Civil Rights Act of 1964 (as amended), the federal Americans with Disabilities Act of 1990 (as amended), the federal Age Discrimination in Employment Act of 1967 (as amended) (the "ADEA"), the federal Family and Medical Leave Act ("FMLA"), the California Labor Code (as amended), and the California Fair Employment and Housing Act (as amended).

**Excluded Claims**. Notwithstanding the foregoing, the following are not included in the Released Claims (the "*Excluded Claims*"): (1) any rights or claims for indemnification I may have pursuant to any written indemnification agreement with the Company to which I am a party, the Company's bylaws, or applicable law; and (2) any rights which are not waivable as a matter of law. In addition, nothing in this Release and Waiver prevents me from filing, cooperating with, or participating in any investigation or proceeding before the Equal Employment Opportunity Commission, the Department of Labor, the California Department of Fair Employment and Housing, or any other government agency, except that I hereby waive my right to any monetary benefits in connection with any such claim, charge, investigation or proceeding. I hereby represent and warrant that, other than the Excluded Claims, I am not aware of any claims I have or might have against any of the Released Parties that are not included in the Released Claims.

**ADEA Waiver**. I acknowledge that I am knowingly and voluntarily waiving and releasing any rights I may have under the ADEA ("*ADEA Waiver*"). I also acknowledge that the consideration given for the ADEA Waiver is in addition to anything of value to which I was already entitled. I further acknowledge that I have been advised by this writing, as required by the ADEA, that: (a) my ADEA Waiver does not apply to any rights or claims that arise after the date I sign this Release and Waiver; (b) I should consult with an attorney prior to signing this Release and Waiver; (c) I have twenty-one (21) days to consider this

Release and Waiver (although I may choose to voluntarily sign it sooner); (d) I have seven (7) days following the date I sign this Release and Waiver to revoke the ADEA Waiver; and (e) the ADEA Waiver will not be effective until the date upon which the revocation period has expired unexercised, which will be the eighth day after I sign this Release and Waiver.

Section 1542 Waiver. In giving the general release herein, which includes claims which may be unknown to me at present, I acknowledge that I have read and understand Section 1542 of the California Civil Code, which reads as follows: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." I hereby expressly waive and relinquish all rights and benefits under that section and any law of any other jurisdiction of similar effect with respect to my release of claims, including but not limited to any unknown or unsuspected claims herein.

Other Agreements and Representations. I further agree: (a) not to disparage the Company, its officers, directors, employees, shareholders, and agents, in any manner likely to be harmful to its or their business, business reputations, or personal reputations; (b) not to voluntarily (except in response to legal compulsion) assist any third party in bringing or pursuing any proposed or pending litigation, arbitration, administrative claim or other formal proceeding against the Company, its parent or subsidiary entities, investors, affiliates, officers, directors, employees or agents; (c) to cooperate fully with the Company, by voluntarily (without legal compulsion) providing accurate and complete information, in connection with the Company's actual or contemplated defense, prosecution, or investigation of any claims or demands by or against third parties, or other matters, arising from events, acts, or failures to act that occurred during the period of my employment by the Company; and (d) I hereby acknowledge and reaffirm my continuing obligations under the terms of my Proprietary Information and Inventions Agreement. In addition, I hereby represent that I have been paid all wages earned owed and for all hours worked, I have received all the leave and leave benefits and protections for which I am eligible, pursuant to FMLA, the California Family Rights Act, or any applicable law or Company policy, and I have not suffered any on-the-job injury for which I have not already filed a workers' compensation claim.

I acknowledge my continuing obligations under my employee Proprietary Information and Inventions Agreement with the Company (the "*PIIA*").

This Release and Waiver attached to the Agreement as Exhibit A, along with the PIIA, constitutes the complete, final and exclusive embodiment of the entire agreement between the Company and me with regard to the subject matter hereof. I am not relying on any promise or representation by the Company that is not expressly stated herein. This Release and Waiver may only be modified by a writing signed by both me and a duly authorized officer of the Company.

Name:	
Date:	

**Understood and Agreed:** 

### XENCOR, INC.

## SEVERANCE AGREEMENT

This Severance Agreement (the "*Agreement*") is dated as of May 26, 2016 (the "*Effective Date*"), by and between John R. Desjarlais, Ph.D. ("*Executive*") and Xencor, Inc., a Delaware corporation (the "*Company*"). This Agreement is intended to provide Executive with certain severance and change in control benefits as described herein.

## **RECITALS**

- A. The Company and Executive are parties that certain Amended and Restated Severance Agreement, dated September 5, 2013, by and between the Company and Executive (the "*Prior Agreement*"), which provides for certain severance benefits upon specific termination events.
- C. The Company's Board of Directors (the "*Board*") believes it is in the best interests of the Company and its stockholders to provide Executive with certain benefits upon a termination of Executive's employment under certain circumstances, including in connection with a Change in Control (as defined below), which benefits are intended to provide Executive with financial security and provide sufficient income and encouragement to Executive to remain with the Company, notwithstanding the possibility of a termination of Executive's employment with the Company.
- E. To accomplish the foregoing objectives, the Board desires to provide the opportunity for severance and change in control benefits to Executive on the terms provided in this Agreement, which shall supersede and replace the Prior Agreement.

Now therefore, in consideration of the mutual promises, covenants and agreements contained herein, and in consideration of the continuing employment of Executive by the Company, the parties hereto agree as follows:

- **1. Effectiveness and Term of Agreement.** This Agreement shall become effective as of the date indicated above, subject to execution by each of Executive and the Company. This Agreement shall remain in effect until it is terminated pursuant to its terms or until the Company has met all of its obligations under this Agreement following a termination of Executive's employment with the Company.
- **2. Definitions Used in This Agreement**. The terms used in this Agreement shall have the following meanings.
- **(a) Annual Base Salary.** "*Annual Base Salary*" means Executive's annual base salary in effect immediately prior to Executive's Involuntary Termination, ignoring any decrease that forms the basis for Executive's termination for Good Reason, if applicable.
  - **(b) Cause**. "*Cause*" means Executive's:
- (i) gross negligence or willful misconduct in the performance of Executive's duties to the Company as an employee of the Company (other than a failure resulting from Executive's complete or partial incapacity due to physical or mental illness or impairment);
- (ii) material and willful violation of any federal or state law or regulation applicable to the business of the Company;

- (iii) refusal or failure to act in accordance with any lawful specific direction or order of the Board;
  - **(iv)** commission of any act of fraud with respect to the Company;
- **(v)** breach of any material provision of Executive's PIIA, including without limitation, Executive's theft or other misappropriation of the Company's proprietary information or trade secrets; or
- **(vi)** conviction of, or entry of plea of *nolo contendere* to, a felony or a crime involving moral turpitude.

Whether or not the actions or omissions of Executive constitute "Cause" shall be decided by the Board based upon a reasonable good faith investigation and determination.

- **(c) Change in Control.** "*Change in Control*" has the meaning set forth in the Company's 2013 Equity Incentive Plan.
- **(d) Change in Control Period.** "*Change in Control Period*" means the period of time beginning one (1) month prior to the execution of a definitive written agreement that if consummated in accordance with its terms would result in a Change in Control and ending on the earlier of (i) the termination of such agreement or (ii) twelve (12) months following the consummation of a Change in Control pursuant to such agreement.
- **(e) Code.** "*Code*" means the Internal Revenue Code of 1986, as amended, and the treasury regulations and other guidance promulgated thereunder.
- (f) Good Reason. "Good Reason" for Executive to terminate Executive's employment hereunder means the occurrence of any of the following events without Executive's consent; provided, however, that any resignation by Executive due to any of the following conditions shall only be deemed for Good Reason if: (i) Executive gives the Company written notice of the intent to terminate for Good Reason within sixty (60) days following the first occurrence of the condition(s) that Executive believes constitutes Good Reason, which notice shall describe such condition(s); (ii) the Company fails to remedy, if remediable, such condition(s) within thirty (30) days following receipt of the written notice (the "Cure Period") of such condition(s) from Executive; and (iii) Executive actually resigns his employment within the first fifteen (15) days after expiration of the Cure Period:
- (i) a material reduction in Executive's authority or job responsibilities as an employee of the Company or successor to the Company;
- (ii) a material reduction in Executive's annual base salary other than pursuant to a Company-wide reduction of annual base salaries for employees of the Company generally; or
- **(iii)** the relocation of the Executive's offices by a distance of fifty (50) miles or more, which relocation requires an increase in Executive's one-way driving distance by more than twenty-five (25) miles.
- **(g) Involuntary Termination**. *"Involuntary Termination"* means Executive's employment with the Company is terminated either (i) by the Company without Cause (and other than as a result of Executive's death or disability) or (ii) by Executive's resignation for Good Reason, and provided in either case such termination constitutes a Separation from Service.

- **(h) PIIA.** "*PIIA*" means the Proprietary Information and Inventions Agreement (or similar type agreement) between Executive and the Company.
- **(i) Section 409A.** "*Section 409A*" means Section 409A of the Code and any state law of similar effect.
- **(j) Separation from Service**. "*Separation from Service*" means a "separation from service", as defined under Treasury Regulation Section 1.409A-1(h).

# 3. Benefits Upon Termination of Employment.

- (a) General. If Executive's employment is terminated for any reason or no reason, the Company shall pay to Executive or to Executive's legal representatives, if applicable: (i) any base salary earned, but unpaid and (ii) any unreimbursed business expenses payable to Executive and any accrued but unused personal time off or vacation benefits and any other payments or benefits required by applicable law (collectively "Accrued Amounts"), which amounts shall be promptly paid in a lump sum to Executive, or in the case of Executive's death to Executive's estate. Other than the Accrued Amounts, Executive or Executive's legal representatives shall not be entitled to any additional compensation or benefits if Executive's employment is terminated for any reason other than by reason of Executive's Involuntary Termination as more fully provided below. If Executive's employment terminates due to an Involuntary Termination, Executive will be eligible to receive the compensation and benefits described in Sections 3(b) and 3(c), as applicable.
- **(b) Involuntary Termination.** Upon Executive's Involuntary Termination at any time, in addition to the Accrued Amounts, and provided that Executive signs and allows to become effective a release and waiver of claims substantially in the form attached hereto as **Exhibit A** (the "**Release**") within the time period provided therein (as further described in Section 4(a) below), then the Company shall provide Executive with the following severance benefits (the "**Severance Benefits**"):
- equivalent to the sum of (A) Executive's then-current Annual Base Salary for twelve (12) months and (B) an amount equal to Executive's then-current target bonus percentage of Executive's then-current Annual Base Salary as most recently established for Executive by the Board (or its compensation committee), assuming achievement of 100% of corporate and, as applicable, Executive's personal goals established for such year (or if no such goals have been established for such year, then the most recently established goals for a prior year) and without regard to any "stretch" or similar goals; provided that the portion of such payment that is payable under this clause (B) shall be pro rated based on the ratio that the number of days from the beginning of the calendar year in which such Involuntary Termination occurs through the date of Involuntary Termination bears to 365. The lump sum cash payment described in this Section 3(b)(i) shall be paid to Executive on the first regular payroll date of the Company following the effective date of the Release, and in any event no later than March 15 of the year following the year in which the Involuntary Termination occurs.
- (ii) Accelerated Vesting of Stock Awards. The vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) as if Executive had completed an additional twelve (12) months of service with the Company as of the date of Executive's Involuntary Termination.
- (iii) Payment of Continued Group Health Plan Benefits. If Executive is eligible for and timely elects continued group health plan coverage under the Consolidated Omnibus Budget

Reconciliation Act of 1985 ("COBRA") following Executive's termination, the Company will pay the COBRA group health insurance premiums for Executive and Executive's eligible dependents until the earliest of (A) the close of the twelve (12) month period following the termination of Executive's employment (the "COBRA Payment Period"), (B) the expiration of Executive's eligibility for the continuation coverage under COBRA, or (C) the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment. References to COBRA premiums shall not include any amounts payable by Executive under a health care reimbursement plan pursuant to Section 125 of the Code. Notwithstanding the foregoing, if at any time the Company determines, in its sole discretion, that it cannot pay the COBRA premiums without potentially incurring financial costs or penalties under applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then regardless of whether Executive elects continued health coverage under COBRA, in lieu of providing the COBRA premiums, the Company will instead pay to Executive, on the last day of each remaining month of the COBRA Payment Period, a fully taxable cash payment equal to the COBRA premiums for that month, subject to applicable tax withholdings (such amount, the "Special Severance Payment"), which payments shall continue until the earlier of expiration of the COBRA Payment Period or the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment.

- **(c)** Additional Benefits Upon Involuntary Termination During the Change in Control Period. If Executive's Involuntary Termination occurs during the Change in Control Period, in addition to the Accrued Amounts and the Severance Benefits described in Section 3(b) above, Executive shall also be eligible for the following additional vesting acceleration benefit (collectively, with the Severance Benefits, the "Covered Severance Benefits"):
- Executive's Involuntary Termination or the effective date of the Change in Control, the vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) in full. Executive's stock awards shall remain outstanding following Executive's Involuntary Termination if and to the extent necessary to give effect to this Section 3(c)(i). For the avoidance of doubt, this vesting acceleration is conditioned upon the actual consummation of a Change in Control and in the event such Change in Control is not consummated, Executive shall receive the vesting acceleration benefits set forth in Section 3(b)(ii) above instead of the vesting acceleration benefits set forth in this Section 3(c)(i).

For the avoidance of doubt, the benefits set forth in this Section 3(c) shall in all events be subject to Executive's execution of an effective Release as described in Section 3(b) above.

## 4. Limitations and Conditions on Covered Severance Benefits.

(a) No Covered Severance Benefits shall be payable until the effectiveness of the Release. The Board, in its sole discretion, may modify the form of the required Release to comply with applicable law and individual circumstance, and may incorporate the Release into a termination agreement or other agreement with Executive. In any case, the Release shall specifically relate to all of Executive's rights and claims in existence at the time of such execution and shall confirm Executive's continuing obligations to the Company (including but not limited to obligations under the PIIA and any confidentiality and/or non-solicitation agreement, as applicable, with the Company) and in all cases the Release by its terms must be signed and become effective no later than sixty (60) days following the Executive's Separation from Service.

- **(b)** All payments under this Agreement will be subject to applicable withholding for federal, state and local taxes. If Executive is indebted to the Company as of Executive's Involuntary Termination, the Company reserves the right to offset any Covered Severance Benefits by the amount of such indebtedness.
- **(c)** As a condition to receipt of any Covered Severance Benefits, Executive acknowledges and agrees to resign following termination of employment with the Company from all Company positions, including membership on the Board, if applicable, unless otherwise requested by the Company.
- **5. Successors.** Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company described in the foregoing sentence or any other successor who becomes bound by this Agreement by operation of law. The terms of this Agreement and all of Executive's rights hereunder and thereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.
- **6. At-Will Employment; No Employment Rights.** Executive acknowledges, affirms and agrees that Executive's employment with the Company is "at will," and may be terminated at any time and for any reason whatsoever by Executive or the Company, with or without Cause and with or without advance notice. This "at-will" employment relationship cannot be changed except in a writing signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board.
- 7. **Section 409A**. It is intended that all of the benefits and other payments payable under this Agreement satisfy, to the greatest extent possible, an exemption from the application of Section 409A, and this Agreement will be construed to the greatest extent possible as consistent with those provisions, and to the extent no so exempt, this Agreement (and any definitions hereunder) will be construed in a manner that complies with Section 409A, and any ambiguities herein shall be interpreted accordingly. Specifically, the benefits under this Agreement are intended to satisfy the exemptions from application of Section 409A provided under Treasury Regulations Sections 1.409A-1(b)(4), 1.409A-1(b)(5) and 1.409A-1(b)(9) and each installment of severance benefits, if any, is a separate "payment" for purposes of Treasury Regulations Section 1.409A-2(b)(2)(i). However, if such exemptions are not available and Executive is, upon Separation from Service, a "specified employee" for purposes of Section 409A, then, solely to the extent necessary to avoid adverse personal tax consequences under Section 409A, the timing of the severance benefits payments shall be delayed until the earlier of (i) six (6) months and one day after Executive's Separation from Service, or (ii) Executive's death. Severance benefits shall not commence until Executive has a Separation from Service. If the severance benefits are not covered by one or more exemptions from the application of Section 409A and the Release could become effective in the calendar year following the calendar year in which Executive's Separation from Service occurs, the Release will not be deemed effective, for purposes of payment of severance, any earlier than the first day of the second calendar year. Except to the minimum extent that payments must be delayed because Executive is a "specified employee" or until the effectiveness of the Release, all severance amounts will be paid as soon as practicable in accordance with this Agreement and the Company's normal payroll practices.

# 8. Parachute Payments.

**(a)** If any payment or benefit Executive will or may receive from the Company or otherwise (a "280G Payment") would (i) constitute a "parachute payment" within the meaning of Section

280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then any such 280G Payment (a "Payment") shall be equal to the Reduced Amount. The "Reduced Amount" shall be either (x) the largest portion of the Payment that would result in no portion of the Payment (after reduction) being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount (i.e., the amount determined by clause (x) or by clause (y)), after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in Executive's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in a Payment is required pursuant to the preceding sentence and the Reduced Amount is determined pursuant to clause (x) of the preceding sentence, the reduction shall occur in the manner (the "Reduction Method") that results in the greatest economic benefit for Executive. If more than one method of reduction will result in the same economic benefit, the items so reduced will be reduced pro rata (the "Pro Rata Reduction Method").

- **(b)** Notwithstanding the foregoing, if the Reduction Method or the Pro Rata Reduction Method would result in any portion of the Payment being subject to taxes pursuant to Section 409A that would not otherwise be subject to taxes pursuant to Section 409A, then the Reduction Method and/or the Pro Rata Reduction Method, as the case may be, shall be modified so as to avoid the imposition of taxes pursuant to Section 409A as follows: (A) as a first priority, the modification shall preserve to the greatest extent possible, the greatest economic benefit for Executive as determined on an after-tax basis; (B) as a second priority, Payments that are contingent on future events (e.g., being terminated without cause), shall be reduced (or eliminated) before Payments that are not contingent on future events; and (C) as a third priority, Payments that are "deferred compensation" within the meaning of Section 409A shall be reduced (or eliminated) before Payments that are not deferred compensation within the meaning of Section 409A.
- (c) Unless Executive and the Company agree on an alternative accounting firm or law firm, the accounting firm engaged by the Company for general tax compliance purposes as of the day prior to the effective date of the Change in Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change in Control, the Company shall appoint a nationally recognized accounting or law firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting or law firm required to be made hereunder. The Company shall use commercially reasonable efforts to cause the accounting or law firm engaged to make the determinations hereunder to provide its calculations, together with detailed supporting documentation, to Executive and the Company within fifteen (15) calendar days after the date on which Executive's right to a 280G Payment becomes reasonably likely to occur (if requested at that time by Executive or the Company) or such other time as requested by Executive or the Company.
- (d) If Executive receives a Payment for which the Reduced Amount was determined pursuant to clause (x) of Section 8(a) and the Internal Revenue Service determines thereafter that some portion of the Payment is subject to the Excise Tax, Executive agrees to promptly return to the Company a sufficient amount of the Payment (after reduction pursuant to clause (x) of Section 8(a) so that no portion of the remaining Payment is subject to the Excise Tax. For the avoidance of doubt, if the Reduced Amount was determined pursuant to clause (y) of Section 8(a), Executive shall have no obligation to return any portion of the Payment pursuant to the preceding sentence.

# 9. Miscellaneous Provisions.

**(a) Waiver**. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other

party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.

- **(b) Whole Agreement.** No agreements, representations or understandings (whether oral or written and whether express or implied) which are not expressly set forth in this Agreement have been made or entered into by either party with respect to the subject matter hereof. This Agreement supersedes the Prior Agreement in its entirety and any other agreement (or portion thereof), representation or understanding (whether oral or written and whether express or implied) with respect to Executive's severance and change in control benefits, dated or made prior to the date of this Agreement, and by execution of this Agreement both parties agree that any such predecessor agreement (or applicable portion thereof), representation or understanding shall be deemed null and void.
- **(c) Choice of Law.** The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California without reference to conflict of laws provisions, and the parties hereto submit to the exclusive jurisdiction of the state and federal courts of the State of California.
- **(d) Severability.** If any term or provision of this Agreement or the application thereof to any circumstance shall, in any jurisdiction and to any extent, be invalid or unenforceable, such term or provision shall be ineffective as to such jurisdiction to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining terms and provisions of this Agreement or the application of such terms and provisions to circumstances other than those as to which it is held invalid or unenforceable, and a suitable and equitable term or provision shall be substituted therefor to carry out, insofar as may be valid and enforceable, the intent and purpose of the invalid or unenforceable term or provision.
- **Dispute Resolution**. To ensure the rapid and economical resolution of disputes that may arise in connection with Executive's employment with the Company, Executive and the Company agree that any and all disputes, claims, or causes of action, in law or equity, including but not limited to statutory claims, arising from or relating to the enforcement, breach, performance, or interpretation of this Agreement, Executive's employment with the Company, or the termination of Executive's employment from the Company, shall be resolved, to the fullest extent permitted by law, by final, binding and confidential arbitration conducted before a single arbitrator by JAMS, Inc. ("JAMS") or its successor, under JAMS' then procedures for employment rules and disputes (which can http://www.jamsadr.com/rules-clauses/, and which will be provided to Executive on request). The arbitration shall take place in the county (or comparable governmental unit) in which Executive was last employed by the Company, as determined by the arbitrator; provided that if the arbitrator determines there will be an undue hardship to Executive to have the arbitration in such location, the arbitrator will choose an alternative appropriate location. Executive and the Company each acknowledge that by agreeing to this arbitration procedure, they waive the right to resolve any such dispute through a trial by jury or judge or administrative proceeding. Executive will have the right to be represented by legal counsel at any arbitration proceeding. The arbitrator shall: (a) have the authority to compel adequate discovery for the resolution of the dispute and to award such relief as would otherwise be available under applicable law in a court proceeding; and (b) issue a written statement signed by the arbitrator regarding the disposition of each claim and the relief, if any, awarded as to each claim, the reasons for the award, and the arbitrator's essential findings and conclusions on which the award is based. The arbitrator, and not a court, shall also be authorized to determine whether the provisions of this section apply to a dispute, controversy, or claim sought to be resolved in accordance with these arbitration procedures. The Company shall pay all arbitration fees and costs in excess of the administrative fees that Executive would be required to incur if the dispute were filed or decided in a court of law. Nothing in this Agreement is intended to prevent either Executive or the Company from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration.

- **(f) Legal Fees and Expenses**. The parties shall each bear their own expenses, legal fees and other fees incurred in connection with the execution of this Agreement.
- **(g) No Assignment**. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, including (without limitation) bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 9(g) shall be void.
- **(h) Counterparts**. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

EXECUTIVE:

/S/
Name: John R. Desjarlais, Ph.D.

COMPANY:

XENCOR, INC.

By:
/S/

Name: Bassil Dahiyat

Title: President and CEO

In Witness Whereof, each of the parties has executed this Agreement as of the Effective Date.

### Exhibit A

### RELEASE AND WAIVER OF CLAIMS

In consideration of the receipt of benefits set forth in the Severance Agreement dated \_\_\_\_\_\_\_\_, 2016 (the "Agreement") to which this form is attached, I, \_\_\_\_\_\_\_\_, hereby furnish Xencor, Inc. and any and all affiliated, subsidiary, related, or successor corporations (collectively, the "Company"), with the following release and waiver ("Release and Waiver"). I understand that if I timely sign, date and return this Release and Waiver, and I do not revoke it, I will receive certain benefits pursuant to the terms and conditions of the Agreement. I understand that I am not entitled to such benefits unless I timely sign this Release and Waiver and allow it to become effective.

General Release and Waiver. In exchange for the consideration to be provided to me under the Agreement that I am not otherwise entitled to receive, I hereby generally and completely Release and Waiver, acquit and forever discharge the Company and its parent, subsidiary, and affiliated entities, and investors, along with its and their predecessors and successors and their respective directors, officers, employees, shareholders, partners, agents, attorneys, insurers, affiliates and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities and obligations, both known and unknown, that arise from or are in any way related to events, acts, conduct, or omissions occurring at any time prior to and including the date that I sign this Release and Waiver (collectively, the "Released Claims"). The Released Claims include, but are not limited to: (1) all claims arising out of or in any way related to my employment with the Company, or the termination of that employment; (2) all claims related to my compensation or benefits from the Company, including salary, bonuses, commissions, other incentive compensation, vacation pay and the redemption thereof, expense reimbursements, fringe benefits, stock, stock options, or any other ownership or equity interests in the Company; (3) all claims for breach of contract, wrongful termination, and breach of the implied covenant of good faith and fair dealing; (4) all tort claims, including but not limited to claims for fraud, defamation, emotional distress, and discharge in violation of public policy; and (5) all federal, state, and local statutory claims, including but not limited to claims for discrimination, harassment, retaliation, attorneys' fees, penalties, or other claims arising under the federal Civil Rights Act of 1964 (as amended), the federal Americans with Disabilities Act of 1990 (as amended), the federal Age Discrimination in Employment Act of 1967 (as amended) (the "ADEA"), the federal Family and Medical Leave Act ("FMLA"), the California Labor Code (as amended), and the California Fair Employment and Housing Act (as amended).

**Excluded Claims**. Notwithstanding the foregoing, the following are not included in the Released Claims (the "*Excluded Claims*"): (1) any rights or claims for indemnification I may have pursuant to any written indemnification agreement with the Company to which I am a party, the Company's bylaws, or applicable law; and (2) any rights which are not waivable as a matter of law. In addition, nothing in this Release and Waiver prevents me from filing, cooperating with, or participating in any investigation or proceeding before the Equal Employment Opportunity Commission, the Department of Labor, the California Department of Fair Employment and Housing, or any other government agency, except that I hereby waive my right to any monetary benefits in connection with any such claim, charge, investigation or proceeding. I hereby represent and warrant that, other than the Excluded Claims, I am not aware of any claims I have or might have against any of the Released Parties that are not included in the Released Claims.

**ADEA Waiver**. I acknowledge that I am knowingly and voluntarily waiving and releasing any rights I may have under the ADEA ("*ADEA Waiver*"). I also acknowledge that the consideration given for the ADEA Waiver is in addition to anything of value to which I was already entitled. I further acknowledge that I have been advised by this writing, as required by the ADEA, that: (a) my ADEA Waiver does not apply to any rights or claims that arise after the date I sign this Release and Waiver; (b) I should consult with an attorney prior to signing this Release and Waiver; (c) I have twenty-one (21) days to consider this

Release and Waiver (although I may choose to voluntarily sign it sooner); (d) I have seven (7) days following the date I sign this Release and Waiver to revoke the ADEA Waiver; and (e) the ADEA Waiver will not be effective until the date upon which the revocation period has expired unexercised, which will be the eighth day after I sign this Release and Waiver.

Section 1542 Waiver. In giving the general release herein, which includes claims which may be unknown to me at present, I acknowledge that I have read and understand Section 1542 of the California Civil Code, which reads as follows: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." I hereby expressly waive and relinquish all rights and benefits under that section and any law of any other jurisdiction of similar effect with respect to my release of claims, including but not limited to any unknown or unsuspected claims herein.

Other Agreements and Representations. I further agree: (a) not to disparage the Company, its officers, directors, employees, shareholders, and agents, in any manner likely to be harmful to its or their business, business reputations, or personal reputations; (b) not to voluntarily (except in response to legal compulsion) assist any third party in bringing or pursuing any proposed or pending litigation, arbitration, administrative claim or other formal proceeding against the Company, its parent or subsidiary entities, investors, affiliates, officers, directors, employees or agents; (c) to cooperate fully with the Company, by voluntarily (without legal compulsion) providing accurate and complete information, in connection with the Company's actual or contemplated defense, prosecution, or investigation of any claims or demands by or against third parties, or other matters, arising from events, acts, or failures to act that occurred during the period of my employment by the Company; and (d) I hereby acknowledge and reaffirm my continuing obligations under the terms of my Proprietary Information and Inventions Agreement. In addition, I hereby represent that I have been paid all wages earned owed and for all hours worked, I have received all the leave and leave benefits and protections for which I am eligible, pursuant to FMLA, the California Family Rights Act, or any applicable law or Company policy, and I have not suffered any on-the-job injury for which I have not already filed a workers' compensation claim.

I acknowledge my continuing obligations under my employee Proprietary Information and Inventions Agreement with the Company (the "*PIIA*").

This Release and Waiver attached to the Agreement as Exhibit A, along with the PIIA, constitutes the complete, final and exclusive embodiment of the entire agreement between the Company and me with regard to the subject matter hereof. I am not relying on any promise or representation by the Company that is not expressly stated herein. This Release and Waiver may only be modified by a writing signed by both me and a duly authorized officer of the Company.

Name:	
Date:	

**Understood and Agreed:** 

### XENCOR, INC.

## SEVERANCE AGREEMENT

This Severance Agreement (the "*Agreement*") is dated as of May 26, 2016 (the "*Effective Date*"), by and between Lloyd A. Rowland, Jr. ("*Executive*") and Xencor, Inc., a Delaware corporation (the "*Company*"). This Agreement is intended to provide Executive with certain severance and change in control benefits as described herein.

## **RECITALS**

- A. The Company and Executive are parties that certain Employment Agreement, dated August 29, 2014, by and between the Company and Executive (the "*Employment Agreement*"), which provides for certain severance benefits upon specific termination events.
- C. The Company's Board of Directors (the "*Board*") believes it is in the best interests of the Company and its stockholders to provide Executive with certain benefits upon a termination of Executive's employment under certain circumstances, including in connection with a Change in Control (as defined below), which benefits are intended to provide Executive with financial security and provide sufficient income and encouragement to Executive to remain with the Company, notwithstanding the possibility of a termination of Executive's employment with the Company.
- E. To accomplish the foregoing objectives, the Board desires to provide the opportunity for severance and change in control benefits to Executive on the terms provided in this Agreement, which shall supersede and replace the terms of the Employment Agreement with respect to Executive's severance and change in control benefits, as described in more detail in Section 9(b) of this Agreement.

Now therefore, in consideration of the mutual promises, covenants and agreements contained herein, and in consideration of the continuing employment of Executive by the Company, the parties hereto agree as follows:

- **1. Effectiveness and Term of Agreement.** This Agreement shall become effective as of the date indicated above, subject to execution by each of Executive and the Company. This Agreement shall remain in effect until it is terminated pursuant to its terms or until the Company has met all of its obligations under this Agreement following a termination of Executive's employment with the Company.
- **2. Definitions Used in This Agreement.** The terms used in this Agreement shall have the following meanings.
- **(a) Annual Base Salary.** "*Annual Base Salary*" means Executive's annual base salary in effect immediately prior to Executive's Involuntary Termination, ignoring any decrease that forms the basis for Executive's termination for Good Reason, if applicable.
  - **(b) Cause**. "*Cause*" means Executive's:
- **(i)** gross negligence or willful misconduct in the performance of Executive's duties to the Company as an employee of the Company (other than a failure resulting from Executive's complete or partial incapacity due to physical or mental illness or impairment);

- **(ii)** material and willful violation of any federal or state law or regulation applicable to the business of the Company;
- (iii) refusal or failure to act in accordance with any lawful specific direction or order of the Board;
  - **(iv)** commission of any act of fraud with respect to the Company;
- **(v)** breach of any material provision of Executive's PIIA, including without limitation, Executive's theft or other misappropriation of the Company's proprietary information or trade secrets; or
- **(vi)** conviction of, or entry of plea of *nolo contendere* to, a felony or a crime involving moral turpitude.

Whether or not the actions or omissions of Executive constitute "Cause" shall be decided by the Board based upon a reasonable good faith investigation and determination.

- **(c) Change in Control.** "*Change in Control*" has the meaning set forth in the Company's 2013 Equity Incentive Plan.
- **(d) Change in Control Period.** "*Change in Control Period*" means the period of time beginning one (1) month prior to the execution of a definitive written agreement that if consummated in accordance with its terms would result in a Change in Control and ending on the earlier of (i) the termination of such agreement or (ii) twelve (12) months following the consummation of a Change in Control pursuant to such agreement.
- **(e) Code.** "*Code*" means the Internal Revenue Code of 1986, as amended, and the treasury regulations and other guidance promulgated thereunder.
- (f) Good Reason. "Good Reason" for Executive to terminate Executive's employment hereunder means the occurrence of any of the following events without Executive's consent; provided, however, that any resignation by Executive due to any of the following conditions shall only be deemed for Good Reason if: (i) Executive gives the Company written notice of the intent to terminate for Good Reason within sixty (60) days following the first occurrence of the condition(s) that Executive believes constitutes Good Reason, which notice shall describe such condition(s); (ii) the Company fails to remedy, if remediable, such condition(s) within thirty (30) days following receipt of the written notice (the "Cure Period") of such condition(s) from Executive; and (iii) Executive actually resigns his employment within the first fifteen (15) days after expiration of the Cure Period:
- (i) a material reduction in Executive's authority or job responsibilities as an employee of the Company or successor to the Company;
- (ii) a material reduction in Executive's annual base salary other than pursuant to a Company-wide reduction of annual base salaries for employees of the Company generally; or
- **(iii)** the relocation of the Executive's offices by a distance of fifty (50) miles or more, which relocation requires an increase in Executive's one-way driving distance by more than twenty-five (25) miles.
- **(g) Involuntary Termination**. "*Involuntary Termination*" means Executive's employment with the Company is terminated either (i) by the Company without Cause (and other than as a

result of Executive's death or disability) or (ii) by Executive's resignation for Good Reason, and provided in either case such termination constitutes a Separation from Service.

- **(h) PIIA.** "*PIIA*" means the Proprietary Information and Inventions Agreement (or similar type agreement) between Executive and the Company.
- **(i) Section 409A.** "*Section 409A*" means Section 409A of the Code and any state law of similar effect.
- **(j) Separation from Service**. *"Separation from Service"* means a "separation from service", as defined under Treasury Regulation Section 1.409A-1(h).

# 3. Benefits Upon Termination of Employment.

- (a) General. If Executive's employment is terminated for any reason or no reason, the Company shall pay to Executive or to Executive's legal representatives, if applicable: (i) any base salary earned, but unpaid and (ii) any unreimbursed business expenses payable to Executive and any accrued but unused personal time off or vacation benefits and any other payments or benefits required by applicable law (collectively "Accrued Amounts"), which amounts shall be promptly paid in a lump sum to Executive, or in the case of Executive's death to Executive's estate. Other than the Accrued Amounts, Executive or Executive's legal representatives shall not be entitled to any additional compensation or benefits if Executive's employment is terminated for any reason other than by reason of Executive's Involuntary Termination as more fully provided below. If Executive's employment terminates due to an Involuntary Termination, Executive will be eligible to receive the compensation and benefits described in Sections 3(b) and 3(c), as applicable.
- **(b) Involuntary Termination.** Upon Executive's Involuntary Termination at any time, in addition to the Accrued Amounts, and provided that Executive signs and allows to become effective a release and waiver of claims substantially in the form attached hereto as **Exhibit A** (the "*Release*") within the time period provided therein (as further described in Section 4(a) below), then the Company shall provide Executive with the following severance benefits (the "*Severance Benefits*"):
- equivalent to the sum of (A) Executive's then-current Annual Base Salary for twelve (12) months and (B) an amount equal to Executive's then-current target bonus percentage of Executive's then-current Annual Base Salary as most recently established for Executive by the Board (or its compensation committee), assuming achievement of 100% of corporate and, as applicable, Executive's personal goals established for such year (or if no such goals have been established for such year, then the most recently established goals for a prior year) and without regard to any "stretch" or similar goals; provided that the portion of such payment that is payable under this clause (B) shall be pro rated based on the ratio that the number of days from the beginning of the calendar year in which such Involuntary Termination occurs through the date of Involuntary Termination bears to 365. The lump sum cash payment described in this Section 3(b)(i) shall be paid to Executive on the first regular payroll date of the Company following the effective date of the Release, and in any event no later than March 15 of the year following the year in which the Involuntary Termination occurs.
- (ii) Accelerated Vesting of Stock Awards. The vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) as if Executive had completed an additional twelve (12) months of service with the Company as of the date of Executive's Involuntary Termination.

(iii) Payment of Continued Group Health Plan Benefits. If Executive is eligible for and timely elects continued group health plan coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985 ("COBRA") following Executive's termination, the Company will pay the COBRA group health insurance premiums for Executive and Executive's eligible dependents until the earliest of (A) the close of the twelve (12) month period following the termination of Executive's employment (the "COBRA Payment Period"), (B) the expiration of Executive's eligibility for the continuation coverage under COBRA, or (C) the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment. References to COBRA premiums shall not include any amounts payable by Executive under a health care reimbursement plan pursuant to Section 125 of the Code. Notwithstanding the foregoing, if at any time the Company determines, in its sole discretion, that it cannot pay the COBRA premiums without potentially incurring financial costs or penalties under applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then regardless of whether Executive elects continued health coverage under COBRA, in lieu of providing the COBRA premiums, the Company will instead pay to Executive, on the last day of each remaining month of the COBRA Payment Period, a fully taxable cash payment equal to the COBRA premiums for that month, subject to applicable tax withholdings (such amount, the "Special Severance Payment"), which payments shall continue until the earlier of expiration of the COBRA Payment Period or the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment.

(c) Additional Benefits Upon Involuntary Termination During the Change in Control Period. If Executive's Involuntary Termination occurs during the Change in Control Period, in addition to the Accrued Amounts and the Severance Benefits described in Section 3(b) above, Executive shall also be eligible for the following additional vesting acceleration benefit (collectively, with the Severance Benefits, the "Covered Severance Benefits"):

**(i)** Full Accelerated Vesting of Stock Awards. Effective as of the later of Executive's Involuntary Termination or the effective date of the Change in Control, the vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) in full. Executive's stock awards shall remain outstanding following Executive's Involuntary Termination if and to the extent necessary to give effect to this Section 3(c)(i). For the avoidance of doubt, this vesting acceleration is conditioned upon the actual consummation of a Change in Control and in the event such Change in Control is not consummated, Executive shall receive the vesting acceleration benefits set forth in Section 3(b)(ii) above instead of the vesting acceleration benefits set forth in this Section 3(c)(i).

For the avoidance of doubt, the benefits set forth in this Section 3(c) shall in all events be subject to Executive's execution of an effective Release as described in Section 3(b) above.

## 4. Limitations and Conditions on Covered Severance Benefits.

(a) No Covered Severance Benefits shall be payable until the effectiveness of the Release. The Board, in its sole discretion, may modify the form of the required Release to comply with applicable law and individual circumstance, and may incorporate the Release into a termination agreement or other agreement with Executive. In any case, the Release shall specifically relate to all of Executive's rights and claims in existence at the time of such execution and shall confirm Executive's continuing obligations to the Company (including but not limited to obligations under the PIIA and any confidentiality and/or non-solicitation agreement, as applicable, with the Company) and in all cases the Release by its terms must be signed and become effective no later than sixty (60) days following the Executive's Separation from Service.

- **(b)** All payments under this Agreement will be subject to applicable withholding for federal, state and local taxes. If Executive is indebted to the Company as of Executive's Involuntary Termination, the Company reserves the right to offset any Covered Severance Benefits by the amount of such indebtedness.
- **(c)** As a condition to receipt of any Covered Severance Benefits, Executive acknowledges and agrees to resign following termination of employment with the Company from all Company positions, including membership on the Board, if applicable, unless otherwise requested by the Company.
- **5. Successors.** Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company described in the foregoing sentence or any other successor who becomes bound by this Agreement by operation of law. The terms of this Agreement and all of Executive's rights hereunder and thereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.
- **6. At-Will Employment; No Employment Rights.** Executive acknowledges, affirms and agrees that Executive's employment with the Company is "at will," and may be terminated at any time and for any reason whatsoever by Executive or the Company, with or without Cause and with or without advance notice. This "at-will" employment relationship cannot be changed except in a writing signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board.
- 7. **Section 409A**. It is intended that all of the benefits and other payments payable under this Agreement satisfy, to the greatest extent possible, an exemption from the application of Section 409A, and this Agreement will be construed to the greatest extent possible as consistent with those provisions, and to the extent no so exempt, this Agreement (and any definitions hereunder) will be construed in a manner that complies with Section 409A, and any ambiguities herein shall be interpreted accordingly. Specifically, the benefits under this Agreement are intended to satisfy the exemptions from application of Section 409A provided under Treasury Regulations Sections 1.409A-1(b)(4), 1.409A-1(b)(5) and 1.409A-1(b)(9) and each installment of severance benefits, if any, is a separate "payment" for purposes of Treasury Regulations Section 1.409A-2(b)(2)(i). However, if such exemptions are not available and Executive is, upon Separation from Service, a "specified employee" for purposes of Section 409A, then, solely to the extent necessary to avoid adverse personal tax consequences under Section 409A, the timing of the severance benefits payments shall be delayed until the earlier of (i) six (6) months and one day after Executive's Separation from Service, or (ii) Executive's death. Severance benefits shall not commence until Executive has a Separation from Service. If the severance benefits are not covered by one or more exemptions from the application of Section 409A and the Release could become effective in the calendar year following the calendar year in which Executive's Separation from Service occurs, the Release will not be deemed effective, for purposes of payment of severance, any earlier than the first day of the second calendar year. Except to the minimum extent that payments must be delayed because Executive is a "specified employee" or until the effectiveness of the Release, all severance amounts will be paid as soon as practicable in accordance with this Agreement and the Company's normal payroll practices.

## 8. Parachute Payments.

**(a)** If any payment or benefit Executive will or may receive from the Company or otherwise (a "280G Payment") would (i) constitute a "parachute payment" within the meaning of Section

280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then any such 280G Payment (a "Payment") shall be equal to the Reduced Amount. The "Reduced Amount" shall be either (x) the largest portion of the Payment that would result in no portion of the Payment (after reduction) being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount (i.e., the amount determined by clause (x) or by clause (y)), after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in Executive's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in a Payment is required pursuant to the preceding sentence and the Reduced Amount is determined pursuant to clause (x) of the preceding sentence, the reduction shall occur in the manner (the "Reduction Method") that results in the greatest economic benefit for Executive. If more than one method of reduction will result in the same economic benefit, the items so reduced will be reduced pro rata (the "Pro Rata Reduction Method").

- **(b)** Notwithstanding the foregoing, if the Reduction Method or the Pro Rata Reduction Method would result in any portion of the Payment being subject to taxes pursuant to Section 409A that would not otherwise be subject to taxes pursuant to Section 409A, then the Reduction Method and/or the Pro Rata Reduction Method, as the case may be, shall be modified so as to avoid the imposition of taxes pursuant to Section 409A as follows: (A) as a first priority, the modification shall preserve to the greatest extent possible, the greatest economic benefit for Executive as determined on an after-tax basis; (B) as a second priority, Payments that are contingent on future events (e.g., being terminated without cause), shall be reduced (or eliminated) before Payments that are not contingent on future events; and (C) as a third priority, Payments that are "deferred compensation" within the meaning of Section 409A shall be reduced (or eliminated) before Payments that are not deferred compensation within the meaning of Section 409A.
- (c) Unless Executive and the Company agree on an alternative accounting firm or law firm, the accounting firm engaged by the Company for general tax compliance purposes as of the day prior to the effective date of the Change in Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change in Control, the Company shall appoint a nationally recognized accounting or law firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting or law firm required to be made hereunder. The Company shall use commercially reasonable efforts to cause the accounting or law firm engaged to make the determinations hereunder to provide its calculations, together with detailed supporting documentation, to Executive and the Company within fifteen (15) calendar days after the date on which Executive's right to a 280G Payment becomes reasonably likely to occur (if requested at that time by Executive or the Company) or such other time as requested by Executive or the Company.
- (d) If Executive receives a Payment for which the Reduced Amount was determined pursuant to clause (x) of Section 8(a) and the Internal Revenue Service determines thereafter that some portion of the Payment is subject to the Excise Tax, Executive agrees to promptly return to the Company a sufficient amount of the Payment (after reduction pursuant to clause (x) of Section 8(a) so that no portion of the remaining Payment is subject to the Excise Tax. For the avoidance of doubt, if the Reduced Amount was determined pursuant to clause (y) of Section 8(a), Executive shall have no obligation to return any portion of the Payment pursuant to the preceding sentence.

# 9. Miscellaneous Provisions.

**(a) Waiver**. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other

party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.

- **(b) Whole Agreement.** No agreements, representations or understandings (whether oral or written and whether express or implied) which are not expressly set forth in this Agreement have been made or entered into by either party with respect to the subject matter hereof. This Agreement supersedes the Employment Agreement, with respect to Sections 6, 7 and 8 thereof, and any other agreement (or portion thereof), representation or understanding (whether oral or written and whether express or implied) with respect to Executive's severance and change in control benefits, dated or made prior to the date of this Agreement, and by execution of this Agreement both parties agree that any such predecessor agreement (or applicable portion thereof), representation or understanding shall be deemed null and void.
- **(c) Choice of Law.** The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California without reference to conflict of laws provisions, and the parties hereto submit to the exclusive jurisdiction of the state and federal courts of the State of California.
- **(d) Severability.** If any term or provision of this Agreement or the application thereof to any circumstance shall, in any jurisdiction and to any extent, be invalid or unenforceable, such term or provision shall be ineffective as to such jurisdiction to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining terms and provisions of this Agreement or the application of such terms and provisions to circumstances other than those as to which it is held invalid or unenforceable, and a suitable and equitable term or provision shall be substituted therefor to carry out, insofar as may be valid and enforceable, the intent and purpose of the invalid or unenforceable term or provision.
- **Dispute Resolution**. To ensure the rapid and economical resolution of disputes that may arise in connection with Executive's employment with the Company, Executive and the Company agree that any and all disputes, claims, or causes of action, in law or equity, including but not limited to statutory claims, arising from or relating to the enforcement, breach, performance, or interpretation of this Agreement, Executive's employment with the Company, or the termination of Executive's employment from the Company, shall be resolved, to the fullest extent permitted by law, by final, binding and confidential arbitration conducted before a single arbitrator by JAMS, Inc. ("JAMS") or its successor, under JAMS' then procedures for employment rules and disputes (which can http://www.jamsadr.com/rules-clauses/, and which will be provided to Executive on request). The arbitration shall take place in the county (or comparable governmental unit) in which Executive was last employed by the Company, as determined by the arbitrator; provided that if the arbitrator determines there will be an undue hardship to Executive to have the arbitration in such location, the arbitrator will choose an alternative appropriate location. Executive and the Company each acknowledge that by agreeing to this arbitration procedure, they waive the right to resolve any such dispute through a trial by jury or judge or administrative proceeding. Executive will have the right to be represented by legal counsel at any arbitration proceeding. The arbitrator shall: (a) have the authority to compel adequate discovery for the resolution of the dispute and to award such relief as would otherwise be available under applicable law in a court proceeding; and (b) issue a written statement signed by the arbitrator regarding the disposition of each claim and the relief, if any, awarded as to each claim, the reasons for the award, and the arbitrator's essential findings and conclusions on which the award is based. The arbitrator, and not a court, shall also be authorized to determine whether the provisions of this section apply to a dispute, controversy, or claim sought to be resolved in accordance with these arbitration procedures. The Company shall pay all arbitration fees and costs in excess of the administrative fees that Executive would be required to incur if the dispute were filed or decided in a court of law. Nothing in this Agreement is intended to prevent either

Executive or the Company from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration.

- **(f) Legal Fees and Expenses**. The parties shall each bear their own expenses, legal fees and other fees incurred in connection with the execution of this Agreement.
- **(g) No Assignment**. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, including (without limitation) bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 9(g) shall be void.
- **(h) Counterparts**. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

**IN WITNESS WHEREOF,** each of the parties has executed this Agreement as of the Effective Date.

EXECUTIVE:
/S/ Name: Lloyd A. Rowland, Jr.
COMPANY:
XENCOR, INC.
By: /S/
Name: Bassil Dahiyat
Title: President and CEO
9

#### Exhibit A

#### RELEASE AND WAIVER OF CLAIMS

In consideration of the receipt of benefits set forth in the Severance Agreement dated \_\_\_\_\_\_\_\_, 2016 (the "Agreement") to which this form is attached, I, \_\_\_\_\_\_\_\_, hereby furnish Xencor, Inc. and any and all affiliated, subsidiary, related, or successor corporations (collectively, the "Company"), with the following release and waiver ("Release and Waiver"). I understand that if I timely sign, date and return this Release and Waiver, and I do not revoke it, I will receive certain benefits pursuant to the terms and conditions of the Agreement. I understand that I am not entitled to such benefits unless I timely sign this Release and Waiver and allow it to become effective.

General Release and Waiver. In exchange for the consideration to be provided to me under the Agreement that I am not otherwise entitled to receive, I hereby generally and completely Release and Waiver, acquit and forever discharge the Company and its parent, subsidiary, and affiliated entities, and investors, along with its and their predecessors and successors and their respective directors, officers, employees, shareholders, partners, agents, attorneys, insurers, affiliates and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities and obligations, both known and unknown, that arise from or are in any way related to events, acts, conduct, or omissions occurring at any time prior to and including the date that I sign this Release and Waiver (collectively, the "Released Claims"). The Released Claims include, but are not limited to: (1) all claims arising out of or in any way related to my employment with the Company, or the termination of that employment; (2) all claims related to my compensation or benefits from the Company, including salary, bonuses, commissions, other incentive compensation, vacation pay and the redemption thereof, expense reimbursements, fringe benefits, stock, stock options, or any other ownership or equity interests in the Company; (3) all claims for breach of contract, wrongful termination, and breach of the implied covenant of good faith and fair dealing; (4) all tort claims, including but not limited to claims for fraud, defamation, emotional distress, and discharge in violation of public policy; and (5) all federal, state, and local statutory claims, including but not limited to claims for discrimination, harassment, retaliation, attorneys' fees, penalties, or other claims arising under the federal Civil Rights Act of 1964 (as amended), the federal Americans with Disabilities Act of 1990 (as amended), the federal Age Discrimination in Employment Act of 1967 (as amended) (the "ADEA"), the federal Family and Medical Leave Act ("FMLA"), the California Labor Code (as amended), and the California Fair Employment and Housing Act (as amended).

**Excluded Claims**. Notwithstanding the foregoing, the following are not included in the Released Claims (the "*Excluded Claims*"): (1) any rights or claims for indemnification I may have pursuant to any written indemnification agreement with the Company to which I am a party, the Company's bylaws, or applicable law; and (2) any rights which are not waivable as a matter of law. In addition, nothing in this Release and Waiver prevents me from filing, cooperating with, or participating in any investigation or proceeding before the Equal Employment Opportunity Commission, the Department of Labor, the California Department of Fair Employment and Housing, or any other government agency, except that I hereby waive my right to any monetary benefits in connection with any such claim, charge, investigation or proceeding. I hereby represent and warrant that, other than the Excluded Claims, I am not aware of any claims I have or might have against any of the Released Parties that are not included in the Released Claims.

**ADEA Waiver**. I acknowledge that I am knowingly and voluntarily waiving and releasing any rights I may have under the ADEA ("*ADEA Waiver*"). I also acknowledge that the consideration given for the ADEA Waiver is in addition to anything of value to which I was already entitled. I further acknowledge that I have been advised by this writing, as required by the ADEA, that: (a) my ADEA Waiver does not apply to any rights or claims that arise after the date I sign this Release and Waiver; (b) I should consult with an attorney prior to signing this Release and Waiver; (c) I have twenty-one (21) days to consider this

Release and Waiver (although I may choose to voluntarily sign it sooner); (d) I have seven (7) days following the date I sign this Release and Waiver to revoke the ADEA Waiver; and (e) the ADEA Waiver will not be effective until the date upon which the revocation period has expired unexercised, which will be the eighth day after I sign this Release and Waiver.

Section 1542 Waiver. In giving the general release herein, which includes claims which may be unknown to me at present, I acknowledge that I have read and understand Section 1542 of the California Civil Code, which reads as follows: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." I hereby expressly waive and relinquish all rights and benefits under that section and any law of any other jurisdiction of similar effect with respect to my release of claims, including but not limited to any unknown or unsuspected claims herein.

Other Agreements and Representations. I further agree: (a) not to disparage the Company, its officers, directors, employees, shareholders, and agents, in any manner likely to be harmful to its or their business, business reputations, or personal reputations; (b) not to voluntarily (except in response to legal compulsion) assist any third party in bringing or pursuing any proposed or pending litigation, arbitration, administrative claim or other formal proceeding against the Company, its parent or subsidiary entities, investors, affiliates, officers, directors, employees or agents; (c) to cooperate fully with the Company, by voluntarily (without legal compulsion) providing accurate and complete information, in connection with the Company's actual or contemplated defense, prosecution, or investigation of any claims or demands by or against third parties, or other matters, arising from events, acts, or failures to act that occurred during the period of my employment by the Company; and (d) I hereby acknowledge and reaffirm my continuing obligations under the terms of my Proprietary Information and Inventions Agreement. In addition, I hereby represent that I have been paid all wages earned owed and for all hours worked, I have received all the leave and leave benefits and protections for which I am eligible, pursuant to FMLA, the California Family Rights Act, or any applicable law or Company policy, and I have not suffered any on-the-job injury for which I have not already filed a workers' compensation claim.

I acknowledge my continuing obligations under my employee Proprietary Information and Inventions Agreement with the Company (the "*PIIA*").

This Release and Waiver attached to the Agreement as Exhibit A, along with the PIIA, constitutes the complete, final and exclusive embodiment of the entire agreement between the Company and me with regard to the subject matter hereof. I am not relying on any promise or representation by the Company that is not expressly stated herein. This Release and Waiver may only be modified by a writing signed by both me and a duly authorized officer of the Company.

#### **UNDERSTOOD AND AGREED:**

Name:			
Date:			

#### XENCOR, INC.

### SEVERANCE AGREEMENT

This Severance Agreement (the "*Agreement*") is dated as of July 29, 2016 (the "*Effective Date*"), by and between Edgardo Baracchini, Jr., Ph.D. ("*Executive*") and Xencor, Inc., a Delaware corporation (the "*Company*"). This Agreement is intended to provide Executive with certain severance and change in control benefits as described herein.

### **RECITALS**

- A. The Company and Executive are parties that certain Offer Letter Agreement, dated September 5, 2013, by and between the Company and Executive (the "*Employment Agreement*"), which provides for certain severance benefits upon specific termination events.
- C. The Company's Board of Directors (the "*Board*") believes it is in the best interests of the Company and its stockholders to provide Executive with certain benefits upon a termination of Executive's employment under certain circumstances, including in connection with a Change in Control (as defined below), which benefits are intended to provide Executive with financial security and provide sufficient income and encouragement to Executive to remain with the Company, notwithstanding the possibility of a termination of Executive's employment with the Company.
- E. To accomplish the foregoing objectives, the Board desires to provide the opportunity for severance and change in control benefits to Executive on the terms provided in this Agreement, which shall supersede and replace the terms of the Employment Agreement with respect to Executive's severance and change in control benefits, as described in more detail in Section 9(b) of this Agreement.

Now therefore, in consideration of the mutual promises, covenants and agreements contained herein, and in consideration of the continuing employment of Executive by the Company, the parties hereto agree as follows:

- **1. Effectiveness and Term of Agreement.** This Agreement shall become effective as of the date indicated above, subject to execution by each of Executive and the Company. This Agreement shall remain in effect until it is terminated pursuant to its terms or until the Company has met all of its obligations under this Agreement following a termination of Executive's employment with the Company.
- **2. Definitions Used in This Agreement**. The terms used in this Agreement shall have the following meanings.
- **(a) Annual Base Salary.** "*Annual Base Salary*" means Executive's annual base salary in effect immediately prior to Executive's Involuntary Termination, ignoring any decrease that forms the basis for Executive's termination for Good Reason, if applicable.
  - **(b) Cause**. "*Cause*" means Executive's:
- **(i)** gross negligence or willful misconduct in the performance of Executive's duties to the Company as an employee of the Company (other than a failure resulting from Executive's complete or partial incapacity due to physical or mental illness or impairment);

- (ii) material and willful violation of any federal or state law or regulation applicable to the business of the Company;
- (iii) refusal or failure to act in accordance with any lawful specific direction or order of the Board;
  - **(iv)** commission of any act of fraud with respect to the Company;
- **(v)** breach of any material provision of Executive's PIIA, including without limitation, Executive's theft or other misappropriation of the Company's proprietary information or trade secrets; or
- **(vi)** conviction of, or entry of plea of *nolo contendere* to, a felony or a crime involving moral turpitude.

Whether or not the actions or omissions of Executive constitute "Cause" shall be decided by the Board based upon a reasonable good faith investigation and determination.

- **(c) Change in Control.** "*Change in Control*" has the meaning set forth in the Company's 2013 Equity Incentive Plan.
- **(d) Change in Control Period.** "*Change in Control Period*" means the period of time beginning one (1) month prior to the execution of a definitive written agreement that if consummated in accordance with its terms would result in a Change in Control and ending on the earlier of (i) the termination of such agreement or (ii) twelve (12) months following the consummation of a Change in Control pursuant to such agreement.
- **(e) Code.** "*Code*" means the Internal Revenue Code of 1986, as amended, and the treasury regulations and other guidance promulgated thereunder.
- (f) Good Reason. "Good Reason" for Executive to terminate Executive's employment hereunder means the occurrence of any of the following events without Executive's consent; provided, however, that any resignation by Executive due to any of the following conditions shall only be deemed for Good Reason if: (i) Executive gives the Company written notice of the intent to terminate for Good Reason within sixty (60) days following the first occurrence of the condition(s) that Executive believes constitutes Good Reason, which notice shall describe such condition(s); (ii) the Company fails to remedy, if remediable, such condition(s) within thirty (30) days following receipt of the written notice (the "Cure Period") of such condition(s) from Executive; and (iii) Executive actually resigns his employment within the first fifteen (15) days after expiration of the Cure Period:
- (i) a material reduction in Executive's authority or job responsibilities as an employee of the Company or successor to the Company;
- (ii) a material reduction in Executive's annual base salary other than pursuant to a Company-wide reduction of annual base salaries for employees of the Company generally; or
- **(iii)** the relocation of the Executive's offices by a distance of fifty (50) miles or more, which relocation requires an increase in Executive's one-way driving distance by more than twenty-five (25) miles.
- **(g) Involuntary Termination**. "*Involuntary Termination*" means Executive's employment with the Company is terminated either (i) by the Company without Cause (and other than as a

result of Executive's death or disability) or (ii) by Executive's resignation for Good Reason, and provided in either case such termination constitutes a Separation from Service.

- **(h) PIIA.** "*PIIA*" means the Proprietary Information and Inventions Agreement (or similar type agreement) between Executive and the Company.
- **(i) Section 409A.** "*Section 409A*" means Section 409A of the Code and any state law of similar effect.
- **(j) Separation from Service**. *"Separation from Service"* means a "separation from service", as defined under Treasury Regulation Section 1.409A-1(h).

# 3. Benefits Upon Termination of Employment.

- (a) General. If Executive's employment is terminated for any reason or no reason, the Company shall pay to Executive or to Executive's legal representatives, if applicable: (i) any base salary earned, but unpaid and (ii) any unreimbursed business expenses payable to Executive and any accrued but unused personal time off or vacation benefits and any other payments or benefits required by applicable law (collectively "Accrued Amounts"), which amounts shall be promptly paid in a lump sum to Executive, or in the case of Executive's death to Executive's estate. Other than the Accrued Amounts, Executive or Executive's legal representatives shall not be entitled to any additional compensation or benefits if Executive's employment is terminated for any reason other than by reason of Executive's Involuntary Termination as more fully provided below. If Executive's employment terminates due to an Involuntary Termination, Executive will be eligible to receive the compensation and benefits described in Sections 3(b) and 3(c), as applicable.
- **(b) Involuntary Termination.** Upon Executive's Involuntary Termination at any time, in addition to the Accrued Amounts, and provided that Executive signs and allows to become effective a release and waiver of claims substantially in the form attached hereto as **Exhibit A** (the "*Release*") within the time period provided therein (as further described in Section 4(a) below), then the Company shall provide Executive with the following severance benefits (the "*Severance Benefits*"):
- equivalent to the sum of (A) Executive's then-current Annual Base Salary for twelve (12) months and (B) an amount equal to Executive's then-current target bonus percentage of Executive's then-current Annual Base Salary as most recently established for Executive by the Board (or its compensation committee), assuming achievement of 100% of corporate and, as applicable, Executive's personal goals established for such year (or if no such goals have been established for such year, then the most recently established goals for a prior year) and without regard to any "stretch" or similar goals; provided that the portion of such payment that is payable under this clause (B) shall be pro rated based on the ratio that the number of days from the beginning of the calendar year in which such Involuntary Termination occurs through the date of Involuntary Termination bears to 365. The lump sum cash payment described in this Section 3(b)(i) shall be paid to Executive on the first regular payroll date of the Company following the effective date of the Release, and in any event no later than March 15 of the year following the year in which the Involuntary Termination occurs.
- (ii) Accelerated Vesting of Stock Awards. The vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) as if Executive had completed an additional twelve (12) months of service with the Company as of the date of Executive's Involuntary Termination.

(iii) Payment of Continued Group Health Plan Benefits. If Executive is eligible for and timely elects continued group health plan coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985 ("COBRA") following Executive's termination, the Company will pay the COBRA group health insurance premiums for Executive and Executive's eligible dependents until the earliest of (A) the close of the twelve (12) month period following the termination of Executive's employment (the "COBRA Payment Period"), (B) the expiration of Executive's eligibility for the continuation coverage under COBRA, or (C) the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment. References to COBRA premiums shall not include any amounts payable by Executive under a health care reimbursement plan pursuant to Section 125 of the Code. Notwithstanding the foregoing, if at any time the Company determines, in its sole discretion, that it cannot pay the COBRA premiums without potentially incurring financial costs or penalties under applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then regardless of whether Executive elects continued health coverage under COBRA, in lieu of providing the COBRA premiums, the Company will instead pay to Executive, on the last day of each remaining month of the COBRA Payment Period, a fully taxable cash payment equal to the COBRA premiums for that month, subject to applicable tax withholdings (such amount, the "Special Severance Payment"), which payments shall continue until the earlier of expiration of the COBRA Payment Period or the date when Executive becomes eligible for substantially equivalent health insurance coverage in connection with new employment.

(c) Additional Benefits Upon Involuntary Termination During the Change in Control Period. If Executive's Involuntary Termination occurs during the Change in Control Period, in addition to the Accrued Amounts and the Severance Benefits described in Section 3(b) above, Executive shall also be eligible for the following additional vesting acceleration benefit (collectively, with the Severance Benefits, the "Covered Severance Benefits"):

**(i)** Full Accelerated Vesting of Stock Awards. Effective as of the later of Executive's Involuntary Termination or the effective date of the Change in Control, the vesting and exercisability of all outstanding stock options and other stock awards covering the Company's common stock that are held by Executive as of immediately prior to the Involuntary Termination, to the extent such awards are subject to time-based vesting requirements, shall be accelerated (and lapse, in the case of reacquisition or repurchase rights) in full. Executive's stock awards shall remain outstanding following Executive's Involuntary Termination if and to the extent necessary to give effect to this Section 3(c)(i). For the avoidance of doubt, this vesting acceleration is conditioned upon the actual consummation of a Change in Control and in the event such Change in Control is not consummated, Executive shall receive the vesting acceleration benefits set forth in Section 3(b)(ii) above instead of the vesting acceleration benefits set forth in this Section 3(c)(i).

For the avoidance of doubt, the benefits set forth in this Section 3(c) shall in all events be subject to Executive's execution of an effective Release as described in Section 3(b) above.

### 4. Limitations and Conditions on Covered Severance Benefits.

(a) No Covered Severance Benefits shall be payable until the effectiveness of the Release. The Board, in its sole discretion, may modify the form of the required Release to comply with applicable law and individual circumstance, and may incorporate the Release into a termination agreement or other agreement with Executive. In any case, the Release shall specifically relate to all of Executive's rights and claims in existence at the time of such execution and shall confirm Executive's continuing obligations to the Company (including but not limited to obligations under the PIIA and any confidentiality and/or non-solicitation agreement, as applicable, with the Company) and in all cases the Release by its terms must be signed and become effective no later than sixty (60) days following the Executive's Separation from Service.

- **(b)** All payments under this Agreement will be subject to applicable withholding for federal, state and local taxes. If Executive is indebted to the Company as of Executive's Involuntary Termination, the Company reserves the right to offset any Covered Severance Benefits by the amount of such indebtedness.
- **(c)** As a condition to receipt of any Covered Severance Benefits, Executive acknowledges and agrees to resign following termination of employment with the Company from all Company positions, including membership on the Board, if applicable, unless otherwise requested by the Company.
- **5. Successors.** Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company described in the foregoing sentence or any other successor who becomes bound by this Agreement by operation of law. The terms of this Agreement and all of Executive's rights hereunder and thereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.
- **6. At-Will Employment; No Employment Rights.** Executive acknowledges, affirms and agrees that Executive's employment with the Company is "at will," and may be terminated at any time and for any reason whatsoever by Executive or the Company, with or without Cause and with or without advance notice. This "at-will" employment relationship cannot be changed except in a writing signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board.
- 7. **Section 409A**. It is intended that all of the benefits and other payments payable under this Agreement satisfy, to the greatest extent possible, an exemption from the application of Section 409A, and this Agreement will be construed to the greatest extent possible as consistent with those provisions, and to the extent no so exempt, this Agreement (and any definitions hereunder) will be construed in a manner that complies with Section 409A, and any ambiguities herein shall be interpreted accordingly. Specifically, the benefits under this Agreement are intended to satisfy the exemptions from application of Section 409A provided under Treasury Regulations Sections 1.409A-1(b)(4), 1.409A-1(b)(5) and 1.409A-1(b)(9) and each installment of severance benefits, if any, is a separate "payment" for purposes of Treasury Regulations Section 1.409A-2(b)(2)(i). However, if such exemptions are not available and Executive is, upon Separation from Service, a "specified employee" for purposes of Section 409A, then, solely to the extent necessary to avoid adverse personal tax consequences under Section 409A, the timing of the severance benefits payments shall be delayed until the earlier of (i) six (6) months and one day after Executive's Separation from Service, or (ii) Executive's death. Severance benefits shall not commence until Executive has a Separation from Service. If the severance benefits are not covered by one or more exemptions from the application of Section 409A and the Release could become effective in the calendar year following the calendar year in which Executive's Separation from Service occurs, the Release will not be deemed effective, for purposes of payment of severance, any earlier than the first day of the second calendar year. Except to the minimum extent that payments must be delayed because Executive is a "specified employee" or until the effectiveness of the Release, all severance amounts will be paid as soon as practicable in accordance with this Agreement and the Company's normal payroll practices.

## 8. Parachute Payments.

**(a)** If any payment or benefit Executive will or may receive from the Company or otherwise (a "280G Payment") would (i) constitute a "parachute payment" within the meaning of Section

280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then any such 280G Payment (a "Payment") shall be equal to the Reduced Amount. The "Reduced Amount" shall be either (x) the largest portion of the Payment that would result in no portion of the Payment (after reduction) being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount (i.e., the amount determined by clause (x) or by clause (y)), after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in Executive's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in a Payment is required pursuant to the preceding sentence and the Reduced Amount is determined pursuant to clause (x) of the preceding sentence, the reduction shall occur in the manner (the "Reduction Method") that results in the greatest economic benefit for Executive. If more than one method of reduction will result in the same economic benefit, the items so reduced will be reduced pro rata (the "Pro Rata Reduction Method").

- **(b)** Notwithstanding the foregoing, if the Reduction Method or the Pro Rata Reduction Method would result in any portion of the Payment being subject to taxes pursuant to Section 409A that would not otherwise be subject to taxes pursuant to Section 409A, then the Reduction Method and/or the Pro Rata Reduction Method, as the case may be, shall be modified so as to avoid the imposition of taxes pursuant to Section 409A as follows: (A) as a first priority, the modification shall preserve to the greatest extent possible, the greatest economic benefit for Executive as determined on an after-tax basis; (B) as a second priority, Payments that are contingent on future events (e.g., being terminated without cause), shall be reduced (or eliminated) before Payments that are not contingent on future events; and (C) as a third priority, Payments that are "deferred compensation" within the meaning of Section 409A shall be reduced (or eliminated) before Payments that are not deferred compensation within the meaning of Section 409A.
- (c) Unless Executive and the Company agree on an alternative accounting firm or law firm, the accounting firm engaged by the Company for general tax compliance purposes as of the day prior to the effective date of the Change in Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change in Control, the Company shall appoint a nationally recognized accounting or law firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting or law firm required to be made hereunder. The Company shall use commercially reasonable efforts to cause the accounting or law firm engaged to make the determinations hereunder to provide its calculations, together with detailed supporting documentation, to Executive and the Company within fifteen (15) calendar days after the date on which Executive's right to a 280G Payment becomes reasonably likely to occur (if requested at that time by Executive or the Company) or such other time as requested by Executive or the Company.
- (d) If Executive receives a Payment for which the Reduced Amount was determined pursuant to clause (x) of Section 8(a) and the Internal Revenue Service determines thereafter that some portion of the Payment is subject to the Excise Tax, Executive agrees to promptly return to the Company a sufficient amount of the Payment (after reduction pursuant to clause (x) of Section 8(a) so that no portion of the remaining Payment is subject to the Excise Tax. For the avoidance of doubt, if the Reduced Amount was determined pursuant to clause (y) of Section 8(a), Executive shall have no obligation to return any portion of the Payment pursuant to the preceding sentence.

# 9. Miscellaneous Provisions.

**(a) Waiver**. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive) or member of the Board. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other

party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.

- **(b) Whole Agreement.** No agreements, representations or understandings (whether oral or written and whether express or implied) which are not expressly set forth in this Agreement have been made or entered into by either party with respect to the subject matter hereof. This Agreement supersedes the Employment Agreement, with respect to Sections 4.1, 4.2, 4.8.1, 4.8.2, 4.8.3 and 4.9 thereof, and any other agreement (or portion thereof), representation or understanding (whether oral or written and whether express or implied) with respect to Executive's severance and change in control benefits, dated or made prior to the date of this Agreement, and by execution of this Agreement both parties agree that any such predecessor agreement (or applicable portion thereof), representation or understanding shall be deemed null and void.
- **(c) Choice of Law.** The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California without reference to conflict of laws provisions, and the parties hereto submit to the exclusive jurisdiction of the state and federal courts of the State of California.
- **(d) Severability.** If any term or provision of this Agreement or the application thereof to any circumstance shall, in any jurisdiction and to any extent, be invalid or unenforceable, such term or provision shall be ineffective as to such jurisdiction to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable the remaining terms and provisions of this Agreement or the application of such terms and provisions to circumstances other than those as to which it is held invalid or unenforceable, and a suitable and equitable term or provision shall be substituted therefor to carry out, insofar as may be valid and enforceable, the intent and purpose of the invalid or unenforceable term or provision.
- **Dispute Resolution**. To ensure the rapid and economical resolution of disputes that may arise in connection with Executive's employment with the Company, Executive and the Company agree that any and all disputes, claims, or causes of action, in law or equity, including but not limited to statutory claims, arising from or relating to the enforcement, breach, performance, or interpretation of this Agreement, Executive's employment with the Company, or the termination of Executive's employment from the Company, shall be resolved, to the fullest extent permitted by law, by final, binding and confidential arbitration conducted before a single arbitrator by JAMS, Inc. ("JAMS") or its successor, under JAMS' then procedures for employment rules and disputes (which can http://www.jamsadr.com/rules-clauses/, and which will be provided to Executive on request). The arbitration shall take place in the county (or comparable governmental unit) in which Executive was last employed by the Company, as determined by the arbitrator; provided that if the arbitrator determines there will be an undue hardship to Executive to have the arbitration in such location, the arbitrator will choose an alternative appropriate location. Executive and the Company each acknowledge that by agreeing to this arbitration procedure, they waive the right to resolve any such dispute through a trial by jury or judge or administrative proceeding. Executive will have the right to be represented by legal counsel at any arbitration proceeding. The arbitrator shall: (a) have the authority to compel adequate discovery for the resolution of the dispute and to award such relief as would otherwise be available under applicable law in a court proceeding; and (b) issue a written statement signed by the arbitrator regarding the disposition of each claim and the relief, if any, awarded as to each claim, the reasons for the award, and the arbitrator's essential findings and conclusions on which the award is based. The arbitrator, and not a court, shall also be authorized to determine whether the provisions of this section apply to a dispute, controversy, or claim sought to be resolved in accordance with these arbitration procedures. The Company shall pay all arbitration fees and costs in excess of the administrative fees that Executive would be required to incur if the dispute were filed or decided in a court of law. Nothing in this Agreement is intended to prevent either

Executive or the Company from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration.

- **(f) Legal Fees and Expenses**. The parties shall each bear their own expenses, legal fees and other fees incurred in connection with the execution of this Agreement.
- **(g) No Assignment**. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, including (without limitation) bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 9(g) shall be void.
- **(h) Counterparts**. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

In Witness Whereof, each of the parties has executed this Agreement as of the Effective Date.

EXECUTIVE:

/S/
Name: Edgardo Baracchini, Jr., Ph.D.

COMPANY:

XENCOR, INC.

By:
/S/

Name: Lloyd Rowland

Title: General Counsel

#### Exhibit A

### RELEASE AND WAIVER OF CLAIMS

In consideration of the receipt of benefits set forth in the Severance Agreement dated \_\_\_\_\_\_\_\_, 2016 (the "Agreement") to which this form is attached, I, \_\_\_\_\_\_\_\_, hereby furnish Xencor, Inc. and any and all affiliated, subsidiary, related, or successor corporations (collectively, the "Company"), with the following release and waiver ("Release and Waiver"). I understand that if I timely sign, date and return this Release and Waiver, and I do not revoke it, I will receive certain benefits pursuant to the terms and conditions of the Agreement. I understand that I am not entitled to such benefits unless I timely sign this Release and Waiver and allow it to become effective.

**General Release and Waiver.** In exchange for the consideration to be provided to me under the Agreement that I am not otherwise entitled to receive, I hereby generally and completely Release and Waiver, acquit and forever discharge the Company and its parent, subsidiary, and affiliated entities, and investors, along with its and their predecessors and successors and their respective directors, officers, employees, shareholders, partners, agents, attorneys, insurers, affiliates and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities and obligations, both known and unknown, that arise from or are in any way related to events, acts, conduct, or omissions occurring at any time prior to and including the date that I sign this Release and Waiver (collectively, the "Released Claims"). The Released Claims include, but are not limited to: (1) all claims arising out of or in any way related to my employment with the Company, or the termination of that employment; (2) all claims related to my compensation or benefits from the Company, including salary, bonuses, commissions, other incentive compensation, vacation pay and the redemption thereof, expense reimbursements, fringe benefits, stock, stock options, or any other ownership or equity interests in the Company; (3) all claims for breach of contract, wrongful termination, and breach of the implied covenant of good faith and fair dealing; (4) all tort claims, including but not limited to claims for fraud, defamation, emotional distress, and discharge in violation of public policy; and (5) all federal, state, and local statutory claims, including but not limited to claims for discrimination, harassment, retaliation, attorneys' fees, penalties, or other claims arising under the federal Civil Rights Act of 1964 (as amended), the federal Americans with Disabilities Act of 1990 (as amended), the federal Age Discrimination in Employment Act of 1967 (as amended) (the "ADEA"), the federal Family and Medical Leave Act ("FMLA"), the California Labor Code (as amended), and the California Fair Employment and Housing Act (as amended).

**Excluded Claims**. Notwithstanding the foregoing, the following are not included in the Released Claims (the "*Excluded Claims*"): (1) any rights or claims for indemnification I may have pursuant to any written indemnification agreement with the Company to which I am a party, the Company's bylaws, or applicable law; and (2) any rights which are not waivable as a matter of law. In addition, nothing in this Release and Waiver prevents me from filing, cooperating with, or participating in any investigation or proceeding before the Equal Employment Opportunity Commission, the Department of Labor, the California Department of Fair Employment and Housing, or any other government agency, except that I hereby waive my right to any monetary benefits in connection with any such claim, charge, investigation or proceeding. I hereby represent and warrant that, other than the Excluded Claims, I am not aware of any claims I have or might have against any of the Released Parties that are not included in the Released Claims.

**ADEA Waiver**. I acknowledge that I am knowingly and voluntarily waiving and releasing any rights I may have under the ADEA ("*ADEA Waiver*"). I also acknowledge that the consideration given for the ADEA Waiver is in addition to anything of value to which I was already entitled. I further acknowledge that I have been advised by this writing, as required by the ADEA, that: (a) my ADEA Waiver does not apply to any rights or claims that arise after the date I sign this Release and Waiver; (b) I should consult with an attorney prior to signing this Release and Waiver; (c) I have twenty-one (21) days to consider this

Release and Waiver (although I may choose to voluntarily sign it sooner); (d) I have seven (7) days following the date I sign this Release and Waiver to revoke the ADEA Waiver; and (e) the ADEA Waiver will not be effective until the date upon which the revocation period has expired unexercised, which will be the eighth day after I sign this Release and Waiver.

Section 1542 Waiver. In giving the general release herein, which includes claims which may be unknown to me at present, I acknowledge that I have read and understand Section 1542 of the California Civil Code, which reads as follows: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." I hereby expressly waive and relinquish all rights and benefits under that section and any law of any other jurisdiction of similar effect with respect to my release of claims, including but not limited to any unknown or unsuspected claims herein.

Other Agreements and Representations. I further agree: (a) not to disparage the Company, its officers, directors, employees, shareholders, and agents, in any manner likely to be harmful to its or their business, business reputations, or personal reputations; (b) not to voluntarily (except in response to legal compulsion) assist any third party in bringing or pursuing any proposed or pending litigation, arbitration, administrative claim or other formal proceeding against the Company, its parent or subsidiary entities, investors, affiliates, officers, directors, employees or agents; (c) to cooperate fully with the Company, by voluntarily (without legal compulsion) providing accurate and complete information, in connection with the Company's actual or contemplated defense, prosecution, or investigation of any claims or demands by or against third parties, or other matters, arising from events, acts, or failures to act that occurred during the period of my employment by the Company; and (d) I hereby acknowledge and reaffirm my continuing obligations under the terms of my Proprietary Information and Inventions Agreement. In addition, I hereby represent that I have been paid all wages earned owed and for all hours worked, I have received all the leave and leave benefits and protections for which I am eligible, pursuant to FMLA, the California Family Rights Act, or any applicable law or Company policy, and I have not suffered any on-the-job injury for which I have not already filed a workers' compensation claim.

I acknowledge my continuing obligations under my employee Proprietary Information and Inventions Agreement with the Company (the "*PIIA*").

This Release and Waiver attached to the Agreement as Exhibit A, along with the PIIA, constitutes the complete, final and exclusive embodiment of the entire agreement between the Company and me with regard to the subject matter hereof. I am not relying on any promise or representation by the Company that is not expressly stated herein. This Release and Waiver may only be modified by a writing signed by both me and a duly authorized officer of the Company.

**Understood and Agreed:** 

Name:		
Date:		
	2	

\*\*\*Text Omitted and Filed Separately with the Securities and Exchange Commission. Confidential Treatment Requested Under 17 C.F.R. Sections 200.80(b)(4) and 240.24b-2.

**EXECUTION VERSION** 

# COLLABORATION AND LICENSE AGREEMENT

by and between

XENCOR, INC.

and

NOVARTIS INSTITUTES FOR BIOMEDICAL RESEARCH, INC.

# Confidential

# **TABLE OF CONTENTS**

			Page
1.		DEFINITIONS	1
	1.1.	Definitions.	1
	1.1.		31
2.		GOVERNANCE	
	2.1.	Alliance Manager	31
	2.2.	Joint Steering Committee.	31
	2.3.	Joint Development Committee.	33
	2.4.	Joint Commercialization Committee.	36
	2.5.	Resolution of Committee Disputes.	38
	2.6.	General Committee Authority	40
	2.7.	Withdrawal	41 41
3.		GLOBAL TARGET RESEARCH ACTIVITIES; FC TARGETS	41
	3.1.	Global Target Pairs.	41
	3.2.	Fc Targets.	43
4.		OPT-IN RIGHT	45
	4.1.	Grant, Exercise Mechanism.	45
5.		DEVELOPMENT	46
	5.1.	Regional Licensed Antibodies and Regional Licensed Products.	46
	5.2.	Global Licensed Antibodies and Global Licensed Products.	54
	5.3.	Optioned Licensed Antibodies and Optioned Licensed Products.	56
	5.4.	Fc Licensed Antibodies and Fc Licensed Products.	60
			60
6.		COMMERCIALIZATION	
	6.1.	Regional Licensed Products.	60
	6.2.	Global Licensed Products.	64
	6.3.	Optioned Licensed Products.	65
	6.4.	Fc Licensed Products.	70
	6.5.	Commercialization Activities, Reports	71
7.		REGULATORY	71
	7.1.	Regional Licensed Products.	71
	7.2.	Global Licensed Products, Optioned Licensed Products and Fc Licensed Products.	74
	7.3.	Right of Reference	75
	7.4.	Pharmacovigilance	75
		-i-	

8.	MA	ANUFACTURE	76
	8.1.	Overview	76
	8.2.	Manufacturing for Research	76
	8.3.	Clinical Supply of Regional Licensed Antibodies and Regional Licensed Products	76
	8.4.	Clinical Supply of other Licensed Antibodies and Licensed Products.	77
	8.5.	Commercial Supply of Regional Licensed Products	77
	8.6.	Commercial Supply of other Licensed Products	78
	8.7.	Third Parties	78
9.	LI	CENSES	78
		521.626	
	9.1.	Regional Target Pairs.	78
	9.2.	Global Target Pairs and Optioned Target Pairs.	81
	9.3.	Fc Targets.	83
	9.4.	Joint Collaboration IP	84
	9.5.	In-Licenses.	84
	9.6.	Bankruptcy	86
	9.7.	No Other Rights	86
10.	PA	YMENTS	87
	10.1	Levial Lieuwe Per	0.7
	10.1.	Initial License Fee	87
	10.2.	Profit Share Payments  Profit Share Payments and Payment of Profit Share Payments and	87 87
	10.3.	Reconciliation and Payment of Profit Share Payments and Costs.	
	10.4.	Development Milestone Payments	88
	10.5.	Sales Milestone Payments	92
	10.6.	Royalties	95
	10.7.	Additional Royalty Terms.	97
	10.8.	Other Amounts Payable	98
	10.9.	Payment Terms.	98
	10.10.	Records and Audits	99
11.	CC	ONFIDENTIALITY AND PUBLICATION	100
	11 1	New Jealance Obleration	100
	11.1.	Nondisclosure Obligation.	100
	11.2.	Publication and Publicity.	102
	11.3.	Press Release.	103
12.	RE	EPRESENTATIONS, WARRANTIES AND COVENANTS	104
	12.1.	Mutual Representations and Warranties as of the Effective Date	104

	Confi	dential
12.2	. Representations and Warranties by Xencor	104
12.3		106
12.4	. Warranty Disclaimer	106
12.5	. Certain Covenants.	107
12.6	. Exclusivity.	108
10	INDEMNITE CATION LIMITATION OF LIABILITY INCLIDANCE	113
13.	INDEMNIFICATION, LIMITATION OF LIABILITY, INSURANCE	
13.1	<i>5</i>	113
13.2	$\mathbf{J}$	113
13.3	3	113
13.4		114
13.5	<b>y</b>	115
13.6		115
13.7	. Disclaimer	115
14.	INTELLECTUAL PROPERTY	115
14.1	. Inventorship.	115
14.2	<u>.</u>	116
14.3	1	117
14.4		120
14.5	3 0	126
14.6		126
14.7	0	126
14.8		127
45	TEDM AND TEDMINATION	128
15.	TERM AND TERMINATION	
15.1		128
15.2		129
15.3		129
15.4	J	129
15.5		130
15.6	<u>.</u>	130
15.7	Effect of Termination by Xencor for Cause or [***] or by Novartis for Convenience	130
15.8	. Effect of Termination by Novartis for Cause	133
15.9	. Effect of Expiration or Termination, Survival	133
16.	MISCELLANEOUS	134
16.1	. Effect of a Xencor Change of Control	134
16.2		134
16.3		134

#### Confidential 134 16.4. Arbitration. 16.5. Entire Agreement, Amendments 137 16.6. Severability 137 Headings 16.7. 137 Waiver of Rule of Construction 137 16.8. 16.9. Interpretation 137 No Implied Waivers, Rights Cumulative 138 16.10. Notices 138 16.11. Compliance with Export Regulations 139 16.12. Force Majeure 139 16.13. Independent Parties 16.14. 139 16.15. Counterparts 139 Performance by Affiliates 139 16.16. Binding Effect, No Third Party Beneficiaries 140 16.17.

## **COLLABORATION and license AGREEMENT**

THIS COLLABORATION AND LICENSE AGREEMENT (this "**Agreement**"), entered into as of June 26, 2016 (the "**Effective Date**"), is entered into by and between Xencor, Inc., a corporation organized and existing under the Laws of the State of Delaware ("**Xencor**"), and Novartis Institutes for BioMedical Research, Inc., a corporation organized and existing under the Laws of the State of Delaware ("**Novartis**"). Xencor and Novartis are referred to in this Agreement individually as a "**Party**" and collectively as the "**Parties**".

### **RECITALS:**

**WHEREAS**, Xencor has developed expertise in engineering Antibodies (as defined below);

**WHEREAS**, Xencor has developed and controls Patents and Know-How directed to its half-life extension, IIb immune inhibitor, and cytotoxic Fc Region-related technologies, which are point mutations to the constant region of an antibody that can be introduced to extend the half-life in vivo of an Antibody, inhibit immune cell function and clear bound antigens from the circulation, or increase antibody-dependent cellular cytotoxicity of an Antibody (respectively);

**WHEREAS**, Novartis and its Affiliates possess expertise in discovering, developing, manufacturing, marketing, and selling pharmaceutical products worldwide, including with respect to discovering, developing, and manufacturing Antibodies and has developed and controls the Novartis Biological Material IP;

**WHEREAS**, Novartis desires to obtain from Xencor, and Xencor desires to grant to Novartis, a non-exclusive license to incorporate Xencor's Fc Region-related technologies into certain biopharmaceuticals of Novartis; and

**WHEREAS**, Novartis desires to obtain from Xencor, and Xencor desires to grant to Novartis, an exclusive license to Research, Develop, Manufacture and Commercialize Licensed Antibodies and Licensed Products (each, as defined below), subject to the terms and conditions of this Agreement.

**NOW, THEREFORE**, in consideration of the foregoing premises and the mutual covenants herein contained, the Parties hereby agree as follows:

### 1. **DEFINITIONS**

### 1.1. Definitions.

Unless specifically set forth to the contrary herein, the following terms, whether used in the singular or plural, will have the respective meanings set forth below:

**1.1.1.** "AAA" has the meaning set forth in Section 16.4.3.

- **1.1.2.** "Accounting Standards" means, GAAP, with respect to Xencor and IFRS, with respect to Novartis, in each case, as generally and consistently applied throughout the Party's organization. Each Party shall promptly notify the other in the event that it changes the Accounting Standards pursuant to which its records are maintained, <u>provided</u> however, that each Party may only use internationally recognized accounting principles (e.g., IFRS, GAAP, etc.).
- **1.1.3.** "Acquirer" means, collectively, the Third Party referenced in the definition of Change of Control and such Third Party's Affiliates, other than the applicable Party in the definition of Change of Control and such Party's Affiliates, determined as of immediately prior to the closing of such Change of Control.
  - **1.1.4.** "Additional Development Activities" has the meaning set forth in Section 5.1.2.4(a).
  - **1.1.5.** "Additional Development Data Package" has the meaning set forth in Section 5.1.2.4(d).
  - **1.1.6.** "Additional Development Opt-In Date" has the meaning set forth in Section 5.1.2.4(d).
  - **1.1.7.** "Additional Development Opt-In Notice" has the meaning set forth in Section 5.1.2.4(d).
  - **1.1.8.** "Additional Development Proposal" has the meaning set forth in Section 5.1.2.4(a).
- **1.1.9.** "Affiliate" means, with respect to a Person, any other Person that controls, is controlled by, or is under common control with such Person. For purposes of this Agreement, a Person will be deemed to control another Person if it owns or controls, directly or indirectly, more than 50% of the equity securities of such other Person entitled to vote in the election of directors (or, in the case that such other Person is not a corporation, for the election of the corresponding managing authority), or otherwise has the power to direct the management and policies of such other Person. The Parties acknowledge that in the case of certain entities organized under the laws of certain countries outside the United States, the maximum percentage ownership permitted by law for a foreign investor may be less than 50%, and that in such case such lower percentage will be substituted in the preceding sentence, <u>provided</u> that such foreign investor has the power to direct the management and policies of such entity.
  - **1.1.10.** "Alliance Manager" has the meaning set forth in Section 2.1.
  - **1.1.11.** "Annual Net Sales" means Net Sales recorded in a given Calendar Year.
- **1.1.12.** "**Antibody**" means a protein comprising an [...\*\*\*...]. For clarity, an Antibody that differs in [...\*\*\*...] will be treated as a

separate Antibody. Notwithstanding the foregoing, solely with respect to Fc Licensed Antibodies, the term Antibody shall include [...\*\*\*...].

- **1.1.13.** "**Antitrust Laws**" means any federal, state or foreign law, regulation or decree, including the HSR Act, designed to prohibit, restrict or regulate actions for the purpose or effect of monopolization or restraint of trade.
  - **1.1.14.** "Audited Party" has the meaning set forth in Section 10.10.
  - **1.1.15.** "Auditing Party" has the meaning set forth in Section 10.10.
  - **1.1.16.** "Auditor" has the meaning set forth in Section 10.10.
  - **1.1.17.** "Available" means, with respect to a:

```
1.1.17.1. [...***...].
1.1.17.2. [...***...].
```

- **1.1.18.** "Availability" has the correlative meaning to "Available".
- **1.1.19.** "Bankruptcy Code" has the meaning set forth in Section 15.4.
- **1.1.20.** "Bankrupt Party" has the meaning set forth in Section 9.6.
- **1.1.21.** "[...\*\*\*...]" means the human [...\*\*\*...] protein [...\*\*\*...].
- **1.1.22.** "Biosimilar Application" means an application submitted to the FDA under subsection (k) of Section 351 of the PHSA, or any analogous application submitted to a Regulatory Authority in the United States or in another country in the world.
  - **1.1.23.** "Biosimilar Product" means, [...\*\*\*...]

[...\*\*\*...].

- **1.1.24.** "Bispecific Antibody(ies)" means an Antibody that [...\*\*\*...].
- **1.1.25.** "BLA" means (a) a Biologics License Application as defined in the FD&C Act and the regulations promulgated thereunder, (b) a Marketing Authorization Application ("MAA") in the EU, or (c) any equivalent or comparable application, registration or certification in any other country or region.
  - **1.1.26.** "Brief" has the meaning set forth in Section 16.4.4.2(b).
- **1.1.27.** "**Business Day**" means a day other than a Saturday, Sunday or a bank or other public holiday in California or Massachusetts in the United States or Basel, Switzerland.
- **1.1.28.** "Calendar Quarter" means the respective periods of three consecutive calendar months ending on March 31, June 30, September 30 and December 31 of each Calendar Year.
- **1.1.29.** "Calendar Year" means each successive period of 12 months commencing on January 1 and ending on December 31.
  - **1.1.30.** [...\*\*\*...].
  - **1.1.31.** [...\*\*\*...].
  - **1.1.32.** [...\*\*\*...].

- **1.1.33.** "cGCP" means the then-current ethical, scientific and quality standards required by FDA for designing, conducting, recording and reporting trials that involve the participation of human subjects, as set forth in FDA regulations in 21 C.F.R. Parts 11, 50, 54, 56, and 312 and related FDA guidance documents, and by the International Conference on Harmonization E6: Good Clinical Practices Consolidated Guideline, or as otherwise required by Laws.
- **1.1.34.** "cGLP" means the then-current good laboratory practice as required by the FDA under 21 C.F.R. Part 58 and all applicable FDA rules, regulations, orders and guidances, and the requirements with respect to current good laboratory practices prescribed by the European Community, the OECD (Organization for Economic Cooperation and Development Council) and the ICH Guidelines, or as otherwise required by Laws.
- **1.1.35.** "**cGMP**" means the then-current good manufacturing practices as required by the FDA under provisions of 21 C.F.R. Parts 210 and 211 and all applicable FDA rules, regulations, orders and guidances, and the requirements with respect to current good manufacturing practices prescribed by the European Community under provisions of "The Rules Governing Medicinal Products in the European Community, Volume 4, Good Manufacturing Practices, Annex 13, Manufacture of Investigational Medicinal Products, July 2003," or as otherwise required by Laws.
- **1.1.36.** "Change of Control" means, with respect to a Party, (a) a merger or consolidation of such Party with a Third Party that results in the voting securities of such Party outstanding immediately prior thereto, or any securities into which such voting securities have been converted or exchanged, ceasing to represent at least 50% of the combined voting power of the surviving entity or the parent of the surviving entity immediately after such merger or consolidation, (b) a transaction or series of related transactions in which a Third Party, together with its Affiliates, becomes the direct or indirect beneficial owner of more than 50% of the combined voting power of the outstanding securities of such Party, or (c) the sale or other transfer to a Third Party of all or substantially all of such Party's and its controlled Affiliates' assets.
- **1.1.37.** "Clinical Study" means a Phase 1 Study, Phase 2 Study, Phase 3 Study, Post-Marketing Study, Supplemental Study or other study (including a non-interventional study) in humans to obtain information regarding a product, including information relating to the safety, tolerability, pharmacological activity, pharmacokinetics, dose ranging or efficacy of the product.
  - **1.1.38.** "Clinical Trial Budget" has the meaning set forth in Section 5.1.2.2.
  - **1.1.39.** "CMC" means chemistry, manufacturing and controls.
  - **1.1.40.** [...\*\*\*...].
  - **1.1.41.** "Co-Detail Notice" has the meaning set forth in Section 6.3.2.1.

- **1.1.42.** "Co-Detail Option Exercise Notice" has the meaning set forth in Section 6.3.2.1.
- **1.1.43.** "Co-Detail Option" has the meaning set forth in Section 6.3.2.1.
- **1.1.44.** "Co-Detailed Optioned Licensed Product" has the meaning set forth in Section 6.3.2.1.
- 1.1.45. [...\*\*\*...].
- **1.1.46.** "Collaboration" means the collaboration of the Parties under this Agreement, including the Research, Development, Commercialization and Manufacture of Licensed Antibodies and Licensed Products (other than Fc Licensed Antibodies and Fc Licensed Products) in the Field.
  - **1.1.47.** "Collaboration In-License" has the meaning set forth in Section 9.5.1.4(a).
- **1.1.48.** "Collaboration IP" means Novartis Collaboration IP, Xencor Collaboration IP, and each Party's interest in Joint Collaboration IP.
- **1.1.49.** "Combination Product" means any pharmaceutical preparations, in any dosage strengths, formulations and methods of administration, that combine the Licensed Product and one or more other active ingredients (other than another Licensed Product) in fixed dose combination, whether co-formulated or co-packaged. The Parties agree that nothing in this definition is intended to limit the possibility of the Parties from Developing, Manufacturing or Commercializing, as part of this Agreement, any product comprising a combination of two or more Licensed Products.
- **1.1.50.** "Combination Therapy" means a therapy that includes a Licensed Product and at least one additional active ingredient that is not a Licensed Product (an "Other Product") sold separately but approved (or being Developed for approval) for use in combination, whether sold at a single price point or under separate price points or as part of a course of treatment.
- **1.1.51.** "Commercialization" or "Commercialize" means any and all activities directed to marketing, promoting, distributing, importing, exporting, using, offering to sell or selling a product, and activities directed to obtaining Pricing Approvals, as applicable.
  - 1.1.52. "Commercialization Costs" [...\*\*\*...].

- **1.1.53.** "Commercialization Cost Calculation Report" has the meaning set forth in Section 6.3.4.
- **1.1.54.** "Commercialization Plans" means, collectively, the Novartis Territory Commercialization Plans, the Xencor Territory Commercialization Plans and the OLP Commercialization Plan.
- **1.1.55.** "Commercially Reasonable Efforts" means, with respect to a Party, the efforts and resources typically used by biotechnology or pharmaceutical companies similar in size and scope to such Party to perform the obligation at issue, which efforts will not be less than those efforts made with respect to other products at a similar stage of development or in a similar stage of product life, with similar developmental risk profiles, of similar market and commercial potential, taking into account the proprietary position of the products relative to the products of Third Parties, the regulatory structure involved, Regulatory Authority-approved labeling, product profile, the profitability of the applicable products, issues of safety and efficacy, the likely timing of the product's entry into the market, the likelihood of receiving Regulatory Approval, and other relevant scientific, technical and commercial factors. [...\*\*\*...].
- **1.1.56.** "Committee" means the Joint Steering Committee, Joint Development Committee or Joint Commercialization Committee, or any other subcommittee established under Section 2.2.3.13, as applicable.
- **1.1.57.** "Competing Program" means a Xencor Competing Program or a Novartis Competing Program, as applicable.
- 1.1.58. "Confidential Information" means any and all confidential or proprietary information and data and all other scientific, pre-clinical, clinical, regulatory, manufacturing, marketing, financial and commercial information or data, whether communicated in writing or orally or by any other method, which is or has been provided by or on behalf of one Party to the other Party or its designee in connection with this Agreement. "Confidential Information" includes the following, which are transferred, disclosed or made available by the disclosing Party: confidential and proprietary technical and commercial information, Know-How, amino acid and nucleic acid sequences, biochemical, cell based and animal assays, animal models, dosages, dosage schedules, drawings, specifications, models and/or designs relating to development, manufacture, production, registration, promotion, distribution, marketing, performance or sale(s); experimental, manufacturing, process, analytical, packaging, product,

warehousing, quality control and quality assurance and marketing specifications, standards, procedures, processes, methods, instructions and techniques, samples, prototypes, formulae, writings of any kind, opinions or otherwise unwritten data or in the form of computer software or computer programs; biological, chemical or physical materials provided under this Agreement; and reports provided under this Agreement.

**1.1.59.** "Control" means, with respect to any Patents, Know-How or other intellectual property right, the possession, legal authority or right (whether by ownership, license or sublicense, other than by a license, sublicense or other right granted (but not assignment) pursuant to this Agreement) by a Party of the ability to assign or grant to the other Party the licenses, sublicenses or rights to access and use or disclose such Patents or Know-How as provided for in this Agreement, without, other than with respect to any In-Licenses, paying any consideration to any Third Party (now or in the future) or violating the terms of any agreement or other arrangement with any Third Party in existence as of the time such Party would be required hereunder to grant such license, sublicense or rights of access and use. Notwithstanding anything in this Agreement to the contrary, a Party will be deemed not to Control any Patents or Know-How that are owned or in-licensed by an Acquirer except (a) with respect to any such Patents or Know-How arising from active participation by employees or consultants of the Acquirer in the Collaboration after such Change of Control, which Patents and Know-How shall be owned in accordance with Section 14.2, (b) to the extent that any such Patents or Know-How are included in or used in furtherance of the Collaboration by the Acquirer after such Change of Control, which Patents and Know-How shall be owned in accordance with Section 14.2, or (c) for Know-How and Patents constituting improvements (or direct improvements to such improvements) to the Xencor Technology or the Novartis Technology (as applicable) in existence prior to such Change of Control developed or conceived by any employees or consultants of the Acquirer in connection with the Collaboration, which Patents and Know-How shall be owned in accordance with Section 14.2.

**1.1.60.** "Cost of Goods Sold" or "COGS" means costs incurred by a Party in the actual Manufacturing of the Optioned Licensed Product, established on a regular, standard basis in accordance with the Manufacturing Party's Accounting Standards as consistently applied by such Party, expressed on a per unit basis. COGS shall include the following elements:

```
1.1.60.1. [...***...];
1.1.60.2. [...***...];
1.1.60.3. [...***...]
```

```
[...***...];

1.1.60.4. [...***...];

1.1.60.5. [...***...]; and

1.1.60.6. [...***...].
```

- **1.1.61.** "Cover", "Covering" or "Covered" means, with respect to each Licensed Product, that, but for a license granted to a Person under a claim included in a Patent, the Research, Development, Manufacture, or Commercialization of such Licensed Product in the Field in the Territory by such Person would infringe, or contribute to or induce the infringement of, such claim, or with respect to a Patent application, as if such claim was contained in an issued Patent.
  - **1.1.62.** "CREATE Act" has the meaning set forth in Section 14.1.2.
- **1.1.63.** "Current ex-US RLP Patents" mean those Patents within Xencor Technology in existence as of the Effective Date that solely and specifically Cover the sale, offer for sale, manufacture, use or import of any Regional Licensed Product in the Novartis Territory, including, to the extent possible, Xencor Patents bifurcated after the Effective Date in accordance with Section 14.2.4.

-9-

\*\*\*Confidential Treatment Requested

- **1.1.64.** "**Detail"** or "**Detailing**" means, with respect to each Optioned Licensed Product launched by Novartis in the United States, a selling presentation for such product by a representative of each Party's sales force, or another employee of each Party who may be deemed to be part of the Commercialization effort for such Optioned Licensed Product (e.g., such as a key account manager, etc.).
- **1.1.65.** "**Develop**" and "**Development**" means any and all clinical drug development activities conducted before or after obtaining Regulatory Approval that are reasonably related to or leading to the development, preparation and submission of data and information to a Regulatory Authority for the purpose of obtaining, supporting or expanding Regulatory Approval or to the appropriate body for obtaining, supporting or expanding Pricing Approval, including all activities related to pharmacokinetic profiling, design and conduct of Clinical Studies, regulatory affairs, statistical analysis, report writing, and regulatory filing creation and submission (including the services of outside advisors and consultants in connection therewith).
- **1.1.66.** "Development Budget" means the OLP Development Budget and the RLP Development Budget.
- 1.1.67. "Development Costs" means, with respect to each of the Regional Target Pairs and the Optioned Target Pair, those costs and expenses directly incurred in connection with the performance of any Development activities for the applicable Licensed Antibodies or Licensed Products in accordance with the applicable Development Plans, including FTE Costs, fees charged by Third Party service providers, costs related to study drug, placebo, or comparator supply, and other Out-of-Pocket Costs reasonably incurred in connection with the performance of any Clinical Study with respect to such Licensed Antibodies or Licensed Products, costs related to preparing and filing applications for Regulatory Approval or submissions to Regulatory Authorities (including associated filing fees, translation expenses and legal and other professional service fees), but excluding (a) overhead costs and capital expenditures, and (b) costs associated with Supplemental Studies (other than for Supplemental Studies added to the RLP Development Plan for such Regional Target pursuant to Section 5.1.2.4).
- **1.1.68.** "Development Plan" means each of the RLP Development Plan and the OLP Development Plan.
  - **1.1.69.** "Developmental Milestone Event" has the meaning set forth in Section 10.4.
  - **1.1.70.** "Developmental Milestone Payment" has the meaning set forth in Section 10.4.
  - **1.1.71.** "**Disputes**" has the meaning set forth in Section 16.4.1.
  - **1.1.72.** "**Dollars**" or "\$" means the legal tender of the United States of America.
  - **1.1.73.** "Effective Date" has the meaning set forth in the preamble.

- **1.1.74.** "EMA" means the European Medicines Agency and any successor Governmental Authority having substantially the same function.
- **1.1.75.** "EU" means the European Union, as its membership may be constituted from time to time, and any successor thereto; provided, that, for purposes of this Agreement the EU will be deemed to include any of France, Germany, Italy, Spain, or the United Kingdom, should any such country leave the European Union.
  - **1.1.76.** "Excluded Target Pairs" means each of those Target pairs set forth on Schedule 1.1.76.
  - **1.1.77.** "Exclusivity Period" means, [...\*\*\*...].
- **1.1.78.** "Executive Officer" means, for Xencor, its Chief Executive Officer, and for Novartis, its President or another senior executive designee with responsibilities and seniority comparable thereto, provided that any of the foregoing individuals may designate the Chief Financial Officer of his/her Party as his/her designee for financial related matters. In the event that the position of any of the Executive Officers no longer exists due to a Change of Control, corporate reorganization, corporate restructuring or the like, the applicable Executive Officer will be replaced with another executive officer with responsibilities and seniority comparable to the eliminated Executive Officer.
- **1.1.79.** "Existing Novartis In-License" means (a) any agreements entered into by Novartis or an Affiliate with a Third Party prior to the Effective Date pursuant to which Novartis or any of its Affiliates Controls any Novartis Technology, and (b) any agreement entered into by Novartis or an Affiliate with a Third Party pursuant to Sections 3.1.9, 5.1.6, 5.2.5 and 5.3.6, which are deemed to be Existing Novartis In-Licenses under such Sections; and, with respect to both (a) and (b) any amendments or restatements thereto during the Term in accordance with Section 12.5.3, but excluding any Collaboration In-License to which Novartis or its Affiliates is a party.
- **1.1.80.** "Existing Xencor In-License" means (a) any agreements entered into by Xencor or an Affiliate with a Third Party prior to the Effective Date as set forth on Exhibit A-3 pursuant to which Xencor or any of its Affiliates Controls any Xencor Technology, and (b) any agreement entered into by Xencor or an Affiliate with a Third Party pursuant to Sections 3.1.9, 5.1.6 and 5.3.6, which are deemed to be Existing Xencor In-Licenses under such Sections; and, with respect to both (a) and (b) any amendments or restatements thereto during the Term in

accordance with Section 12.5.3, but excluding any Collaboration In-License to which Xencor or its Affiliates is a party.

- **1.1.81.** "Expedited Arbitration" has the meaning set forth in Section 16.4.4.1.
- **1.1.82.** "Expedited Dispute" has the meaning set forth in Section 16.4.4.1.
- **1.1.83.** "Fc Gatekeeper" has the meaning set forth in 3.2.1.1.
- **1.1.84.** "Fc GLP Tox Milestone" has the meaning set forth in Section 10.4.4.
- **1.1.85.** "Fc Know-How" means Know-How Controlled by Xencor or its Affiliates during the Fc Target Selection Period that is reasonably necessary to Research, Develop, Manufacture or Commercialize Fc Licensed Antibodies or Fc Licensed Products in the Field in the Territory.
- **1.1.86.** "Fc Licensed Antibody" means an Antibody (other than a Regional Licensed Antibody, Global Licensed Antibody or Optioned Licensed Antibody) for which all of the following hold true: (a) Novartis has provided timely written notice under Section 3.2.1.3 identifying a Target (including a Target that would be Specifically Bound by a Bispecific Antibody) that it wishes to designate as a Fc Target, (b) the Antibody must contain an Fc Licensed Component, and (c) the Antibody must not Specifically Bind any other Target other than the Fc Target that is the subject of the foregoing subclause (a).
  - 1.1.87. "Fc Licensed Component" means [...\*\*\*...].
- **1.1.88.** "Fc Licensed Product" means on an Fc Target-by-Fc Target basis, a product containing an Fc Licensed Antibody as an active ingredient, and all formulations, dosages and delivery systems thereof; provided that, an Fc Licensed Product does not include any Antibodies, compounds or products of Xencor or any of its Affiliates other than the Fc License Antibody, Global Licensed Product or Regional Licensed Product, as applicable, contained in or comprising such Fc Licensed Product.
  - **1.1.89.** "Fc Notice" has the meaning set forth in Section 3.2.1.3.
- **1.1.90.** "Fc Patents" means those Patents listed on Exhibit A-1 and any Patents Controlled by Xencor or its Affiliates during the Term that Cover any Fc Licensed Component in the Field in the Territory.

- **1.1.91. "Fc Region"** means the [...\*\*\*...] and any variant, fragment or portion thereof, including naturally occurring fragments, naturally occurring variants of such fragments and non-naturally occurring variants of such fragments.
- **1.1.92.** "Fc Target" means a target that is selected by Novartis and confirmed by the Fc Gatekeeper pursuant to Section 3.2.1.
- **1.1.93.** "Fc Target Selection Period" means the period beginning on the Effective Date and ending upon the [...\*\*\*...] anniversary of the Effective Date.
  - **1.1.94.** "Fc Target Unavailable List" has the meaning set forth in Section 3.2.1.2.
  - **1.1.95.** "Fc Technology" means the Fc Know-How and Fc Patents.
- **1.1.96.** "FDA" means the United States Food and Drug Administration or any successor agency thereto.
  - **1.1.97.** "FD&C Act" means the United States Federal Food, Drug and Cosmetic Act, as amended.
- **1.1.98.** "Field" means the treatment, diagnosis (including companion diagnostics) or prevention of all human or animal diseases or disorders.
  - **1.1.99.** "**Finance Officers**" has the meaning set forth in Section 10.3.1.
- **1.1.100.** "First Commercial Sale" means, on a Licensed Product-by-Licensed Product, and country-by-country basis, the first commercial sale in an arms'-length transaction of a Licensed Product to a Third Party by a Party or any of its Related Parties in such country following receipt of applicable Regulatory Approval of such Licensed Product in such country. For clarity, the First Commercial Sale shall not include (a) any distribution or other sale solely for patient assistance, named patient use, compassionate use, or test marketing programs or non-registrational studies or similar programs or studies where the Licensed Product is supplied without charge or at the actual manufacturing cost thereof (without allocation of indirect costs or any markup); or (b) any sale by a Party to its Affiliates or Sublicensees.
- **1.1.101. "FTE**" means a full-time employee, or in the case of less than a full-time employee, a full-time equivalent employee year, carried out by an appropriately qualified employee of a Party or its Related Parties, based on […\*\*\*...] person-hours per year. For clarity, indirect personnel (including support functions such as managerial, financial, legal or business development) shall not constitute FTEs.

- **1.1.102.** "FTE Costs" means, for any period, the FTE Rate multiplied by the number of FTEs in such period.
- **1.1.103.** "FTE Rate" means \$[...\*\*\*...] per one full FTE per full 12 month Calendar Year, which rate includes all direct and indirect costs of a Party's FTE, including personnel and travel expenses, <u>provided</u> that, starting January 1, 2017, such rate will adjust on January 1 of each Calendar Year by an amount equal to the change, if any, in the Consumer Price Index for All Urban Consumers (CPI-U) for the U.S. City Average, 1982-84 = 100, calculated by the Bureau of Labor Statistics during the immediately preceding Calendar Year. Notwithstanding the foregoing, for any Calendar Year during the Term that is less than a full year, the above referenced rate will be proportionately reduced to reflect such portion of FTEs for such full Calendar Year.
  - **1.1.104.** "**Fv Region**" means an [...\*\*\*...].
- **1.1.105.** "GAAP" means generally accepted accounting principles as practiced in the United States, as consistently applied.
- **1.1.106.** "Global Licensed Antibody" means, on a Global Target Pair-by-Global Target Pair basis, any Bispecific Antibody that (a) Specifically Binds to such Global Target Pair and (b) is (i) Researched by or on behalf of Xencor or its Related Parties under a Research Program, and/or (ii) Researched, Developed or Commercialized by or on behalf of Novartis or its Related Parties pursuant to this Agreement, <u>provided</u> that after a Change of Control of a Party, [ ...\*\*\*...].
- **1.1.107.** "Global Licensed Product" means, on a Global Target Pair-by-Global Target Pair basis, a product containing or comprising (or using as part of a Combination Product) a Global Licensed Antibody as an active ingredient, and all formulations, dosages and delivery systems thereof; <u>provided</u> that, a Global Licensed Product does not include any Antibodies, compounds or products of Xencor or any of its Affiliates other than the Global Licensed Antibody contained in or comprising such Global Licensed Product.
- **1.1.108.** "Global Target Pair" means [...\*\*\*...] or such other Target pair as is identified or substituted pursuant to Section 3.1, subject to a total of four such Target pairs at

any one time or, in the case where Xencor has exercised its Opt-In Right and not exercised its Opt-Out Right pursuant to Section 5.3.7, three such Target pairs at any one time.

- **1.1.109. "Global Target Pair Selection/Replacement Period**" means the period beginning on the Effective Date and ending upon the [...\*\*\*...] of the Effective Date.
- **1.1.110.** "GLP Toxicology Study" means a toxicology study, in species that satisfies applicable regulatory requirements, using applicable cGLP that meets the standard necessary for submission as part of an IND Filing with the applicable Regulatory Authority.
- **1.1.111.** "Governmental Authority" means any applicable government authority, court, tribunal, arbitrator, agency, department, legislative body, commission or other instrumentality of (a) any government of any country or territory, (b) any nation, state, province, county, city or other political subdivision thereof or (c) any supranational body.
- **1.1.112.** "**Gross Profits**" means Net Sales in the United States less Cost of Goods Sold in the United States for sales of Optioned Licensed Products by Novartis or its Related Parties to Third Parties in the United States.
- **1.1.113.** "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.
- **1.1.114.** "**IFRS**" means International Financial Reporting Standards, the set of accounting standards and interpretations as promulgated by the International Standards Accounting Board and as they may be updated for time to time, as consistently applied.
- **1.1.115.** "**IND**" means an investigational new drug application, clinical trial application or similar application or submission for approval to conduct human clinical investigations filed with or submitted to a Regulatory Authority in conformance with the requirement of such Regulatory Authority, and any amendments thereto.
- **1.1.116.** "**IND Acceptance**" means the acceptance by a Regulatory Authority in a Major Market Country of an IND.
- **1.1.117.** "**IND Filing**" means the filing with a Regulatory Authority in a Major Market Country of an IND.
  - **1.1.118.** "Indemnified Party" has the meaning set forth in Section 13.4.
  - **1.1.119.** "**Indemnifying Party**" has the meaning set forth in Section 13.4
- **1.1.120.** "**Indication**" means a disease or pathological condition for which a BLA application or an efficacy supplement (or other addition) to an existing BLA application is required for the purpose of obtaining Regulatory Approval in a country.

- **1.1.121.** "Infringement" means, where the making, using, selling, offering for sale or importing by any Third Party (other than any Sublicensee or authorized purchaser or other transferee of a Licensed Product) of any pharmaceutical product is Covered by a Valid Claim of a Patent licensed under Xencor Technology or Novartis Technology. For clarity, the filing of a Biosimilar Application with respect to a Licensed Product as the reference product by any such Third Party will be deemed to be Infringement.
- **1.1.122.** "**Initiation**" means, with respect to a Clinical Study of a Licensed Product, the first dosing of the first human subject pursuant to the protocol for such Clinical Study.
- **1.1.123.** "**In-Licenses**" means, collectively, all Existing Xencor In-Licenses and all Existing Novartis In-Licenses.
  - **1.1.124.** "**Internal Program**" means, [...\*\*\*...].
  - **1.1.125.** "**Invention Period**" means, [...\*\*\*...].
- **1.1.126.** "**IP Committee**" means the intellectual property advisory committee as more fully described in Section 14.3.1.
- **1.1.127.** "JCC" or "Joint Commercialization Committee" has the meaning set forth in Section 2.4.1.
  - **1.1.128.** "JDC" or "Joint Development Committee" has the meaning set forth in Section 2.3.1.
  - **1.1.129.** "Joint Collaboration IP" has the meaning set forth in Section 14.2.3.
  - **1.1.130.** "JSC" has the meaning set forth in Section 2.2.1.
- **1.1.131.** "Know-How" means all commercial, technical, scientific and other know-how and information, amino acid and nucleic acid sequences, biochemical, cellular and animal assays, animal models, trade secrets, knowledge, technology, methods, processes, practices, formulae, instructions, skills, techniques, procedures, experiences, ideas, technical

assistance, designs, drawings, assembly procedures, computer programs, specifications, data and results (including biological, chemical, pharmacological, toxicological, pharmaceutical, physical and analytical, preclinical, clinical, safety, manufacturing and quality control data and know-how, including regulatory data, study designs and protocols), and Materials, in all cases, whether or not confidential, proprietary, patented or patentable, in written, electronic or any other form now known or hereafter developed.

- 1.1.132. "Landed Costs" mean, [...\*\*\*...].
- **1.1.133.** "Laws" means all applicable laws, statutes, rules, regulations, orders, judgments, injunctions, ordinances or other pronouncements having the binding effect of law of any Governmental Authority.
  - **1.1.134.** "Lead Party" has the meaning set forth in Section 14.4.2.4(a).
- **1.1.135.** "Licensed Antibody" means any Regional Licensed Antibody, Global Licensed Antibody, Optioned Licensed Antibody or Fc Licensed Antibody. For clarity, a Licensed Antibody shall be synonymous with the Licensed Product containing or comprising such Licensed Antibody.
- **1.1.136.** "Licensed Product" means any Regional Licensed Product, Global Licensed Product, Optioned Licensed Product or Fc Licensed Product.
- **1.1.137.** "Licensed Target Pair" means any Regional Target Pair, Global Target Pair or Optioned Target Pair.
  - **1.1.138.** "Losses" has the meaning set forth in Section 13.1.
- **1.1.139.** "Loss of Market Exclusivity" means, with respect to any Licensed Product in any country, the following has occurred (a) the Net Sales of such Licensed Product in that country in any Calendar Year are less than [...\*\*\*...] and (b) the decline in such sales is [...\*\*\*...].
  - **1.1.140.** "MAA" has the meaning set forth in Section 1.1.25.
- **1.1.141. "Major Market Countries"** means the United States, France, Germany, Italy, Spain, the United Kingdom and Japan.
- **1.1.142.** "Manufacturing" or "Manufacture" means all activities related to the manufacture of Licensed Antibodies or Licensed Products, including manufacturing supplies for Research, Development or Commercialization, packaging, in-process and finished product

testing, release of product or any component or ingredient thereof, quality assurance and quality control activities related to manufacturing and release of product, ongoing stability tests, storage, shipment, and regulatory activities related to any of the foregoing.

- **1.1.143.** "Manufacturing Proposing Party" has the meaning set forth in Section 8.3.
- **1.1.144.** "Materials" mean all tangible compositions of matter, devices, articles of manufacture, assays, biological, chemical or physical materials and other similar materials.
  - **1.1.145.** "material communications" has the meaning set forth in Section 7.1.1.1(b).
- **1.1.146.** "Medical Affairs Activities" means design, strategies, oversight and implementation of activities designed to ensure or improve appropriate medical use of, conduct medical education of, or further research regarding, a Licensed Product, including by way of example (i) activities of medical liaisons, (ii) grants to support continuing independent medical education (including independent symposia and congresses), and (iii) development, publication and dissemination of publications in support of an approved indication for a Licensed Product, as well as medical information services (and the content thereof) provided in response to inquiries communicated via the sales representatives or received by letter, phone call or email.
  - **1.1.147.** "Milestone Payments" has the meaning set forth in Section 10.5.
- **1.1.148.** "Net Sales" means the net sales recorded by or on behalf of Novartis or any of its Affiliates or Sublicensees, excluding distributors and wholesalers, for any Licensed Product sold to Third Parties other than Sublicensees, as determined in accordance with Novartis' Accounting Standards as consistently applied, less a deduction of [...\*\*\*...] percent ([...\*\*\*...]%) for direct expenses related to the sales of such Licensed Product, distribution and warehousing expenses and uncollectible amounts on previously sold products. The deductions booked on an accrual basis by Novartis and its Affiliates and Sublicensees under its Accounting Standards to calculate the recorded net sales from gross sales include the following:

```
1.1.148.1. [...***...],
1.1.148.2. [...***...],
1.1.148.3. [...***...],
1.1.148.4. [...***...],
1.1.148.5. [...***...],
```

```
1.1.148.6. [...***...],
1.1.148.7. [...***...]; and
1.1.148.8. [...***...].
1.1.148.9. [...***...];
(a) [...***...];
(b) [...***...];
```

In the event the Licensed Product is sold as a Combination Product, the Net Sales of the Licensed Product, for the purposes of determining royalty payments, shall be determined in good faith by multiplying the Net Sales of the Combination Product by the fraction A/(A+B) where A is the weighted (by sales volume) average sale price in a particular country of the Licensed Product containing the Licensed Antibody as the sole active ingredient when sold separately in finished form and B is the weighted (by sales volume) average sale price in that country of the products containing the other component(s) as the sole active ingredient(s) when sold separately in finished form. Regarding prices comprised in the weighted average price when sold separately referred to above, if these are available for different dosages from the dosages of Licensed Antibody and other active ingredient components that are included in the Combination Product, then Novartis shall be entitled to make a proportional adjustment to such prices in calculating the royalty-bearing Net Sales of the Combination Product. In the event that such weighted average sale price cannot be determined for both the Licensed Product and the other product(s) in combination, the calculation of Net Sales for purposes of determining royalty payments shall be agreed by the Parties based on the relative value contributed by each component, such agreement not to be unreasonably withheld or delayed.

(d) [...\*\*\*...]

[...\*\*\*...].

- **1.1.149.** "New RLP Patents" mean those Patents within Xencor Technology created after the Effective Date that solely and specifically Cover the sale, offer for sale, manufacture, use or import of any Regional Licensed Product in the Novartis Territory.
  - **1.1.150.** "Non-Bankrupt Party" has the meaning set forth in Section 9.6.
  - **1.1.151.** "Non-Proposing Party" has the meaning set forth in Section 5.1.2.4(a).
  - **1.1.152.** "Non-Withholding Party" has the meaning set forth in Section 10.9.4.
- **1.1.153.** "Novartis Biological Material Inventions" means, all Know-How (and Patents specifically claiming inventions therein) first developed or conceived as part of the Collaboration during the Term, whether by on behalf of employee(s), agent(s) or consultant(s) of Xencor or its Affiliates or Novartis or its Affiliates, individually or jointly, that primarily relates to any Novartis Biological Material IP, including any Patent that claims inventions therein.
- **1.1.154.** "Novartis Biological Material IP" means all Know-How, including all tangible biological materials, vectors, cell banks, cells (including any derivatives or progeny thereof), cell culture processes and purification processes, reference standards, and cell culture media (including any Patents covering such Know-How) used in Manufacturing products that is Controlled by Novartis or its Affiliates as of the Effective Date, or any improvement developed thereto outside of the Collaboration.
  - **1.1.155.** "Novartis Collaboration IP" has the meaning set forth in Section 14.2.2.
  - **1.1.156.** "Novartis Competing Program" has the meaning set forth in Section 12.6.2.2(a).
- **1.1.157.** "Novartis Core Inventions" mean, (a) Novartis Biological Material Inventions; and (b) all other Know-How first developed or conceived as part of the Collaboration during the applicable Invention Period, whether by or on behalf of employee(s), agent(s) or consultant(s) of Xencor or its Affiliates or Novartis or its Affiliates, individually or jointly, that specifically and solely relates to a Licensed Target Pair, Licensed Antibody or Licensed Product, including any Patent that specifically and solely claims such Know-How therein.

- **1.1.158.** "Novartis GLP and OLP Infringement" means any Infringement by a Third Party with respect to a Global Licensed Product or Optioned Licensed Product in the Novartis Territory.
  - **1.1.159.** "Novartis Indemnitees" has the meaning set forth in Section 13.2.
- **1.1.160.** "**Novartis In-Licenses**" means any Existing Novartis In-License or any Collaboration In-License to which Novartis is a party.
- **1.1.161.** "Novartis Know-How" means Know-How Controlled by Novartis or its Affiliates as of the Effective Date or during the applicable Invention Period to the extent contributed to the Collaboration by Novartis and that is reasonably necessary for Xencor to Research, Develop or Commercialize Licensed Antibodies or Licensed Products (other than Fc Licensed Antibodies and Fc Licensed Products) in the Field, other than Novartis' interest in Joint Collaboration IP and Novartis Collaboration IP. For clarity, Novartis will be deemed to have contributed Know-How to the Collaboration to the extent that it is (i) provided to Xencor or (ii) used by Novartis in the performance of its obligations or exercise of its rights under this Agreement.
- **1.1.162.** "Novartis Patents" means those Patents Controlled by Novartis or its Affiliates as of the Effective Date or during the applicable Invention Period to the extent contributed to the Collaboration by Novartis and that are reasonably necessary to Research, Develop or Commercialize Licensed Antibodies or Licensed Products in the Field. Novartis Patents excludes Patents included in Novartis Collaboration IP and Novartis' interest in Joint Collaboration IP. For clarity, Novartis will be deemed to have contributed Patents to the Collaboration to the extent that it is (i) provided to Xencor or (ii) used by Novartis in the performance of its obligations or exercise of its rights under this Agreement.
- **1.1.163.** "**Novartis RLP Product Infringement**" means any Infringement by a Third Party with respect to a Regional Licensed Product (excluding any Fc Licensed Product) in the Novartis Territory.
  - **1.1.164.** "Novartis RLP Trademarks" has the meaning set forth in Section 14.8.1.2.
- **1.1.165.** "Novartis Technology" means, collectively, Novartis Know-How, Novartis Patents, Novartis Collaboration IP and Novartis' interest in Joint Collaboration IP; <u>provided</u> however, that Novartis Technology shall only include Novartis Biological Material IP to the extent contributed to the Collaboration by Novartis. For clarity, Novartis will be deemed to have contributed Novartis Biological Material IP to the Collaboration to the extent that it is (i) provided to Xencor or (ii) used by Novartis in the performance of its obligations or exercise of its rights under this Agreement.
- **1.1.166.** "Novartis Territory" means (a) with respect to any Global Licensed Product, Optioned Licensed Product or Fc Licensed Product, worldwide, and (b) with respect to

any Regional Licensed Product, all countries and territories of the world other than the Xencor Territory.

- **1.1.167.** "Novartis Territory Commercialization Plan" has the meaning set forth in Section 6.1.3.
- **1.1.168.** "**OLP Commercialization Plan**" has the meaning set forth in Section 6.3.3.
- **1.1.169.** "**OLP Development Budget**" has the meaning set forth in Section 5.3.2.2.
- **1.1.170.** "OLP Development Plan" has the meaning set forth in Section 5.3.2.1.
- **1.1.171.** "**Opt-In Right**" has the meaning set forth in Section 4.1.1.
- **1.1.172.** "**Opt-Out Right**" means each of Xencor's rights to opt-out of the Development and Commercialization of all Regional Licensed Antibodies and Regional Licensed Products for a Regional Target Pair in accordance with Section 5.1.8 or all Optioned Licensed Antibodies and Optioned Licensed Products for the Optioned Target Pair in accordance with Section 5.3.7.
- **1.1.173.** "**Option Package**" means, with respect to a given Global Licensed Product that Specifically Binds the applicable Global Target Pair, (a) the documents and information provided by or on behalf of Novartis or its Affiliates to the Regulatory Authority or from such Regulatory Authority to Novartis in connection with any IND Filing for such Global Licensed Product; and (b) the proposed development plan [...\*\*\*...] in connection with such Global Target Pair.
- **1.1.174.** "**Optioned Licensed Antibody**" means any Bispecific Antibody that (a) Specifically Binds to the Optioned Target Pair, and (b) (i) was Researched by or on behalf of Xencor or its Related Parties pursuant to this Agreement under the applicable Research Program prior to being selected as an Optioned Target Pair and/or (ii) is Researched, Developed or Commercialized by or on behalf of Novartis or its Related Parties pursuant to this Agreement, <u>provided</u> that after a Change of Control of a Party, [...\*\*\*...]

- **1.1.175.** "Optioned Licensed Product" means a product containing or comprising (or using as part of a Combination Product) an Optioned Licensed Antibody as an active ingredient, and all formulations, dosages and delivery systems thereof; <u>provided</u> that, an Optioned Licensed Product does not include any Antibodies, compounds or products of Xencor or any of its Affiliates other than the Optioned Licensed Antibody contained in or comprising such Optioned Licensed Product.
  - **1.1.176.** "Optioned Target Pair" has the meaning set forth in Section 4.1.1.
  - **1.1.177.** "Other Combination Therapy Product" has the meaning set forth in Section 5.1.7.
  - **1.1.178.** "Other Product" has the meaning set forth in Section 1.1.50.
- 1.1.179. "Out-of-Pocket Costs" means, with respect to certain activities hereunder, direct expenses paid or payable by either Party or its Affiliates to Third Parties and specifically identifiable and incurred to conduct such activities for a Regional Licensed Antibody or a Regional Licensed Product, or an Optioned Licensed Antibody or an Optioned Licensed Product, as applicable, including payments to contract personnel (including contractors, consultants and subcontractors) in each case, pursuant to the applicable Development Plan or Commercialization Plan.
- **1.1.180."Patent**" means all patents and patent applications and all substitutions, divisions, continuations, continuations-in-part, any patent issued with respect to any such patent applications, any reissue, reexamination, utility models or designs, renewal or extension (including any supplementary protection certificate) of any such patent, and any confirmation patent or registration patent or patent of addition based on any such patent, and all counterparts thereof in any country.
- **1.1.181.** "Patent Costs" means the out-of-pocket costs and expenses paid to outside legal counsel and other Third Parties (including to any licensor pursuant to any in-license), and filing and maintenance expenses, incurred in Prosecuting and Maintaining Patents and enforcing and defending them.
- **1.1.182.** "**Person**" means any natural person, corporation, unincorporated organization, partnership, association, sole proprietorship, joint stock company, joint venture, limited liability company, trust or government, or Governmental Authority, or any other similar entity.
- **1.1.183.** "**Phase 1 Study**" means a clinical study of an investigational product in human patients with the primary objective of characterizing its safety, tolerability, and

pharmacokinetics and identifying a recommended dose and regimen for future studies as described in 21 C.F.R. 312.21(a), or a comparable Clinical Study prescribed by the relevant Regulatory Authority in a country other than the United States. The investigational product can be administered to patients as a single agent or in combination with other investigational or marketed agents and a Phase 1 Study shall be deemed commenced when Initiated.

- **1.1.184.** "Phase 2 Study" means a clinical study of an investigational product in human patients with the primary objective of characterizing its activity in a specific disease state as well as generating more detailed safety, tolerability, and pharmacokinetics information as described in 21 C.F.R. 312.21(b), or a comparable Clinical Study prescribed by the relevant Regulatory Authority in a country other than the United States including a human clinical trial that is also designed to satisfy the requirements of 21 C.F.R. 312.21(a) or corresponding foreign regulations and is subsequently optimized or expanded to satisfy the requirements of 21 C.F.R. 312.21(b) (or corresponding foreign regulations) or otherwise to enable a Phase 3 Clinical Study (e.g., a phase 1/2 trial). The investigational product can be administered to patients as a single agent or in combination with other investigational or marketed agents and a Phase 2 Study shall be deemed commenced when Initiated.
- **1.1.185.** "Phase 2b Study" means a phase 2b Study carried out prior to the initiation of pivotal Phase 3 Studies that is intended to be the definitive dose range finding study in patients with efficacy as primary endpoint, as well as safety, initiated after completion of a Phase I Clinical Study (or phase IIa Clinical Study, if performed), that will evaluate the dose-dependent effectiveness of a pharmaceutical product for a particular indication or indications in patients with the disease or condition under study, as well as to collect further adverse effects and safety data to assess the risks associated with the pharmaceutical product, and further pharmacokinetic data.
- **1.1.186.** "Phase 3 Study" means a clinical study of an investigational product in human patients that incorporates accepted endpoints for confirmation of statistical significance of efficacy and safety with the aim to obtain Regulatory Approval in any country as described in 21 C.F.R. 312.21I, or a comparable Clinical Study prescribed by the relevant Regulatory Authority in a country other than the United States. The investigational product can be administered to patients as a single agent or in combination with other investigational or marketed agents and a Phase 3 Study shall be deemed commenced when Initiated.
  - **1.1.187.** "PHSA" means the United States Public Health Service Act, as amended.
  - **1.1.188.** "Potential In-License" has the meaning set forth in Section 9.5.1.4(a).
- **1.1.189.** "Post-Marketing Study" means a non-human or human clinical study of a Licensed Product initiated after receipt of Regulatory Approval for such Licensed Product in a country or territory, that is required by the Regulatory Authority in such country or territory to maintain the Regulatory Approval for such Licensed Product in such country or territory, but excluding any Supplemental Study.

- **1.1.190.** "**Pricing Approval**" means such governmental approval, agreement, determination or decision establishing prices for a Licensed Product that can be charged or reimbursed in regulatory jurisdictions where the applicable Governmental Authorities approve or determine the price or reimbursement of pharmaceutical products.
- **1.1.191.** "**Pricing Matters**" means all issues and decisions regarding (a) price, price terms and other contract terms with respect to Licensed Product sales, including discounts, rebates, other price concessions and service fees to payors and purchasers and (b) reimbursement programs applicable to a Licensed Product. For clarity, Pricing Matters includes all financial issues and financial decisions with respect to contracting with managed care entities, hospitals, pharmacies, group purchasing organizations, pharmacy benefit managers and government, and specifically includes issues and decisions about the offer of discounts or rebates for formulary placement for Licensed Products.
  - **1.1.192.** "**Primary RLP Supplier**" has the meaning set forth in Section 8.3.
- **1.1.193.** "**Product Global Trademarks**" means the Trademarks used, or intended for use, in connection with the distribution, marketing, promotion and sale of the Global Licensed Products, Optioned Licensed Products and Fc Licensed Products by Novartis and its Related Parties in the Novartis Territory. Product Global Trademarks specifically exclude the corporate names and logos of the Parties and their Affiliates.
  - **1.1.194.** "**Promotional Materials**" has the meaning set forth in Section 6.1.5.2.
  - **1.1.195.** "**Proposed Fc Target**" has the meaning set forth in Section 3.2.1.3.
  - **1.1.196.** "**Proposed Global Target Pair**" has the meaning set forth in Section 3.1.2.
  - **1.1.197.** "**Proposing Party**" has the meaning set forth in Section 5.1.2.4(a).
  - **1.1.198.** "**Proprietary Product**" has the meaning set forth in Section 5.1.7.
  - **1.1.199.** "**Proposed Xencor Sublicense**" has the meaning set forth in Section 9.1.2.2(d).
- **1.1.200.** "Prosecution and Maintenance" means, with regard to a particular Patent, the preparation, filing, prosecution and maintenance of such Patent in any jurisdictions, as well as re-examinations, reviews, reissues and the like with respect to that Patent, together with the conduct of interferences, the defense of oppositions, defending post-grant reviews, defending inter partes reviews, and other similar proceedings with respect to that Patent and further including Patent management and litigation strategy. For clarity, Prosecution and Maintenance does not include instituting post-grant reviews or inter partes review with respect to Patents of Third Parties.

- **1.1.201.** "Public Official" means (i) any officer, employee (including physicians, hospital administrators, or other healthcare professionals), agent, representative, department, agency, de facto official, representative, corporate entity, instrumentality or subdivision of any government, military or international organization, including any ministry or department of health or any state-owned or affiliated company or hospital, or (ii) any candidate for political office, any political party or any official of a political party.
- **1.1.202.** "Regional Licensed Antibody" means, on a Regional Target Pair-by-Regional Target Pair basis, any Bispecific Antibody that (a) Specifically Binds to such Regional Target Pair, and (b) is (i) Controlled by Xencor or its Related Parties, including those Antibodies known as XmAb13676 and XmAb14045, and/or (ii) Researched, Developed or Commercialized by or on behalf of Novartis or its Related Parties pursuant to this Agreement, <u>provided</u> that after a Change of Control of a Party, no Antibodies of the Acquirer will be included as Regional Licensed Antibodies except to the extent [...\*\*\*...].
- **1.1.203.** "Regional Licensed Product" means, on a Regional Target Pair-by-Regional Target Pair basis, a product containing or comprising (or using as part of a Combination Product) a Regional Licensed Antibody as an active ingredient, and all formulations, dosages and delivery systems thereof; provided that, a Regional Licensed Product does not include any Antibodies, compounds or products of Xencor or any of its Affiliates other than the Regional Licensed Antibody contained in or comprising such Regional Licensed Product.
- **1.1.204.** "Regional Target Pair" means each of (a) the Target pair CD3xCD20, and (b) the Target pair CD3xCD123.
- **1.1.205.** "**Regulatory Approval**" means a BLA, together with all other approvals or establishment licenses, registrations or authorizations (including marketing authorizations) of any Regulatory Authority that may be necessary for the marketing, sale and commercialization of a pharmaceutical product in any country or region in the Territory.
- **1.1.206.** "Regulatory Authority" means any Governmental Authority involved in granting approvals for the Development, Manufacturing and Commercialization, including Pricing Approval of Licensed Products, including the FDA, the EMA, the Japanese Ministry of Health, Labour and Welfare and the Pharmaceuticals and Medical Devices Agency in Japan.

- **1.1.207.** "**Regulatory Materials**" means any regulatory application, submission, notification, communication, correspondence, registration, Regulatory Approvals and other filings made to, received from or otherwise conducted with a Regulatory Authority related to Developing, Manufacturing, obtaining marketing authorization, marketing, selling or otherwise Commercializing a pharmaceutical product in a particular country or jurisdiction.
  - **1.1.208.** "Related Party" means a Party's Affiliates and permitted Sublicensees.
- **1.1.209.** "Research" or "Researching" means activities, other than Development, related to the design, discovery, generation, identification, profiling, characterization, production, process development, cell line development, pre-clinical development or non-clinical or pre-clinical studies of drug candidates and products.
  - **1.1.210.** "Research Failure" means, [...\*\*\*...].
- **1.1.211.** "Research Plan" means the plan for conducting the Research Programs as set forth in Exhibit C.
- **1.1.212.** "**Research Program**" means each of the programs for conducting Research in connection with the Global Licensed Antibodies for each of the Global Target Pairs.
- **1.1.213.** "Research Term" means the time period commencing on the Effective Date and ending on the fifth anniversary of such date.
  - 1.1.214. "Restricted Licensed Antibody" means [...\*\*\*...].
- **1.1.215.** "**Restricted Technology**" means Xencor Technology, Novartis Technology (excluding Novartis Biological Material IP), and any Confidential Information of either Party or its Related Parties provided under or developed in connection with the Collaboration.
  - 1.1.216. [...\*\*\*...].
    1.1.217. [...\*\*\*...].
    1.1.218. [...\*\*\*...].
  - **1.1.219.** "**Right of Reference**" has the meaning set forth in Section 7.1.3.

- **1.1.220.** "**RLP Branding Strategy**" has the meaning set forth in Section 6.1.5.1.
- **1.1.221.** "RLP Commercial Strategy" has the meaning set forth in Section 6.1.2.
- **1.1.222.** "RLP Development Activities" means collectively, (a) those Development activities that are reasonably necessary to obtain initial Regulatory Approval of the applicable Regional Licensed Products in the applicable Major Market Country, and (b) any Additional Development Activities included in the initial RLP Development Plan or added pursuant to Section 5.1.2.4.
  - **1.1.223.** "RLP Development Budget" has the meaning set forth in Section 5.1.2.2.
  - **1.1.224.** "RLP Development Plan" has the meaning set forth in Section 5.1.2.1.
  - 1.1.225. "RLP Patents" means Current ex-US RLP Patents and New RLP Patents.
- **1.1.226.** "**RLP Trademarks**" means the Trademarks used, or intended for use, in connection with the Development and Commercialization of the Regional Licensed Products. RLP Trademarks specifically exclude the corporate names and logos of the Parties and their Affiliates and Sublicensees. RLP Trademarks include both the Xencor RLP Trademarks and the Novartis RLP Trademarks.
- **1.1.227.** "Royalty Patents" means Xencor Patents, Patents within Xencor Collaboration IP, Patents within Novartis Collaboration IP (for clarity, including Patents that Cover Novartis Core Inventions), and Patents within Joint Collaboration IP, but excluding Patents within Novartis Biological Material IP.
  - **1.1.228.** "Royalty Term" has the meaning set forth in Section 10.7.1.
- **1.1.229.** "Sales & Profit Share Report" means, with respect to a Calendar Quarter, a written report showing each of (a) the Gross Profits on sales of each Optioned Licensed Product in the United States during such Calendar Quarter, including Novartis' calculation to arrive at Gross Profits which comprise Net Sales less COGS, and (b) the amounts payable by Novartis under Section 10.2, in United States Dollars, which shall have accrued hereunder with respect to such Gross Profits, including the exchange rates used to calculate such amounts.
  - **1.1.230.** "Sales Milestone Event" has the meaning set forth in Section 10.5.
  - **1.1.231.** "Sales Milestone Payment" has the meaning set forth in Section 10.5.
- **1.1.232.** "SDEA" means the Safety Data Exchange Agreement to be entered into by the Parties in accordance with Section 7.4.

- **1.1.233.** "Secondary RLP Supplier" has the meaning set forth in Section 8.3.
- **1.1.234.** "**Specifically Bind**" means, [...\*\*\*...].
- **1.1.235.** "Sublicensee" means a Third Party to which a Party or its Affiliate has granted or grants rights, as permitted under this Agreement, to Research, Develop, Manufacture or Commercialize any Licensed Antibody or Licensed Product, or any further permitted sublicensee of such rights (regardless of the number of tiers, layers or levels of sublicenses of such rights).
- **1.1.236.** "Supplemental Study" is any Clinical Study (other than any Post-Marketing Study) for an additional indication or other label expansion for a Regional Licensed Product beyond the initial Indication(s) contemplated by the applicable Development Plan or involving the Regional Licensed Product and one or more other products.
  - **1.1.237.** "Supply Agreement" has the meaning set forth in Section 8.5.2.1(a).
- **1.1.238.** "Target" means a specific human protein that is bound by an Fv Region of an Antibody. For clarity, CD3, CD20, CD123 and [...\*\*\*...], and any specific human protein that is bound by an Fv Region of a Bispecific Antibody and that is added or substituted pursuant to Section 3.1, are each a Target.
- **1.1.239.** "**Tax**" and "**Taxation**" means any form of tax or taxation, levy, duty, charge or withholding (including any related fine, penalty, addition to tax, surcharge or interest) imposed by, or payable to, a governmental authority.
  - **1.1.240.** "Tech Transfer Costs" has the meaning set forth in Section 8.3.
- **1.1.241.** "**Territory**" means (a) with respect to Xencor, the Xencor Territory and (b) with respect to Novartis, the Novartis Territory.
  - **1.1.242.** "**Term**" has the meaning set forth in Section 15.1.
  - **1.1.243.** "Third Party" means any Person other than Novartis, Xencor or their respective Affiliates.
  - **1.1.244.** "Third Party Acquisition" has the meaning set forth in Section 12.6.2.1(a).
  - **1.1.245.** "**Third Party Payment**" has the meaning set forth in Section 9.5.1.

- **1.1.246.** "**Trademark**" means any trademark, trade name, service mark, service name, brand, domain name, trade dress, logo, slogan or other indicia of origin or ownership, including the goodwill and activities associated with each of the foregoing.
- **1.1.247.** "**United States**" or "**U.S.**" means the United States of America and its territories, possessions and commonwealths.
- **1.1.248.** "Valid Claim" means a claim of a Patent that (a) has not been rejected, revoked or held to be invalid or unenforceable by a court or other authority of competent jurisdiction, from which no appeal can be further taken, or (b) has not been finally abandoned, disclaimed or admitted to be invalid or unenforceable through reissue or disclaimer. In order to be a Valid Claim, any claim being prosecuted in a pending patent application must be prosecuted in good faith and not have been pending for more than [...\*\*\*...] years from the filing date of the first utility patent application (or equivalent concept in any such country) in the patent application family in the country in question, in which case it will cease to be considered a Valid Claim until the patent issues and recites said claim (from and after which time the same would be deemed a Valid Claim).
  - **1.1.249.** "Withdrawal Notice" has the meaning set forth in Section 2.7.
  - **1.1.250.** "Withholding Party" has the meaning set forth in Section 10.9.4.
  - **1.1.251.** [...\*\*\*...].
  - **1.1.252.** [...\*\*\*...].
  - **1.1.253.** "**Xencor Collaboration IP**" has the meaning set forth in Section 14.2.1.
  - **1.1.254.** "Xencor Competing Program" has the meaning set forth in Section 12.6.2.1(a).
- **1.1.255.** "Xencor Core Inventions" means, excluding Novartis Core Inventions, all Know-How first developed or conceived as part of the Collaboration during the applicable Invention Period, whether by on behalf of employee(s), agent(s) or consultant(s) of Xencor or its Affiliates or Novartis or its Affiliates, individually or jointly, that primarily relates to Xencor Know-How in existence as of the Effective Date, including any Patent that primarily claims such Know-How therein.
  - **1.1.256.** "**Xencor Indemnitees**" has the meaning set forth in Section 13.1.
- **1.1.257.** "**Xencor In-Licenses**" means any Existing Xencor In-License or any Collaboration In-License to which Xencor is a party.

- **1.1.258.** "Xencor Know-How" means Know-How Controlled by Xencor or its Affiliates as of the Effective Date or during the Invention Period that is reasonably necessary to Research, Develop, Manufacture or Commercialize any Licensed Antibody (other than an Fc Licensed Antibody) or Licensed Product (other than an Fc Licensed Product) in the Field, other than Xencor's interest in Joint Collaboration IP and Xencor Collaboration IP. For clarity, and notwithstanding the foregoing, in all cases, Xencor Know-How [...\*\*\*...]. Xencor Know-How excludes Know-How included in Fc Know-How.
- **1.1.259.** "Xencor Patents" means those Patents listed on Exhibit A-2 and those Patents Controlled by Xencor or its Affiliates as of, the Effective Date or during the applicable Invention Period that are reasonably necessary to Research, Develop, Commercialize or Manufacture any Licensed Antibody (other than an Fc Licensed Antibody) or Licensed Product (other than an Fc Licensed Product) in the Field, including for clarity, RLP Patents. Xencor Patents excludes Patents included in Fc Patents, Xencor Collaboration IP and Xencor's interest in Joint Collaboration IP, and excludes Patents that cover Fv Regions other than CD3, CD20 or CD123.
- **1.1.260.** "Xencor RLP Product Infringement" means any Infringement by a Third Party with respect to a Regional Licensed Product in the Xencor Territory.
  - **1.1.261.** "Xencor RLP Trademarks" has the meaning set forth in Section 14.8.1.2.
- **1.1.262.** "Xencor Technology" means Xencor Know-How, Xencor Patents, Xencor Collaboration IP and Xencor's interest in Joint Collaboration IP.
- **1.1.263.** "Xencor Territory" means, with respect to any Regional Licensed Product, the United States, unless Xencor exercises its Opt-Out Right with respect to such Regional Licensed Product as set forth in Section 5.1.8.
  - **1.1.264.** "Xencor Territory Commercialization Plan" has the meaning set forth in Section 6.1.4.

### 2. GOVERNANCE

2.1. <u>Alliance Manager</u>. Promptly following the Effective Date, each Party will designate an individual to facilitate communication and coordination of the Parties' activities under this Agreement relating to Licensed Target Pairs, Licensed Antibodies and Licensed Products and to provide support and guidance to the JSC (each, an "Alliance Manager"). Each Alliance Manager may also serve as a representative of its respective Party on one or more Committees.

### 2.2. <u>Joint Steering Committee</u>.

- **2.2.1.** Purpose, Formation. Within [...\*\*\*...] after the Effective Date, the Parties will establish a joint steering committee (the "JSC") that will monitor, make decisions, and provide strategic oversight of the activities under this Agreement and facilitate communications between the Parties with respect to the Research, Development, Manufacture and Commercialization of Licensed Antibodies and Licensed Products, all in accordance with this Section 2.2.
- **2.2.2.** Composition. Each Party will initially appoint three representatives to the JSC, all of whom will have sufficient seniority within the applicable Party to make decisions arising within the scope of the JSC's responsibilities. The Parties' initial representatives to the JSC are set forth on Exhibit B. The JSC may change its size from time to time by mutual consent of its members, provided that the JSC will consist at all times of an equal number of representatives of each of Xencor and Novartis. Each Party may replace its JSC representatives at any time upon written notice to the other Party. The JSC may invite non-members to participate in the discussions and meetings of the JSC, provided that such participants have no voting authority at the JSC and are bound under written obligation of confidentiality no less protective of the Parties' Confidential Information than those set forth in this Agreement. For clarity, the Alliance Managers should attend meetings of the JSC. The JSC will be co-chaired, with one chairperson designated by Xencor and one chairperson designated by Novartis, whose responsibilities will include conducting meetings, including, when feasible, ensuring that objectives for each meeting are set and achieved. Responsibility for running each meeting of the JSC will alternate between the chairpersons from meeting-to-meeting, with Xencor's chairperson running the first meeting. The Alliance Managers will work with the chairpersons to prepare and circulate agendas and to ensure the preparation of minutes. The chairpersons have no additional powers or rights beyond those held by the other JSC representatives.
- **2.2.3.** Specific Responsibilities. In addition to its overall responsibility for monitoring and providing strategic oversight with respect to the Parties' activities under this Agreement, the JSC will in particular have the following responsibilities:
  - **2.2.3.1.** discuss, prepare and approve amendments to the Research Plan,
  - **2.2.3.2.** review, discuss and oversee the conduct of the Research Programs,
  - **2.2.3.3.** review, discuss and oversee the Development of Regional Licensed Antibodies, including to discuss a plan for the Development of backups to the lead Regional Licensed Antibodies,
  - **2.2.3.4.** review, discuss and oversee the Commercialization of Regional Licensed Products and any other ongoing related activities, including approving the Commercialization Plans submitted by the JCC,

- **2.2.3.5.** review and discuss the Commercialization of Regional Licensed Products and any other ongoing related activities,
- **2.2.3.6.** review, discuss and oversee the Manufacturing for the Licensed Antibodies (other than Fc Licensed Antibodies), including the supply chain,
- **2.2.3.7.** facilitate the flow of information between the Parties with respect to Licensed Antibodies and Licensed Products,
- **2.2.3.8.** review and discuss reports from the JDC and JCC, provide guidance thereto, direct the activities of such Committees, and review and approve each RLP Development Plan and OLP Development Plan, and, in each case, all amendments thereto,
- **2.2.3.9.** review, discuss and approve the entry into any Collaboration In-Licenses with respect to the Research, Development, Manufacture or Commercialization of any Regional Licensed Antibodies, Regional Licensed Products, Optioned Licensed Antibodies and Licensed Products,
- **2.2.3.10.** review, discuss and coordinate the Parties' scientific presentation and publication strategy relating to the Regional Licensed Products,
- **2.2.3.11.** review and facilitate discussion of proposed publications and resolve disputes with respect thereto pursuant to Section 11.2.1,
- **2.2.3.12.** attempt to resolve issues presented to it by, and disputes within, the JDC or JCC, or any other subcommittee,
- **2.2.3.13.** establish such additional joint subcommittees as it deems necessary to achieve the objectives and intent of this Agreement, and
- **2.2.3.14.** perform such other functions as appropriate, and direct each other Committee to perform such other functions as appropriate, to further the purposes of this Agreement, in each case as agreed in writing by the Parties or as expressly provided in this Agreement.
- **2.2.4.** Meetings. The JSC will meet at least once per [...\*\*\*...] during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...\*\*\*...] prior to any meeting of the JSC (or such shorter time period as the Parties may agree), the Alliance Managers will prepare and circulate an agenda for such meeting, <u>provided</u> however, that either Party may propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JSC (by videoconference, teleconference or in person) by providing at least [...\*\*\*...] prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party will work

with the co-chairs of the JSC and the Alliance Managers of both Parties to provide the members of the JSC no later than [...\*\*\*...] prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The JSC may meet in person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one meeting per Calendar Year will be in person unless the Parties mutually agree in writing to waive such requirement. In-person JSC meetings will be held at locations in the United States alternately selected by Xencor and by Novartis. Each Party will bear the expense of its respective JSC members' participation in JSC meetings. Meetings of the JSC will be effective only if at least one representative of each Party (which representative is not such Party's Alliance Manager) is present or participating in such meeting. The Alliance Managers will be responsible for preparing reasonably detailed written minutes of all JSC meetings that reflect material decisions made and action items identified at such meetings. The Alliance Managers will send draft meeting minutes to each member of the JSC for review and approval within [...\*\*\*...] after each JSC meeting. Such minutes will be deemed approved unless [...\*\*\*...] members of the JSC objects to the accuracy of such minutes within [...\*\*\*...] of receipt. Minutes will be officially endorsed by the JSC at the next JSC meeting, including reflecting any differences noted by the Parties, and will be signed by the Alliance Managers.

# 2.3. <u>Joint Development Committee</u>.

**2.3.1.** Formation, Composition, Dissolution. No later than [...\*\*\*...] after the Effective Date the Parties will establish a committee to (a) oversee the Development of Regional Licensed Antibodies and Regional Licensed Products and Optioned Licensed Antibodies and Optioned Licensed Products in accordance with the applicable Development Plans and to coordinate the Development activities of the Parties with respect thereto, and (b) facilitate the flow of information between the Parties with respect to, and provide a forum to discuss, the Development of Optioned Licensed Antibodies, Optioned Licensed Products, Global Licensed Antibodies and Global Licensed Products (the "JDC"). Each Party will initially appoint [...\*\*\*...] representatives to the JDC, with each representative having knowledge and expertise in the Development of compounds and products similar to the Licensed Antibodies and Licensed Products and having sufficient seniority within the applicable Party to make decisions arising within the scope of the JDC's responsibilities. The JDC may change its size from time to time, provided that the JDC will consist at all times of an equal number of representatives of each of Xencor and Novartis. Each Party may replace its JDC representatives at any time upon written notice to the other Party. The JDC may invite non-members to participate in the discussions and meetings of the JDC, provided that such participants have no voting authority at the JDC and are bound under written obligation of confidentiality no less protective of the Parties' Confidential Information than those set forth in this Agreement. For clarity, the Alliance Managers should attend all meetings of the JDC. The JDC will be co-chaired, with one chairperson designated by Xencor and one chairperson designated by Novartis, whose responsibilities will include conducting meetings, including, when feasible, ensuring that objectives for each meeting are set and achieved. Responsibility for running each meeting of the JDC will alternate between the chairpersons from meeting-to-meeting, with Novartis'

chairperson running the first meeting. Upon the later of (a) first commercial sale of the final Regional Licensed Product or Optioned Licensed Product, or (b) removal of any requirement for any Clinical Study established by a Regulatory Authority upon Regulatory Approval for the final Regional Licensed Product or Optioned Licensed Product in any country, the Parties agree that the JDC will be automatically dissolved with no further action required by either Party.

- **2.3.2.** Specific Responsibilities of the JDC. The JDC has the following responsibilities:
- **2.3.2.1.** oversee and review Development responsibilities for each Regional Licensed Antibody, Regional Licensed Product, Optioned Licensed Antibody and Optioned Licensed Product,
- **2.3.2.2.** discuss, prepare and approve for submission to the JSC all RLP Development Plans and OLP Development Plans, and all amendments thereto,
- **2.3.2.3.** oversee and direct the conduct of all RLP Development Plans and OLP Development Plans,
- **2.3.2.4.** create, implement and review the overall strategy for Development, including the design of all Clinical Studies for Regional Licensed Antibodies, Regional Licensed Products, Optioned Licensed Antibodies and Optioned Licensed Products,
- **2.3.2.5.** decide whether and when to initiate or discontinue any Clinical Study under each RLP Development Plan or OLP Development Plan, as applicable,
- **2.3.2.6.** allocate budgeted resources and determine priorities for each Clinical Study included under each RLP Development Plan or OLP Development Plan,
- **2.3.2.7.** oversee and direct the conduct of all Clinical Studies under each RLP Development Plan or OLP Development Plan,
- **2.3.2.8.** facilitate the flow of information between the Parties with respect to the Development of Licensed Antibodies and Licensed Products,
- **2.3.2.9.** allocate primary responsibility as between the Parties for tasks relating to the Development of Regional Licensed Antibodies and Optioned Licensed Antibodies where not already specified in the applicable RLP Development Plans or OLP Development Plan,
- **2.3.2.10.** review, discuss, oversee and direct Manufacturing for the Development of Regional Licensed Antibodies, Regional Licensed Products, Optioned Licensed Antibodies and Optioned Licensed Products including the supply chain,
- **2.3.2.11.** determine and oversee the implementation of the overall strategy regarding, and facilitate the flow of information between the Parties with respect to

obtaining, Regulatory Approval of Regional Licensed Products and Optioned Licensed Products,

- **2.3.2.12.** without limitation to Section 2.3.2.1, review the regulatory strategy with respect to discussions with and commitments to or agreements with Regulatory Authorities (including post-approval commitments) with respect to Regional Licensed Product and Optioned Licensed Product labeling, risk management or Clinical Studies, and
- **2.3.2.13.** perform such other functions as may be appropriate to further the purposes of this Agreement, as directed by the JSC in accordance with Section 2.2.3 or as expressly provided in this Agreement.
- **2.3.3.** Meetings. The JDC will meet at least once per [...\*\*\*...], unless the Parties mutually agree in writing to a different frequency. No later than [...\*\*\*...] prior to any meeting of the JDC (or such shorter time period as the Parties may agree), the Alliance Managers will prepare and circulate an agenda for such meeting, provided however, that either Party will be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JDC (by videoconference, teleconference or in person) by providing at least [...\*\*\*...] prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party will work with the Alliance Manager to provide the members of the JDC no later than [...\*\*\*...] prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision. The JDC may meet in person, by videoconference, or by teleconference. In-person JDC meetings will be held at locations in the United States alternately selected by Xencor and by Novartis or at any other location mutually agreed by the members of the JDC. Each Party will report to the JDC on all material issues relating to the Development of Licensed Antibodies and Licensed Products for and in the Territory promptly after such issues arise. Each Party will bear the expense of its respective JDC members' participation in JDC meetings. The JDC co chairs will be responsible for preparing reasonably detailed written minutes of JDC meetings that reflect all decisions made and action items identified at such meetings. The JDC co chairs will send meeting minutes to each member of the JDC for review and approval within [...\*\*\*...] after each JDC meeting. Minutes will be deemed approved unless one or more members of the JDC objects to the accuracy of such minutes within [...\*\*\*...] of receipt. Minutes will be officially endorsed by the JDC at the next JDC meeting, including reflecting any differences noted by the Parties, and will be signed by the Alliance Managers.
- **2.3.4.** Decision-Making. Subject to the remainder of this Section 2.3.4 and Section 2.5, the JDC will act by unanimous agreement. The representatives from each Party have, collectively, [...\*\*\*...] vote on behalf of that Party. If the JDC cannot reach unanimous agreement on an issue that comes before the JDC within [...\*\*\*...] of the meeting such issue was raised and over which the JDC has oversight, then the Parties will refer such matter to the JSC for resolution in accordance with Section 2.5.

#### 2.4. <u>Joint Commercialization Committee</u>.

- **2.4.1.** General. The Parties will establish a committee, no later than completion of the first Phase 2 Clinical Study for the first Regional Licensed Product or Optioned Licensed Product, to (a) oversee Commercialization of Regional Licensed Products and Optioned Licensed Products in the Territory, and (b) facilitate the flow of information between the Parties with respect to, and provide a forum to discuss, the Commercialization of Regional Licensed Products and Optioned Licensed Products (the "**JCC**").
- **2.4.2.** Formation, Composition. Each Party will initially appoint [...\*\*\*...] representatives to the JCC, with each representative having knowledge and expertise in the commercialization of products similar to the Regional Licensed Products or Optioned Licensed Products and having sufficient seniority within the applicable Party to make decisions arising within the scope of the JCC's responsibilities. The JCC may change its size from time to time by mutual consent of its members, <u>provided</u> that the JCC will consist at all times of an equal number of representatives of each of Xencor and Novartis. Each Party may replace its JCC representatives at any time upon written notice to the other Party. The JCC may invite non-members to participate in the discussions and meetings of the JCC, <u>provided</u> that such participants have no voting authority at the JCC and are bound under written obligation of confidentiality no less protective of the Parties' Confidential Information than those set forth in this Agreement. For clarity, the Alliance Managers should attend all meetings of the JCC. The JCC will be co-chaired, with one chairperson designated by Xencor and one chairperson designated by Novartis, whose responsibilities will include conducting meetings, including, when feasible, ensuring that objectives for each meeting are set and achieved. Responsibility for running each meeting of the JCC will alternate between the chairpersons from meeting-to-meeting, with Novartis' chairperson running the first meeting.
- **2.4.3.** Specific Responsibilities of the JCC. Subject to any limitations under Law, including Antitrust Laws, the JCC has the following responsibilities:
  - **2.4.3.1.** review, update and approve for submission to the JSC the RLP Commercial Strategy for each Regional Licensed Product and Optioned Licensed Product,
  - **2.4.3.2.** discuss, prepare and approve for submission to the JSC the Commercialization Plan for each Regional Licensed Product and Optioned Licensed Product, including, in each case, any amendments thereto,
  - **2.4.3.3.** oversee implementation of each Commercialization Plan for each Regional Licensed Product, in accordance with the RLP Commercial Strategy, and Optioned Licensed Product,
  - **2.4.3.4.** discuss, prepare, update and approve for submission to the JSC the RLP Commercialization Budget,

- **2.4.3.5.** review, discuss and direct Commercialization activities with respect to Regional Licensed Products and Optioned Licensed Products,
- **2.4.3.6.** facilitate the flow of information between the Parties with respect to the Commercialization of Regional Licensed Products and Optioned Licensed Products,
- **2.4.3.7.** oversee forecasting and market planning with respect to Regional Licensed Products and Optioned Licensed Products,
- **2.4.3.8.** review and discuss strategies with respect to Pricing Matters for Regional Licensed Products and Optioned Licensed Products in the Territory, to the extent operationally feasible and not prohibited by Law,
- **2.4.3.9.** review, discuss and oversee Manufacturing for the Commercialization of Regional Licensed Products and Optioned Licensed Products, including the supply chain,
  - 2.4.3.10. manage Trademarks as contemplated by Section 14.8, and
- **2.4.3.11.** perform such other functions as appropriate to further the purposes of this Agreement, as directed by the JSC in accordance with Section 2.2.3 or as expressly provided in this Agreement.
- **2.4.4.** Meetings. The JCC will meet at least once per [...\*\*\*...], unless the Parties mutually agree in writing to a different frequency. No later than [...\*\*\*...] prior to any meeting of the JCC (or such shorter time period as the Parties may agree), the Alliance Managers will prepare and circulate an agenda for such meeting, provided however, that either Party will be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JCC (by videoconference, teleconference or in person) by providing at least [...\*\*\*...] prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party will work with the chairperson of the JCC to provide the members of the JCC no later than [...\*\*\*...] prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision. The JCC may meet in person, by videoconference, or by teleconference. In-person JCC meetings will be held at locations in the United States alternately selected by Xencor and by Novartis or at any other location mutually agreed by the members of the JCC. Meetings of the JCC will be effective only if at least one representative of each Party is present or participating in such meeting. Each Party will report to the JCC on all material issues relating to the Commercialization of Regional Licensed Products and Optioned Licensed Products promptly after such issues arise. Each Party will bear the expense of its respective JCC members' participation in JCC meetings. The JCC co-chairs will be responsible for preparing reasonably detailed written minutes of JCC meetings that reflect all decisions made and action items identified at such meetings. The JCC co-chairs will send meeting minutes to each member of the JCC for review and approval within [...\*\*\*...]

after each JCC meeting. Minutes will be deemed approved unless one or more members of the JCC object to the accuracy of such minutes within [...\*\*\*...] of receipt. Minutes will be officially endorsed by the JCC at the next JCC meeting, including reflecting any differences noted by the Parties, and will be signed by the Alliance Managers.

**2.4.5.** Decision-Making. Subject to the remainder of this Section 2.4.5 and Section 2.5, the JCC will act by unanimous agreement. The representatives from each Party have, collectively, one vote on behalf of that Party. If the JCC cannot reach unanimous agreement on an issue that comes before the JCC within [...\*\*\*...] of the meeting such issue was raised and over which the JCC has oversight, then the Parties will refer such matter to the JSC for resolution in accordance with Section 2.5, <u>provided</u> that any issues arising under Section 2.4.3.8 shall not be subject to such escalation or decision-making authority, and instead shall be determined by each Party in its respective Territory. For clarity, any and all such communications or strategy involving the Commercialization activities shall be limited to those permitted under Law, including Antitrust Laws.

# 2.5. Resolution of Committee Disputes.

- **2.5.1.** Within Operating Committees. All decisions within the JDC and JCC will be made by unanimous agreement and all decisions within the other Committees, other than the JSC, will be made by unanimous agreement unless escalated as set forth in this Section 2.5. If a dispute arises which cannot be resolved within the JDC, JCC or such other Committees, then if such dispute relates to a matter within the jurisdiction of the applicable Committee, the representatives of either Party may cause such matter to be referred to the JSC for resolution as provided in Section 2.5.2.
- **2.5.2.** Decision Making Within the JSC. In addition to resolving issues specifically delegated to it, the JSC has the authority to resolve disputes within the jurisdiction of the JDC, JCC and any other Committees (excluding the IP Committee) that the Parties may subsequently create to assist in governance of this Agreement, but otherwise has no authority except where expressly specified elsewhere in this Agreement or mutually agreed to by the Parties in writing. The representatives from each Party have, collectively, one vote on behalf of that Party, and all decisions within the JSC (whether originating there, or referred to it by an operating Committee) will be made by unanimous agreement. If a matter is referred by an operating Committee to the JSC, the JSC will use good faith efforts, in compliance with this Section 2.5.2, to resolve promptly such matter. If the JSC is unable to reach unanimous agreement, within [...\*\*\*...] after a Party affirmatively states that a decision needs to be made, on any issue for which it is responsible, either Party may elect to submit such issue to the Parties' Executive Officers in accordance with Section 2.5.3.
- **2.5.3.** Referral to Executive Officers. If a Party makes an election under Section 2.5.2 to refer a matter to the Executive Officers, the JSC will submit in writing the respective positions of the Parties to their respective Executive Officers. Such Executive Officers will use good faith efforts, in compliance with this Section 2.5.3, to resolve promptly such matter, which good faith efforts will include at least one meeting between such Executive Officers within [...\*\*\*...]

[...\*\*\*...] after the JSC's submission of such matter to them. If the Executive Officers are unable to reach unanimous agreement on any such matter within [...\*\*\*...] of such matter being referred to them, the matter will be decided in accordance with Section 2.5.4.

- **2.5.4.** Dispute Resolution Within the JSC. If the Parties' Executive Officers are unable to reach unanimous agreement in accordance with Section 2.5.3, then:
  - **2.5.4.1.** the following matters will be decided by Novartis' Executive Officer subject to the remainder of this Section 2.5.4 and Section 2.5.5.
    - (a) [...\*\*\*...];
    - (b) [...\*\*\*...]; and
    - (c) [...\*\*\*...]; and
    - (d) [...\*\*\*...].
  - **2.5.4.2.** the following matters will be decided by Xencor's Executive Officer subject to the remainder of this Section 2.5.4 and Section 2.5.5, subject to Section 2.5.4.1[...\*\*\*...]; and
    - **2.5.4.3.** the following matters will be decided pursuant to Expedited Arbitration:
      - (a) [...\*\*\*...];
      - (b) [...\*\*\*...]; and

1	(c)	١	Г				***				
١		,	ı	٠	٠	•		٠	•	•	

- **2.5.4.4.** with respect to disputes regarding the conduct of Additional Development Activities, as set forth in Section 5.1.2.4,
- **2.5.4.5.** with respect to the approval of any Potential In-License, such Potential In-License will be deemed not to have been approved, with the consequences as provided in Section 9.5.1.4,
  - **2.5.4.6.** with respect to disputes arising out of Section 14, as set forth therein, and
- **2.5.4.7.** with respect to any other matter before the JSC that is not expressly subject to a dispute resolution process identified in this Section 2.5.4, the status quo will be maintained unless and until the Parties mutually agree on such matter.

Notwithstanding anything herein to the contrary, (i) no exercise of a Party's decision-making authority (which, for clarity does not include decision-making by Expedited Arbitration) on any such matters may, without the other Party's prior written consent, result in a material decrease or increase in the other Party's or its Related Parties' obligations, costs or expenses under this Agreement, the Research Plan, any Development Plan or Commercialization Plan or require the other Party to perform additional activities not contemplated by this Agreement, and (ii) no exercise of decision-making authority on any such matters may, whether made by a Party or Expedited Arbitration, conflict with or amend this Agreement without both Parties' prior written consent.

- **2.5.5.** Good Faith. In conducting themselves on committees, and in exercising their rights under this Section 2.5, all representatives of both Parties will consider diligently, reasonably and in good faith all input received from the other Party, and will use reasonable efforts to reach unanimous agreement on all matters before them. In exercising any decision-making authority granted to it under this Section 2.5, each Party will act based on its good faith judgment taking into consideration such Party's obligations to use Commercially Reasonable Efforts with respect to Research, Development or Commercialization activities and [...\*\*\*...] as provided in this Agreement.
- 2.6. <u>General Committee Authority</u>. Each Committee has solely the powers expressly assigned to it in this Section 2. No Committee will have any power to amend, modify, or waive compliance with this Agreement. It is expressly understood and agreed that the control of decision-making authority by Xencor or Novartis, as applicable, pursuant to Section 2.5, so as to resolve a disagreement or deadlock on a Committee for any matter will not authorize either Party to perform any function or exercise any decision-making right not delegated to a Committee or such Party, and that neither Xencor nor Novartis has any right to unilaterally modify, amend or waive its own compliance with, the terms of this Agreement.

2.7. Withdrawal 2.8. . At any time during the Term and for any reason, Xencor shall have the right to withdraw from participation in each Committee upon written notice to Novartis, which notice shall be effective immediately upon receipt ("Withdrawal Notice"). Following the issuance of a Withdrawal Notice, Xencor's representatives to the applicable Committee shall not participate in any meetings of such Committee. Following Xencor's issuance of a Withdrawal Notice: (i) all meetings of the applicable Committee shall be held at Novartis' facilities; (ii) Xencor shall have the right to continue to receive the minutes of such Committee but shall not have the right to approve the minutes for, or vote on any matter before, such Committee with respect to any meeting held after Xencor's issuance of a Withdrawal Notice and (iii) all decisions specified herein to be made by a Committee shall be made by Novartis.

# 3. GLOBAL TARGET RESEARCH ACTIVITIES; FC TARGETS

## 3.1. Global Target Pairs.

- **3.1.1.** Initial Global Target. The Parties agree that the Target pair known as [...\*\*\*...] is hereby designated as the first Global Target Pair.
- **3.1.2.** Maximum Number of Global Target Pairs, Selection Right. At any given time under this Agreement, Novartis may only hold rights to a maximum of four Global Target Pairs (or, in the case where Xencor has exercised its Opt-In Right and not exercised its Opt-Out Right pursuant to Section 5.3.7, three such Global Target Pairs). From time to time during the Global Target Selection /Replacement Period, Novartis may notify Xencor in writing of Targets that Novartis desires to designate as a Global Target Pair for the purposes of this Agreement (each, a "**Proposed Global Target Pair**") and, within [...\*\*\*...], Xencor will advise Novartis whether such Target pairs are Available. If the Target pair is available, then such Target pair will be deemed to be a Global Target Pair. If the Proposed Global Target Pair is not Available, then such Proposed Global Target Pair will not be deemed a Global Target Pair. If Novartis questions why a Proposed Global Target Pair is not Available, upon request, Xencor shall promptly provide reasonable evidence of such un-Availability, which evidence may be provided to outside counsel or outside consultant engaged by Novartis (and who will enter an appropriate confidentiality agreement with Xencor prior to receipt of such evidence) to confirm such status, with such outside counsel or consultant engaged by Novartis permitted to disclose to Novartis only whether or not it agrees with the Xencor determination as to Availability. In the event of a dispute with regard to Availability, such dispute shall be resolved by Expedited Arbitration.
- **3.1.3.** Substitution Rights. For each Global Target Pair, Novartis shall have the one-time right, during the Global Target Pair Selection/Replacement Period, to substitute another Target pair that are Available Target pairs for such Global Target Pair, <u>provided</u> that (i) Novartis may only make such substitution once per Global Target Pair (i.e., Novartis may make four such substitutions) and (ii) a given Global Target Pair cannot be substituted once a Global Licensed Antibody against such Global Target Pair achieves [...\*\*\*...]. Without limiting the foregoing, Novartis shall also have the right to substitute up to [...\*\*\*...] Global Target Pairs at any time due to a Research Failure. Any Global Target Pairs that are abandoned by Novartis as a result of such a substitution shall no longer be considered as Global Target Pairs for the

purposes of this Agreement unless subsequently substituted for a Global Target Pair pursuant to this Section 3.1.3. In the event that Novartis desires to undertake a substitution, Novartis will notify Xencor in writing regarding its desired Target pair and Xencor, within [...\*\*\*...] will advise Novartis whether such Target pair is an Available Target. If such Target pair is available, then such Target pair will be deemed to be a Global Target Pair (subject to the four Global Target Pair limitation). If such Target pair is not Available, then the substitution shall not take effect and the Global Target Pair proposed to be substituted for will remain a Global Target Pair unless and until substituted for in accordance with this Section 3.1.3. If Novartis questions why a Proposed Global Target Pair is not Available, upon request, Xencor shall promptly provide reasonable evidence of such un-Availability, which evidence may be provided to outside counsel or outside consultant engaged by Novartis (and who will enter an appropriate confidentiality agreement with Xencor prior to receipt of such evidence) to confirm such status, with such outside counsel or consultant engaged by Novartis permitted to disclose to Novartis only whether or not it agrees with the Xencor determination as to Availability. In the event of a dispute with regard to Availability, such dispute shall be resolved by Expedited Arbitration.

- **3.1.4.** Research Program for Global Target Pairs. Each Party will be responsible for performing the Research Programs in accordance with this Agreement and the Research Plan during the Research Term. In the event of any inconsistency between the Research Plan and this Agreement, the terms of this Agreement will prevail. For clarity, Xencor's obligations with respect to the performance of the Research Programs and Development of the Global Target Pairs and, as applicable, the Optioned Target Pair, is expressly limited to the activities set forth in the Research Plan and shall conclusively expire upon the expiration of the Research Term.
- **3.1.5.** Diligence, Standards of Conduct. During the Research Term, each Party (itself or through its Affiliates or by permitted subcontracting pursuant to Section 3.1.9) will use Commercially Reasonable Efforts to Research Global Licensed Antibodies in accordance with the Research Plan. Each Party will conduct its activities under the Research Plan in a good scientific manner and in compliance with Law.
- **3.1.6.** Research Costs. During the Research Term, each Party will be responsible for [...\*\*\*...]% of all costs and expenses incurred by it or on its behalf or its Related Parties in connection with each Research Program. After the expiration of the Research Term, Novartis will be responsible for [...\*\*\*...]% of all costs and expenses incurred by or on behalf of Novartis or its Related Parties in connection with any further Research of Global Licensed Antibodies.
- **3.1.7.** Research Reports. Each Party will keep the JSC informed regarding the progress of Research activities for each Research Program, including a review of results and progress against timelines in the Research Plan on a [...\*\*\*...] basis.
- **3.1.8.** Research Records. Each Party will maintain scientific records in sufficient detail and in good scientific manner appropriate for patent and regulatory purposes, and with respect to activities that require cGLP compliance to be submitted in regulatory filings, in compliance with cGLPs, which records will fully and accurately reflect work done and results

achieved in the performance of the Research activities by or on behalf of such Party with respect to Global Licensed Antibodies.

- Third Parties. Each Party will be entitled to utilize the services of Third Parties to perform its Research activities under this Section 3.1, provided that (a) a Party will require that such Third Party operates in a manner consistent with the terms of this Agreement, and (b) such Party will remain at all times fully liable for its responsibilities hereunder, and such Party hereby expressly waives any requirement that the other Party exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against such Party. The subcontracting Party will require that any Third Party agreement entered into pursuant to this Section 3.1.9 (i) include confidentiality and nonuse provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (ii) obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) under and to, any Know-How and Patents that are developed by such Third Party in the performance of its obligations under such agreement and are reasonably necessary to Research, Develop, Manufacture or Commercialize Global Licensed Antibodies or Global Licensed Products in the Field, and any such license shall be deemed to be a Novartis Existing In-License or Xencor Existing In-License, as applicable, for all purposes of this Agreement. For clarity, the foregoing requirement to obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) shall not apply to any improvements to the proprietary core or platform technology owned or in-licensed by any such Third Party or its Affiliates, unless such improvements are reasonably necessary to Research, Develop, Manufacture or Commercialize those Global Licensed Antibodies or Global Licensed Products with respect to which such Third Party or its Affiliate conducted its activities under such Third Party agreement. The subcontracting Party will be solely responsible for direction of and communications with such Third Party.
- **3.1.10.** Material Transfer. From time to time during, and within [...\*\*\*...] following the earlier of the expiration of the Research Term generally or the completion of a Research Program for a given Global Target Pair, Xencor shall transfer to Novartis all tangible Materials, and copies of all material data, results, and research records in its Control relating to such Global Licensed Antibodies, including any such tangible Materials, data, results, and records developed by a Third Party on behalf of Xencor pursuant to Section 3.1.9.

# 3.2. <u>Fc Targets</u>.

## **3.2.1.** Selection Process.

**3.2.1.1.** Fc Gatekeeper. The Parties shall jointly engage a mutually agreeable law firm, to serve as the "Fc Gatekeeper" under this Agreement. The Parties shall endeavor to sign an engagement letter with the Fc Gatekeeper within [...\*\*\*...] after the Effective Date. The engagement letter shall be in a form reasonably acceptable to both Parties and consistent with this Agreement. The engagement letter will require the Fc Gatekeeper to keep all Confidential Information related to Novartis' inquiries

concerning potential reservation of potential Targets, including the identity of any Target proposed to be an Fc Target in confidence and the Fc Gatekeeper will not disclose any such information to Xencor, its Related Parties, or to any Third Party unless specifically instructed to do so by Novartis. Novartis shall be solely responsible for all invoices by the Fc Gatekeeper and shall timely pay the Fc Gatekeeper's invoices as set forth in the Fc Gatekeeper engagement letter.

- **3.2.1.2.** Fc Target Unavailable List. [...\*\*\*...].
- **3.2.1.3.** Novartis Selection Notice and Process. From time to time during the Fc Target Selection Period, Novartis may submit to the Fc Gatekeeper a written notice (the "Fc Notice") identifying a Target (including a Target that would be Specifically Bound by a Bispecific Antibody) that it wishes to designate as a Fc Target for the purposes of this Agreement (each, a "**Proposed Fc Target**").
  - (a) Within [...\*\*\*...] after receiving such Fc Notice, the Fc Gatekeeper shall inform Novartis if such Proposed Fc Target is Available. Such Proposed Fc Target shall be confirmed as an Fc Target for the purposes of this Agreement if such Target or Bispecific (including a Target that would be Specifically Bound by a Bispecific Antibody) does not appear on the then-current Fc Target Unavailable List. If such Proposed Fc Target does appear on the then-current Fc Target Unavailable List then the Fc Gatekeeper shall so notify Novartis, without providing additional details, and such Proposed Fc Target shall not become an Fc Target.
  - (b) If Novartis questions why a Proposed Fc Target is on the Fc Target Unavailable List, Xencor shall promptly provide reasonable evidence of such un-Availability, which evidence may be provided to an outside counsel or outside consultant engaged by Novartis who is reasonably acceptable to Xencor (and who will enter an appropriate confidentiality agreement with Xencor prior to receipt of such evidence) to confirm such status, with such outside counsel or consultant permitted to disclose to Novartis only whether or not it agrees with the Xencor determination as to Availability. In the event of a dispute with regard to Availability, such dispute shall be resolved by Expedited Arbitration.

- **3.2.1.4.** Maximum Number of Fc Targets, Substitution Rights. Notwithstanding Section 3.2.1.3 above, at any given time during the Fc Target Selection Period, Novartis may only hold rights to a maximum of 10 Fc Targets. Novartis shall have the right, every [...\*\*\*...], on or about [...\*\*\*...], of each Calendar Year, during the Fc Target Selection Period, to substitute an Fc Target for another Target (including a Target that would be Specifically Bound by a Bispecific Antibody) pursuant to the selection mechanism in Section 3.2.1.3. Any Fc Targets that are abandoned by Novartis as a result of such a substitution shall no longer be considered as Fc Targets for the purposes of this Agreement. Any Fc Target for which Novartis has [...\*\*\*...] may not be substituted. For clarity, Novartis shall not have the right to substitute any Fc Target until such time, if ever, as there are 10 Fc Targets.
- **3.2.1.5.** Xencor Confirmed Target Inquiry. From time to time, Xencor may query the Fc Gatekeeper to determine whether a given Target has been confirmed as an Fc Target. The Fc Gatekeeper shall respond within [...\*\*\*...] to each such query by Xencor, specifying whether the Target is or is not an Fc Target. Xencor shall only make such queries in good faith, in planning for the grant of an exclusive license grant related to the applicable Target or the initiation of a discovery program for Antibody(ies) that Specifically Bind the applicable Target.
- **3.2.1.6.** Fc Target Identification. Novartis will identify each Fc Target to Xencor in writing upon the earlier of (i) expiration of the Fc Target Selection Period or (ii) the date on which an Fc Product that Specifically Binds such Fc Target [...\*\*\*...].

#### 4. OPT-IN RIGHT

#### 4.1. Grant, Exercise Mechanism.

- **4.1.1.** Grant. Xencor shall have the right, but not the obligation, to share in 50% of Gross Profits, Commercialization Costs and Medical Affairs Activities Costs in the United States (and 25% of the global Development Costs) for one Global Target Pair (other than the Global Target Pair CD3xBCMA) (the "**Opt-In Right**"). The Global Target Pair that is the subject of an exercised Opt-In Right is referred to herein as the "**Optioned Target Pair.**"
  - **4.1.2.** Exercise Mechanism. The mechanism for exercise of such Opt-in Right is as follows:
  - **4.1.2.1.** [...\*\*\*...] First Global Target Pair. Promptly after [...\*\*\*...] with respect to a Global Licensed Product that Specifically Binds the first Global Target Pair, Novartis will either (a) notify Xencor that such Global Target Pair is not eligible for Xencor's Opt-in Right, or (b) in Novartis' sole discretion, provide Xencor with an Option Package with respect to such Global Target Pair, and afford reasonable access during normal business hours to Novartis' personnel by Xencor and its representatives as Xencor may reasonably

request to assist Xencor in deciding whether to exercise the Opt-In Right for such Global Target Pair. In the event that Novartis provides the Option Package as set forth in subsection (b), then Xencor shall have [...\*\*\*...] from the receipt of such Option Package to exercise its Opt-In Right by providing written notice to Novartis. Upon receipt of such written notice, such Global Target Pair shall be deemed to be an Optioned Target Pair for the purposes of this Agreement, and Xencor shall not have any further right to Opt-In to any additional Global Target Pairs. If Xencor does not exercise its Opt-In Right with respect to such Global Target Pair, such Global Target Pair shall remain a Global Target Pair and shall thereafter not be subject to Xencor's Opt-In Right.

- **4.1.2.2.** [...\*\*\*...] Second Global Target Pair. If Xencor has not exercised its Opt-In Right, then, promptly after [...\*\*\*...] with respect to a Global Licensed Product that Specifically Binds the second Global Target Pair, Novartis will provide Xencor with an Option Package with respect to such Global Target Pair, and afford reasonable access during normal business hours to Novartis' personnel by Xencor and its representatives as Xencor may reasonably request to assist Xencor in deciding whether to exercise the Opt-In Right for such Global Target Pair. Xencor shall have [...\*\*\*...] from the receipt of such Option Package to exercise its Opt-In Right by providing written notice to Novartis. Upon receipt of such written notice, such Global Target Pair shall be deemed to be an Optioned Target Pair for the purposes of this Agreement, and Xencor shall not have any further right to Opt-In to any additional Global Target Pairs. If Xencor does not exercise its Opt-In Right with respect to such Global Target Pair, such Global Target Pair shall remain a Global Target Pair and shall thereafter not be subject to Xencor's Opt-In Right.
- **4.1.2.3.** [...\*\*\*...] Third Global Target Pair. If Xencor has not exercised its Opt-In Right, then, promptly after [...\*\*\*...] with respect to a Global Licensed Product that Specifically Binds the third Global Target Pair, Novartis will provide Xencor with an Option Package with respect to such Global Target Pair, and afford reasonable access during normal business hours to Novartis' personnel by Xencor and its representatives as Xencor may reasonably request to assist Xencor in deciding whether to exercise the Opt-In Right for such Global Target Pair. Xencor shall have [...\*\*\*...] from the receipt of such Option Package to exercise its Opt-In Right by providing written notice to Novartis. Upon receipt of such written notice, such Global Target Pair shall be deemed to be an Optioned Target Pair for the purposes of this Agreement, and Xencor shall not have any further right to Opt-In to any additional Global Target Pairs. If Xencor does not exercise its Opt-In Right with respect to such Global Target Pair, such Global Target Pair shall remain a Global Target Pair, and Xencor's Opt-In Right shall terminate.

#### 5. DEVELOPMENT

-47-

## 5.1. Regional Licensed Antibodies and Regional Licensed Products.

**5.1.1.** Overview. Subject to the oversight of the JSC and the JDC, on a Regional Target Pair-by-Regional Target Pair basis, the Parties will collaborate on further global Development of Regional Licensed Antibodies and Regional Licensed Products in accordance with this Agreement and the RLP Development Plan with each Party's responsibility for Development activities specifically related to obtaining Regulatory Approval in its Territory.

# **5.1.2.** Development Plans, Budgets.

- RLP Development Plans. On a Regional Target Pair-by-Regional Target Pair basis, the Development activities that are anticipated to be necessary or useful to be undertaken over the next [...\*\*\*...] for the applicable Regional Licensed Antibodies and Regional Licensed Products to achieve Regulatory Approvals will be set forth in a written work plan and time table to be agreed by the Parties within [...\*\*\*...] of the Effective Date, along with a preliminary RLP Development Budget that satisfies the requirements of Section 5.1.2.2 (each, as updated from time to time, a "RLP **Development Plan**"). Each RLP Development Plan will (i) allocate responsibility for the performance of each RLP Development Activity to one or both of the Parties in their respective Territories or on a global basis, which allocation shall provide for each Party to be responsible for the performance of certain Phase 2 Studies and certain Phase 3 Studies, and (ii) focus on obtaining Regulatory Approval of the Regional Licensed Products. The terms of, and Development activities set forth in each RLP Development Plan will at all times be designed to be in compliance with all Laws and in accordance with professional and ethical standards customary in the pharmaceutical industry. The Parties will update the applicable RLP Development Plan for such Regional Licensed Antibodies and Regional Licensed Products [...\*\*\*...] and will provide such updated RLP Development Plan to the JDC. The JDC will review and approve each RLP Development Plan submitted to it for submission to the JSC in accordance with Section 2.3.
- 5.1.2.2. RLP Development Budgets. Each RLP Development Plan will contain (i) a [...\*\*\*...] rolling budget covering costs associated with, and not subject to the following subclause (ii), the anticipated RLP Development Activities to be performed during the then-current Calendar Year (broken down by Calendar Quarter) and the next Calendar Year (broken down by Calendar Quarter), and a forecast of the budgets for each subsequent Calendar Year thereafter through completion of all RLP Development Activities set forth in any such RLP Development Plan and (ii) a specific budget for the entirety of each Clinical Trial that the Parties include in the RLP Development Plan ("Clinical Trial Budget"), provided that each initial RLP Development Plan will also include such a budget for the partial Calendar Year commencing as of the date of such RLP Development Plan and ending December 31 of such Calendar Year (each such budget plus any such partial Calendar Year is a "RLP Development Budget"). Each RLP Development Budget will be updated [...\*\*\*...] by the JDC in accordance with Section 5.1.2.3; provided that, a Clinical Trial Budget shall

not be further revised thereafter until the next [...\*\*\*...] update without the Parties' mutual agreement. The initial RLP Development Budget for a Regional Target Pair, and each update thereto, will be prepared by the JDC, based on (a) the Parties' good faith estimation of the anticipated RLP Development Activities to be conducted during the relevant [...\*\*\*...] and (b) information prepared by the Parties in good faith for their own internal planning processes relating to anticipated RLP Development Activities for such Regional Target Pair, a summary of which will be provided to the JDC for review and incorporation into the RLP Development Budget. Each RLP Development Budget will include an itemized list of the applicable RLP Development Activities to be performed during the [...\*\*\*...] covered by such RLP Development Budget, with detailed line item entries for each RLP Development Activity setting forth the costs directly related to such RLP Development Activity (broken out to show Out-of-Pocket Costs and FTE Costs for FTEs directly engaged to perform such RLP Development Activity) and specifying what Party or Third Party is responsible for performing the applicable RLP Development Activity, which itemized list may include:

- (i) Any material preclinical, non-clinical studies or GLP toxicology studies, itemized by study,
- (ii) Clinical Studies, including (A) the following costs itemized by Clinical Study, if applicable: itemized cost of study drug (if available), per-patient costs (if available), study initiation costs, per-protocol costs, investigator payments, monthly study fixed costs, statistical analysis costs, study report development costs and study close-out expenses and (B) the following information itemized by Clinical Study: number of study sites and number of patients, and
- (iii) allocation of responsibility for Manufacturing the applicable Regional Licensed Antibodies and Regional Licensed Products.

Each RLP Development Budget will also include a high level summary of the Parties' plans with respect to the Development of such Regional Target Pair as they relate to such budget for the years following the [...\*\*\*...] of such RLP Development Budget.

5.1.2.3. Managing and Amending RLP Development Plans and RLP Development Budgets. The JDC will update and amend the applicable RLP Development Plan from time-to-time as it deems necessary and, until such time as no further RLP Development Activities are occurring or expected to occur with respect to a given Regional Target Pair. Each RLP Development Budget will be updated [...\*\*\*...] by the JDC and agreed to at least [...\*\*\*...] prior to the commencement of the then-next Calendar Year or sooner should there be a change in the RLP Development Plan; provided that, a Clinical Trial Budget shall not be further revised thereafter until the next [...\*\*\*...] update without the Parties' mutual agreement.

## **5.1.2.4.** Supplemental Studies.

- (a) Additional Development Proposals. If a Party desires to conduct a Supplemental Study of a Regional Licensed Product for a Regional Target Pair such Party (the "Proposing Party" and such other Party, the "Non-Proposing Party") will submit to the JDC a proposal to add such Supplemental Study to the applicable RLP Development Plan (an "Additional Development Proposal"). Each Additional Development Proposal will describe in reasonable detail the applicable Regional Target Pair and the Supplemental Study(ies) that the Proposing Party desires to conduct, including a synopsis of the trial or activities, the proposed enrollment criteria, number of patients to be included, endpoints to be measured, estimated drug supply requirements and statistical design and powering (the "Additional Development Activities"), as well as a proposed timeline and budget and an analysis of the business opportunity and revenue potential for such Additional Development Activities.
- (b) JDC Decision Regarding Additional Development Activities. The JDC will approve or reject an Additional Development Proposal within [...\*\*\*...] after receipt thereof from the Proposing Party as set forth in this Section 5.1.2.4.
  - (i) If the JDC approves an Additional Development Proposal, upon such an approval, the applicable RLP Development Plan will be amended to include the Additional Development Activities, including the proposed timeline and budget for such Additional Development Activities, set forth in such Additional Development Proposal (as may be amended by the JDC) upon such approval. Any Additional Development Activities included in a RLP Development Plan pursuant to this Section 5.1.2.4 will be deemed to be RLP Development Activities for all purposes under Section 5.1.4.
  - (ii) If the JDC fails to approve an Additional Development Proposal, the Supplemental Study proposed in the Additional Development Proposal will not be deemed an RLP Development Activity for any purpose under this Agreement, and Sections 5.1.2.4(c) and 5.1.2.4(d) will apply.
  - (c) Independent Performance of Additional Development Activities.
  - (i) If the JDC fails to approve for inclusion in the RLP Development Plan a Proposing Party's Additional Development Proposal for a Supplemental Study(ies) for such Regional Licensed Product, the Proposing Party may, upon notice to the Non-Proposing Party, conduct the proposed Supplemental Study(ies) at its own expense, provided however, that if the Non-Proposing Party determines

reasonably and in good faith that the performance of such proposed Supplemental Study(ies) would pose safety concerns or other ethical concerns, then the Proposing Party will not undertake such Supplemental Study(ies) unless and until the Non-Proposing Party determines that such Additional Development Activities should be permitted.

- (ii) Notwithstanding anything in Sections 7.1.3 and 7.3 to the contrary, if the JDC does not approve an Additional Development Proposal, unless and until the Non-Proposing Party delivers an Additional Development Opt-In Notice with respect to such Additional Development Activity, as described in Section 5.1.2.4(d), the Non-Proposing Party will not have any rights under Section 7.1.3 and 7.3, as applicable, with respect to any information or data generated from any Supplemental Study that was the subject of the unapproved Additional Development Proposal or from any future information or data generated from any future Clinical Studies with respect to the same Indication, other than (A) to use such information or data to determine whether to deliver an Additional Development Opt-In Notice in accordance with Section 5.1.2.4(d), (B) as permitted pursuant to the SDEA, (C) as required by Law and (D) [...\*\*\*...].
- (d) Opt-In for Additional Development Activities. In the event that the Proposing Party conducts a Supplemental Study pursuant to Section 5.1.2.4(c), the Proposing Party will provide to the Non-Proposing Party (i) a summary of the information or data generated from such Supplemental Study, including the final Clinical Study report, (ii) a summary of the proposed Development activities, which shall be required to include the next Clinical Study Phase after such Supplemental Study for such Regional Licensed Product, (iii) the costs and expenses specified in clause (1) below, and (iv) a budget for such proposed new Clinical Study for such Regional Licensed Product (the "Additional Development Data Package"). The Non-Proposing Party shall have the [...\*\*\*...] right to elect, in its sole discretion and upon written notice to the Proposing Party no later than [...\*\*\*...] after the date the Additional Development Data Package is made available to the Non-Proposing Party (an "Additional Development Opt-In Notice"), to opt in with respect to any Supplemental Study that was the subject of such Additional Development

Proposal that the Proposing Party elected to conduct in accordance with Section 5.1.2.4(c), and then (A) such Supplemental Study will be deemed to be an RLP Development Activity under the RLP Development Plan for the applicable Regional Target Pair from and after the date on which such Additional Development Opt-In Notice is received by the Proposing Party (the "Additional Development Opt-In Date"), (B) the then-current plan and budget of the Proposing Party with respect to such Supplemental Study will be deemed to be included within, and part of, the RLP Development Plan for such Regional Licensed Product as of the Additional Development Opt-In Date, and will control with respect to such Supplemental Study unless and until an amendment to the RLP Development Plan providing for a different or modified plan and budget is approved by the JDC, and (C) the Non-Proposing Party will have all rights granted to it under Section 7.1.3 and 7.3 with respect to the information and data generated from such Supplemental Study as if such Supplemental Study was conducted under the RLP Development Plan for such Regional Licensed Product, provided that, (1) the Non-Proposing Party's right to so opt-in with respect to such Additional Development Activities, triggering the results described in the foregoing clauses (A) through (C), is conditioned on the payment by the Non-Proposing Party to the Proposing Party of a payment of [...\*\*\*...]% of those costs and expenses incurred by the Proposing Party prior to the Additional Development Opt-in Date that the Non-Proposing Party should have paid in connection with such Additional Development Activities had such Additional Development Activities been included in the RLP Development Plan pursuant to Section 5.1.2.4(b)(i), and (2) any future Development Costs with respect to such Regional Licensed Product, including any future Clinical Studies, will be allocated in accordance with Section 5.1.4.

- **5.1.3.** Diligence. On a Regional Target Pair-by-Regional Target Pair basis, (a) Novartis will use Commercially Reasonable Efforts to Develop Regional Licensed Antibodies and Regional Licensed Products for such Regional Target Pair and obtain Regulatory Approval therefor in the Novartis Territory, and (b) Novartis and Xencor will each use Commercially Reasonable Efforts to perform the RLP Development Activities allocated to it under the RLP Development Plan for such Regional Licensed Antibodies and Regional Licensed Products for such Regional Target Pair in accordance with the RLP Development Plan.
- **5.1.4.** Development Costs. With respect to each Regional Target Pair, Xencor will be responsible for 50% of all Development Costs and Novartis will be responsible for 50% of all Development Costs, in each case for the Development of all Regional Licensed Antibodies or Regional Licensed Products. For clarity, Development Costs will be reconciled and paid in accordance with the procedure described in Section 10.3.
- **5.1.5.** Records, Reports, Information Sharing. On a Regional Target Pair-by-Regional Target Pair basis, each Party will provide to the JDC, on a [...\*\*\*...] basis, or more frequently as reasonably requested by the JDC, an update regarding Development activities

conducted by or on behalf of such Party, as well as any Supplemental Studies and Post-Marketing Studies conducted by or on behalf of such Party. In addition, each Party will promptly share with the other Party all material developments and information that it comes to possess relating to the Development of any Regional Licensed Antibodies and Regional Licensed Products, including (a) safety concerns, and (b) study reports and data generated from Clinical Studies, provided however, that excluding (i) safety concerns or as required under the SDEA,) or (ii) as required under Section 5.1.2.4(c)(ii)(D) a Party as Proposing Party will not be obligated to share any study reports and data generated from Clinical Studies for any Additional Development Activities conducted by or on behalf of the Proposing Party where the Non-Proposing Party has not exercised an Additional Development Opt-in Notice other than to permit the Non-Proposing Party data to determine whether to deliver an Additional Development Opt-In Notice in accordance with Section 5.1.2.4(d) or to exercise its rights under Section 5.1.2.4(c)(ii)(D).

- **5.1.5.1.** Scientific Records. Each Party will maintain scientific records, in sufficient detail and in sound scientific manner appropriate for Patent and regulatory purposes and in compliance with cGLP with respect to activities that require cGLP compliance to be submitted in regulatory filings (including INDs and BLAs), which will accurately reflect work done and results achieved in the performance of the Development activities, Clinical Studies, and Supplemental Studies.
- Information Exchange and Development Assistance. Until the expiration or termination of the final RLP Development Plan, upon the reasonable request of the other Party, each Party will provide to the other Party, without additional compensation and in a commercially reasonable format, Know-How Controlled by such Party or its Related Parties that is licensed to the other Party under this Agreement to the extent that it is reasonably necessary for Development of Regional Licensed Antibodies or Regional Licensed Products in the requesting Party's Territory in accordance with the applicable RLP Development Plan or for obtaining or maintaining Regulatory Approval for Regional Licensed Products in the requesting Party's Territory, including copies of (a) all material scientific information and data related to such Regional Licensed Antibodies or Regional Licensed Products (including all material data made, collected or otherwise generated in the conduct of any pre-clinical studies, Clinical Studies, or Supplemental Studies for which a Party as Non-Proposing Party has exercised its Additional Development Opt-In Notice, or early access/named patient programs, as well as CMC information), and (b) protocols and investigator brochures, in each case, that are reasonably necessary for the other Party (or its Related Parties) to perform its obligations or exploit its rights under this Agreement with respect to such Regional Licensed Antibodies or Regional Licensed Products. Notwithstanding the foregoing, or anything to the contrary herein, neither Party shall be required to disclose information that is subject to bona fide confidentiality obligations to a Third Party; provided that if the rights granted to the other Party hereunder would reasonably be expected to be adversely affected or such Party otherwise would be prejudiced hereunder by such failure to disclose, the Party bound by such confidentiality obligations will use commercially reasonable efforts to obtain the consent of such Third Party to disclose such information.

- **5.1.5.3.** Personnel. Each Party may request, through the JDC or the other Party's Alliance Manager, that the other Party reasonably make available for consultation regarding the Development of such Regional Licensed Antibodies or Regional Licensed Products certain of its employees engaged in Development activities and Supplemental Studies for which a Party as Non-Proposing Party has exercised its Additional Development Opt-In Notice. The JDC or the Alliance Managers will reasonably coordinate, upon reasonable notice during normal business hours and at their respective places of employment, consultation between the Parties on the progress of the Development, including any Supplemental Studies for which a Party as Non- Proposing Party has exercised its Additional Development Opt-In Notice.
- Third Parties. The Parties will be entitled to utilize the services of Third Parties to perform their respective Development under this Section 5.1, provided that (a) each Party will require that such Third Party operates in a manner consistent with this Agreement, (b) each Party will remain at all times fully liable for its respective responsibilities, and such Party hereby expressly waives any requirement that the other Party exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against the Party who engaged such Third Party and (c) the Parties will make reasonable efforts to share, through the JDC, information regarding any prior experience with specific contract research organizations that are anticipated to be engaged to perform work under the RLP Development Plan. Each Party will require that any Third Party agreement entered into pursuant to this Section 5.1.6 (x) include confidentiality and non-use provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (y) obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) under and to, any Know-How and Patents that are developed by such Third Party in the performance of such agreement and are reasonably necessary or useful to Research, Develop, Manufacture or Commercialize Regional Licensed Antibodies or Regional Licensed Products in the Field, and any such license shall be deemed to be a Novartis Existing In-License or Xencor Existing In-License, as applicable, for all purposes of this Agreement, unless otherwise agreed by the JDC. For clarity, the foregoing requirement to obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) shall not apply to any improvements to the proprietary core or platform technology owned or in-licensed by any such Third Party or its Affiliates unless such improvements are reasonably necessary to Research, Develop, Manufacture or Commercialize those Licensed Antibodies or Licensed Products with respect to which such Third Party or its Affiliate conducted its activities under such Third Party agreement. The Party utilizing the services of a Third Party service provider will be solely responsible for direction of and communications with such Third Party.
- **5.1.7.** Combination Therapy. Notwithstanding anything to the contrary herein, on a Regional Licensed Product-by-Regional Licensed Product basis, in the event a Party desires to

Develop, Manufacture and Commercialize a Combination Therapy containing or comprising a Regional Licensed Product and one or more other products (an "Other Combination Therapy Product(s)"), such Party shall provide notice of its interest with respect thereto to the other Party and a proposed plan for the Development of such Combination Therapy and the Parties shall discuss such proposal. In the event that the Development of such Combination Therapy is approved by the JSC, it shall be included in the RLP Development Plan; <u>provided</u> that in the event that the Other Combination Therapy Product(s) is proprietary to one of the Parties (a "Proprietary Product"), then the Development Costs associated with such Combination Therapy shall be apportioned [...\*\*\*...]. For clarity, in the event that the Development of such Combination Therapy is not approved by the JSC, such Development of a Combination Therapy may be conducted as Additional Development Activities pursuant to Section 5.1.2.4(c).

**5.1.8.** Opt-Out Right. On a Regional Target Pair-by-Regional Target Pair basis, Xencor has the right, at its sole discretion, to opt-out of further Development and Commercialization of all Regional Licensed Antibodies or Regional Licensed Products for such Regional Target Pair upon [...\*\*\*...] prior written notice to Novartis. Upon the delivery of such notice, (a) Xencor's then on-going funding commitments and Development Activities will continue until [...\*\*\*...] of the date of the notice; provided that [...\*\*\*...], (b) the Novartis Territory for such Regional Target Pair will be worldwide for all purposes of this Agreement and the term Xencor Territory shall no longer have any applicability for such Regional Target Pair, (c) all Regional Licensed Antibodies and Regional Licensed Products with respect to such Regional Target Pair will convert to Global Licensed Antibodies and Global Licensed Products (subject to sub-clause (a) above and royalty payments in accordance with Section 10.6.1.2 and not 10.6.2), respectively, and (d) the licenses set forth in Section 9.1.2.1 will terminate. For the sake of clarity, Xencor will have no further obligations with respect to the Development of, or Development Costs associated with, such Regional Target Pair from and after the end of [...\*\*\*...] described in sub-clause (a) above; provided that the Parties will use Commercially Reasonable Efforts to transition on-going activities for which Xencor was responsible to Novartis during [...\*\*\*...]

### 5.2. Global Licensed Antibodies and Global Licensed Products.

- **5.2.1.** Overview. On a Global Target Pair-by-Global Target Pair basis, Novartis will be responsible for all Development of Global Licensed Antibodies and Global Licensed Products.
- **5.2.2.** Diligence, Standards of Conduct. On a Global Target Pair-by-Global Target Pair basis, Novartis will use Commercially Reasonable Efforts to Develop Global Licensed Antibodies and Global Licensed Products for such Global Target Pair and obtain Regulatory Approval thereof in the Novartis Territory.
- **5.2.3.** Development Costs. With respect to each Global Target Pair, Novartis will be responsible for 100% of all Development Costs for the Development of Global Licensed Antibodies and Global Licensed Products for such Global Target Pair.

## **5.2.4.** Records, Reports, Information Sharing.

- 5.2.4.1. Development Activities, Reports. On a Global Target Pair-by-Global Target Pair basis, Novartis will provide to the JDC, on a [...\*\*\*...] basis, an update regarding Development activities for such Global Target Pair, including any material Research or pre-clinical Development activities or Clinical Studies conducted by Novartis. Without limiting the foregoing, on a Global Target Pair-by-Global Target Pair basis, Novartis shall provide Xencor with annual written reports within [...\*\*\*...] after the end of each Calendar Year, summarizing Novartis' and its Affiliates' and Sublicensees' Development efforts with respect to such Global Target Pair during such Calendar Year, including significant preclinical, clinical and regulatory events such as initiation and results of Clinical Studies and filing of significant Regulatory Materials and a description of planned activities for the following Calendar Year. In addition to such report and during such same period, at Xencor's reasonable request, Xencor and Novartis will meet in-person or remotely by telephone or video-conference at least [...\*\*\*...] in order for Novartis to update Xencor regarding the matters described in this Section 5.2.4.1.
- **5.2.4.2.** Information Exchange and Development Assistance. Upon the reasonable request of Novartis, Xencor will provide to Novartis, without additional compensation and in a commercially reasonable format, a copy of Know-How Controlled by Xencor or its Related Parties that is licensed to Novartis under this Agreement (i.e., Know-How included in Xencor Technology for Xencor) to the extent that it is reasonably necessary for Development of Global Licensed Antibodies or Global Licensed Products or for obtaining or maintaining Regulatory Approval or Pricing Approval for Global Licensed Products, including copies of all scientific information and data related to such Global Licensed Antibodies or Global Licensed Products (including all data made, collected or otherwise generated in the conduct of any pre-clinical studies). Notwithstanding the foregoing, or anything to the contrary herein, Xencor shall not be required to disclose information that is subject to bona fide

confidentiality obligations to a Third Party; <u>provided</u> that if the rights granted to Novartis hereunder would reasonably be expected to be adversely affected or Novartis otherwise would be prejudiced hereunder by such failure to disclose, Xencor will use commercially reasonable efforts to obtain the consent of such Third Party to disclose such information.

- **5.2.4.3.** Personnel. Novartis may request that Xencor reasonably make available for consultation regarding the Development of Global Licensed Antibodies and Global Licensed Products certain of its employees engaged in Research and Development activities with respect to such Global Licensed Antibodies and Global Licensed Products. Xencor will reasonably cooperate (and subject to capacity restraints at Xencor) with Novartis to provide (a) up to [...\*\*\*...] of consultation without charge to Novartis, and (b) any additional hours of consultation as Novartis may reasonably request, for which Novartis will pay Xencor a rate of [...\*\*\*...] of such consultation services.
- Third Parties. Novartis will be entitled to utilize the services of Third Parties to perform its 5.2.5. Development activities under this Section 5.2.5, provided that (a) Novartis will require that such Third Party operates in a manner consistent with the terms of this Agreement and (b) Novartis will remain at all times fully liable for its responsibilities and Novartis hereby expressly waives any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis. Novartis will require that any such Third Party agreement entered into pursuant to this Section 5.2.5 (x) include confidentiality and non-use provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (y) obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) under and to, any Know-How and Patents that are developed by such Third Party in the performance of such agreement and are reasonably necessary or useful to Research, Develop, Manufacture or Commercialize Global Licensed Antibodies or Global Licensed Products in the Field, and any such license shall be deemed to be a Novartis Existing In-License for all purposes of this Agreement, unless otherwise agreed by the JDC. For clarity, the foregoing requirement to obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) shall not apply to any improvements to the proprietary core or platform technology owned or in-licensed by any such Third Party or its Affiliates. Novartis will be solely responsible for direction of and communications with such Third Party.

## 5.3. Optioned Licensed Antibodies and Optioned Licensed Products.

- **5.3.1.** Overview. Subject to the oversight of the JSC and the JDC, the Parties will collaborate on further global Development of Optioned Licensed Antibodies and Optioned Licensed Products in accordance with this Agreement and the OLP Development Plan.
  - **5.3.2.** Development Plan, Budgets.

- **5.3.2.1.** OLP Development Plan. Promptly upon exercise by Xencor of its Opt-In Right with respect to an Optioned Target Pair, the JDC will generate a written work plan and time table that sets forth the Development activities that are necessary or useful to be undertaken over the next [...\*\*\*...] for the applicable Optioned Licensed Antibodies and Optioned Licensed Products to achieve initial Regulatory Approval (as updated from time to time, the "**OLP Development Plan**"). The terms of, and Development activities set forth in, each OLP Development Plan will at all times be designed to be in compliance with all Laws and in accordance with professional and ethical standards customary in the pharmaceutical industry. The Parties will update the applicable OLP Development Plan for such Optioned Licensed Antibodies and Optioned Licensed Products [...\*\*\*...] and will provide such updated OLP Development Plan to the JDC. The JDC will review and approve the OLP Development Plan submitted to it in accordance with Section 2.3.4.
- OLP Development Budgets. The OLP Development Plan will contain a [...\*\*\*...] rolling budget covering the anticipated Development activities to be performed during the thencurrent Calendar Year (broken down by Calendar Ouarter) and the next Calendar Year (broken down by Calendar Quarter), and a forecast of the budgets for each subsequent Calendar Year thereafter through completion of all Development activities set forth in any such OLP Development Plan, provided that the initial OLP Development Plan will also include such a budget for the partial Calendar Year commencing as of the date of such OLP Development Plan and ending December 31 of such Calendar Year (each such one year budget plus any such partial Calendar Year is a "OLP **Development Budget**"). Each OLP Development Budget will be updated [...\*\*\*...] by the JDC in accordance with Section 5.3.2.1. The OLP Development Budget, and each update thereto, will be prepared by the JDC, based on (a) the Parties' good faith estimation of the anticipated OLP Development Activities to be conducted during the relevant [...\*\*\*...] period and (b) information prepared by the Parties in good faith for their own internal planning processes relating to anticipated Development activities, a summary of which will be provided to the JDC for review and incorporation into the OLP Development Budget. The OLP Development Budget will include an itemized list of the applicable Development activities to be performed during the [...\*\*\*...] covered by such OLP Development Budget, with detailed line item entries for each Development activity setting forth the costs directly related to such Development activity (broken out to show Out-of-Pocket Costs and FTE Costs for FTEs directly engaged to perform such Development activity) and specifying what Party or Third Party is responsible for performing the applicable Development activity, which itemized list may include:
  - (a) Any material preclinical, non-Clinical Studies or GLP toxicology studies, itemized by study, and
  - (b) Clinical Studies, including (A) the following costs itemized by Clinical Study, if applicable: itemized cost of study drug (if available), per-

patient costs (if available), study initiation costs, per-protocol costs, investigator payments, monthly study fixed costs, statistical analysis costs, study report development costs and study close-out expenses and (B) the following information itemized by Clinical Study: number of study sites and number of patients.

- **5.3.2.3.** Managing and Amending OLP Development Plans and OLP Development Budgets. The JDC will update and amend the OLP Development Plan from time-to-time as it deems necessary and, until such time as no further Development activities are occurring or expected to occur with respect to such Optioned Target Pair.
- **5.3.3.** Diligence. Novartis will use Commercially Reasonable Efforts to (a) Develop Optioned Licensed Antibodies and Optioned Licensed Products for the Optioned Target Pair and obtain Regulatory Approval therefor in the Novartis Territory, and (b) perform the Development activities pursuant to the OLP Development Plan for such Optioned Licensed Antibodies and Optioned Licensed Products for such Optioned Target Pair in accordance with the OLP Development Plan.
- **5.3.4.** Development Costs. With respect to each Optioned Target Pair, Xencor will be responsible for 25% of all Development Costs and Novartis will be responsible for 75% of all Development Costs, in each case for the Development of all Optioned Licensed Antibodies or Optioned Licensed Products from and after the date that Xencor exercises its Opt-In Right with respect thereto. For clarity, Development Costs will be reconciled and paid in accordance with the procedure described in Section 10.3.
  - **5.3.5.** Records, Reports, Information Sharing.
  - **5.3.5.1.** Development Activities Reports. Novartis will provide to the JDC, on a [...\*\*\*...] basis, or more frequently as reasonably requested by the JDC, an update regarding Development activities conducted by or on its behalf. In addition, Novartis will promptly share with Xencor all material developments and information that it comes to possess relating to the Development of any Optioned Licensed Antibodies and Optioned Licensed Products, including (a) safety concerns, and (b) study reports and data generated from Clinical Studies.
  - **5.3.5.2.** Scientific Records. Novartis will maintain scientific records, in sufficient detail and in sound scientific manner appropriate for Patent and regulatory purposes and in compliance with cGLP with respect to activities that require cGLP compliance to be submitted in regulatory filings (including INDs and BLAs), which will fully and accurately reflect all work done and results achieved in the performance of the Development activities and Clinical Studies.
  - **5.3.5.3.** Information Exchange and Development Assistance. Until the expiration or termination of the final OLP Development Plan, upon the reasonable request of Novartis, Xencor will provide to Novartis, without additional compensation

and in a commercially reasonable format, Know-How Controlled by Xencor or its Related Parties that is licensed to Novartis under this Agreement to the extent that it is reasonably necessary for Development of Optioned Licensed Antibodies or Optioned Licensed Products in accordance with the applicable OLP Development Plan or for obtaining or maintaining Regulatory Approval for Optioned Licensed Products, including copies of all material scientific information and data related to such Optioned Licensed Antibodies or Optioned Licensed Products (including all material data made, collected or otherwise generated in the conduct of any pre-clinical studies). Notwithstanding the foregoing, or anything to the contrary herein, neither Party shall be required to disclose information that is subject to *bona fide* confidentiality obligations to a Third Party; provided that if the rights granted to the other Party hereunder would reasonably be expected to be adversely affected or such Party otherwise would be prejudiced hereunder by such failure to disclose, the Party bound by such confidentiality obligations will use commercially reasonable efforts to obtain the consent of such Third Party to disclose such information.

- **5.3.5.4.** Personnel. Novartis may request that Xencor reasonably make available for consultation regarding the Development of Optioned Licensed Antibodies and Optioned Licensed Products certain of its employees engaged in Research and Development activities with respect to such Optioned Licensed Antibodies and Optioned Licensed Products. Xencor will reasonably cooperate (and subject to capacity restraints at Xencor) with Novartis to provide (a) up to [...\*\*\*...] of consultation without charge to Novartis, and (b) any additional hours of consultation as Novartis may reasonably request, for which Novartis will pay Xencor a rate of [...\*\*\*...] of such consultation services.
- Third Parties. Novartis will be entitled to utilize the services of Third Parties to perform its Development activities under this Section 5.3, provided that (a) Novartis will require that such Third Party operates in a manner consistent with this Agreement, (b) Novartis will remain at all times fully liable for its respective responsibilities. Novartis hereby expressly waives any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis and (c) Novartis will make reasonable efforts to share, through the JDC, information regarding any prior experience with specific contract research organizations that are anticipated to be engaged to perform work under the OLP Development Plan. Novartis will (x) require that any Third Party agreement entered into pursuant to this Section 5.3.6 include confidentiality and non-use provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (y) obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) under and to, any Know-How and Patents that are developed by such Third Party in the performance of such agreement and are reasonably necessary or useful to Research, Develop, Manufacture or Commercialize Optioned Licensed Antibodies or Optioned Licensed Products in the Field, and any such license shall be deemed to be a Novartis Existing In-License for all purposes of this Agreement, unless otherwise agreed by the JDC. For clarity, the

foregoing requirement to obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) shall not apply to any improvements to the proprietary core or platform technology owned or in-licensed by any such Third Party or its Affiliates unless such improvements are reasonably necessary to Research, Develop, Manufacture or Commercialize those Optioned Licensed Antibodies or Optioned Licensed Products with respect to which such Third Party or its Affiliate conducted its activities under such Third Party agreement. Novartis will be solely responsible for direction of and communications with such Third Party.

Opt-Out Right. Xencor has the right, at its sole discretion, to opt-out of further cost sharing and any activities, if any, allocated to it pursuant to the OLP Development Plan with respect to all Optioned Licensed Antibodies and Optioned Licensed Products for the Optioned Target Pair upon [...\*\*\*...] prior written notice to Novartis. Upon the delivery of such notice, (a) Xencor's then on-going funding commitments and Development Activities will continue until the [...\*\*\*...] anniversary of the date of the notice; provided that during such [...\*\*\*...] period, Xencor's responsibility for Development Costs with respect to such Optioned Target Pair shall not exceed [...\*\*\*...]% of the Development Costs included in the OLP Development Budget in effect as of the date of Xencor's notice, notwithstanding any increases that may be made to such OLP Development Budget during such [...\*\*\*...] period (for clarity, if Novartis reduces the OLP Development Budget during such [...\*\*\*...] then Xencor shall only be responsible for [...\*\*\*...]% of the Development Costs included in such updated OLP Development Budget), (b) the Optioned Target Pair will convert to a Global Target Pair, (c) all Optioned Licensed Antibodies and Optioned Licensed Products will convert to Global Licensed Antibodies and Global Licensed Products, respectively, and (d) the licenses set forth in Section 9.2 will apply to such new Global Target Pair, Global Licensed Antibodies and Global Licensed Products; provided that, notwithstanding the foregoing, in the event that Xencor exercises its Opt-Out Right with respect to the Optioned Licensed Antibodies and Optioned Licensed Products within [...\*\*\*...] days of Xencor's receipt of the [...\*\*\*...] updated OLP Development Plan, then Xencor will have no obligation to share in any further Development Costs with respect thereto beyond those committed in the previously applicable OLP Development Budget for the next year. For example, if the prior OLP Development Budget provided for a single Clinical Study budgeted for \$[...\*\*\*...] and the updated OLP Development budget provides for such Clinical Study, but with a budget of \$[...\*\*\*...] and adds a new Clinical Study budgeted at \$[...\*\*\*...], Xencor would remain obligated to share its portion of the Development Costs for the first Clinical Study only and only at the originally budgeted amount (i.e., [...\*\*\*...]% of \$[...\*\*\*...] or \$[...\*\*\*...]). For the sake of clarity, Xencor will have no further obligations with respect to the Development Costs associated with such Optioned Target Pair from and after the end of such [...\*\*\*...] notice period.

### 5.4. Fc Licensed Antibodies and Fc Licensed Products.

**5.4.1.** Overview. Novartis will be responsible for all Development of Fc Licensed Antibodies and Fc Licensed Products.

- **5.4.2.** Diligence, Standards of Conduct. Novartis will use Commercially Reasonable Efforts to Develop Fc Licensed Antibodies and Fc Licensed Products and obtain Regulatory Approval thereof in the Novartis Territory.
- **5.4.3.** Development Costs. Novartis will be responsible for 100% of all Development Costs for the Development of Fc Licensed Antibodies and Fc Licensed Products.
- **5.4.4.** Reports. On an Fc Licensed Antibodies by Fc Licensed Antibodies and Fc Licensed Product-by-Fc Licensed Product basis, Novartis shall provide Xencor with annual written reports within [...\*\*\*...] after the end of each Calendar Year, summarizing Novartis' and its Affiliates' and Sublicensees' Development efforts with respect to such Fc Licensed Antibody or Fc Licensed Product during such Calendar Year, including significant preclinical, clinical and regulatory events such as initiation and results of Clinical Studies and filing of significant Regulatory Materials and a description of planned activities for the following Calendar Year.
- 5.4.5. Third Parties. Novartis will be entitled to utilize the services of Third Parties to perform its Development activities under this Section 5.4.5, provided that (a) Novartis will require that such Third Party operates in a manner consistent with the terms of this Agreement and (b) Novartis will remain at all times fully liable for its responsibilities and Novartis hereby expressly waives any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis. Novartis will require that any such Third Party agreement entered into pursuant to this Section 5.4.5 include confidentiality and non-use provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose). Novartis will be solely responsible for direction of and communications with such Third Party.

### 6. COMMERCIALIZATION

## **Regional Licensed Products**.

- **6.1.1.** Responsibility, Cost and Diligence. On a Regional Target Pair-by-Regional Target Pair basis, each Party will be solely responsible, at its expense, for all Commercialization activities relating to Regional Licensed Products in the Field in its Territory. On a Regional Target Pair-by-Regional Target Pair basis, Novartis will use Commercially Reasonable Efforts to (a) Commercialize each Regional Licensed Product for which it has obtained Regulatory Approval (including where required by Law to Commercialize the Regional Licensed Product, Pricing Approval) in its Territory, and (b) perform all Commercialization activities for such Regional Licensed Product in accordance with the RLP Commercial Strategy.
- **6.1.2.** RLP Commercial Strategy. Within [...\*\*\*...] before the anticipated Initiation of the first Phase 3 Study for a Regional Licensed Product, Novartis will provide and within [...\*\*\*...] after such provision the JCC will, subject to Law including Antitrust Laws, review, update and approve a written summary of the global Commercial strategy for such Regional Licensed Product (the "**RLP Commercial Strategy**"); provided that, until the RLP Commercial Strategy

is approved by the JCC or, if no such approval is obtained, neither Party shall be bound by the global Commercial strategy proposed by Novartis. For clarity, any and all such communications and strategy involving the Commercialization of Regional Licensed Products shall be limited to those permitted under Law, including Antitrust Laws.

- **6.1.3.** Novartis Territory Commercialization Plan. No less than [...\*\*\*...] in advance of the reasonably expected first Regulatory Approval in the Novartis Territory with respect to a Regional Licensed Product, Novartis will prepare and deliver to the JCC for review a reasonable written plan that summarizes the Commercialization activities to be undertaken with respect to such Regional Licensed Product in the Novartis Territory in the next Calendar Year (the "**Novartis Territory Commercialization Plan**"). The Novartis Territory Commercialization Plan for a Regional Licensed Product will subsequently be updated and modified by Novartis, from time to time at its discretion and no less frequently than [...\*\*\*...], based upon, among other things, Novartis' Commercialization activities with respect to such Regional Licensed Product in the Novartis Territory, a copy of which updated plan Novartis will provide to the JCC. Notwithstanding the foregoing, in the event of any disagreement between the Parties regarding the Novartis Territory Commercialization Plan for a Regional Licensed Product pursuant to Section 2.4.5, the Novartis representatives on the JCC will have final decision-making authority over the preparation and updating of such Novartis Territory Commercialization Plan, provided that such decisions do not materially adversely affect the Commercialization of such Regional Licensed Product in the Xencor Territory.
- **6.1.4.** Xencor Territory Commercialization Plan. No less than [...\*\*\*...] in advance of the reasonably expected first Regulatory Approval in the Xencor Territory with respect to a Regional Licensed Product, Xencor will prepare and deliver to the JCC for review a reasonable written plan that summarizes the Commercialization activities to be undertaken with respect to such Regional Licensed Product in the Xencor Territory in the next Calendar Year (the "**Xencor Territory Commercialization Plan**"). The Xencor Territory Commercialization Plan for a Regional Licensed Product will subsequently be updated and modified by Xencor, from time to time at its discretion and no less frequently than [...\*\*\*...] per [...\*\*\*...], based upon, among other things, Xencor's Commercialization activities with respect to such Regional Licensed Product in the Xencor Territory, a copy of which updated plan Xencor will provide to the JCC. Notwithstanding the foregoing, in the event of any disagreement between the Parties regarding the Xencor Territory Commercialization Plan for a Regional Licensed Product pursuant to Section 2.4.5, the Xencor representatives on the JCC will have final decision-making authority over the preparation and updating of such Xencor Territory Commercialization Plan, provided that such decisions do not materially adversely affect the Commercialization of such Regional Licensed Product in the Novartis Territory.
  - **6.1.5.** Advertising and Promotional Materials.
  - **6.1.5.1.** RLP Branding. Each Party will use Commercially Reasonable Efforts to develop (and thereafter modify and update) a branding strategy (including positioning, colors, other visual branding elements and Novartis RLP Trademarks and Xencor RLP Trademarks in accordance with Section 14.8.1) for each Regional

Licensed Product for use in the Field for such Party's Territory (each a "RLP Branding Strategy"), which the JCC will, in accordance with Sections 2.4.3.5 and 2.4.5, review, coordinate and approve, and which the Parties will, following such review and approval, implement. Each Party will submit its RLP Branding Strategy for a Regional Licensed Product to the JCC at least [...\*\*\*...] (or more frequently if agreed to by the Parties). Each Party will consider in good faith any timely comments by the other Party with respect to its RLP Branding Strategy, but will have final decision-making authority with respect to such its RLP Branding Strategy in its Territory. Notwithstanding the foregoing, each Party will use Commercially Reasonable Efforts to ensure that (a) its RLP Branding Strategy complies with Laws in its Territory, and (b) that any branding elements selected for inclusion in its RLP Branding Strategy do not infringe any Third Party trademarks or other intellectual property rights. If any such RLP Branding Strategy infringes Third Party trademarks or other intellectual property rights or otherwise does not comply with Law in the Territory in which such RLP Branding Strategy is used, the affected Party will take action to end such infringement or other noncompliance (including by modifying its RLP Branding Strategy) in its Territory and the other Party will not be obligated to implement its RLP Branding Strategy in its Territory pursuant to this Section 6.1.5.1 unless and until such infringement or noncompliance is ended.

- **6.1.5.2.** Xencor Advertising & Promotion. Xencor will be responsible for the creation, preparation, production, reproduction and filing with the applicable Regulatory Authorities, of relevant written sales, promotion and advertising materials ("**Promotional Materials**") relating to each Regional Licensed Product for use in the Xencor Territory. All such Promotional Materials will be compliant with applicable Law and, if applicable, consistent in all material respects with the Xencor Territory Commercialization Plan and, if applicable, consistent in all material respects with the RLP Branding Strategy for such Regional Licensed Product in the Xencor Territory. Xencor will submit representative samples of its Promotional Materials developed by it for use in the Xencor Territory to the JCC at least [...\*\*\*...] (or more frequently if agreed to by the Parties). Xencor will consider in good faith any timely comments Novartis may have with respect to any such Promotional Materials, but will have final decision-making authority in the Xencor Territory with respect to such Promotional Materials.
- **6.1.5.3.** Novartis Advertising & Promotion. Novartis will be responsible for the creation, preparation, production, reproduction and filing with the applicable Regulatory Authorities, of relevant Promotional Materials relating to each Regional Licensed Product for use in the Novartis Territory. All such Promotional Materials will be compliant with Law, consistent in all material respects with the Novartis Territory Commercialization Plan and, if applicable, consistent in all material respects with the RLP Branding Strategy for such Regional Licensed Product in the Novartis Territory. Novartis will submit representative samples of its Promotional Materials developed by it for use in the Novartis Territory to the JCC at least [...\*\*\*...] thereafter (or more frequently if agreed to by the Parties). Novartis will consider in good faith any timely

comments Xencor may have with respect to any such Promotional Materials, but will have final decision-making authority in the Novartis Territory with respect to such Promotional Materials.

- **6.1.5.4.** Reporting Obligations. Each Party will report to the JCC in writing, on an [...\*\*\*...] basis in the first [...\*\*\*...] of the [...\*\*\*...] following the first Regulatory Approval of such Regional Licensed Product in the Field in such Party's Territory (for the period ending December 31 of the prior Calendar Year), summarizing in reasonable detail such Party's Commercialization activities for such Regional Licensed Product performed to date (or updating such report for activities performed since the last such report was given hereunder, as applicable). In addition, each Party will provide the other Party with written notice of the First Commercial Sale of each Regional Licensed Product in such Party's Territory as soon as reasonably practicable after such event, provided however, that such Party will inform the other Party of such event prior to general public disclosure of such event by such Party. Subject to Law, each Party will provide such other information to the JCC as the other Party may reasonably request with respect to Commercialization of such Regional Licensed Product and will keep such JCC reasonably informed of such Party's Commercialization activities with respect to such Regional Licensed Product.
- **6.1.6.** Commercialization Obligations. Each Party and its Related Parties will be responsible for booking sales of the Regional Licensed Products sold in its Territory. Each Party and its Related Parties may warehouse Regional Licensed Products both inside and outside of such Party's Territory, <u>provided</u> that any sales with respect to such Regional Licensed Products are booked in such Party's Territory. If a Party receives any orders for any Regional Licensed Product in the other Party's Territory, it will refer such orders to the other Party, to the extent it is not prohibited from doing so under Law. Moreover, each Party and its Related Parties will, using Commercially Reasonable Efforts, be solely responsible for handling all returns of any Regional Licensed Product sold in its Territory, as well as all aspects of Regional Licensed Product order processing, invoicing and collection, distribution, inventory and receivables of Regional Licensed Products sold in its Territory.
- **6.1.7.** Recalls, Market Withdrawals or Corrective Actions. In the event that any Regulatory Authority issues or requests a recall or takes a similar action in connection with a Regional Licensed Product in a Territory, or in the event either Party determines that an event, incident or circumstance has occurred that may result in the need for a recall or market withdrawal of a Regional Licensed Product in its Territory, the Party notified of such recall or similar action, or the Party that desires such recall or similar action, will as promptly as possible, notify the other Party's Alliance Manager thereof by telephone or e-mail. Each Party, in consultation with the other Party, will decide whether to conduct a recall of a Regional Licensed Product in its own Territory and the manner in which any such recall will be conducted (except in the case of a government mandated recall, when such Party may act without such advance notice but will notify the other Party as soon as possible thereafter). Except as may otherwise be agreed to by the Parties, and subject to the Parties' respective indemnification obligations

hereunder or under any ancillary agreements between the Parties, each Party will bear the expense of any such recall in its own Territory. Each Party will make available all of its pertinent records that may be reasonably requested by the other Party that are necessary for a Party to effect a recall of a Regional Licensed Product in its Territory. The Parties' rights and obligations under this Section 6.1.7 will be subject to the terms of any supply agreement(s), SDEA or quality related agreements entered into between the Parties. In the event of a conflict between the provisions of any such supply agreement, SDEA or quality related agreements and this Section 6.1.7, the provisions of such supply agreement, SDEA or quality related agreements will govern.

## **6.1.8.** Ex-Territory Sales, Export Monitoring.

- **6.1.8.1.** Ex-Territory Sales. Subject to Law, neither Party will engage in any advertising or promotional activities relating to any Regional Licensed Product directed primarily to customers or other buyers or users of such Regional Licensed Product located outside its Territory or accept orders for Regional Licensed Products from or sell Regional Licensed Products into such other Party's Territory for its own account, and if a Party receives any order for any Regional Licensed Product in the other Party's Territory, it will refer such orders to the other Party. The Parties expressly acknowledge and agree that Law may prevent or limit a Party from taking action to prevent exports from one EU country to another.
- **6.1.8.2.** Export Monitoring. Each Party and its Related Parties will use Commercially Reasonable Efforts to monitor and prevent exports of Regional Licensed Products from its own Territory for Commercialization in the other Party's Territory using methods permitted under Law that are commonly used in the industry for such purpose (if any), and will promptly inform the other Party of any such exports of Regional Licensed Products from its Territory, and any actions taken to prevent such exports. Each Party agrees to take reasonable actions requested in writing by the other Party that are consistent with Law to prevent exports of Regional Licensed Products from its Territory for Commercialization in the other Party's Territory. The Parties expressly acknowledge and agree that Law may prevent or limit a Party from taking action to prevent exports from one EU country to another.

#### **6.2.** Global Licensed Products.

- **6.2.1.** Responsibility, Cost and Diligence. Novartis will be solely responsible, at its expense, for all Commercialization activities relating to Global Licensed Products in the Field in the Novartis Territory. Novartis will use Commercially Reasonable Efforts to Commercialize each Global Licensed Product for which Novartis has obtained Regulatory Approval within the Novartis Territory.
  - **6.2.2.** Advertising and Promotional Materials.
  - **6.2.2.1.** Global Licensed Product. Novartis will have the sole right, from time to time during the Term, to develop (and thereafter modify and update) a global branding strategy (including global positioning, messages, logo, colors and other visual

branding elements) for each Global Licensed Product for use in the Field throughout the Novartis Territory.

- **6.2.2.2.** Promotional Materials. Novartis will be responsible for the creation, preparation, production, reproduction and filing with the applicable Regulatory Authorities, of relevant written sales, promotion and advertising materials relating to each Global Licensed Product for use in the Novartis Territory. All such Promotional Materials will be compliant with Law. Unless prohibited by Law, Novartis will include a reference in such Promotional Materials to such Global Licensed Product as being sold under license from Xencor.
- **6.2.3.** Sales and Distribution. Novartis and its Related Parties will be solely responsible for booking sales and will warehouse and distribute Global Licensed Products throughout the Novartis Territory.
- **6.2.4.** Recalls, Market Withdrawals or Corrective Actions. In the event that any Regulatory Authority issues or requests a recall or takes a similar action in connection with a Global Licensed Product, Novartis will have the sole right to decide whether to conduct a recall and the manner in which any such recall will be conducted. Novartis will bear the expense of any such recall.

## **6.3.** Optioned Licensed Products.

- **6.3.1.** Responsibility, Cost and Diligence. Novartis will be solely responsible for all Commercialization activities relating to Optioned Licensed Products in the Field in the Novartis Territory; provided that Xencor shall have the right to co-detail Optioned Licensed Products in the U.S. in accordance with Section 6.3.2. Xencor will be responsible for 50% of all Commercialization Costs and costs of Medical Affairs Activities in the United States and 100% of the Commercialization Costs and costs of Medical Affairs Activities in the remainder of the Novartis Territory, in each case for the Commercialization of all Optioned Licensed Antibodies or Optioned Licensed Products. For clarity, Commercialization Costs and costs of Medical Affairs Activities in the United States will be reconciled and paid in accordance with the procedure described in Section 10.3. Novartis will use Commercially Reasonable Efforts to (a) Commercialize each Optioned Licensed Product for which Novartis has obtained Regulatory Approval (including where required by Law to Commercialize the Optioned Licensed Product, Pricing Approval) within the Novartis Territory, and (b) perform all Commercialization activities for Optioned Licensed Products in accordance with the OLP Commercialization Plan.
  - **6.3.2.** Xencor Co-Detail Right.

- **6.3.2.1.** Co-Detail Option of Xencor. With respect to each Optioned Licensed Product launched by Novartis in the United States, Xencor will have an option (the "Co-Detail Option") to Co-Detail such Optioned Licensed Product in the United States for each indication for which such Optioned Licensed Product has received Regulatory Approval according to the terms and conditions set forth in this Section 6.3.2. This Co-Detail Option may be exercised, at Xencor's discretion, on an [...\*\*\*...] basis. Novartis will provide a notice (the "Co-Detail Notice") at least [...\*\*\*...] before the anticipated approval of an application for Regulatory Approval to the FDA for each Optioned Licensed Product. In the event that Xencor wishes to Co-Detail such Optioned Licensed Product in the United States, it shall provide notice in writing to Novartis no later than [...\*\*\*...] after the giving of the Co-Detail Notice containing the information set forth below in Section 6.3.2.2 (the "Co-Detail Option Exercise Notice"). Each such Optioned Licensed Product for which Xencor exercises the Co-Detail Option in accordance with this Section 6.3.2.1 is referred to as a "Co-Detailed Optioned Licensed Product").
- **6.3.2.2.** Co-Detail Option Exercise Notice. The Co-Detail Option Exercise Notice provided by Xencor will include the following:
  - (a) [...\*\*\*...].
  - (b) [...\*\*\*...]:
    - (i) [...\*\*\*...]; and

- (ii) [...\*\*\*...].
- (c) [...\*\*\*...].
- (d) [...\*\*\*...].
- (e) [...\*\*\*...]

-69-

\*\*\*Confidential Treatment Requested

[...\*\*\*...].

- (f) [...\*\*\*...]:
  - (i) [...\*\*\*...].
  - (ii) [...\*\*\*...].
- (g) [...\*\*\*...].
- (h) [...\*\*\*...].

-70-

\*\*\*Confidential Treatment Requested

- **6.3.3.** OLP Commercialization Plan. No less than [...\*\*\*...] in advance of the reasonably expected first Regulatory Approval in the Novartis Territory with respect to an Optioned Licensed Product, Novartis will prepare and deliver to the JCC for review a reasonable written plan that summarizes the Commercialization activities, including pre-marketing activities, to be undertaken with respect to such Optioned Licensed Product in the Novartis Territory [...\*\*\*...] and the associated budget for such activities, which budget shall be allocated between the United States and the rest of the world (the "**OLP Commercialization Plan**"). The OLP Commercialization Plan for an Optioned Licensed Product will subsequently be updated and modified by [...\*\*\*...] by the JCC, based upon, among other things, Novartis' Commercialization activities with respect to such Optioned Licensed Product in the Novartis Territory. Notwithstanding the foregoing, in the event of any disagreement between the Parties regarding the OLP Commercialization Plan for an Optioned Licensed Product pursuant to Section 2.4.5, the Novartis representatives on the JCC will consider any comments provided by Xencor in good faith, but have final decision-making authority over the preparation and updating of such OLP Commercialization Plan.
- **6.3.4.** Commercialization Costs and Medical Affairs Activities Costs Records. Novartis shall keep accurate records of its Commercialization Costs and Medical Affairs Activities Costs, including appropriately allocating such Commercialization Costs between the United States and the rest of the world. Likewise, to the extent Xencor exercises its Co-Detailing Option, it too shall keep such records. Beginning with the [...\*\*\*...] Calendar Quarter after the First Commercial Sale of an Optioned Licensed Product, Novartis shall report to Xencor, pursuant to Section 10.3, within [...\*\*\*...] after the end of each Calendar Quarter a report of its Commercialization Costs, by applicable region (i.e., United States and the rest of the world), incurred in such Calendar Quarter of all of such Commercialization Costs and Medical Affairs Activities Costs (and Xencor will provide same to the extent it exercises its Co-Detailing Option) (each such report, a "Commercialization Cost Calculation Report").
  - **6.3.5.** Advertising and Promotional Materials.
  - **6.3.5.1.** Optioned Licensed Product. Novartis will have the sole right, from time to time during the Term, to develop (and thereafter modify and update) a global branding strategy (including global positioning, messages, logo, colors and other visual branding elements) for each Optioned Licensed Product for use in the Field throughout the Novartis Territory.
  - **6.3.5.2.** Promotional Materials. Novartis will be responsible for the creation, preparation, production, reproduction and filing with the applicable Regulatory

Authorities, of relevant written sales, promotion and advertising materials relating to each Optioned Licensed Product for use in the Novartis Territory. All such Promotional Materials will be compliant with Law. Unless prohibited by Law, Novartis will include a reference in such Promotional Materials to such Optioned Licensed Product as being sold under license from Xencor; <u>provided</u> that, to the extent Xencor exercises its Co-Detailing Option, both Parties' logos shall be displayed equally prominently in the United States on Promotional Materials and Product branding.

- **6.3.6.** Sales and Distribution. Novartis and its Related Parties will be solely responsible for booking sales and will warehouse and distribute Optioned Licensed Products in the Novartis Territory.
- **6.3.7.** Recalls, Market Withdrawals or Corrective Actions. In the event that any Regulatory Authority issues or requests a recall or takes a similar action in connection with a Optioned Licensed Product, Novartis will have the sole right to decide whether to conduct a recall and the manner in which any such recall will be conducted. Novartis will bear the expense of any such recall.

## **6.4.** Fc Licensed Products.

- **6.4.1.** Responsibility, Cost and Diligence. Novartis will be solely responsible, at its expense, for all Commercialization activities relating to Fc Licensed Products in the Field in the Novartis Territory. Novartis will use Commercially Reasonable Efforts to Commercialize each Fc Licensed Product for which Novartis has obtained Regulatory Approval (including where required by Law to Commercialize the Fc Licensed Product, Pricing Approval) in the Novartis Territory.
  - **6.4.2.** Advertising and Promotional Materials.
  - **6.4.2.1.** Fc Licensed Product. Novartis will have the sole right, from time to time during the Term, to develop (and thereafter modify and update) a global branding strategy (including global positioning, messages, logo, colors and other visual branding elements) for each Fc Licensed Product for use in the Field throughout the Novartis Territory.
  - **6.4.2.2.** Promotional Materials. Novartis will be responsible for the creation, preparation, production, reproduction and filing with the applicable Regulatory Authorities, of relevant written sales, promotion and advertising materials relating to each Fc Licensed Product for use in the Novartis Territory. All such Promotional Materials will be compliant with Law. Unless prohibited by Law, Novartis will include a reference in such Promotional Materials to such Fc Licensed Product as being sold under license from Xencor.

- **6.4.3.** Sales and Distribution. Novartis and its Related Parties will be solely responsible for booking sales and will warehouse and distribute Fc Licensed Products in the Novartis Territory.
- **6.4.4.** Recalls, Market Withdrawals or Corrective Actions. In the event that any Regulatory Authority issues or requests a recall or takes a similar action in connection with a Fc Licensed Product, Novartis will have the sole right to decide whether to conduct a recall and the manner in which any such recall will be conducted. Novartis will bear the expense of any such recall.
- 6.5. <u>Commercialization Activities, Reports.</u> <u>Without limiting the foregoing, on an Fc Licensed Product-by-Fc Licensed Product basis, Novartis shall provide Xencor with annual written reports within [...\*\*\*...] after the end of each Calendar Year, summarizing <u>Novartis</u>' and its Affiliates' and Sublicensees' Commercialization efforts with respect to such Fc Licensed Product during such Calendar Year and a description of planned activities for the following Calendar Year.</u>

### 7. REGULATORY

## 7.1. Regional Licensed Products.

**7.1.1.** Regulatory Filings and Interactions.

## **7.1.1.1.** Responsibilities.

Pursuant to the RLP Development Plan for a Regional Licensed Product and, except as otherwise provided in such RLP Development Plan, or set forth in Section 7.1.1.1(b) below, each Party will be solely responsible for all regulatory matters relating to such Regional Licensed Product in its Territory and will own all INDs, BLAs, Regulatory Materials and related regulatory documents in its Territory with respect to such Regional Licensed Product, including any drug master files maintained by or on behalf of such Party solely with respect thereto in such Territory, which will be and remain such Party's sole responsibility; provided that, in the case of any Supplemental Study conducted by a Party with respect to a Combination Therapy involving a Regional Licensed Product in the other Party's Territory, unless prohibited by the applicable Regulatory Authority, the Party conducting the Supplemental Study shall be entitled to own the IND. At Novartis' reasonable request, Xencor will use Commercially Reasonable Efforts to promptly assign and transfer to Novartis all INDs, Regulatory Materials and other regulatory documentation in the Novartis Territory with respect to such Regional Licensed Product that is in the possession and Control of Xencor, and each Party will submit to the applicable Regulatory Authority all filings, letters and other documentation necessary to effect such assignment and transfer as soon as practicable and no later than [...\*\*\*...] after such request for such Regional Licensed Product, in each case, including any drug master files maintained by or on behalf of Xencor

solely with respect thereto. For clarity, Xencor will not be required to transfer any drug master files maintained by or on behalf of any Third Party, including any contract manufacturer, provided that Novartis has access to or rights to cross-reference those drug master files pursuant to Section 7.3 to permit Novartis to comply with its regulatory obligation in connection with the Research, Development, Manufacture, and Commercialization of Regional Licensed Products. Subject to the remainder of this Section 7.1.1, each Party will have the sole right to (i) oversee, monitor and coordinate all regulatory actions, communications and filings with, and submissions to, each Regulatory Authority in its Territory with respect to such Regional Licensed Product, (ii) interface, correspond and meet with each Regulatory Authority in its Territory with respect to such Regional Licensed Product, and (iii) seek and maintain all regulatory filings in its Territory with respect to such Regional Licensed Product. Notwithstanding the foregoing, each Party will have the right to review and comment on any material submissions, regulatory actions, communications, or filings proposed to be made with respect to a Regional Licensed Product by the other Party in such other Party's Territory at least [...\*\*\*...] prior to submission of such materials, unless the Party is required to respond sooner to the applicable Regulatory Authority.

- (b) Communications with Regulatory Authorities. Each Party will notify the JDC, including a brief description in English, of the principal issues raised in each material communication with Regulatory Authorities with respect to such Regional Licensed Product within [...\*\*\*...] after receipt thereof. Upon request, each Party will provide to the other Party (i) at the requesting Party's expense, a summary translation of such material communications in English, (ii) at the requesting Party's expense, complete copies of the original correspondence in their native language, or (iii) at the requesting Party's expense, a full translation of such material communications in English, in each case of (i) through (iii) within a reasonable period of time following such request. For the purposes of this Agreement, "material communications" with Regulatory Authorities include meetings with Regulatory Authorities and Regulatory Authority questions or concerns with respect to significant issues, including those stemming from global health authority phone conversations, regulatory meetings, regulatory filings and Regulatory Approvals, whether initiated by a Party or its Related Parties or the Regulatory Authority.
- **7.1.1.2.** Regulatory Meetings. Each Party will provide the other Party with reasonable advance notice of all substantive meetings with the Governmental Authorities in its Territory pertaining to each Regional Licensed Product, or with as much advance notice as practicable under the circumstances. Each Party will use Commercially Reasonable Efforts, to the extent reasonably practicable, to permit the other Party to have, at the other Party's expense, mutually acceptable representatives of the other Party attend, solely as a non-participating observer, material, substantive

meetings, including pre-IND, pre-BLA and end of Phase 2 Study meetings, with the Governmental Authorities within either its or the other Party's Territory pertaining to such Regional Licensed Product, provided however, that (a) if not prohibited by the Governmental Authority, attendance by the other Party will be permitted, (b) attendance by the representatives of the other Party will not prevent participation of a representative of the Party in charge of its Territory due to restrictions imposed by Regulatory Agencies on the number of attendees, and (c) neither Party will be obligated to change the schedule of such meeting in order to accommodate the schedule of the other Party's representatives.

- **7.1.1.3.** Submissions. Each Party will provide the other Party with written notice of each of the following events with regard to each Regional Licensed Product (a) within a reasonable period of time following the occurrence thereof (i) the submission of any filings or applications for Regulatory Approval of such Regional Licensed Product in such Party's Territory to any Regulatory Authority and (ii) receipt or denial of Regulatory Approval, as well as any applicable withdrawals, for such Regional Licensed Product, and (b) on a [...\*\*\*...] basis, a summary of all regulatory applications, including INDs, pediatric and orphan drug designation requests and applications therefor) that were filed for such Regional Licensed Product during such preceding Calendar Quarter and those anticipated to be filed within the upcoming Calendar Quarter, provided however, that each Party will inform the other Party of such event under (a) or (b) prior to public disclosure of such event by such Party.
- **7.1.2.** Costs of Regulatory Affairs. Except as provided in Section 5.1.4, each Party will be responsible for all costs and expenses incurred in connection with applying for, obtaining and maintaining Regulatory Approval with respect to Regional Licensed Products in its Territory, and related regulatory affairs activities.
- 7.1.3. Right of Reference. Each Party hereby grants to the other Party, and at the request of the other Party will grant to the other Party's Related Parties, a "Right of Reference," as that term is defined in 21 C.F.R. 314.3(b) (or any successor rule or analogous Law recognized outside of the United States), to, and a right to copy, access, and otherwise use, all information and data (including all CMC information as well as data made, collected or otherwise generated in the conduct of any Clinical Studies or upon exercise of the Additional Development Opt-In Right, Supplemental Studies, or early access/named patient programs for the Regional Licensed Products) included in or used in support of any regulatory filing, Regulatory Approval, drug master file or other regulatory documentation (including pediatric and orphan drug applications and designations) maintained on behalf of such Party (or its Related Parties) that relates to any Regional Licensed Product, to the extent necessary or useful to obtain Regulatory Approval of a Regional Licensed Product in the Novartis Territory or the Xencor Territory, as applicable, and such Party will provide a signed statement to this effect, if requested by the other Party, in accordance with 21 C.F.R. 314.50(g)(3) (or any successor or analogous Law outside of the United States). In addition, upon reasonable request of either Party (on behalf of itself or a Sublicensee), the other Party will obtain and provide to the requesting Party certificates or other

formal or official attestations concerning the regulatory status of the Regional Licensed Products in the Novartis Territory or the Xencor Territory, as applicable (e.g., Certificates of Free Sale, Certificates for Export, Certificates to Foreign Governments), at the requesting Party's request, and <u>provided</u> that such attestations are reasonably necessary for the requesting Party to exercise its rights under this Agreement. Notwithstanding anything to the contrary in this Agreement other than for safety concerns, neither Party will voluntarily withdraw or inactivate any regulatory filing that the other Party references or otherwise uses pursuant to this Section 7.1.3. For clarity, the benefit of any regulatory vouchers with respect to Regional Licensed Products will be allocated to the Parties on a [...\*\*\*...] basis.

## 7.2. Global Licensed Products, Optioned Licensed Products and Fc Licensed Products.

- **7.2.1.** Regulatory Filings and Interactions.
- Ownership of Regulatory Filings. Novartis will own all INDs, BLAs, Regulatory Materials and related regulatory documentation with respect to any Global Licensed Product, Optioned Licensed Product and Fc Licensed Product, including any drug master files maintained by or on behalf of Xencor solely with respect thereto. At Novartis' request, Xencor will promptly assign and transfer to Novartis all INDs, Regulatory Materials and other regulatory documentation in the Novartis Territory with respect to such Global Licensed Product, Optioned Licensed Product or Fc Licensed Product that is in the possession and Control of Xencor, including any drug master files maintained by or on behalf of Xencor solely with respect thereto, and each Party will submit all filings, letters and other documentation necessary to effect such assignment and transfer to the applicable Regulatory Authority as soon as reasonably practicable, but no later than [...\*\*\*...] after such request for such Global Licensed Product, Optioned Licensed Product or Fc Licensed Product. For clarity, Xencor will not be required to transfer any drug master files maintained by or on behalf of any Third Party, including any contract manufacturer, provided that Novartis has access to or rights to cross-reference those drug master files pursuant to Section 7.3 to permit Novartis to comply with its regulatory obligation in connection with the Research, Development, Manufacture, and Commercialization of Global Licensed Products, Optioned Licensed Products and Fc Licensed Products.
- **7.2.1.2.** Responsibilities for Regulatory Matters. Novartis will be solely responsible for all regulatory matters relating to Global Licensed Products, Optioned Licensed Products and Fc Licensed Products in the Novartis Territory, including (a) overseeing, monitoring and coordinating all regulatory actions, communications and filings with, and submissions to, each Regulatory Authority in the Novartis Territory, (b) interfacing, corresponding and meeting with each Regulatory Authority in the Novartis Territory, and (c) seeking and maintaining all Regulatory Materials in the Novartis Territory.
- **7.2.1.3.** Communications with Regulatory Authorities. Novartis will provide Xencor, through the JDC, as part of the quarterly updates regarding Development

activities described in Section 2.3.2, with a description in English, of the principal issues raised in any material communication with any Regulatory Authority in the Novartis Territory with respect to any Global Licensed Product, Optioned Licensed Product or Fc Licensed Product during the preceding Calendar Quarter.

- **7.2.1.4.** Regulatory Meeting Updates. Novartis will provide Xencor, through the JDC, with quarterly updates of substantive meetings with the Governmental Authorities in the Novartis Territory pertaining to the Development of each Global Licensed Product, Optioned Licensed Product and Fc Licensed Product.
- **7.2.1.5.** Submissions. With respect to each Global Licensed Product, Optioned Licensed Product or Fc Licensed Product, Novartis will provide Xencor with written notice of each of the following events (a) within a reasonable period of time after the occurrence of such event in the Novartis Territory, (i) the submission of any filings or applications for Regulatory Approval to any Regulatory Authority, and (ii) receipt or denial of Regulatory Approval, or any applicable withdrawals, and (b) on a quarterly basis, all regulatory applications, including INDs, pediatric and orphan drug designation requests and applications therefor that were filed during such preceding Calendar Quarter and those anticipated to be filed within the upcoming Calendar Quarter, provided however, that Novartis will inform Xencor of any such events under (a) or (b) prior to public disclosure of such event by Novartis.
- **7.2.2.** Costs of Regulatory Affairs. Novartis will be responsible for all costs and expenses incurred by either Party in connection with applying for Regulatory Approval with respect to Global Licensed Products, Optioned Licensed Products and Fc Licensed Products in the Novartis Territory, and related regulatory affairs activities, <u>provided</u> that Xencor shall not be reimbursed for any such costs or expenses that are not (i) contemplated by the Development Plan or this Agreement or (ii) without the prior written consent of Novartis.
- 7.3. **Right of Reference**. Xencor hereby grants to Novartis, and at the request of Novartis will grant to Novartis' Related Parties, a Right of Reference to, and a right to copy, access, and otherwise use, all information and data (including all CMC information as well as data made, collected or otherwise generated in the conduct of any preclinical studies, Clinical Studies or early access/named patient programs) included in or used in support of any drug master file maintained by or on behalf of Xencor (including its Related Parties) that solely relates to any Global Licensed Product or Optioned Licensed Product (or a Fc Licensed Product) to the extent necessary or useful to Research, Develop, Manufacture or Commercialize Global Licensed Products or Optioned Licensed Products in the Novartis Territory. Notwithstanding anything to the contrary in this Agreement, Xencor will not voluntarily withdraw or inactivate any regulatory filing that Novartis or its Related Parties reference or otherwise use pursuant to this Section 7.3. The benefit of any regulatory vouchers with respect to Optioned Licensed Products will be allocated to Novartis.
- 7.4. **Pharmacovigilance**. The Parties shall cooperate with regard to the reporting and handling of safety information involving the Regional Licensed Antibodies and Regional Licensed Products in accordance with the applicable regulatory Laws and regulations on pharmacovigilance and

clinical safety. The Parties agree that Novartis will take the lead for safety and hold the global safety database on behalf of both Parties for the Regional Licensed Antibodies and Regional Licensed Products for the whole time that both Parties are involved in development of Regional Licensed Antibodies and Regional Licensed Products. Xencor agrees to provide Novartis with such historical safety data for the Regional Licensed Antibodies and Regional Licensed Products as is required for Novartis to fulfill its function as the holder of the global safety database. Following the Effective Date, and in time to ensure that all regulatory requirements are met, the Parties shall enter into the SDEA which will define the pharmacovigilance responsibilities of the Parties and safety data exchange procedures to enable each Party to comply with all of its legal and regulatory obligations related to the Regional Licensed Antibodies and Regional Licensed Products.

### 8. MANUFACTURE

- 8.1. **Overview**. The Parties shall be responsible for the Manufacture and supply of each Licensed Antibody and Licensed Product for Research, Development and Commercialization, with Novartis taking the lead as indicated herein.
- 8.2. <u>Manufacturing for Research</u>. Subject to the oversight of the JSC and in accordance with the Research Plan, the Parties will allocate responsibility for Manufacture of Global Licensed Antibodies for each of the Research Programs in accordance with the Research Plan. Xencor will Manufacture such Global Licensed Antibodies in accordance with quality standards in the industry for research purposes. Novartis shall have the sole responsibility, at its own expense, to Manufacture Fc Licensed Antibodies and Fc Licensed Products for Research.
  - 8.3. Clinical Supply of Regional Licensed Antibodies and Regional Licensed Products. Subject to the terms and conditions of this Agreement, the JDC will consider and approve the manufacturing chain for clinical supplies of Regional Licensed Antibodies and Regional Licensed Products, including allocating responsibility for the Manufacture thereof (such Manufacturer(s) approved by the JDC, the "Primary RLP Supplier(s)"). The costs associated with Manufacture by the Primary RLP Supplier(s) shall be allocated between the Parties as Development Costs in accordance with Section 5.1.4. Such supply chain will meet all relevant regulatory requirements and guidelines and be intended to ensure that each Party has supply sufficient for Development of Regional Licensed Antibodies and Regional Licensed Products under this Agreement, provided that in the event of a supply shortage, available supply shall be allocated first to Clinical Studies contemplated by the RLP Development Plan, with any remaining supply allocated to any Supplemental Studies to be conducted by a Party. Notwithstanding the foregoing, each Party (the "Manufacturing Proposing Party") shall have the right to Manufacture or have Manufactured, itself or using a Third Party, clinical supplies of Regional Licensed Antibodies and Regional Licensed Products (any such additional manufacturer, a "Secondary RLP Supplier"), provided that [...\*\*\*...]

[...\*\*\*...].

**8.4.** Clinical Supply of other Licensed Antibodies and Licensed Products. Novartis shall be solely responsible for Manufacturing or having Manufactured clinical supplies of Global Licensed Antibodies and Global Licensed Products, any Optioned Licensed Antibodies and Optioned Licensed Products, and Fc Licensed Antibodies and Fc Licensed Products. The costs associated with Manufacture of such Licensed Products shall be borne by Novartis except that costs associated with Manufacturing Optioned Licensed Antibodies and Optioned Licensed Products shall be allocated between the Parties as Development Costs in accordance with Section 5.3.4.

## 8.5. <u>Commercial Supply of Regional Licensed Products</u>.

- **8.5.1.** Novartis Territory. Novartis shall be solely responsible at its cost for Manufacturing or having Manufactured Regional Licensed Products for Commercialization in the Novartis Territory.
  - **8.5.2.** Xencor Territory.
    - **8.5.2.1.** Novartis shall have the right to manufacture such Regional Licensed Products for Commercialization in the Xencor Territory in accordance with the following procedure:
    - (a) Novartis may submit a term sheet to the JSC proposing financial and other key terms under which it would propose to Manufacture and supply Regional Licensed Products for Commercialization in the Xencor Territory. [...\*\*\*...].
    - (b) If Novartis elects not to exercise its Manufacturing rights or the Novartis proposal is deemed not commercially reasonable, then Xencor shall be entitled to engage a Third Party contract manufacturer to Manufacture Regional

Licensed Products for Commercialization in the Xencor Territory, provided that [...\*\*\*...].

- **8.5.2.2.** Xencor shall be responsible for [...\*\*\*...]% of the costs associated with the Manufacture of Regional Licensed Products for Commercialization in the Xencor Territory, whether the supplier is Novartis or a Third Party.
- **8.6.** Commercial Supply of other Licensed Products. Novartis shall be solely responsible for Manufacturing or having Manufactured Commercial supplies of Global Licensed Products, any Optioned Licensed Products and Fc Licensed Products. The costs associated with Manufacture of such Licensed Products shall be borne by Novartis except that Landed Costs associated with Manufacturing Optioned Licensed Products shall be allocated between the Parties as Commercialization Costs in accordance with Section 6.3.1.
- **Third Parties**. The Parties will be entitled to utilize the services of Third Parties to perform their respective Manufacturing activities under this Section 8, subject to in the case of any Third Party selected by Xencor, Novartis' prior written approval as and to the extent set forth above and, in the case of any Third Party so selected by Novartis, quality standards that satisfy Novartis' quality standards. [...\*\*\*...]. Each Party will (a) require that any such Third Party operates in a manner consistent with the terms of this Agreement and (b) remain at all times fully liable for its respective responsibilities contracted to such Third Party and such Party hereby expressly waives any requirement that the other Party exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against the Party who engaged such Third Party. Each Party will require that any such Third Party agreement entered into pursuant to this Section 8.7 (x) includes confidentiality and non-use provisions that are no less stringent than those set forth in Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (y) obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) under and to, any Know-How and Patents that are developed by such Third Party in the performance of such agreement and are reasonably necessary or useful to Research, Develop, Manufacture or Commercialize Licensed Antibodies or Licensed Products in the Field. For clarity, the foregoing requirement to obtain ownership of, or a fully sublicensable license (or an exclusive option to obtain such license) shall not apply to any improvements to the background or foundational Know-How or Patents owned or in-licensed by a Third Party contract manufacturer or its Affiliates unless such improvements are reasonably necessary to Research, Develop, Manufacture or Commercialize those Licensed Antibodies or Licensed Products with respect to which such Third Party or its Affiliate conducted its activities under such Third Party agreement. The Party utilizing the services of a Third

Party service provider will be solely responsible for direction of and communications with such Third Party.

#### 9. LICENSES

## 9.1. Regional Target Pairs.

#### **9.1.1.** License Grants to Novartis.

**9.1.1.1.** Subject to the terms and conditions of this Agreement, on a Regional Target Pairby-Regional Target Pair basis, effective upon the Effective Date, Xencor hereby grants to Novartis and its Affiliates a non-transferable (except as provided in Section 16.12), sublicensable (in accordance with Section 9.1.1.2), royalty-bearing, exclusive license under the Xencor Technology to (a) Research, Develop and Manufacture Regional Licensed Antibodies and Regional Licensed Products in the Field anywhere in the world for Commercialization in the Field in the Novartis Territory, and (b) Commercialize Regional Licensed Antibodies and Regional Licensed Products in the Field solely in the Novartis Territory. Notwithstanding the foregoing, Xencor retains the right under the Xencor Technology, with the right to grant licenses through multiple tiers, to perform its obligations and exercise its rights under this Agreement.

# **9.1.1.2.** Sublicensing Terms.

- (a) Novartis will have the right to sublicense any of its rights under Section 9.1.1.1 to any of its Affiliates or to any Third Party (which sublicensed rights may be further sublicensable through multiple tiers) without the prior consent of Xencor, subject to the requirements of this Section 9.1.1.2.
- (b) Each sublicense granted by Novartis pursuant to Section 9.1.1.2(a) will be subject and subordinate to this Agreement and will contain provisions consistent with the terms and conditions of this Agreement. Novartis will, within [...\*\*\*...] thereafter, provide Xencor with a copy of any executed sublicense agreement covering a material sublicense granted hereunder (which copy may be redacted to remove provisions which are not related to Novartis' obligations under this Agreement), and each such sublicense agreement will contain the following provisions, (i) a requirement that the Sublicensee comply with the confidentiality and non-use provisions of Section 11.1, (ii) a requirement that the Sublicensee submit applicable sales or other reports to Novartis to the extent necessary or relevant to the reports required to be made or records required to be maintained under this Agreement, (iii) the audit requirement set forth in Section 10.10, and (iv) a requirement that the Sublicensee comply with the applicable provisions under any Xencor In-License. For clarity, the obligation to provide a copy of each sublicense agreement includes the agreements granted through multiple tiers.

(c) Notwithstanding any sublicense, Novartis will remain primarily liable to Xencor for the performance of all of Novartis' obligations under, and Novartis' compliance with all provisions of, this Agreement. Novartis hereby waives any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis.

#### **9.1.2.** License Grants to Xencor.

**9.1.2.1.** Subject to the terms and conditions of this Agreement, on a Regional Target Pairby-Regional Target Pair basis, effective upon the Effective Date, Novartis hereby grants to Xencor and its Affiliates a non-transferable (except as provided in Section 16.12), sublicensable (in accordance with Section 9.1.2.2), exclusive license under the Novartis Technology to (a) Research, Develop and, subject to Section 8, Manufacture Regional Licensed Antibodies and Regional Licensed Products in the Field anywhere in the world for Commercialization in the Field in the Xencor Territory, and (b) Commercialize Regional Licensed Antibodies and Regional Licensed Products in the Field solely in the Xencor Territory. Notwithstanding the foregoing, Novartis retains the right under the Novartis Technology, with the right to grant licenses through multiple tiers, to perform its obligations and exercise its rights under this Agreement.

## **9.1.2.2.** Sublicensing Terms.

- (a) Xencor will have the right to sublicense any of its rights under Section 9.1.2.1 to any of its Affiliates or to any Third Party (which sublicensed rights may be further sublicensable through multiple tiers) without the prior consent of Novartis, subject to the requirements of this Section 9.1.2.2.
- (b) Each sublicense granted by Xencor pursuant to Section 9.1.2.2(a) will be subject and subordinate to this Agreement and will contain provisions consistent with the terms and conditions of this Agreement. Xencor will, within [...\*\*\*...] thereafter, provide Novartis with a copy of any executed sublicense agreement covering a material sublicense granted hereunder (which copy may be redacted to remove provisions which are not related to Xencor's obligations under this Agreement), and each such sublicense agreement will contain the following provisions: (i) a requirement that the Sublicensee comply with the confidentiality and non-use provisions of Section 11.1 and (ii) a requirement that the Sublicensee comply with the applicable provisions under any Novartis In-License. For clarity, the obligation to provide a copy of each sublicense agreement includes the agreements granted through multiple tiers.
- (c) Notwithstanding any sublicense, Xencor will remain primarily liable to Novartis for the performance of all of Xencor's obligations under, and Xencor's compliance with all provisions of, this Agreement. Xencor hereby waives any requirement that Novartis exhaust any right, power or remedy, or

proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Xencor.

(d) Notwithstanding the other provisions of this Section 9.1.2, if Xencor is considering entering a collaborative transaction with a Third Party with respect to the Research, Development, Manufacture or Commercialization of any Regional Licensed Antibodies or Regional Licensed Product, which agreement includes the grant of rights to Commercialize any Regional Licensed Product in the Xencor Territory (any such agreement, a "**Proposed Xencor Sublicense**"), Xencor will so notify Novartis in writing. Novartis will have a non-exclusive right of negotiation for a period of [...\*\*\*...]. For clarity, this Section 9.1.2.2(d) will not apply to (a) any permitted assignment of this Agreement under Section 16.2, or (b) any *bona fide* agreement with a Third Party contract sales organization, contract research organization or contract manufacturer, under which such Third Party performs contract services on behalf of Xencor or any of its Affiliates for the Research, Development, or Manufacture of any Regional Licensed Antibody or Regional Licensed Product as permitted under this Agreement on a fee-for-services basis.

# 9.2. <u>Global Target Pairs and Optioned Target Pairs</u>.

#### **9.2.1.** License Grants to Novartis.

**9.2.1.1.** Subject to the terms and conditions of this Agreement, on a Global Target Pair-by-Global Target Pair and Optioned Target Pair-by-Optioned Target Pair basis, effective upon the Effective Date, Xencor hereby grants to Novartis and its Affiliates a non-transferable (except as provided in Section 16.2), sublicensable (in accordance with Section 9.2.1.2), royalty-bearing, exclusive license under the Xencor Technology to Research, Develop, Manufacture and Commercialize Global Licensed Antibodies, Global Licensed Products, Optioned Licensed Antibodies and Optioned Licensed Products in the Field anywhere in the world. Notwithstanding the foregoing, Xencor retains the right under the Xencor Technology, with the right to grant licenses through multiple tiers, to perform its obligations and exercise its rights under this Agreement.

# **9.2.1.2.** Sublicensing Terms.

(a) Novartis will have the right to sublicense any of its rights under Section 9.2.1.1 to any of its Affiliates or to any Third Party (which sublicensed

rights may be further sublicensable through multiple tiers) without the prior consent of Xencor, subject to the requirements of this Section 9.2.1.2.

- (b) Each sublicense granted by Novartis pursuant to Section 9.2.1.2(a) will be subject and subordinate to this Agreement and will contain provisions consistent with the terms and conditions of this Agreement. Novartis will within [...\*\*\*...] thereafter, provide Xencor with a copy of any executed sublicense agreement covering a material sublicense granted hereunder (which copy may be redacted to remove provisions which are not related to Novartis' obligations under this Agreement), and each such sublicense agreement will contain the following provisions, (i) a requirement that the Sublicensee comply with the confidentiality and non-use provisions of Section 11.1, (ii) a requirement that the Sublicensee submit applicable sales or other reports to Novartis to the extent necessary or relevant to the reports required to be made or records required to be maintained under this Agreement, (iii) the audit requirement set forth in Section 10.10, and (iv) a requirement that the Sublicensee comply with the applicable provisions under any Xencor In-License. For clarity, the obligation to provide a copy of each sublicense agreement includes the agreements granted through multiple tiers.
- (c) Notwithstanding any sublicense, Novartis will remain primarily liable to Xencor for the performance of all of Novartis' obligations under, and Novartis' compliance with all provisions of, this Agreement. Novartis hereby waives any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis.

#### **9.2.2.** License Grants to Xencor.

**9.2.2.1.** Subject to the terms and conditions of this Agreement, on a Global Target Pair-by-Global Target Pair basis, and with respect to the Optioned Target Pair, effective upon the date when Xencor exercises its Opt-in Right, Novartis hereby grants to Xencor and its Affiliates solely during the Research Term a non-transferable (except as provided in Section 16.2), sublicensable (in accordance with Section 9.2.2.2), non-exclusive license under the Novartis Technology to Research Global Licensed Antibodies, Global Licensed Products, Optioned Licensed Antibodies and Optioned Licensed Products in the Field in accordance with the Research Plan. Notwithstanding the foregoing, Novartis retains the right under the Novartis Technology, with the right to grant licenses through multiple tiers, to perform its obligations and exercise its rights under this Agreement.

#### **9.2.2.2.** Sublicensing Terms.

- (a) Xencor will have the right to sublicense any of its rights under Section 9.2.2.1 to any of its Affiliates or to any Third Party without the prior consent of Novartis, subject to the requirements of this Section 9.2.2.2.
- (b) Each sublicense granted by Xencor pursuant to this Section 9.2.2.2(a) will be subject and subordinate to this Agreement and will contain provisions consistent with the terms and conditions of this Agreement. Xencor will, within [...\*\*\*...] thereafter, provide Novartis with a copy of any executed sublicense agreement covering a material sublicense granted hereunder (which copy may be redacted to remove provisions which are not related to Xencor's obligations under this Agreement), and each such sublicense agreement will contain the following provisions, (i) a requirement that the Sublicensee comply with the confidentiality and non-use provisions of Section 11.1 and (ii) a requirement that the Sublicensee comply with the applicable provisions under any Novartis In-License. For clarity, the obligation to provide a copy of each sublicense agreement includes the agreements granted through multiple tiers.
- (c) Notwithstanding any sublicense, Xencor will remain primarily liable to Novartis for the performance of all of Xencor's obligations under, and Xencor's compliance with all provisions of, this Agreement. Xencor hereby waives any requirement that Novartis exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Xencor.

## 9.3. Fc Targets.

### **9.3.1.** License Grants to Novartis.

**9.3.1.1.** Subject to the terms and conditions of this Agreement, on a Fc Target by Fc Target basis, effective upon the date that a given Fc Target passes the Gatekeeper function pursuant to Section 3.2.1.3(a), Xencor hereby grants to Novartis and its Affiliates a non-transferable (except as provided in Section 16.2), sublicensable (in accordance with Section 9.3.1.2), royalty-bearing, non-exclusive license under the Fc Technology to Research, Develop, Commercialize and Manufacture Fc Licensed Antibodies and Fc Licensed Products that Specifically Bind such Fc Target in the Field anywhere in the world.

### **9.3.1.2.** Sublicensing Terms.

(a) Novartis will have the right to sublicense any of its rights under Section 9.3.1.1 to any of its Affiliates or to any Third Party (which sublicensed rights may be further sublicensable through multiple tiers) without the prior consent of Xencor, subject to the requirements of this Section 9.3.1.2.

- (b) Each sublicense granted by Novartis pursuant to Section 9.3.1.2(a) will be subject and subordinate to this Agreement and will contain provisions consistent with the terms and conditions of this Agreement. Novartis will within [...\*\*\*...] thereafter, provide Xencor with a copy of any executed sublicense agreement covering a material sublicense granted hereunder (which copy may be redacted to remove provisions which are not related to Novartis' obligations under this Agreement), and each such sublicense agreement will contain the following provisions, (i) a requirement that the Sublicensee comply with the confidentiality and non-use provisions of Section 11.1, (ii) a requirement that the Sublicensee submit applicable sales or other reports to Novartis to the extent necessary or relevant to the reports required to be made or records required to be maintained under this Agreement, (iii) the audit requirement set forth in Section 10.10, and (iv) a requirement that the Sublicensee comply with the applicable provisions under any Xencor In-License. For clarity, the obligation to provide a copy of each sublicense agreement includes the agreements granted through multiple tiers.
- (c) Notwithstanding any sublicense, Novartis will remain primarily liable to Xencor for the performance of all of Novartis' obligations under, and Novartis' compliance with all provisions of, this Agreement. Novartis hereby waive any requirement that Xencor exhaust any right, power or remedy, or proceed against such Third Party, for any obligation or performance hereunder prior to proceeding directly against Novartis.
- 9.4. <u>Joint Collaboration IP</u>. Subject to the rights and licenses granted to, and the obligations (including royalty obligations) of, each Party under this Agreement, including any exclusivity obligations, either Party is entitled to practice Joint Collaboration IP for all purposes on a worldwide basis without consent of and without a duty of accounting to the other Party. Each Party will grant and hereby does grant all permissions, consents and waivers with respect to, and all licenses under, the Joint Collaboration IP, throughout the world, necessary to provide the other Party with such rights of use and exploitation of the Joint Collaboration IP, and will execute documents as necessary to accomplish the foregoing.

## 9.5. <u>In-Licenses</u>.

- **9.5.1.** In-Licenses. The Parties agree that all upfront, milestone, royalty and other payments to any Third Party in respect of any Collaboration In-License, Xencor Existing In-Licenses or Novartis Existing In-License will be deemed a "**Third Party Payment**" and subject to this Section 0. Responsibility for Collaboration In-Licenses, Xencor Existing In-Licenses, Novartis Existing In-License and Third Party Payments will be as follows:
  - **9.5.1.1.** Xencor will be responsible for Third Party Payments under the Xencor Existing In-Licenses or Novartis Existing In-Licenses in connection with Xencor's sale of Regional Licensed Products in the Xencor Territory.

- **9.5.1.2.** Novartis will be responsible for Third Party Payments under the Novartis Existing In-Licenses or Xencor Existing In-Licenses in connection with Novartis' sale of Licensed Products in the Novartis Territory; <u>provided</u> that, any such payments with respect to an Optioned Licensed Product in the U.S. will be deemed to be Commercialization Costs.
- **9.5.1.3.** Any Development-related milestone, including regulatory milestone, Third Party Payments under the Xencor Existing In-Licenses or the Novartis Existing In-Licenses will be deemed to be Development Costs and shared by the Parties accordingly or borne solely by Novartis, as applicable, depending on which Party(ies) are responsible for the relevant Development Costs hereunder.

#### **9.5.1.4.** Potential In-Licenses and Collaboration In-Licenses.

- (a) The Parties acknowledge that during the Term, the JSC may determine that Research, Development, Manufacture or Commercialization of any Licensed Antibodies or Licensed Products may require or benefit from a license acquired or entered into after the Effective Date with respect to additional Patents or Know-How of Third Parties (a "Potential **In-License**"). If a Party desires to acquire or otherwise enter into any Potential In-License after the Effective Date with respect to the Research, Development, Manufacture, or Commercialization of any Regional Licensed Antibodies, Regional Licensed Products, Optioned Licensed Antibodies, and Optioned Licensed Products, such Party will bring such Potential In-License to the attention of the JSC and the Parties will, through the JSC, discuss in good faith whether such Potential In-License should be obtained and made available for use by the Parties pursuant to this Agreement. For any Potential In-License that the JSC approves pursuing and for use by the Parties pursuant to this Agreement, (i) the JSC will remain involved throughout the negotiation of such Potential In-License and will approve its final terms prior to execution, (ii) upon execution, if ever, such Potential In-License will be deemed to be a "Collaboration In-License" hereunder, (iii) the Patents and Know-How inlicensed under such Collaboration In-License will be deemed "Controlled" under this Agreement as Xencor Patents or Xencor Know-How (as applicable) or Novartis Patents or Novartis Know-How (as applicable) for purposes of Research, Development, Manufacture, or Commercialization of any Licensed Antibodies or Licensed Products, (iv) any allocated payments for Regional Licensed Antibodies or Regional Licensed Products that are nonterritory specific will be allocated as Development Costs under Section 5.1.4, (v) any allocated payments for Optioned Licensed Antibodies or Optioned Licensed Products that are non-territory specific will be allocated as Development Costs under Section 5.3.4, and (vi) any allocated payments for a Party's Territory will be borne by the applicable Party.
- (b) If the JSC does not approve a Potential In-License, then the applicable Party may proceed to enter into such Potential In-License, <u>provided</u>

- that (i) such Potential In-License will not be deemed to be a Collaboration In-License hereunder, (ii) the Patents and Know-How in-licensed under such Potential In-License will not be deemed Xencor Patents or Xencor Know-How (as applicable) or Novartis Patents or Novartis Know-How (as applicable) and will not be deemed "Controlled" for purposes of this Agreement, (iii) each Party will have the right to enter into such Potential In-License solely with respect to its own Territory and, subject to Section 10.7.3, shall bear [...\*\*\*...]% of any Third Party Payment thereunder, and (iv) the other Party will not be entitled to use any Patents or Know-How in-licensed under such Potential In-License in connection with the performance of this Agreement. For clarity, Novartis has the right to enter into any Potential In-License covering a Global Licensed Antibody or Global Licensed Product without bringing such Potential In-License to the JSC for approval.
- (c) Neither Party will negotiate for or agree to economic terms in any such Potential In-License in a manner that [...\*\*\*...].
- **9.5.2.** Compliance with In-Licenses. All licenses and other rights granted to Novartis under this Section 9 are subject to the rights and obligations of Xencor under the Xencor In-Licenses. All licenses and other rights granted to Xencor under this Section 9 are subject to the rights and obligations of Novartis under the Novartis In-Licenses. Each Party will comply with all applicable provisions of the In-Licenses, and will perform and take such actions as may be reasonably required to allow the Party that is party to such In-License to comply with its obligations thereunder, including obligations relating to sublicensing, patent matters, confidentiality, reporting, audit rights, indemnification and diligence. Without limiting the foregoing, each Party will prepare and deliver to the other Party any additional reports reasonably required under the applicable In-Licenses and reasonably requested by such other Party, in each case sufficiently in advance to enable the Party that is party to such In-License to comply with its obligations under the applicable In-Licenses. Each Party agrees, upon the other Party's reasonable request, to provide the other Party with copies of any In-Licenses to which it is a party. Confidential Information of the providing Party or its counterparty may be redacted from such copies to the extent necessary to comply with such In-Licenses.
- 9.6. **Bankruptcy**. All rights and licenses granted under or pursuant to this Agreement by a Party to the other are and will otherwise be deemed to be, for purposes of Section 365(n) of the Bankruptcy Code, licenses of right to "intellectual property" as defined under Section 101 of the Bankruptcy Code. The Parties agree that the Parties and their respective Sublicensees, as Sublicensees of such rights under this Agreement, will retain and may fully exercise all of their rights and elections under the Bankruptcy Code and any foreign counterpart thereto. The Parties further agree that upon

commencement of a bankruptcy proceeding by or against a Party (the "Bankrupt Party") under the Bankruptcy Code, the other Party (the "Non-Bankrupt Party") will be entitled to a complete duplicate of, or complete access to (as the Non-Bankrupt Party deems appropriate), all such intellectual property and all embodiments of such intellectual property. Such intellectual property and all embodiments of such intellectual property will be promptly delivered to the Non-Bankrupt Party (a) upon any such commencement of a bankruptcy proceeding and upon written request by the Non-Bankrupt Party, unless the Bankrupt Party elects to continue to perform all of its obligations under this Agreement, or (b) if not delivered under (a) above, upon the rejection of this Agreement by or on behalf of the Bankrupt Party and upon written request by the Non-Bankrupt Party. The Bankrupt Party (in any capacity, including debtor-in-possession) and its successors and assigns (including any trustee) agree not to interfere with the exercise by the Non-Bankrupt Party or its Related Parties of its rights and licenses to such intellectual property and such embodiments of intellectual property in accordance with this Agreement, and agrees to assist the Non-Bankrupt Party and its Related Parties in obtaining such intellectual property and such embodiments of intellectual property in the possession or control of Third Parties as are reasonably necessary or desirable for the Non-Bankrupt Party to exercise such rights and licenses in accordance with this Agreement. The foregoing provisions are without prejudice to any rights the Non-Bankrupt Party may have arising under the Bankruptcy Code or other Laws.

9.7. **No Other Rights**. Except as otherwise expressly provided in this Agreement, under no circumstances will a Party or any of its Affiliates, as a result of this Agreement, obtain any ownership interest, license or other right in or to any Know-How, Patents or other intellectual property rights of the other Party, including tangible or intangible items owned, controlled or developed by the other Party, or provided by the other Party to the receiving Party at any time, pursuant to this Agreement.

## 10. PAYMENTS

- 10.1. <u>Initial License Fee</u>. Novartis will pay to Xencor within [...\*\*\*...] after receipt of an invoice from Xencor, which invoice shall be substantially in the form of Exhibit D and issued promptly following the Effective Date, a one-time payment of \$150,000,000. Such payment will be non-refundable, non-creditable and not subject to set-off.
- 10.2. **Profit Share Payments.** Following the First Commercial Sale of an Optioned Licensed Product in the United States and throughout the remainder of the Term for such Optioned Licensed Product, Xencor shall be entitled to receive [...\*\*\*...]% of Gross Profits on sales of such Optioned Licensed Product in the United States in accordance with the procedure set forth in Section 10.3 below.

## 10.3. Reconciliation and Payment of Profit Share Payments and Costs.

10.3.1. For each Regional Target Pair and the Optioned Target Pair, commencing upon the first Calendar Quarter immediately following the approval of the first applicable Development Plan and continuing thereafter so long as a Party incurs Development Costs under this Agreement for which reconciliation will be provided, Xencor and Novartis will, within [...\*\*\*...] of such Calendar Quarter submit to a finance officer designated by Xencor and a finance officer designated by Novartis (the "Finance Officers") a report setting forth the Development Costs it incurred in such Calendar Quarter with respect to each Regional Target Pair, and the

Optioned Target Pair, if applicable, in accordance with the applicable Development Plan as approved by the JSC. Following the First Commercial Sale of an Optioned Licensed Product in the United States and throughout the remainder of the Term for such Optioned Licensed Product, such report will also include (i) sales of such Optioned Licensed Product in the United States made by or behalf of Novartis, including a Sales & Profit Share Report for the sales of such Optioned Licensed Products in the United States and (ii) a Commercialization Cost Calculation Report. Each such report will specify in reasonable detail all applicable sales and costs, and, if requested by Xencor or Novartis, any such invoices or other supporting documentation for any payments to a Third Party or with respect to which documentation is otherwise reasonably requested will be promptly provided. Within [...\*\*\*...] after receipt of such reports, the Finance Officers will confer and agree in writing on whether a reconciliation payment is due from Xencor to Novartis or Novartis to Xencor, and if so, the amount of such reconciliation payment, so that Xencor and Novartis share (ii) Development Costs in accordance with Sections 5.1.4 and 5.3.4, (ii) Gross Profits on sales of the Optioned Licensed Product in the United States in accordance with 10.2 and (iii) Commercialization Costs and Medical Affairs Activities Costs in accordance with Sections 6.3.1 and 6.3.4, as applicable. Xencor or Novartis, as applicable, if required to pay such reconciliation payment, will submit such payment to Novartis or Xencor, respectively, as applicable, within [...\*\*\*...] of receipt of the other Party's invoice for such amount, provided however, that in the event of any disagreement with respect to the calculation of such reconciliation payment, any undisputed portion of such reconciliation payment will be paid in accordance with the foregoing timetable and the remaining, disputed portion will be paid within [...\*\*\*...] after the date on which Xencor and Novartis, using good faith efforts, resolve the dispute. In addition, each Party will consider in good faith other reasonable procedures proposed by the other Party for sharing financial information in order to permit each Party to close its books periodically in a timely manner. All payments under this Section 10.3 will be non-refundable, non-creditable and not subject to set-off.

**10.3.2.** Any expenses incurred by a Party for Development activities related to a Regional Licensed Antibody, Regional Licensed Product, Optioned Licensed Antibody or Optioned Licensed Product, as applicable, that do not fall within the definitions of Development Costs (as the case may be) will be borne solely by such Party unless the JDC determines otherwise.

Target basis, Novartis will make one-time milestone payments to Xencor (each, a "Developmental Milestone Payment") upon the first achievement of the development and regulatory milestone events set forth in this Section 10.4 (each, a "Developmental Milestone Event") with respect to a Licensed Target as set forth in the applicable table below for Regional Licensed Products, Global Licensed Products, Optioned Licensed Products or Fc Licensed Products, as applicable. Such payments will be non-refundable, non-creditable and not subject to set-off. For clarity, and without limitation, references to Licensed Product include a Combination Product. Notwithstanding any other provision of this Agreement, each series of Developmental Milestone Payments will be payable only once with respect to the specified Licensed Target Pair, notwithstanding the number of Licensed

Products (or the number of times a Licensed Product) may achieve the applicable Developmental Milestone Event.

**10.4.1.** Regional Licensed Products. On a Regional Target Pair-by-Regional Target Pair basis, Novartis will make the following Developmental Milestone Payments to Xencor upon the first achievement of the corresponding Developmental Milestone Event for each Regional Target Pair:

Developmental Milestone Event	Developmental Milestone Payment
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
Total potential Development and Regulatory Milestone Payments per Regional Target Pair	\$200,000,000

**10.4.2.** Global Licensed Products. On a Global Target Pair-by-Global Target Pair basis, Novartis will make the following Developmental Milestone Payments to Xencor upon the first achievement of the corresponding Developmental Milestone Event for each Global Target Pair:

Developmental Milestone Event	Developmental Milestone Payment
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]

[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
Total potential Development and Regulatory Milestone Payments per Global Target Pair	\$150,000,000

**10.4.3.** Optioned Licensed Products. Novartis will make the following Developmental Milestone Payments to Xencor upon the first achievement of the corresponding Developmental Milestone Event for the Optioned Target Pair:

Developmental Milestone Event	Developmental Milestone
	Payment
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
Total potential Development and Regulatory Milestone Payments for Optioned Target Pair	\$[***]

**10.4.4.** Fc Licensed Products. On a Fc Target-by-Fc Target basis, with respect to each Fc Target selected by Novartis pursuant to Section 3.2, Novartis will pay to Xencor \$[...\*\*\*...] upon completion of the first GLP Toxicology Study with respect to the first Fc Licensed Product Specifically Binding such Fc Target (each, an "Fc GLP Tox Milestone"). For clarity, the total potential Fc GLP Tox Milestones

payable with respect to Fc Targets hereunder shall be \$[...\*\*\*...]. In addition, subject to Section 10.4.5.5, on a Fc Target-by-Fc Target basis, Novartis will make the following Developmental Milestone Payments to Xencor upon the first achievement of the corresponding Developmental Milestone Event for each Fc Target:

Developmental Milestone Event	Developmental Milestone Payment
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
[***]	\$[***]
Total potential Development and Regulatory Milestone Payments per Fc Licensed Product that Specifically Binds such Fc Target	\$46,000,000

**10.4.5.** Additional Developmental Milestone Terms. Notwithstanding the foregoing, for the purpose of construing the Developmental Milestone Payments specified in the above tables:

**10.4.5.1.** For clarity, each Developmental Milestone Payment shall be payable only on the first occurrence of the applicable Developmental Milestone Event for the Licensed Target Pair or Fc Target, as applicable, and none of the Developmental Milestone Payments shall be payable more than once with respect to a Licensed Target Pair or Fc Target, as applicable.

**10.4.5.2.** If Development of a Licensed Product is terminated after it achieves a Developmental Milestone Event, then the corresponding Developmental Milestone Payment will not be due on any subsequent achievement of the same Developmental Milestone Event by a subsequent Licensed Product for the applicable Licensed Target Pair.

**10.4.5.3.** [...\*\*\*...]

[...\*\*\*...].

- **10.4.5.4.** Milestone Payments for Regulatory Approval in the EU are payable upon achievement of [...\*\*\*...].
- **10.4.5.5.** If a Licensed Product is both an Fc Licensed Product and also a Regional Licensed Product, Global Licensed Product or Optioned Licensed Product, then the applicable Developmental Milestone Payments shall be those set forth in Sections 10.4.1, 10.4.2 or 10.4.3, as applicable, and no Developmental Milestone Payment shall be due pursuant to Section 10.4.4.
- **10.4.6.** Payment Terms for Developmental Milestone Payments. Novartis shall provide Xencor with written notice of achievement of each Developmental Milestone Event within [...\*\*\*...] after such Developmental Milestone Event is achieved. After receipt of such notice, Xencor shall submit an invoice to Novartis substantially in the form of Exhibit D for the corresponding Developmental Milestone Payment. Novartis shall make the corresponding Developmental Milestone Payment within [...\*\*\*...] after receipt of such invoice. Novartis' failure to provide such a notice will not relieve Novartis of its obligation to make the applicable Developmental Milestone Payment.
- 10.5. Sales Milestone Payments. Subject to Section 10.5.5, on a Licensed Target Pair-by-Licensed Target Pair or Fc Target-by-Fc Target basis, as applicable, Novartis will make one-time payments of each of the sales milestone payments indicated below (each, a "Sales Milestone Payment" and together with the Developmental Milestone Payments, the "Milestone Payments") to Xencor when aggregate Annual Net Sales of all Licensed Products Specifically Binding such Licensed Target Pair or Specifically Binding an Fc Target, as applicable, in the Territory in a given Calendar Year first reach the dollar values indicated on each table below for the applicable Licensed Products (each, a "Sales Milestone Event"). Such payments will be non-refundable, non-creditable and not subject to set-off. Notwithstanding any other provision of this Agreement, each series of Sales Milestone Payments will be payable only once with respect to the specified Licensed Target Pair, notwithstanding the number of Licensed Products (or the number of times a Licensed Product) may achieve the applicable Sales Milestone Event.
  - **10.5.1.** Regional Licensed Products. On a Regional Target Pair-by-Regional Target Pair basis, Novartis will make the following Sales Milestone Payments to Xencor upon achievement of the corresponding Sales Milestone Event for all Regional Licensed Products for such Regional Target Pair in the Novartis Territory:

Annual Net Sales in the Novartis Territory in a Given Calendar Year for all Regional Licensed Products for a Regional Target Pair	Sales Milestone Payment
\$[***]	\$[***]
\$[***]	\$[***]
\$[***]	\$[***]
Total potential Sales Milestone Payments per Regional Target Pair	\$125,000,000

**10.5.2.** Global Licensed Products. On a Global Target Pair-by-Global Target Pair basis, Novartis will make the following Sales Milestone Payments to Xencor upon achievement of the corresponding Sales Milestone Event for all Global Licensed Products for such Global Target Pair:

Annual Net Sales in a Given Calendar Year for all Global Licensed Products for a Global Target Pair	Sales Milestone Payment
\$[***]	\$[***]
\$[***]	\$[***]
\$[***]	\$[***]
Total potential Sales Milestone Payments per Global Target Pair	\$100,000,000

**10.5.3.** Optioned Licensed Products. Novartis will make the following Sales Milestone Payments to Xencor upon achievement of the corresponding Sales Milestone Event for the Optioned Licensed Products for the Optioned Target Pair:

Annual Net Sales in a Given Calendar Year for all Optioned Licensed Products for the Optioned Target Pair	Sales Milestone Payment
\$[***]	\$[***]

\$[***]	\$[***]
\$[***]	\$[***]
Total potential Sales Milestone Payments for the Optioned Target Pair	\$[***]

**10.5.4.** Fc Licensed Products. Subject to Section 10.5.5.3, on a Fc Target-by-Fc Target basis, Novartis will make the following Sales Milestone Payments to Xencor upon achievement of the corresponding Sales Milestone Event for all Fc Licensed Products that Specifically Bind such Fc Target:

Annual Net Sales in the Novartis Territory in a Given Calendar Year of all Fc Licensed Products that Specifically Bind an Fc Target	Sales Milestone Payment
\$[***]	\$[***]
\$[***]	\$[***]
Total potential Sales Milestone Payments per Fc Licensed Target	\$30,000,000

**10.5.5.** Additional Sales Milestone Payment Terms.

**10.5.5.1.** Each Sales Milestone shall be payable only once per Licensed Target Pair or Fc Target, as applicable, the first time Annual Net Sales in the applicable Novartis Territory for all Licensed Products in a Calendar Year for such Licensed Target Pair or Fc Target, as applicable, exceeds the relevant threshold set forth above.

**10.5.5.2.** The Sales Milestone Payments in this Section 10.5 are [...\*\*\*...].

**10.5.5.3.** If a Licensed Product is both an Fc Licensed Product and also a Regional Licensed Product, Global Licensed Product or Optioned Licensed Product, then the applicable Sales Milestone Payments shall be those set forth in Sections 10.5.1, 10.5.2 and 10.5.3, as applicable, and no Sales Milestone Payment shall be due pursuant to Section 10.5.4.

- **10.5.6.** Each Sales Milestone Payment shall be deemed earned upon achievement of the corresponding Sales Milestone, and shall be notified by Novartis to Xencor within [...\*\*\*...] after [...\*\*\*...]. After receipt of such notice, Xencor shall submit an invoice to Novartis substantially in the form of Exhibit D for the corresponding Sales Milestone Payment. Novartis shall make the corresponding Sales Milestone Payment within [...\*\*\*...] after receipt of such invoice. Novartis' failure to provide such a notice will not relieve Novartis of its obligation to make the applicable Developmental Milestone Payment.
- 10.6. <u>Royalties</u>. During the applicable Royalty Term and subject to Section 10.7, Novartis will make royalty payments to Xencor, on a Licensed Product-by-Licensed Product basis, based on Annual Net Sales of the applicable Licensed Product within the Field in the Novartis Territory by Novartis and its Related Parties at the applicable rates set forth below. Such payments will be non-refundable, non-creditable and not subject to set-off.

## 10.6.1. Regional Licensed Products.

**10.6.1.1.** Subject to Section 10.6.1.2, Novartis will pay to Xencor royalties on a Regional Licensed Product-by-Regional Licensed Product basis on Annual Net Sales for each Regional Licensed Product in the Novartis Territory at the royalty rates set forth below.

Annual Net Sales	Royalty Rate
Portion up to and including \$[***]	[***]%
Portion greater than \$[***] up to and including \$[***]	[***]%
Portion greater than \$[***]	[***]%

**10.6.1.2.** In the event that Xencor exercises its Opt-Out Right with respect to a given Regional Target Pair following initiation of a Phase 3 Study for a Licensed Product that Specifically Binds such Regional Target Pair, the royalty rates and tiers set forth in Schedule 10.6.1.2 will apply with respect to such Licensed Target Pair.

The applicable royalty will be calculated by reference to Annual Net Sales of each Regional Licensed Product in the Novartis Territory. See Exhibit E for an example of such calculation.

**10.6.2.** Global Licensed Products. Novartis will pay to Xencor royalties on a Global Licensed Product-by-Global Licensed Product basis on Annual Net Sales for each Global Licensed Product in the Novartis Territory at the royalty rates set forth below.

Annual Net Sales	Royalty Rate
Portion up to and including \$[***]	[***]%
Portion greater than \$[***] up to and including \$[***]	[***]%
Portion greater than \$[***]	[***]%

The applicable royalty will be calculated by reference to the Annual Net Sales of each Global Licensed Product in the Novartis Territory. See Exhibit E for an example of such calculation.

**10.6.3.** Optioned Licensed Products. Subject to Section 10.6.1.2, Novartis will pay to Xencor royalties on an Optioned Licensed Product-by-Optioned Licensed Product basis on Annual Net Sales in the Novartis Territory other than the U.S. for each Optioned Licensed Product at the royalty rates set forth below.

Annual Net Sales	Royalty Rate
Portion up to and including \$[***]	[***]%
Portion greater than \$[***] up to and including \$[***]	[***]%
Portion greater than \$[***]	[***]%

The applicable royalty will be calculated by reference to the Annual Net Sales of each Optioned Licensed Product in the Novartis Territory outside the U.S. See Exhibit E for an example of such calculation.

**10.6.4.** Fc Licensed Products. Novartis will pay to Xencor royalties on an Fc Licensed Product-by-Fc Licensed Product basis on Annual Net Sales for each Fc Licensed Product in the Novartis Territory equal to [...\*\*\*...]% of Annual Net Sales of each Fc Licensed Product in the Novartis Territory.

# 10.7. Additional Royalty Terms.

**10.7.1.** Royalty Term. Subject to this Section 10.7, on a Licensed Product-by-Licensed Product and country-by-country basis, the royalties due under Section 10.6 will be payable as

follows (a) for Regional Licensed Products, Global Licensed Products and Optioned Licensed Products, on Annual Net Sales from the First Commercial Sale of such Licensed Product in a country until the later of (i) [...\*\*\*...], (ii) [...\*\*\*...], or (iii) [...\*\*\*...], and (b) for Fc Licensed Products, on Annual Net Sales from the First Commercial Sale of such Fc Licensed Product in a country until the later of (i) [...\*\*\*...], or (ii) [...\*\*\*...] (in each case, the "Royalty Term").

- **10.7.2.** Know How Royalty for Licensed Products. For any period during the Royalty Term in which the sale of a Licensed Product in a given country is not Covered by a Valid Claim of the Royalty Patents in such country, then the royalty rate applicable to Net Sales of such Licensed Product in such country during such period shall be equal to [...\*\*\*...]% of the weighted average royalty rate described above on worldwide Net Sales. See Exhibit E for an example of such calculation.
- **10.7.3.** Reduction for Third Party Obligations. In the event that intellectual property owned or controlled by a Third Party would be infringed or misappropriated [...\*\*\*...], Novartis shall have the right to negotiate and acquire rights to such intellectual property through a license or otherwise (including pursuant to any settlement agreement) and to deduct from [...\*\*\*...] on such Licensed Product due to Xencor with respect to a given Calendar Quarter [...\*\*\*...]% of the [...\*\*\*...] paid by Novartis to such Third Party with respect to such Licensed Product, subject to the limitation set forth in Section 10.7.6. [...\*\*\*...].
- **10.7.4.** Only One Royalty. Only one royalty will be due (a) with respect to the sale of the same unit of Licensed Product and (b) hereunder on the sale of a Licensed Product even if the manufacture, use, sale, offer for sale or importation of such Licensed Product infringes more than one claim of the Royalty Patents or Fc Patents, as applicable.
- **10.7.5.** Reduction for Loss of Market Exclusivity. Notwithstanding the foregoing, on a country-by-country basis, in the event of Loss of Market Exclusivity with respect to a Licensed Product in a country, the applicable royalty rates for Annual Net Sales of such Licensed Product set forth in Section 10.6 will be reduced by [...\*\*\*...]%.
- **10.7.6.** Royalty Minimum. Notwithstanding the foregoing in this Section 10.7, in no event will the applicable [...\*\*\*...] otherwise due to Xencor in a Calendar Quarter be reduced by more than [...\*\*\*...]% due to the deductions contemplated by Sections 10.7.2, 10.7.3 and 10.7.5, except

that (a) in the case where there is at any time during the Royalty Term [...\*\*\*...], no Valid Claim of the Royalty Patents that Covers such Regional Licensed Product in the Novartis Territory, then the applicable [...\*\*\*...] otherwise due to Xencor in a Calendar Quarter for such Regional Licensed Product may be reduced by up to [...\*\*\*...]% due to the deductions contemplated by Sections 10.7.2, 10.7.3 and 10.7.5 of the amount that would otherwise be due hereunder, and (b) in the case where Novartis takes a [...\*\*\*...]% reduction under Section 14.4.2.1 or Section 14.4.2.2, there shall be no limit to such deductions; provided that, in each of the foregoing circumstances, any such reduction not fully taken as a result of the application of this Section 10.7.6, may be carried forward and applied against future [...\*\*\*...] otherwise owed.

10.8. Other Amounts Payable. With respect to any amounts owed under this Agreement by one Party to the other for which no other invoicing and payment procedure is specified in this Section 10, within [...\*\*\*...] after the end of each Calendar Quarter, each Party will provide an invoice, together with reasonable supporting documentation, to the other Party for such amounts owed in respect of such Calendar Quarter. The owing Party will pay any undisputed amounts within [...\*\*\*...] of receipt of the invoice, and any disputed amounts owed by a Party will be paid within [...\*\*\*...] of resolution of the dispute.

# 10.9. Payment Terms.

- **10.9.1.** Manner of Payment. All payments to be made by a Party hereunder will be made in Dollars by wire transfer to such bank account as the other Party may designate.
- **10.9.2.** Reports and Royalty Payments. For as long as royalties are due under Section 10.6, Novartis will furnish to Xencor a written report, within [...\*\*\*...] after the end of each Calendar Quarter, showing in Dollars, the amount of Annual Net Sales of Licensed Products and royalty due for such Calendar Quarter. Upon receipt of such written report, Xencor shall issue an invoice to Novartis. Royalty payments for each Calendar Quarter will be due within [...\*\*\*...] of receipt of such written invoice for the Calendar Quarter. The report will include, at a minimum, the following information for the applicable Calendar Quarter, each listed by product and by country of sale [...\*\*\*...]. All such reports will be treated as Confidential Information of Novartis.
- **10.9.3.** Currency Exchange. With respect to Annual Net Sales invoiced in Dollars, the Annual Net Sales and the amounts due to Xencor hereunder will be expressed in Dollars. When conversion of payments from any foreign currency is required to be undertaken by Novartis, the Dollar equivalent shall be calculated using Novartis' then-current standard exchange rate methodology as applied in its external reporting for the conversion of foreign currency sales into Dollars.

10.9.4. Taxes. Either Party (a "Withholding Party") may withhold from payments due to the other Party (a "Non-Withholding Party") amounts for payment of any withholding tax that is required by Law to be paid to any taxing authority with respect to such payments, which shall be remitted in accordance with Law. The Withholding Party will provide to the Non-Withholding Party all relevant documents and correspondence, and will also provide to the Non-Withholding Party any other cooperation or assistance on a reasonable basis as may be necessary to enable the Non-Withholding Party to claim exemption from such withholding taxes and to receive a refund of such withholding tax or claim a foreign tax credit. The Withholding Party will give proper evidence from time to time as to the payment of any such tax. The Parties will cooperate with each other in seeking deductions under any double taxation or other similar treaty or agreement from time to time in force. Such cooperation may include the Withholding Party making payments from a single source in the U.S., where possible.

**10.9.5.** Interest Due. Each paying Party will pay the other Party interest on any undisputed payments that are not paid on or before the date such payments are due under this Agreement at the prime rate as reported by Citibank N.A., plus [...\*\*\*...]% per annum or the maximum applicable legal rate, if less, calculated on the total number of days that the payment is delinquent.

**Records and Audits**. Each Party shall keep complete, true and accurate books and records in accordance with its Accounting Standards in relation to this Agreement, including in relation to Development Costs, Commercialization Costs, COGS, Net Sales and royalties. Each Party will keep such books and records for at least [...\*\*\*...] following the Calendar Year to which they pertain. Each Party (the "Auditing Party") may, upon written request, cause an internationally-recognized independent accounting firm (the "Auditor"), which is reasonably acceptable to the other Party (the "Audited Party"), to inspect the relevant records of such Audited Party and its Affiliates to verify the payments made and amounts reported by the Audited Party and the related reports, statements and books of accounts, as applicable. Before beginning its audit, the Auditor shall execute an undertaking acceptable to the Audited Party by which the Auditor agrees to keep confidential all information reviewed during the audit. The Auditor shall have the right to disclose to Auditing Party only its conclusions regarding any payments owed under this Agreement. Each Party and its Affiliates shall make their records available for inspection by the Auditor during regular business hours at such place or places where such records are customarily kept, upon receipt of reasonable advance notice from the Auditing Partner. The records shall be reviewed solely to verify the accuracy of the Audited Party's royalties and other payment obligations and compliance with the financial terms of this Agreement. Such inspection right shall not be exercised more than [ ...\*\*\*...] without cause in any Calendar Year and not more frequently than [...\*\*\*...] without cause with respect to records covering any specific period of time. In addition, Auditing Party shall only be entitled to audit the books and records of Audited Party from the [...\*\*\*...] Calendar Years prior to the Calendar Year in which the audit request is made. The Auditing Party agrees to hold in strict confidence all information received and all information learned in the course of any audit or inspection, except to the extent necessary to enforce its rights under this Agreement or to the extent required to comply with any Law or judicial order. The Auditor shall provide its audit report and basis for any determination to the Audited Party at the time

such report is provided to the Auditing Party before it is considered final. In the event that the final result of the inspection reveals an underpayment or overpayment by either Party, the underpaid or overpaid amount shall be settled promptly. The Auditing Party shall pay for such inspections, as well as its expenses associated with enforcing its rights with respect to any payments hereunder; <u>provided</u> that, if an underpayment of more than [...\*\*\*...]% of the total payments due hereunder for the applicable year is discovered, the fees and expenses charged by the Auditor shall be paid by Audited Party.

#### 11. CONFIDENTIALITY AND PUBLICATION

## 11.1. <u>Nondisclosure Obligation</u>.

- **11.1.1.** All Confidential Information disclosed by or on behalf of one Party to the other Party or its representative under this Agreement will be maintained in confidence by the receiving Party and will not be disclosed to a Third Party or used for any purpose except to exercise its licenses and other rights, to perform its obligations, or as otherwise set forth herein, without the prior written consent of the disclosing Party, except to the extent that such Confidential Information:
  - **11.1.1.1.** is known by the receiving Party at the time of its receipt, and not through a prior disclosure by the disclosing Party, as documented by the receiving Party's business records,
  - **11.1.1.2.** is known to the public before its receipt from the disclosing Party, or thereafter becomes generally known to the public through no breach of this Agreement by the receiving Party,
  - **11.1.1.3.** is subsequently disclosed to the receiving Party by a Third Party who is not under an obligation of confidentiality to the disclosing Party, or
  - **11.1.1.4.** is developed by the receiving Party independently, and without use or application, of Confidential Information received from the disclosing Party, as documented by the receiving Party's business records.

Specific aspects or details of Confidential Information will not be deemed to be within the public domain or in the possession of the receiving Party merely because the Confidential Information is embraced by more general information in the public domain or in the possession of the receiving Party. Further, any combination of Confidential Information will not be considered in the public domain or in the possession of the receiving Party merely because individual elements of such Confidential Information are in the public domain or in the possession of the receiving Party unless the combination and its principles are in the public domain or in the possession of the receiving Party.

**11.1.2.** Notwithstanding the obligations of confidentiality and non-use set forth in this Agreement, a receiving Party may provide Confidential Information disclosed to it, and disclose the existence and terms of this Agreement, as may be reasonably required in order to perform its

obligations and to exploit its licenses and other rights under this Agreement, and specifically to (a) Related Parties, and the receiving Party's employees, directors, agents, consultants, or advisors to the extent necessary for the potential or actual performance of its obligations or exercise of its licenses and other rights under this Agreement in each case who are under an obligation of confidentiality and non-use with respect to such information that is no less stringent than the terms of this Section 11.1, (b) governmental or other Regulatory Authorities in order to obtain patents or perform its obligations or exploit its rights under this Agreement, <u>provided</u> that such Confidential Information will be disclosed only to the extent reasonably necessary to do so, and where permitted, subject to confidential treatment, (c) the extent required by Law, including by the rules or regulations of the United States Securities and Exchange Commission or similar regulatory agency in a country other than the United States or of any stock exchange or listing entity, (d) with respect to the terms of this Agreement only, any bona fide actual or prospective acquirers, underwriters, investors, lenders or other financing sources and any bona fide actual or prospective collaborators, licensors, Sublicensees, licensees or strategic partners and to employees, directors, agents, consultants and advisers of such Third Party, in each case who are under an obligation or confidentiality with respect to such information that is no less stringent than the terms of this Section 11.1 (but of duration customary in confidentiality agreements entered into for a similar purpose), and (e) Third Parties to the extent a Party is required to do so pursuant to the terms of an In-License, provided that such Confidential Information will be disclosed only to the extent reasonably necessary to do so. If a Party is required by Law to disclose Confidential Information of the other Party that is subject to the non-disclosure provisions of this Section 11.1, such Party will promptly inform the other Party of the disclosure that is being sought in order to provide the other Party an opportunity to challenge or limit the disclosure and will reasonably assist in obtaining such treatment at the other Party's reasonable cost. Notwithstanding this Section 11.1.1, Confidential Information that is permitted or required to be disclosed will remain otherwise subject to the confidentiality and non-use provisions of this Section 11.1. If either Party concludes that a copy of this Agreement must be filed with the United States Securities and Exchange Commission or similar regulatory agency in a country other than the United States, such Party will, a reasonable time prior to any such filing, provide the other Party with a copy of such agreement showing any provisions hereof as to which the Party proposes to request confidential treatment, will provide the other Party with an opportunity to comment on any such proposed redactions and to suggest additional redactions, and will take such Party's reasonable comments into consideration before filing such agreement and use Commercially Reasonable Efforts to have terms identified by such other Party afforded confidential treatment by the applicable regulatory agency.

# 11.2. <u>Publication and Publicity</u>.

**11.2.1.** Publication. Except for disclosures permitted pursuant to Section 11.1 and 11.2.2, either Party wishing to make a publication or public presentation that contains the Confidential Information of the other Party or any results of Research and Development activities under the Collaboration will deliver to the other Party a copy of the proposed written publication or presentation at least [...\*\*\*...] prior to submission for publication or presentation. The reviewing Party will have the right (a) to propose modifications to the publication or

presentation for patent reasons or trade secret reasons or to remove Confidential Information of the reviewing Party or its Related Parties, and the publishing Party will remove all Confidential Information of the other Party if requested by the reviewing Party and otherwise take such Party's reasonable comments into consideration, or (b) to request a reasonable delay in publication or presentation in order to protect patentable information. If the reviewing Party requests such a delay, the publishing Party will delay submission or presentation for a period of [...\*\*\*...] (or such shorter period as may be mutually agreed by the Parties) to enable the non-publishing Party to file patent applications protecting such Party's rights in such information. With respect to any proposed publications or disclosures by investigators or academic or non-profit collaborators, such materials will be subject to review under this Section 11.2.1 to the extent that Novartis or Xencor, as the case may be, has the right and ability (after using Commercially Reasonable Efforts to obtain such right and ability) to do so. Notwithstanding the foregoing, (i) in no event will Xencor, its Affiliates or Sublicensees make a publication or public presentation or any other public disclosure with respect to any Regional Licensed Target Pair, Regional Licensed Product, Regional Licensed Antibody, Global Licensed Target Pair, Global Licensed Product, Global Licensed Antibody, Optioned Target Pair, Optioned Licensed Product or Optioned Antibody without the prior written consent of Novartis, and (ii) in no event will Novartis, its Affiliates or Sublicensees make a publication or public presentation or any other public disclosure with respect to any Regional Target Pair, Regional Licensed Product or Regional Licensed Antibody without the prior written consent of Xencor. Further, Xencor shall not submit or publish any article or other publication to or with any scientific journal or other publisher that requires, as a condition of publication, that Xencor agree to make available to the publisher or Third Parties any Antibodies or other Materials which are the subject of the publication.

**11.2.2.** Publicity. Except as set forth in Section 11.1, 11.2.1 and 11.3, the existence of and terms of this Agreement may not be disclosed by either Party, and neither Party will use the name, Trademark, trade name or logo of the other Party or its employees in any publicity, news release or disclosure relating to this Agreement, its subject matter, or the activities of the Parties hereunder without the prior express written permission of the other Party except (a) as may be required by Law, including by the rules or regulations of the United States Securities and Exchange Commission or similar regulatory agency in any country other than the United States or of any stock exchange or listing entity, <u>provided</u> that the Party issuing such press release gives reasonable notice prior to use of such name, Trademark, trade name or logo of the other Party, and otherwise complies with Section 11.1.2, or (b) as expressly permitted by the terms hereof.

## 11.3. Press Release.

11.3.1. Except as provided in Section 11.3.2, neither Party will issue a press release or public announcement relating to this Agreement without the prior written approval of the other Party (such approval not to be unreasonably withheld, conditioned or delayed), except that a Party may (a) once a press release or other public statement is approved in writing by both Parties, make subsequent public disclosure of the information contained in such press release or

other written statement without the further approval of the other Party, and (b) issue a press release or public announcement as required by Law based on the advice of counsel (including a press release corresponding to any securities disclosure, such as pursuant to a Form 8-K or with respect to the achievement of a Development Milestone Event or Sales Milestone Event and the amount of, and receipt of, any Milestone Payment), including by the rules or regulations of the United States Securities and Exchange Commission or similar regulatory agency in a country other than the United States or of any stock exchange or listing entity, provided that, with respect to clauses (a) and (b), the Party issuing such press release gives reasonable prior notice to the other Party of and the opportunity to comment on the press release or public announcement, and otherwise complies with this Section 11. In addition, Xencor may with Novartis' prior written approval, such approval not to be unreasonably withheld, conditioned or delayed, issue a press release regarding the payment or receipt of any milestone payments under this Agreement with respect to any Licensed Products, provided that (i) such press release does not identify the Licensed Antibody or Licensed Target, and (ii) otherwise complies with this Section 11.3.1.

11.3.2. Notwithstanding anything in this Section 11.3 to the contrary, (a) either Party may issue a press release or make a public disclosure relating to such Party's Development, Manufacturing or Commercialization activities under this Agreement with respect to the Regional Target Pairs or Regional Licensed Products in such Party's Territory, and (b) notwithstanding (a) above, Novartis may issue a press release or make a public disclosure relating to the Research, Development, Manufacturing or Commercialization activities under this Agreement with respect to Licensed Target Pairs or Licensed Products other than the Regional Licensed Target Pairs and Regional Licensed Products, provided that such press release or public disclosure pursuant to clause (a) or (b) does not disclose Confidential Information of the other Party. Prior to making any such disclosure under clause (a) of this Section 11.3.2, however, the Party making the disclosure will provide the other Party with a draft of such proposed disclosure within a reasonable time (but at least [...\*\*\*...]) prior to disclosure for the other Party's review and comment, and the disclosing Party will consider in good faith any timely comments provided by the other Party.

## 12. REPRESENTATIONS, WARRANTIES AND COVENANTS

- 12.1. **Mutual Representations and Warranties as of the Effective Date**. Each Party represents and warrants to the other Party that, as of the Effective Date:
  - **12.1.1.** such Party is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or formation,
  - **12.1.2.** such Party has all requisite corporate power and corporate authority to enter into this Agreement and to carry out its obligations under this Agreement,
  - **12.1.3.** all requisite corporate action on the part of such Party, its directors and stockholders required by Law for the authorization, execution and delivery by such Party of this

Agreement, and the performance of all obligations of such Party under this Agreement, has been taken, and

- **12.1.4.** the execution, delivery and performance of this Agreement, and compliance with the provisions of this Agreement, by such Party do not and will not (a) violate any provision of Law or any ruling, writ, injunction, order, permit, judgment or decree of any Governmental Authority, (b) constitute a breach of, or default under (or an event which, with notice or lapse of time or both, would become a default under) or conflict with, or give rise to any right of termination, cancellation or acceleration of, any agreement, arrangement or instrument, whether written or oral, by which such Party or any of its assets are bound, or (c) violate or conflict with any of the provisions of such Party's organizational documents (including any articles or memoranda of organization or association, charter, bylaws or similar documents), and
- **12.1.5.** no consent, approval, authorization or other order of, or filing with, or notice to, any Governmental Authority or other Third Party is required to be obtained or made by such Party in connection with the authorization, execution and delivery by the Company of this Agreement, <u>provided</u> that no representation or warranty is made under this Section 12.1.5 with respect whether any required filing pursuant to the HSR Act.
- 12.2. **Representations and Warranties by Xencor.** Xencor represents and warrants to Novartis as of the Effective Date that:
  - **12.2.1.** Exhibit A-2 forth a complete and accurate list of (a) all Xencor Patents in existence as of the Effective Date, all of which are owned by Xencor,
  - **12.2.2.** Exhibit A-3 sets forth a complete and accurate list of all agreements pursuant to which Xencor Controls any Know-How or Patents that are included within the Xencor Technology,
  - **12.2.3.** to Xencor's knowledge, the Xencor Technology in existence as of the Effective Date comprises all of the intellectual property rights used by or on behalf of Xencor and its Affiliates in the Research, Development and Manufacturing of the Licensed Antibodies and Licensed Products that are in existence as of the Effective Date,
  - **12.2.4.** Xencor (a) owns or has a valid license to, or has a valid option to license, all Xencor Technology in existence as of the Effective Date, including those Licensed Antibodies and Licensed Products in existence as of the Effective Date, and (b) has the right, or an option to obtain the right, and authority to (i) grant to Novartis and its Related Parties, the licenses under the Xencor Technology in existence as of the Effective Date hereunder, and (ii) use, disclose, and commercially exploit, and to enable Novartis and its Related Parties to use, disclose, and commercially exploit (in each case under appropriate conditions of confidentiality) the Xencor Technology in existence as of the Effective Date in the Field and in accordance with the terms and conditions of this Agreement,

- **12.2.5.** Xencor has not granted its Affiliates or any Third Party, including any academic organization or agency, rights that would interfere with Novartis' rights hereunder, and there are no agreements or arrangements other than as set forth in Exhibit A-2 to which Xencor or any of its Affiliates is a party relating to Xencor Technology, Licensed Antibodies, or Licensed Product(s), that would (a) limit the rights granted to Novartis under this Agreement or (b) that restrict or result in a restriction on Novartis' ability to Research, Develop, Manufacture, use or Commercialize the Licensed Antibodies and Licensed Product(s) in the Novartis Territory, in accordance with this Agreement,
- 12.2.6. with respect to any Xencor Technology owned by Xencor, (a) Xencor and its Affiliates have obtained from all individuals who participated in any respect in the invention or authorship thereof, effective, to Xencor's knowledge, assignments of all ownership rights of such individuals in such Xencor Technology, either pursuant to written agreement or by operation of law, and (b) all of its employees, officers, and consultants have executed agreements or have existing obligations under Law requiring assignment to Xencor or its Affiliate, as applicable, of all inventions made during the course of and as the result of the Collaboration, and, to Xencor's knowledge, no officer or employee of Xencor or its Affiliate is subject to any agreement with any other Third Party that requires such officer or employee to assign any interest in any Xencor Technology to any Third Party,
- 12.2.7. all employees, officers, and consultants of Xencor and its Affiliates have executed agreements or have existing obligations under applicable Law obligating the individual to maintain as confidential Xencor's Confidential Information as well as confidential information of other parties (including of Novartis and its Affiliates) that such individual may receive in the conduct of the Collaboration, to the extent required to support Xencor's obligations under this Agreement, and Xencor and its Affiliates have taken commercially reasonable precautions to preserve the confidentiality of Xencor Know-How that is not claimed in a published Xencor Patent or that was not publicly disclosed prior to the Effective Date,
- **12.2.8.** neither Xencor nor any Affiliate has initiated or been involved in any proceedings or other claims in which such Person alleges that any Third Party is or was infringing or misappropriating any Xencor Technology, nor have any such proceedings been threatened in writing by Xencor or its Affiliates,
- **12.2.9.** neither Xencor nor its Affiliates have entered into a government funding relationship that would result in rights to any Licensed Antibody or Licensed Product residing in the US Government, National Institutes of Health, National Institute for Drug Abuse or other agency, and the licenses granted hereunder are not subject to overriding obligations to the US Government as set forth in Public Law 96 517 (35 U.S.C. 200 204), as amended, or any similar obligations under the laws of any other country,
- **12.2.10.** to Xencor's knowledge, the Research, Development, Manufacture, use or Commercialization of the Licensed Antibodies and Licensed Products that exist as of the Effective Date in accordance with the terms and conditions of this Agreement does not infringe or misappropriate the intellectual property rights of any Third Party, nor has Xencor or any Affiliate received in writing any notice alleging such infringement or misappropriation, and

to the knowledge of Xencor, (a) the issued patents in the Xencor Patents as of the Effective Date are valid and enforceable without any claims, challenges, oppositions, nullity actions, interferences, inter-partes reexaminations, inter-partes reviews, post-grant reviews, derivation proceedings, or other proceedings pending or threatened and Xencor has filed and prosecuted patent applications within the Xencor Patents owned by Xencor in good faith and complied with all duties of disclosure with respect thereto, (b) Xencor has not committed any act, or omitted to commit any act, that may cause the Xencor Patents to expire prematurely or be declared invalid or unenforceable, and (c) all application, registration, maintenance and renewal fees in respect of the Xencor Patents as of the Effective Date have been paid and all necessary documents and certificates have been filed with the relevant agencies for the purpose of maintaining the Xencor Patents.

- **12.3. Representations and Warranties by Novartis.** Novartis represents and warrants to Xencor as of the Effective Date that:
  - **12.3.1.** Novartis has not identified any Novartis Technology or Existing Novartis In-Licenses as of the Effective Date; and
  - **12.3.2.** no premerger notification filing is required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules promulgated thereunder in connection with the execution and delivery by Novartis of this Agreement; and
  - 12.3.3. all employees, officers, and consultants of Novartis and its Affiliates have executed agreements or have existing obligations under applicable Law obligating the individual to maintain as confidential Novartis's Confidential Information as well as confidential information of other parties (including of Xencor and its Affiliates) that such individual may receive in the conduct of the Collaboration, to the extent required to support Novartis's obligations under this Agreement.
- 12.4. Warranty Disclaimer. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, NEITHER PARTY MAKES ANY REPRESENTATION OR EXTENDS ANY WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, TO THE OTHER PARTY WITH RESPECT TO ANY PATENTS, KNOW-HOW, MATERIALS, LICENSED ANTIBODY, LICENSED PRODUCT, GOODS, SERVICES, RIGHTS OR OTHER SUBJECT MATTER OF THIS AGREEMENT AND HEREBY DISCLAIMS ALL IMPLIED WARRANTIES OF MERCHANTABILITY, AND FITNESS FOR A PARTICULAR PURPOSE WITH RESPECT TO ANY AND ALL OF THE FOREGOING. WITHOUT EXCUSING EITHER PARTY'S PERFORMANCE OF ITS OBLIGATIONS UNDER THIS AGREEMENT, EACH PARTY HEREBY DISCLAIMS ANY REPRESENTATION OR WARRANTY THAT THE RESEARCH, DEVELOPMENT, MANUFACTURE OR COMMERCIALIZATION OF ANY LICENSED ANTIBODY OR LICENSED PRODUCT PURSUANT TO THIS AGREEMENT WILL BE SUCCESSFUL.
  - 12.5. Certain Covenants.

**12.5.1.** Compliance.

- 12.5.1.1. Each Party and its Related Parties will conduct the Collaboration and the Development, Manufacture and Commercialization of the Licensed Products in accordance with all Laws, including governmental regulations concerning, to the extent applicable to a given activity, cGLP, cGCP and cGMP. In addition, if either Party is or becomes subject to a legal obligation to a Regulatory Authority or other Governmental Authority (such as a corporate integrity agreement or settlement agreement with a Governmental Authority), then the other Party will perform such activities as may be reasonably requested by the obligated Party, at the obligated Party's sole expense, to enable the obligated Party to comply with its legal obligation to such Regulatory Authority with respect to the Licensed Products.
- **12.5.1.2.** During the Term, each Party covenants to the other Party that it and its Affiliates' employees and contractors, shall not, in connection with the performance of their respective obligations under this Agreement, directly or indirectly through Third Parties, pay, promise or offer to pay, or authorize the payment of, any money or give any promise or offer to give, or authorize the giving of anything of value to a Public Official or other Person for purpose of obtaining or retaining business for or with, or directing business to, any Person, including either Party (it being understood that, without any limitation to the foregoing, such Party, and to its knowledge, its and its Affiliates' employees and contractors, has not directly or indirectly promised, offered or provided any corrupt payment, gratuity, emolument, bribe, kickback, illicit gift or hospitality or other illegal or unethical benefit to a Public Official or any other Person in connection with the performance of such Party's obligations under this Agreement, and shall not, directly or indirectly, engage in any of the foregoing).
- **12.5.2.** Conflicting Transactions. During the Term, each of Novartis and Xencor will not, and will cause their Affiliates not to, enter into any agreement granting a license or other right under the Novartis Technology or Xencor Technology, respectively, that is inconsistent with this Agreement.
- 12.5.3. In-Licenses. Each Party will use Commercially Reasonable Efforts to maintain Control of all Patents and Know-How licensed to such Party under the In-Licenses to which such Party is the contracting party. Each Party will use Commercially Reasonable Efforts not to materially breach or be in material default under any of its obligations under any In-License to which such Party is the contracting party that would be necessary or useful for the other Party to Research, Develop, Manufacture and Commercialize any Licensed Antibodies or Licensed Products in the Field in such Party's Territory pursuant to this Agreement. Each Party will not terminate any In-License to which such Party is the contracting party in a manner that would terminate rights that are sublicensed to the other Party. In the event that a Party receives notice of an alleged breach by such Party under an In-License to which it is a party and for which termination of such In-License is being sought by the counterparty, then such Party will

promptly, but in no event less than [...\*\*\*...] thereafter, provide written notice thereof to the other Party and, if possible, grant the other Party the right (but not the obligation) to cure such alleged breach. In the event that a Party intends to materially amend an In-License to which it is a party, then such Party will promptly, but in no event less than [...\*\*\*...] before, provide written notice thereof to the other Party and grant the other Party the right (but not the obligation), acting reasonably, to reject any amendment that would either materially increase the other (i.e., non-party to the In-License) Party's obligations under this Agreement, including any financial obligations or materially decrease the other Party's rights under this Agreement.

12.5.4. No Debarment. Each Party will use reasonable efforts to not use, in any capacity in connection with the Collaboration or the performance of its obligations under this Agreement, any Person that has been debarred pursuant to Section 306 of the FD&C Act, as amended, or that is the subject of a conviction described in such section. Each Party agrees to inform the other Party in writing immediately if it or any Person that is performing activities in the Collaboration or under this Agreement, is debarred or is subject to debarment or is the subject of a conviction described in Section 306 of the FD&C Act, or if any action, suit, claim, investigation or legal or administrative proceeding is pending or, to the best of the notifying Party's knowledge, is threatened, relating to the debarment or conviction of the notifying Party or any Person or entity used in any capacity by such Party or any of its Affiliates in connection with the Collaboration or the performance of its other obligations under this Agreement.

# 12.6. <u>Exclusivity</u>.

## **12.6.1.** Exclusivity.

**12.6.1.1.** On a Licensed Target Pair-by-Licensed Target Pair basis, subject to Section 12.6.2.1, during the Exclusivity Period for a Licensed Target Pair, Xencor and its Affiliates will not, alone or with any Third Parties (including through licensing any Third Party), Research, Develop or Commercialize anywhere in the world any Antibody (other than one or more Licensed Antibodies or Licensed Products that [...\*\*\*...] in accordance with the terms and conditions of this Agreement) that [...\*\*\*...]. For clarity, this Section 12.6.1.1 shall not apply to the Research, Development or Commercialization of an Antibody that [...\*\*\*...].

**12.6.1.2.** On a Licensed Target Pair-by-Licensed Target Pair basis, subject to Section 12.6.2.2, during the Exclusivity Period for a Licensed Target Pair, Novartis and its Affiliates will not, alone or with any Third Parties (including through licensing any Third Party), Research, Develop or Commercialize anywhere in the world any Antibody (other than one or more Licensed Antibodies or Licensed Products that [...\*\*\*...] in accordance with the terms and conditions of this Agreement) that [...\*\*\*...]. For clarity, it is understood that this Section 12.6.1.2 does not restrict the Research, Development or Commercialization by Novartis, its Affiliates or Sublicensees of an Antibody that

[...\*\*\*...], and <u>provided</u> further, for clarity, that this Section 12.6.1.2 shall not permit Novartis to utilize Xencor Technology not otherwise licensed to it under this Agreement.

**12.6.1.3.** For clarity, this Section 12.6.1 shall not apply to the Research, Development or Commercialization of an Antibody that [...\*\*\*...] to only one of the Targets included in a Licensed Target Pair. For clarity, the exercise by Xencor of its Opt-In Right or an Opt-Out Right with respect to a Licensed Target Pair shall not affect the Parties' obligations with respect to such Licensed Target Pair pursuant to Section 12.6.

# **12.6.2.** Other Programs.

#### **12.6.2.1.** Xencor.

- Notwithstanding Section 12.6.1, in the event that Xencor or its Affiliates (a) acquire a Third Party or a portion of the business of a Third Party (whether by merger, stock purchase, purchase of assets, in-license or other means) (a "Third Party Acquisition") that is, prior to such acquisition, conducting a research, development or commercialization program that, if conducted by Xencor at such time, would be a breach of Xencor's exclusivity obligation in Section 12.6.1 (a "Xencor Competing Program"), Xencor will use Commercially Reasonable Efforts to divest such Xencor Competing Program promptly following the closing of such acquisition, and in any event will complete such divestment [...\*\*\*...], <u>provided</u> that (i) [...\*\*\*...] will be extended if, at the expiration of such time period (and any extensions thereto), Xencor provides competent evidence of reasonable on-going efforts to divest such Xencor Competing Program, (ii) Xencor may conduct the Xencor Competing Program independently of Xencor's activities under this Agreement during such time period and without any use of any Restricted Technology, and (iii) Xencor will cease all research, development and commercialization activities with respect to such Xencor Competing Program if Xencor has not completed such divestment within one year after the closing of such acquisition (it being understood that Xencor may thereafter continue its efforts to divest such asset). Xencor will not be deemed in breach of Section 12.6.1 with respect to such Xencor Competing Program so long as Xencor complies with the terms of this Section 12.6.2.1.
- (b) In the event of a Change of Control of Xencor, the exclusivity obligations of Xencor set forth in Section 12.6.1.1 will apply to and bind the Third Party referred to in the definition of Change of Control and its Affiliates. Notwithstanding the foregoing, if any activities of the Third Party or its Affiliates as of the date of the closing of such Change of Control would

constitute a Xencor Competing Program with respect to any Regional Target Pair, Regional Licensed Antibody or Regional Licensed Product, Optioned Target Pair, Optioned Licensed Antibody or Optioned Licensed Product, the provisions of Section 12.6.2.1 shall apply *mutatis mutandis*, unless:

- (i) in the case of a Xencor Competing Program with respect to an Optioned Target Pair, Optioned Licensed Antibody or Optioned Licensed Product, Xencor elects by written notice to Novartis [...\*\*\*...], or
- (ii) in the case of a Xencor Competing Program with respect to a Regional Target Pair, Regional Licensed Antibody or Regional Licensed Product, Xencor elects by written notice to Novartis [...\*\*\*...] such Change of Control to (A) exercise its Opt-Out Right with respect to such Regional Target, (B) maintain such Competing Program outside the Collaboration but cede all decision-making at the JSC to Novartis with respect to the applicable Regional Target Pair, Regional Licensed Antibodies and Regional Licensed Products, in which case, (x) notwithstanding the provisions of Section 2.5.4, if the Parties' Executive Officers are unable to reach unanimous agreement in accordance with Section 2.5.3 with respect to any disputed matter, such matter will be decided by Novartis' Executive Officer, and (y) the provisions of Section 12.6.1.2 shall no longer apply to Novartis with respect to the applicable Regional Target Pair or (C) submit the Competing Program to be part of the Collaboration subject to the Parties negotiating in good faith for up to 90 days any unique terms and conditions that should apply to such contributed program; provided, [...\*\*\*\*....].

[...\*\*\*....].

If any activities of such Third Party or its Affiliates as of the date of the closing of such Change of Control would constitute a Xencor Competing Program with respect to any Global Target Pair, Global Licensed Antibody or Global Licensed Product then, notwithstanding the first sentence of this Section 12.6.2.1(b), such Xencor Competing Program may continue, <u>provided</u> that such Xencor Competing Program is conducted completely independently of Xencor's activities under this Agreement, including without any use of any Restricted Technology.

(c) With respect to Sections 12.6.2.1(a) and 12.6.2.1(b), Xencor and its Affiliates (including such Third Party and its Affiliates under Sections 12.6.2.1(a) and 12.6.2.1(b)) will adopt reasonable procedures (which include appropriate administrative, physical and technical safeguards, including underlying operating system and network security controls and other firewalls) to prevent the use of and access to any Restricted Technology in a manner that is in violation of this Agreement.

#### **12.6.2.2.** Novartis.

- Notwithstanding Section 12.6.1, in the event that Novartis or its (a) (a) Affiliates make a Third Party Acquisition where the applicable Third Party or portion of such Third Party's business, prior to such acquisition, is conducting a research, development or commercialization program that, if conducted by Novartis at such time, would be a breach of Novartis' exclusivity obligations in Section 12.6.1 (a "Novartis Competing Program"), Novartis will use Commercially Reasonable Efforts to divest such Novartis Competing Program promptly following the closing of such acquisition, unless Novartis has exercised its right to terminate this Agreement pursuant to Section 15.2 with respect to the Target of the Novartis Competing Program, and in any event will complete such divestment [...\*\*\*...], provided that (i) [...\*\*\*...] will be extended if, at the expiration of such time period (and any extensions thereto), Novartis provides competent evidence of reasonable on-going efforts to divest such Novartis Competing Program, (ii) Novartis may conduct the Novartis Competing Program independently of Novartis' activities under this Agreement during such time period and without any use of any Restricted Technology, and (iii) Novartis will cease all research, development and commercialization activities with respect to such Novartis Competing Program if Novartis has not completed such divestment within one year after the closing of such acquisition (it being understood that Novartis may thereafter continue its efforts to divest such asset). Novartis will not be deemed in breach of Section 12.6.1 with respect to such Novartis Competing Program so long as Novartis complies with the terms of this Section 12.6.2.1(c).
- (b) (b) In the event of a Change of Control of Novartis, the exclusivity obligations of Novartis set forth in Section 12.6.1.2 will apply to and bind the Third Party referred to in the definition of Change of Control and its Affiliates. Notwithstanding the foregoing, if any activities of the Third Party or its Affiliates as of the date of the closing of such Change of Control would constitute a Novartis Competing Program with respect to any Regional Target Pair, Regional Licensed Antibody or Regional Licensed Product, Optioned Target Pair, Optioned Licensed Antibody or Optioned Licensed Product, the provisions of Section 12.6.2.1 shall apply mutatis mutandis, unless:
  - (i) Novartis provides written notice of its intent to exercise its termination rights with respect to the applicable Regional Target Pair or the Optioned Target Pair [...\*\*\*....].
  - (ii) in the case of a Novartis Competing Program with respect to a Regional Target Pair, Regional Licensed Antibody or Regional Licensed Product,

Novartis elects by written notice to Xencor [...\*\*\*....] such Change of Control to (A) maintain such Competing Program outside the Collaboration but cede all decision-making at the JSC to Xencor with respect to the applicable Regional Target Pair, Regional Licensed Antibodies and Regional Licensed Products, in which case, (x) notwithstanding the provisions of Section 2.5.4, if the Parties' Executive Officers are unable to reach unanimous agreement in accordance with Section 2.5.3 with respect to any disputed matter, such matter will be decided by Xencor's Executive Officer, and (x) the provisions of Section 12.6.1.2 shall no longer apply to Xencor with respect to the applicable Regional Target Pair or (B) submit the Competing Program to be part of the Collaboration subject to the Parties negotiating any good faith for up to 90 days any unique terms and conditions that should apply to such contributed program; provided, [...\*\*\*....].

[...\*\*\*....].

If any activities of such Third Party or its Affiliates as of the date of the closing of such Change of Control would constitute a Novartis Competing Program with respect to any Global Target Pair, Global Licensed Antibody or Global Licensed Product then, notwithstanding the first sentence of this Section 12.6.2.1(b), such Novartis Competing Program may continue, provided that such Novartis Competing Program is conducted completely independently of Novartis' activities under this Agreement, including without any use of any Restricted Technology.

(c) With respect to Sections 12.6.2.2(a) and 12.6.2.1(a), Novartis and its Affiliates (including such Third Party and its Affiliates under Sections 12.6.2.2(a) and 12.6.2.1(a)) will adopt reasonable procedures (which include appropriate administrative, physical and technical safeguards, including underlying operating system and network security controls and other firewalls) to prevent the use of and access to any Restricted Technology in a manner that is in violation of this Agreement.

#### 13. INDEMNIFICATION, LIMITATION OF LIABILITY, INSURANCE

- **General Indemnification by Novartis**. Novartis will indemnify, hold harmless and defend Xencor, 13.1. its Related Parties, and their respective directors, officers, employees and agents ("Xencor Indemnitees") from and against any and all Third Party claims, suits, losses, liabilities, damages, costs, fees and expenses (including reasonable attornevs' fees and litigation expenses) (collectively, "Losses") arising out of or resulting from, directly or indirectly, (a) any breach of, or inaccuracy in, any representation or warranty made by Novartis in this Agreement, or any breach or violation of any covenant or agreement of Novartis, or in the performance of, this Agreement, (b) the negligence or willful misconduct by or of Novartis or its Related Parties (for clarity, excluding Xencor and its Related Parties), and their respective directors, officers, employees and agents in the performance of Novartis' obligations under this Agreement, or (c) subject to Section 13.3, to the extent such Losses arise out of the Research, Development, Manufacturing or Commercialization of Licensed Antibodies or Licensed Products by or on behalf of Novartis or its Related Parties, subject, in the case of Manufacturing, to the terms of any Supply Agreement. Novartis will have no obligation to indemnify the Xencor Indemnitees to the extent that the Losses arise out of or result from, directly or indirectly, any breach of, or inaccuracy in, any representation or warranty made by Xencor in this Agreement, or any breach or violation of any covenant or agreement of Xencor in, or in the performance of, this Agreement, or the negligence or willful misconduct by or of any of the Xencor Indemnitees, or matters for which Xencor is obligated to indemnify Novartis under Section 13.2.
- 13.2. **General Indemnification by Xencor**. Xencor will indemnify, hold harmless, and defend Novartis, its Related Parties and their respective directors, officers, employees and agents ("**Novartis Indemnitees**") from and against any and all Third Party Losses arising out of or resulting from, directly or indirectly, (a) any breach of, or inaccuracy in, any representation or warranty made by Xencor in this Agreement, or any breach or violation of any covenant or agreement of Xencor, or in the performance of, this Agreement, (b) the negligence or willful misconduct by or of Xencor or its Related Parties (for clarity, excluding Novartis and its Related Parties), and their respective directors, officers, employees and agents in the performance of Xencor's obligations under this Agreement, or (c) subject to Section 13.3, to the extent such Losses arise out of the Research, Development, Manufacturing or Commercialization of Licensed Antibodies or Licensed Products by or on behalf of Xencor and its Related Parties. Xencor will have no obligation to indemnify the Novartis Indemnitees to the extent that the Losses arise out of or result from, directly or indirectly, any breach of, or inaccuracy in, any representation or warranty made by Novartis in this Agreement, or any breach or violation of any covenant or agreement of Novartis in or in the performance of this Agreement, or the negligence or willful misconduct by or of any of the Novartis Indemnitees, or matters for which Novartis is obligated to indemnify Xencor under Section 13.1.

## 13.3. **Product Liability**.

**13.3.1.** Any Losses arising out of Third Party product liability claims arising from the Development, Manufacture or Commercialization of Licensed Products will be (a) borne by Novartis, to the extent such Losses arise out of or relate to (i) the Manufacture or Commercialization in or for the Novartis Territory by or on behalf of Novartis or its Related Parties of a Regional Licensed Product, or (ii) a Global Licensed Product or Fc Licensed

Product anywhere in or for the world by or on behalf of Novartis or its Related Parties, and (b) borne by Xencor, to the extent such Losses arise out of or relate to the Manufacture or Commercialization in or for the Xencor Territory by or on behalf of Xencor or its Related Parties of a Regional Licensed Product.

13.3.2. Any Losses arising out of Third Party product liability claims arising from the Development of Regional Licensed Products or any Optioned Licensed Product will be treated as Development Costs in accordance with Section 5.1.4, provided that (x) with respect to any Additional Development Activities, the Proposing Party will be solely responsible for any such liability claims unless and until the Non-Proposing Party delivers an Additional Development Opt-In Notice with respect to such Additional Development Activity, (y) after Xencor exercises its Opt-Out Right, Xencor will remain responsible for its portion of any such liability claims with respect to Development activities occurring up to and through the [...\*\*\*...] period following Novartis' receipt of Xencor's notice exercising the applicable Opt-Out Right, and (z) after termination of this Agreement with respect to any Licensed Target by Novartis pursuant to Section 15.2 or by Xencor pursuant to Section 15.3.1.1 or 15.4, (i) Novartis will remain responsible for its portion of any such liability claims with respect to any ongoing Clinical Studies that Xencor elects to wind down under Section 15.7.1(b) until such wind-down process is complete, and (ii) Xencor will be responsible for any liability claims with respect to any ongoing Clinical Studies that Xencor elects to continue. Any Losses arising out of Third Party product liability claims arising from the Commercialization of Optioned Licensed Products will be treated as Commercialization Costs in accordance with Section 6.1.3. The Party bearing such Losses in accordance with the immediately preceding two sentences will indemnify, hold harmless and defend the other Party and its Related Parties and their respective directors, officers, employees and agents from and against such Losses.

**Indemnification Procedure.** The Party entitled to indemnification under Section 13 (an "Indemnified Party") shall notify the Party potentially responsible for such indemnification (the "Indemnifying Party") in writing promptly upon being notified of or gaining knowledge of any claim or claims asserted or threatened against the Indemnified Party which could give rise to a right of indemnification under this Agreement, provided that the failure to give such notice shall not relieve the Indemnifying Party of its indemnity obligation hereunder except to the extent that such failure materially prejudices the Indemnifying Party. If the Indemnifying Party has acknowledged in writing to the Indemnified Party the Indemnifying Party's responsibility for defending a claim, the Indemnifying Party shall have the right to defend, at its sole cost and expense, such claim by all appropriate proceedings, provided that the Indemnifying Party may not enter into any compromise or settlement unless (a) such compromise or settlement imposes only a monetary obligation on the Indemnifying Party and which includes as an unconditional term thereof, the giving by each claimant or plaintiff to the Indemnified Party of a release from all liability in respect of such claim, or (b) the Indemnified Party consents to such compromise or settlement, which consent shall not be unreasonably withheld, conditioned or delayed unless such compromise or settlement involves (i) any admission of legal wrongdoing by the Indemnified Party, (ii) any payment by the Indemnified Party that is not indemnified hereunder or (iii) the imposition of any equitable relief against the Indemnified Party. If the Indemnifying Party does not elect to assume control of the defense of a claim, the Indemnified

Party shall have the right, at the expense of the Indemnifying Party, upon at least [...\*\*\*...] prior written notice to the Indemnifying Party of its intent to do so, to undertake the defense of such claim for the account of the Indemnifying Party (with counsel reasonably selected by the Indemnified Party and approved by the Indemnifying Party, such approval not to be unreasonably withheld, conditioned or delayed), provided that the Indemnified Party shall keep the Indemnifying Party apprised of all material developments with respect to such claim. The Indemnified Party may not enter into any compromise or settlement without the prior written consent of the Indemnifying Party, such consent not to be unreasonably withheld, conditioned or delayed. The Indemnified Party will cooperate with the Indemnifying Party and may participate in, but not control, any defense or settlement of any claim controlled by the Indemnifying Party pursuant to this Section 13.4 and shall bear its own costs and expenses with respect to such participation, provided that the Indemnifying Party shall bear such costs and expenses if counsel for the Indemnifying Party shall have reasonably determined that such counsel may not properly represent both the Indemnifying Party and the Indemnified Party.

- 13.5. <u>Limitation of Liability</u>. NEITHER PARTY HERETO WILL BE LIABLE FOR SPECIAL, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES ARISING OUT OF THIS AGREEMENT, OR THE EXERCISE OF ITS RIGHTS OR THE PERFORMANCE OF ITS OBLIGATIONS HEREUNDER, INCLUDING LOST PROFITS ARISING FROM OR RELATING TO ANY BREACH OF THIS AGREEMENT, REGARDLESS OF ANY NOTICE OF SUCH DAMAGES, EXCEPT AS A RESULT OF [...\*\*\*...]. NOTHING IN THIS SECTION 13.5 IS INTENDED TO LIMIT OR RESTRICT THE INDEMNIFICATION RIGHTS OR OBLIGATIONS OF EITHER PARTY.
- 13.6. <u>Insurance</u>. Each Party will obtain and maintain insurance during the Term and for a period of at least [...\*\*\*...] after the last commercial sale of any Licensed Product for which it is responsible, with a reputable, solvent insurer in an amount appropriate for its business and products of the type that are the subject of this Agreement, and for its obligations under this Agreement. Specifically, each Party will maintain product liability insurance and clinical trial liability insurance with limits of at least \$[...\*\*\*...] per occurrence and in annual aggregate. Upon request, each Party will provide the other Party with evidence of the existence and maintenance of such insurance coverage. Notwithstanding the foregoing, in the case of Novartis, such obligation may be satisfied by a program of self-insurance.
- 13.7. <u>Disclaimer</u>. The Parties each acknowledge and agree, that (a) Research, Development, and Commercialization is inherently uncertain, (b) no outcome or success of any Licensed Antibodies or Licensed Products is or can be assured and (c) failure to achieve Development and Commercialization of Licensed Products will not in and of itself constitute a breach or default of any obligation in this Agreement.
  - 14. INTELLECTUAL PROPERTY
  - 14.1. <u>Inventorship</u>.

- **14.1.1.** Inventorship for inventions and discoveries (including Know-How) first made during the course of the performance of activities pursuant to the Collaboration will be determined in accordance with United States patent Laws for determining inventorship.
- **14.1.2.** Notwithstanding anything to the contrary in this Agreement, each Party will have the right to invoke the Cooperative Research and Technology Enhancement Act of 2004, 35 U.S.C. § 103(c)(2)-(c)(3) (the "**CREATE Act**") when exercising its rights under this Agreement, but with respect to any Xencor Patents, only with the prior written consent of Xencor in its sole discretion, and with respect to any Patents within Novartis Technology, only with the prior written consent of Novartis in its sole discretion. In the event that a Party intends to invoke the CREATE Act, once agreed to by the other Party if required by the preceding sentence, it will notify the other Party and the other Party will cooperate and coordinate its activities with such Party with respect to any filings or other activities in support thereof. The Parties acknowledge and agree that this Agreement is a "joint research agreement" as defined in the CREATE Act

# 14.2. **Ownership**14.2.1.

- **14.2.1.** Xencor. Excluding Novartis Core Inventions, Xencor will own the entire right, title and interest in and to (a) all Know-How (and Patents claiming inventions therein) first developed or conceived solely by employee(s), agent(s) or consultant(s) of Xencor or its Affiliates in the conduct of the Collaboration and (b) Xencor Core Inventions (collectively, "**Xencor Collaboration IP**"). Novartis, on behalf of itself, its Affiliates and Third Party contractors, hereby assigns to Xencor, any and all right, title and interest in and to any such Xencor Core Inventions developed by them, including all Intellectual Property and proprietary rights therein and all rights of action and claims for damages and benefits arising due to past and present infringement of said rights.
- **14.2.2.** Novartis. Excluding Xencor Core Inventions, Novartis will own the entire right, title and interest in and to (a) all Know-How (and Patents claiming inventions therein) first developed or conceived solely by employee(s), agent(s) or consultant(s) of Novartis or its Affiliates in the conduct of the Collaboration; and (b) Novartis Core Inventions (collectively, "**Novartis Collaboration IP**"). Xencor, on behalf of itself, its Affiliates and Third Party contractors, hereby assigns to Novartis, any and all right, title and interest in and to any such Novartis Core Inventions developed by them, including all Intellectual Property and proprietary rights therein and all rights of action and claims for damages and benefits arising due to past and present infringement of said rights.
- **14.2.3.** Joint Ownership. Excluding Xencor Core Inventions and Novartis Core Inventions, the Parties will jointly own the entire right, title and interest in and to all Know-How (and Patents claiming inventions therein) first developed or conceived jointly by employee(s), agent(s) or consultant(s) acting on behalf of Xencor or its Affiliates, on the one hand, and employee(s), agent(s) or consultant(s) acting on behalf of Novartis or its Affiliates, on the other hand, in the conduct of the Collaboration ("**Joint Collaboration IP**").

**14.2.4.** Current ex-US RLP Patents. Within [...\*\*\*...] of the Effective Date, the Parties will meet to cooperate to bifurcate any Xencor Patents into Patents that (a) solely and specifically Cover the sale, offer for sale, manufacture, use or import of any Regional Licensed Product in the Novartis Territory, which shall be deemed a Current ex-US RLP Patent, and (b) primarily Cover Xencor Know-How in existence as of the Effective Date, which shall be deemed a Patent owned by Xencor.

# 14.3. Prosecution and Maintenance of Patents.

#### 14.3.1. IP Committee.

- **14.3.1.1.** Composition. The IP Committee will be comprised of at least one representative of each Party, which representative shall be either an employee or an outside legal counsel for such Party, provided further that any such outside counsel is bound by confidentiality obligations no less stringent than the requirements of Section 11.1. Each Party will appoint its respective representatives to the IP Committee within [...\*\*\*...] of the Effective Date, and from time to time, may substitute one or more of its representatives, in its sole discretion, but subject to the terms of this Section 14.3.1.1, effective upon notice to the other Party of such change. All IP Committee representatives will have appropriate expertise, seniority, decision-making authority and ongoing familiarity with the Collaboration and each Party's representatives collectively will have relevant expertise in intellectual property portfolio management and licensing matters. Additional representatives or consultants may from time to time, by mutual consent of the Parties, be invited to attend IP Committee meetings, subject to such representatives and consultants (or the representative's or consultant's employer) undertaking confidentiality obligations, whether in a written agreement or by operation of law, no less stringent than the requirements of Section 11.1.
- **14.3.1.2.** Meetings. The IP Committee will meet as necessary to carry out its duties under Section 14.3.1.3, but no more often than once per Calendar Quarter, unless otherwise agreed by its members. The IP Committee will meet in-person at Xencor or Novartis or, alternatively, by means of teleconference, videoconference or other similar communications equipment.
- **14.3.1.3.** IP Committee Responsibilities. The IP Committee will provide input regarding the following solely with respect to Licensed Products (except for Fc Licensed Products):
  - (a) strategies for Prosecuting and Maintaining Patents within Novartis Technology and Xencor Technology (excluding Xencor Core Inventions, [...\*\*\*...]),

- (b) Novartis Collaboration IP (excluding Novartis Biological Material Inventions), Xencor Collaboration IP and the Joint Collaboration IP, and
- (c) such other matters as the Parties agree in writing will be the responsibility of the IP Committee.

In furtherance of the foregoing, each Party will provide appropriate updates to the IP Committee regarding Collaboration IP and other Patents and Know-How licensed hereunder, including with respect to anticipated filing strategies and new inventions.

- **14.3.1.4.** Decision-Making. The IP Committee will be an advisory committee for the Collaboration and to the Parties and will make recommendations by consensus. The IP Committee will not have any final decision-making power; <u>provided</u> that, the Parties will work together in good faith to enable Novartis to file Patents in the Novartis Territory that solely and specifically Cover Licensed Target Pairs, Licensed Antibodies and Licensed Products.
- **14.3.1.5.** Term. Either Party will have the right to terminate the IP Committee upon 30 days advance written notice to the other Party. Notwithstanding the foregoing, on a Regional Licensed Product-by-Regional Licensed Product basis during the Term for such Regional Licensed Product, Xencor will have the right (but not the obligation) to continue to participate in the IP Committee in relation to any such Licensed Product until the twelfth anniversary of the First Commercial Sale of such Regional Licensed Product in the Novartis Territory.
- **14.3.2.** Cooperation. Neither Party will (a) file or assist any Third Party in the filing of any Patents arising from the Collaboration, that in the case of Xencor, claims a Licensed Product, and in the case of Novartis, claims Xencor Know-How, or (b) amend any claims in a Patent arising from the Collaboration, that in the case of Xencor, relates to a Licensed Product, and in the case of Novartis, relates to Xencor Know-How, in each of (a) and (b), without the prior written consent of the other Party, such consent not to be unreasonably withheld, delayed, or conditioned.

# **14.3.3.** Novartis Technology.

- **14.3.3.1.** General. Subject to remainder of this Section 14.3.3, as between the Parties, Novartis will have the sole right to, at Novartis' sole discretion, and sole responsibility for all applicable Patents Costs, Prosecute and Maintain all Patents within Novartis Technology (other than within Joint Collaboration IP), in Novartis' name.
- **14.3.3.2.** Step-In Rights to Novartis Technology. In the event that Novartis elects not to Prosecute and Maintain (or not to continue to Prosecute and Maintain, including filing a Patent claiming priority to a Patent prior to its issuance, or, in the case of foreign deadlines for filing divisionals, prior to such deadline), any Patent within the

Novartis Technology (other than within Joint Collaboration IP, Novartis Biological Material IP or Novartis Biological Material Inventions) that Covers the sale, offer for sale, manufacture, use or import of any Licensed Product, Novartis will notify Xencor at least [...\*\*\*...] before any such Patent would become abandoned, no longer available or otherwise forfeited, and subject to the provisions of any applicable Novartis In-License, Xencor will have the right (but not the obligation), at Xencor's sole discretion, and sole responsibility for all applicable Patent Costs, to Prosecute and Maintain such Patent in the name of Novartis (which right will include the right to file additional Patents claiming priority to such Patent). In such event, Xencor will, upon Novartis' reasonable request, consult with Novartis, through the IP Committee, on the status of its Prosecution and Maintenance activities, and furnish Novartis, copies of documents related to the Prosecution and Maintenance of such Patents.

# 14.3.4. Xencor Technology.

- **14.3.4.1.** General. Subject to remainder of this Section 14.3.4, as between the Parties, (a) Xencor will have the sole right to, at Xencor's sole discretion, and sole responsibility for all applicable Patents Costs, to Prosecute and Maintain all Patents within Xencor Technology (excluding RLP Patents and Patents within Joint Collaboration IP) and Fc Patents in Xencor's name, and (b) subject to the provisions of any applicable Xencor In-License, Novartis will have the sole right, at Novartis' sole discretion, and sole responsibility for all applicable Patents Costs, to Prosecute and Maintain all Patents within RLP Patents in Xencor's name.
- **14.3.4.2.** RLP Patents. In the event that Novartis elects not to Prosecute and Maintain (or not to continue to Prosecute and Maintain, including filing a Patent claiming priority to a Patent prior to its issuance, or, in the case of foreign deadlines for filing divisionals, prior to such deadline), any RLP Patent, Novartis will notify Xencor at least [...\*\*\*...] before any such Patent would become abandoned, no longer available or otherwise forfeited, and the right (but not the obligation), at Xencor's sole discretion, and sole responsibility for all applicable Patent Costs, to Prosecute and Maintain such RLP Patent will revert to Xencor. In such event, Xencor will, upon Novartis' reasonable request, consult with Novartis, through the IP Committee, on the status of its Prosecution and Maintenance activities, and furnish Novartis, copies of documents related to the Prosecution and Maintenance of such RLP Patent.

## 14.3.5. Joint Collaboration IP.

- **14.3.5.1.** Subject to remainder of this Section 14.3.5, as between the Parties, Novartis will have the right (but not the obligation), at Novartis' sole discretion, and sole responsibility for all applicable Patents Costs, to Prosecute and Maintain all Patents within Joint Collaboration IP, in the names of both Novartis and Xencor.
- **14.3.5.2.** In the event that Novartis elects not to Prosecute and Maintain (or not to continue to Prosecute and Maintain, including filing a Patent claiming priority to a

Patent prior to its issuance, or, in the case of foreign deadlines for filing divisionals, prior to such deadline), any Patent within Joint Collaboration IP anywhere in the world, Novartis will notify Xencor at least [...\*\*\*...] before any such Patent within Joint Collaboration IP would become abandoned, no longer available or otherwise forfeited, Xencor will have the right (but not the obligation), at Xencor's sole discretion, and sole responsibility for all applicable Patents Costs, to Prosecute and Maintain such Patent, in the names of both Novartis and Xencor.

14.3.5.3. Each Party when controlling Prosecution and Maintenance of Patents within Joint Collaboration IP will (a) furnish the other Party, via electronic mail or such other method as mutually agreed by the Parties, (i) copies of substantive proposed filings and documents received from outside counsel or generated internally in the course of Prosecuting and Maintaining such Patents within Joint Collaboration IP, and (ii) copies of documents filed with the relevant patent offices and other Governmental Authorities with respect to such Patents within Joint Collaboration IP, and (iii) such other substantive documents related to the Prosecution and Maintenance of such Patents within Joint Collaboration IP, and as applicable in sufficient time prior to filing such document or making any payment due thereunder to allow for review and comment by the other Party and (b) consider in good faith timely comments from the other Party with respect thereon. Each Party will furnish the other Party, via electronic mail or such other method as mutually agreed by the Parties, copies of documents filed with the relevant national patent offices and other Governmental Authorities with respect to such Patents within Joint Collaboration IP.

**14.3.6.** Patent Cooperation. Each Party will (a) use Commercially Reasonable Efforts to make its employees, agents and consultants reasonably available to the other Party (or to the other Party's authorized attorneys, agents or representatives), to the extent reasonably necessary to enable such other Party in exercising its Prosecution and Maintenance rights described herein, and (b) reasonably cooperate in any such Prosecution and Maintenance by the other Party, including signing or having signed all legal documents as are reasonably necessary to Prosecute and Maintain Patents.

### 14.4. Third Party Infringement and Defense.

**14.4.1.** Notices. Each Party will promptly report in writing to the other Party of any Infringement of which such Party (or any of its Affiliates or Sublicensees) becomes aware, and will provide the other Party with all available evidence of such Infringement in such Party's Control. Subject to the terms of this Section 14.4, the JSC will discuss in good faith strategies for abating such Infringement of any Regional Licensed Product within each of the Party's respective Territory.

### **14.4.2.** Rights to Enforce.

**14.4.2.1.** Novartis GLP and OLP Infringement. Subject to the terms of this Section 14.4, as between the Parties, Novartis will have the first right (but not the

obligation), at Novartis' sole discretion, through counsel of its choosing which is reasonably acceptable to Xencor, to seek to abate any Novartis GLP and OLP Infringement by enforcing (including deciding on any litigation strategy) any Patents within the Novartis Patents or Joint Patents in the Novartis Territory. If Novartis does not take steps to abate such Infringement, within [...\*\*\*...] after receipt of written notice of such Infringement (or such shorter period of time as is required to comply with the provisions of Section 14.4.2.4 or any other Law in the United States or any other country in the Territory to not waive any statutory rights), Novartis will provide Xencor with notice of such decision and, subject to the terms of any In-Licenses, [...\*\*\*...], then Xencor shall have the right (but not the obligation) to assume such enforcement (excluding any enforcement of any Patents covering [...\*\*\*...]); and if Novartis' failure to take such steps to abate such Infringement is, [...\*\*\*...]. Novartis will pay all Patent Costs incurred by Novartis for such enforcement, and Xencor will pay all Patent Costs incurred by Xencor if it assumes such enforcement. [...\*\*\*...]

**14.4.2.2.** Novartis RLP Product Infringement. Subject to the terms of this Section 14.4, as between the Parties, Novartis will have the first right (but not the obligation), at Novartis' sole discretion, through counsel of its choosing which is reasonably acceptable to Xencor, to seek to abate any Novartis RLP Product Infringement by enforcing (including deciding on any litigation strategy) any Patents within the RLP Patents, Novartis Patents or Joint Patents in the Novartis Territory[...\*\*\*...]. If Novartis does not take steps to abate such Infringement, within [...\*\*\*...] after receipt of written notice of such Infringement, Novartis will provide Xencor with notice of such decision and, subject to

the terms of any In-Licenses, and if Novartis' failure to take such steps to abate such Infringement is, [...\*\*\*...], then Xencor shall have the right (but not the obligation) to assume such enforcement (excluding any enforcement of any Patents covering [...\*\*\*...]). Novartis will pay all Patent Costs incurred by Novartis for such enforcement, and Xencor will pay all Patent Costs incurred by Xencor if it assumes such enforcement.

- **14.4.2.3.** Xencor RLP Product Infringement. Subject to the terms of this Section 14.4, as between the Parties, Xencor will have the exclusive right (but not the obligation), at Xencor's sole discretion, through counsel of its choosing which is reasonably acceptable to Novartis, to seek to abate any Xencor RLP Product Infringement by enforcing (including deciding on any litigation strategy) any Patents within the Xencor Patents or Joint Patents in the Xencor Territory. If Xencor does not take steps to abate such Xencor RLP Product Infringement, within [...\*\*\*...] after receipt of written notice of such Infringement, Xencor will provide Novartis with notice of such decision. Xencor will pay all Patent Costs incurred by such Xencor for such enforcement.
- **14.4.2.4.** Biosimilar Application. Notwithstanding Sections 14.4.2.1 or 14.4.2.3, if either Party (or any of their Related Parties) receives a copy of a Biosimilar Application naming a Licensed Product as a reference product or otherwise becomes aware that such a Biosimilar Application has been filed (such as in an instance described in Section 351(1)(9)(C) of the PHSA), such Party will promptly notify the other Party. If either Party receives any equivalent or similar certification or notice in the United States or any other jurisdiction, either Party will, promptly, notify and provide the other Party copies of such communication. Regardless of the Party that is the "reference product sponsor" for purposes of such Biosimilar Application:
  - (a) The Party with the enforcement rights under Sections 14.4.2.1 or 14.4.2.3 (the "**Lead Party**" for the remainder of this Section 14.4.2.4) will designate pursuant to Section 351(l)(1)(B)(ii) of the PHSA the outside counsel and in-house counsel who will receive confidential access to the Biosimilar Application. The Lead Party will pay all Patent Costs incurred by such Party for such enforcement under this Section 14.4.2.4.
  - (b) The Lead Party will have the right, after consulting with the other Party, to list any Patents for which the enforcement rights in Sections 14.4.2.1 and 14.4.2.3 are applicable, insofar as they meet the statutory requirements pursuant to Section 351(l)(1)(3)(A), Section 351(l)(5)(b)(i)(II), or Section 351(l)(7) of the PHSA, to respond to any communications with respect to such lists from the filer of the Biosimilar Application, and to negotiate with the filer of the Biosimilar Application as to whether to utilize a different mechanism for information exchange other than that specified in Section 351(l) of the PHSA.

- (c) The Lead Party will have the right, after consulting with the other Party, to identify Patents for which the enforcement rights in Sections 14.4.2.1 and 14.4.2.3 are applicable, or respond to relevant communications under any equivalent or similar listing to those described in the preceding Section 14.4.2.4(b) in any other jurisdiction outside of the United States. If required pursuant to Law, upon the Lead Party's request, the other Party will assist in the preparation of such list and make such response after consulting with the Lead Party.
- (d) The other Party will (1) within [...\*\*\*...] after the Lead Party's written request, provide to the Lead Party all information, including a list of Patents Controlled by such other Party and for which the enforcement rights in Sections 14.4.2.1 and 14.4.2.3 are applicable, that is necessary or reasonably useful to enable the Lead Party to make any lists or communications with respect to such Patents that are described in the foregoing Sections 14.4.2.4(b) or 14.4.2.4(c), and (2) cooperate with the Lead Party's reasonable requests in connection therewith to the extent required or not prohibited by Law. The Lead Party will consult with the other Party prior to identifying any Patents controlled by such other Party as contemplated by this Section 14.4.2.4. The Lead Party will consider in good faith advice and suggestions with respect thereto received from the other Party, and will notify the other Party of any such lists or communications promptly after they are made.
- (e) The Parties recognize that procedures other than those set forth above in this Section 14.4.2.4 may be applicable to Biosimilar Applications that are not governed by the PHSA. As a result, in the event that the Parties acting in good faith mutually determine that certain provisions of Law in the United States or in any other country in the Territory are applicable to actions taken by the Parties with respect to Biosimilar Applications under this Section 14.4.2.4 in such country, the Parties will comply with any such Law in such country (and any relevant and reasonable procedures established by the applicable governing body in such country) in exercising their rights and obligations with respect to Biosimilar Applications under this Section 14.4.2.4.
- **14.4.3.** Defense. As between the Parties, the Party controlling the Prosecution and Maintenance of any Patent under Section 14.3 (i.e., initially, Novartis for Patents contained in Novartis Technology, RLP Patents, and Joint Collaboration IP and Xencor for other Patents contained within Xencor Technology), will have the right (but not the obligation), at its sole discretion, to defend (including deciding on any litigation strategy) against a declaratory judgment action or other action challenging any such Patent, other than with respect to (a) any counter-claims in any enforcement action brought by the other Party pursuant to Section 14.4.2 or (b) any action by a Third Party in response to an enforcement action brought by the other Party, which in both cases ((a) and (b)) will be controlled by such other Party. If the Party controlling such Prosecution and Maintenance of Patents under Section 14.3 is Novartis, and

Novartis does not defend such Patent under this Section 14.4.3 within [...\*\*\*...] (or such shorter period of time as is required to comply with the provisions of Section 14.4.2.4 or any other Law in the United States or any other country in the Territory to not waive any statutory rights), or elects not to continue any such defense (in which case it will promptly provide notice thereof to Xencor), then Xencor will have the right (but not the obligation), at its sole discretion, to defend any such Patent (excluding any Patents covering Novartis Biological Material IP or Novartis Biological Material Inventions). For clarity, only Xencor has a step-in defense right under this Section 14.4.3.

- **14.4.4.** Withdrawal, Cooperation and Participation. With respect to any infringement or defensive action identified above in this Section 14.4 and subject to the terms of this Section:
  - **14.4.4.1.** If the controlling Party ceases to pursue or withdraws from such action, it will promptly notify the other Party (in sufficient time to enable the other Party to meet any deadlines by which any action must be taken to preserve any rights in such infringement or defensive action (including any such period of time as is required to comply with the provisions of Section 14.4.2.4)) and, if permitted under Section 14.4.3, such other Party may substitute itself for the withdrawing Party and proceed under the terms and conditions of this Section 14.4.
  - **14.4.4.2.** The non-controlling Party will cooperate with the Party controlling any such action (as may be reasonably requested by the controlling Party), including, at the controlling Party's sole cost and expense, (1) providing access to relevant documents and other evidence, (2) using reasonable efforts to make its and its Affiliates and licensees and Sublicensees and all of their respective employees, subcontractors, consultants and agents available at reasonable business hours and for reasonable periods of time, but only to the extent relevant to such action, and (3) if reasonably necessary, by being joined as a party, subject for this clause (3) to the controlling Party agreeing to indemnify such non-controlling Party for its involvement as a named party in such action and paying those Patent Costs incurred by such non-controlling Party in connection with such joinder. The Party controlling any such action will keep the other Party reasonably updated with respect to any such action, including providing copies of all materials documents received or filed in connection with any such action.
  - **14.4.4.3.** Each Party will have the right to consult with the other Party regarding any such action controlled by such other Party, in each case at such first Party's sole cost and expense. If a Party elects to so be involved, the controlling Party will provide the other Party and its counsel with an opportunity to consult with the controlling Party and its counsel regarding the prosecution of such action (including reviewing the contents of any correspondence, legal papers or other documents related thereto), and the controlling Party will take into account reasonable and timely requests of the other Party regarding such enforcement or defense. Nothing in this Section 14.4.4.3 will limit the controlling Party's ability to prosecute any such action.

**14.4.5.** Settlement. With respect to any infringement or defensive action identified above in this Section 14.4, the Party controlling such action will have the right to settle or otherwise dispose of such action on such terms as such Party will determine in its sole discretion, including by granting a license or sublicense to a Third Party under the rights granted to such Party in Section 9, <u>provided</u> that, notwithstanding the foregoing, any such settlement that results in a sublicense will be subject to the terms and conditions of this Agreement pertaining to sublicenses and no such settlement or other disposition will (a) impose any monetary restriction or obligation on or admit fault of the other Party and (b) adversely affect the other Party's rights under this Agreement to any such Patent then being enforced or defended, including by amending, invalidating, or otherwise affecting the claims of any Patents licensed to the other Party hereunder, in each case ((a) and (b) without the prior written consent of the other Party, not to be unreasonably withheld, delayed, or conditioned.)

**14.4.6.** Damages. Unless otherwise agreed by the Parties, all monies recovered upon the final judgment or settlement of any action described in Section 14.4 will be used:

```
14.4.6.1. [...***...],
14.4.6.2. [...***...];
14.4.6.3. [...***...];
14.4.6.4. [...***...];
14.4.6.5. [...***...];
```

-127-

**14.4.6.7.** [...\*\*\*...]; **14.4.6.8.** [...\*\*\*...]; and **14.4.6.9.** [...\*\*\*...].

- 14.5. Patent Extensions. For clarity, the Party controlling the Prosecution and Maintenance of any Patent under Section 14.3 (i.e., initially, Novartis for Patents contained in Novartis Technology, RLP Patents, and Joint Collaboration IP and Xencor for other Patents contained within Xencor Technology), will have the right to elect and file for patent term restoration or extension, supplemental protection certificate or any of their equivalents with respect to such Patents with respect to any Licensed Product in their respective Territories provided that, Xencor shall take into account reasonable and timely requests by Novartis to make any such elections or filings with respect to any such other Patents contained within Xencor Technology if Xencor's failure to so act in such country would impair Novartis' ability to obtain any such restoration, extension, supplemental protection certificate or any of their equivalents for the same relating to any Patents for which Novartis controls the Prosecution and Maintenance thereof. The Parties will cooperate and shall take the other Party's reasonable input into account in determining whether to obtain such Patent term restoration, extension, supplemental protection certificate or equivalent thereof. Upon the request by a Party, such other Party will reasonably cooperate in the implementation of such requesting Party's decisions made in a manner with this Section 14.5.
- Patent Listings. With respect to any filings made to Regulatory Authorities with respect to any Patents within the Novartis Technology or the Xencor Technology for any Licensed Product within such Party's Territory, including as required or allowed in connection with in the United States, the FDA's Orange or Purple Book, if applicable, or outside the United States, other international equivalents, but subject to Section 14.4.2.4, (a) the Parties will list any such Patents as may be required by Laws, and (b) otherwise (i) Novartis will have the sole right to make any such decision whether to list for Patents within the Novartis Collaboration IP or Joint Collaboration IP for the Novartis Territory and for any Licensed Product (other than any Regional Licensed Product in the Xencor Territory), (ii) Xencor will have the sole right to make any such decision whether to list for Patents within the Xencor Collaboration IP or Joint Collaboration IP for the Xencor Territory for any Regional Licensed Product, and (iii) each Party will have the sole right to make any such decision whether to list for Patents otherwise within the Novartis Technology for Novartis or the Xencor Technology for Xencor with respect to any Licensed Product, in each case, such consent not to be unreasonably withheld, delayed, or conditioned. Upon the request by a Party, such other Party will

reasonably cooperate in the implementation of such requesting Party's decisions made in a manner with this Section 14.6.

14.7. Common Interest. All information exchanged between the Parties regarding the Prosecution and Maintenance, and enforcement and defense, of the Patents under this Section 14 will be deemed Confidential Information of the disclosing Party. In addition, the Parties acknowledge and agree that, with regard to such Prosecution and Maintenance, and enforcement and defense, the interests of the Parties as collaborators and licensor and licensee are aligned and are legal in nature. The Parties agree and acknowledge that they have not waived, and nothing in this Agreement constitutes a waiver of, any legal privilege concerning the Patents under this Section 14, including privilege under the common interest doctrine and similar or related doctrines. Notwithstanding anything to the contrary contained herein, to the extent a Party has a good faith belief that any information required to be disclosed by such Party to the other Party under this Section 14 is protected by attorney-client privilege or any other applicable legal privilege or immunity, such Party shall not be required to disclose such information and the Parties shall in good faith cooperate to agree upon a procedure (including entering into a specific common interest agreement, disclosing such information on a "for counsel eyes only" basis or similar procedure) under which such information may be disclosed without waiving or breaching such privilege or immunity.

### 14.8. <u>Trademarks</u>.

- 14.8.1. Regional Licensed Products.
- **14.8.1.1.** Each Party has the right to use any Trademark it owns or controls for Regional Licensed Products in its Territory at its sole discretion, and each Party and its Affiliates will retain all right, title and interest in and to its and their respective corporate names and logos.
- **14.8.1.2.** Pursuant to Section 6.1.5, each Party will develop and propose, and the JSC will review and comment on, one or more RLP Trademark(s) for use throughout the Novartis Territory and Xencor Territory, as applicable. Such RLP Trademark(s) considered by the JSC may include, in each Party's sole discretion, the RLP Trademark(s) developed by the other Party with respect to the Commercialization of Regional Licensed Products in such Party's Territory, but may not include other Trademarks owned or controlled by the other Party. Any RLP Trademark(s) that are developed and used by Novartis to promote and sell Regional Licensed Products in the Novartis Territory are hereinafter referred to as the "Novartis RLP Trademarks". Any RLP Trademark(s) that are developed and used by Xencor to promote and sell Regional Licensed Products in the Xencor Territory are hereinafter referred to as the "Xencor RLP Trademarks". As between the Parties, Xencor will own all rights to Xencor RLP Trademarks and all goodwill associated therewith, throughout the Xencor Territory and, if Novartis chooses to use such Xencor RLP Trademarks for the Regional Licensed Products in the Novartis Territory, the Novartis Territory. As between the Parties, Novartis will own all rights to Novartis RLP Trademarks and all goodwill associated therewith, throughout the Novartis Territory and, if Xencor chooses to use such Novartis RLP Trademarks for the Regional Licensed Products in the Xencor Territory, the Xencor Territory. Xencor will also own rights to any Internet domain

names incorporating the applicable Xencor RLP Trademarks or any variation or part of such Xencor RLP Trademarks used as its URL address or any part of such address, and Novartis will also own rights to any Internet domain names incorporating the applicable Novartis RLP Trademarks or any variation or part of such Novartis RLP Trademarks used as its URL address or any part of such address.

- **14.8.1.3.** If a Party determines to use the other Party's RLP Trademarks to promote and sell any Regional Licensed Product in its Territory, then Xencor and Novartis will enter into a separate trademark license agreement containing commercially reasonable and customary terms pursuant to which the Party owning such RLP Trademark will grant the other Party an exclusive, royalty-free license to use the applicable RLP Trademark(s) to Commercialize Regional Licensed Products in the other Party's Territory.
- **14.8.1.4.** In the event either Party becomes aware of any infringement of any RLP Trademark by a Third Party, such Party will promptly notify the other Party and the Parties will consult with each other and jointly determine the best way to prevent such infringement, including by the institution of legal proceedings against such Third Party, <u>provided</u> further that the Party owning such RLP Trademark retains the sole right (but not obligation) to seek to abate any such infringement.
- **14.8.2.** Global Licensed Products, Optioned Licensed Products, and Fc Products.
- **14.8.2.1.** As between the Parties, Novartis has the sole and exclusive right to select and develop one or more Product Global Trademark(s) for use throughout the Novartis Territory for all Global Licensed Products, Optioned Licensed Products and Fc Licensed Products. Such Product Global Trademark(s) may not include Trademarks owned or controlled by Xencor. Novartis will own all rights to Product Global Trademarks and all goodwill associated therewith, throughout the Novartis Territory. Novartis will also own rights to any Internet domain names incorporating the applicable Product Global Trademarks or any variation or part of such Product Global Trademarks used as its URL address or any part of such address.
- **14.8.2.2.** In the event that either Party becomes aware of any infringement of any Product Global Trademark by a Third Party, such Party will promptly notify the other Party and the Parties will reasonably consult with each other and jointly determine the best way to prevent such infringement, including by the institution of legal proceedings against such Third Party, <u>provided</u> further that Novartis retains the sole right (but not obligation) to seek to abate any such infringement.

**14.8.3.** No Other Trademark Rights. For the avoidance of doubt, neither Party will have any right to use the other Party's or the other Party's Affiliates' corporate names or logos in connection with Research, Development, Manufacturing and Commercialization of Regional Licensed Products.

### 15. TERM AND TERMINATION

- 15.1. <u>Term.</u> This Agreement will be effective as of the Effective Date and, unless terminated earlier, this <u>Agreement will continue on Licensed Target Pair-by-Licensed Target Pair</u>, or with respect to any given Fc Licensed Antibody/Fc Licensed Product, basis until the expiration of all payment obligations with respect to such Licensed Target Pair or Fc Licensed Antibody/Fc Licensed Product ("Term"). Upon such expiration (but not any earlier termination), the licenses granted covering such Licensed Target Pair or Fc Licensed Antibody/Fc Licensed Product, as applicable, shall become fully-paid up, non-exclusive, and perpetual.
- 15.2. <u>Termination by Novartis for Convenience</u>. Novartis may terminate this Agreement for any reason or no reason on a Licensed Target Pair-by-Licensed Target Pair or Fc Target-by-Fc Target basis at any time upon [...\*\*\*...] prior written notice.

### 15.3. Termination for Material Breach.

#### 15.3.1. Material Breach.

- **15.3.1.1.** Subject to Section 15.3.2, Xencor will have the right to terminate this Agreement (a) on a Licensed Target Pair-by-Licensed Target Pair or Fc Target-by-Fc Target basis, upon delivery of written notice to Novartis in the event of any material breach by Novartis of this Agreement with respect to such Licensed Target Pair or Fc Target or (b) in its entirety upon delivery of written notice to Novartis in the event of any material breach of this Agreement as an entirety, <u>provided</u> that such termination will not be effective if such breach has been cured within [...\*\*\*...] after written notice thereof is given by Xencor to Novartis specifying the nature of the alleged breach (or, if such default cannot be cured within such [...\*\*\*...] period, within [...\*\*\*...] after such notice if Novartis commences actions to cure such default within such [...\*\*\*...] period and thereafter diligently continues such actions, but fails to cure the default by the end of such [...\*\*\*...]). Notwithstanding the foregoing, the cure period for a material breach of Novartis' obligations to make payments to Xencor shall not exceed [...\*\*\*...].
- **15.3.1.2.** Subject to Section 15.3.2, Novartis will have the right to terminate this Agreement (a) on a Licensed Target Pair-by-Licensed Target Pair or Fc Target-by-Fc Target basis, upon delivery of written notice to Xencor in the event of any material breach by Xencor of this Agreement with respect to such Licensed Target Pair or Fc Target or (b) in its entirety upon delivery of written notice to Xencor in the event of any material breach of this Agreement as an entirety, <u>provided</u> that such termination will not be effective if such breach has been cured within [...\*\*\*...] after written notice thereof is given by Novartis to Xencor specifying the nature of the alleged breach (or, if such

default cannot be cured within such [...\*\*\*...] period, within [...\*\*\*...] after such notice if Xencor commences actions to cure such default within such [...\*\*\*...] period and thereafter diligently continues such actions, but fails to cure the default by the end of such [...\*\*\*...]). Notwithstanding the foregoing, the cure period for a material breach of Xencor's obligations to make payments to Novartis shall not exceed [...\*\*\*...].

- **15.3.2.** Disputed Breach. If the alleged breaching Party disputes in good faith the existence or materiality of a breach specified in a notice provided by the other Party in accordance with Section 15.3.1 and such alleged breaching Party provides the other Party notice of such dispute within such [...\*\*\*...] or [...\*\*\*...], as applicable, then the non-breaching Party will not have the right to terminate this Agreement under Section 15.3.1 unless and until the dispute resolution process set forth in Section 16.4 has be completed (including the tolling and cure periods set forth therein).
- 15.4. **Termination for Insolvency**. If, at any time during the Term (a) a case is commenced by or against either Party under Title 11, United States Code, as amended, or analogous provisions of Law outside the United States (the "**Bankruptcy Code**") and, in the event of an involuntary case under the Bankruptcy Code, such case is not dismissed within 60 days after the commencement thereof, (b) either Party files for or is subject to the institution of bankruptcy, liquidation or receivership proceedings (other than a case under the Bankruptcy Code), (c) either Party assigns all or a substantial portion of its assets for the benefit of creditors, (d) a receiver or custodian is appointed for either Party's business, or (e) a substantial portion of either Party's business is subject to attachment or similar process, then, in any such case ((a), (b), (c), (d) or (e)), the other Party may terminate this Agreement upon written notice to the extent permitted under Law.
  - **15.5. Termination for [...\*\*\*...]**. Xencor shall have the right to terminate this Agreement upon written notice to Novartis if (i) Novartis or any of its Affiliates directly, or indirectly through any Third Party, commences [...\*\*\*...] or (ii) any Sublicensee directly, or indirectly through any Third Party, commences [...\*\*\*...], and (A) Novartis does not cause such Sublicensee [...\*\*\*...] or (B) Novartis does not terminate the [...\*\*\*...]. Notwithstanding the foregoing, Xencor shall have no such right to terminate this Agreement in the case of (I) [...\*\*\*...] as a defense in any lawsuit or administrative proceeding brought by Xencor, its Affiliates or licensees for [...\*\*\*...]; or (II) [...\*\*\*...].
    - **15.6. Post-Termination** [...\*\*\*...]. Upon termination of this Agreement for any reason

other than termination by Novartis pursuant to Section 15.3 or Section 15.4 (and as opposed to expiration of this Agreement) with respect to any Restricted Licensed Antibody [...\*\*\*...], Novartis and its Affiliates will not, alone or with any Third Parties (including through licensing any Third Party), [...\*\*\*...].

- **15.7.** Effect of Termination by Xencor for Cause or [...\*\*\*...] or by Novartis for Convenience. Upon termination of this Agreement with respect to any Licensed Target Pair or Fc Target by Novartis pursuant to Section 15.2 or by Xencor pursuant to Section 15.3.1.1, 15.4 or 15.5:
  - 15.7.1. With respect to terminated Regional Target Pairs or the Optioned Target Pair, Novartis will pay (a) for a period of [...\*\*\*...] after the effective date of termination for the applicable Licensed Target Pair or Optioned Target Pair, its portion of Development Costs for those Licensed Antibodies or Licensed Products against such Regional Target Pair or the Option Target Pair, as applicable in the budget for the applicable Development Plan, and (b) [...\*\*\*...] of reasonable costs and expenses (calculated in the same manner as Development Costs) to wind down those then on-going associated Clinical Studies that Xencor identifies, by written notice to Novartis provided no later than [...\*\*\*...] after the effective date of termination, will not be continued.
  - **15.7.2.** Subject to Section 15.7.1, all licenses granted under Section 9 and other rights and obligations under this Agreement with respect to such Licensed Target Pair or Fc Target will terminate. With respect to terminated Licensed Target Pairs, Novartis shall, at Xencor's discretion, (a) transition such activities, if any, (whether Commercial or Development) pursuant to a transition plan to be mutually agreed by the Parties in good faith or (b) wind down any ongoing activities (whether Commercial- or Development-related) of Novartis and its Affiliates and Sublicensees with respect to such Licensed Target Pairs in an orderly fashion and in accordance with all Laws, accepted pharmaceutical industry norms, ethical practices and, with respect to clinical activities, in a manner not adverse to patient safety.
  - 15.7.3. Solely with respect to Regional Licensed Products, Optioned Licensed Products, and Global Licensed Products [...\*\*\*...], Novartis hereby grants to Xencor [...\*\*\*...] license (it being understood and agreed that with respect to any [...\*\*\*...] that is in-licensed by Novartis or any of its Related Parties, Xencor will be responsible for any payments due to a Third Party with respect thereto and Xencor's rights will be subject to the terms of the applicable Third Party agreement, in each case to the extent that Xencor has been advised in writing regarding any such obligations), with the right to grant sublicenses [...\*\*\*...], under [...\*\*\*...], to the extent reasonably necessary to Research, Develop, subject to Section 15.7.7, Manufacture and Commercialize [...\*\*\*...]

[...\*\*\*...], provided that the Parties agree [...\*\*\*...] and Xencor will bear the reasonable and documented Out-of-Pocket Costs of [...\*\*\*...]. In addition, Novartis hereby grants to Xencor a non-exclusive, perpetual, irrevocable, worldwide, royalty-free license (it being understood and agreed that with respect to any [...\*\*\*...] that is in-licensed by Novartis or any of its Related Parties, Xencor will be responsible for any payments due to a Third Party with respect thereto and Xencor's rights will be subject to the terms of the applicable Third Party agreement, in each case to the extent that Xencor has been advised in writing regarding any such obligations) under [...\*\*\*...], to the extent [...\*\*\*...].

15.7.4. Pursuant to a plan to be established by the Parties working together in good faith, Novartis will as promptly as practicable (a) assign to Xencor or Xencor's designee possession and ownership of all material governmental or regulatory filings and approvals (including all material Regulatory Approvals, Regulatory Materials, Pricing Approvals) and copies of material correspondence and conversation logs relating [...\*\*\*...], (b) provide to Xencor or Xencor's designee copies of all material data, reports, records, and other material sales and marketing related information in Novartis' possession and Control [...\*\*\*...]

-134-

[...\*\*\*...] and (c) at Xencor's request and discretion, assign or transfer or, if subject to the agreement of a Third Party, use Commercially Reasonable Efforts to assign or transfer, any Third Party agreement to which Novartis is a party [...\*\*\*...]. In addition, Novartis will [...\*\*\*...]. In the event of failure to obtain assignment of any of the items required to be assigned under this Section 15.7.4, Novartis hereby consents and grants to Xencor the right to [...\*\*\*...].

- **15.7.5.** Novartis shall, subject to the obligations of confidentiality and non-use set forth in Section 11.1, disclose to Xencor [...\*\*\*...].
- **15.7.6.** If the effective date of termination is after the First Commercial Sale of a [...\*\*\*...] then, to the extent permitted by Laws, Novartis or its Affiliates (or to the extent permitted by the applicable sublicense, its Sublicensees) will appoint Xencor as its exclusive distributor of [...\*\*\*...] in the Novartis Territory and grant Xencor the right to appoint sub-distributors, until such time as all Regulatory Approvals in the Novartis Territory have been transferred to Xencor or its designee but in no event longer than [...\*\*\*...] from the effective date of termination; provided, that such [...\*\*\*...] period shall be extended to the extent Xencor is using Commercially Reasonably Efforts to obtain the approvals necessary to [...\*\*\*...].

## 15.7.7. Manufacturing

15.7.7.1. With respect to any [...\*\*\*...], which, as of the effective date of termination is being Manufactured by Novartis or its Affiliates or a Third Party using Third Party Know-How, including cell lines and Manufacturing processes, Novartis will use Commercially Reasonable Efforts to [...\*\*\*...].

-135-

- 15.7.7.2. With respect to any [...\*\*\*...], which, as of the effective date of termination is being Manufactured by Novartis or its Affiliates (or to the extent permitted by the applicable sublicense, its Sublicensees) using Novartis Biological Material IP or Novartis Biological Material Inventions, then the provisions of Section 8.5.2 shall apply *mutatis mutandis* to the Manufacturing of such Reversion Product, with the license grant to any such Novartis [...\*\*\*...] required to Manufacture [...\*\*\*...] set forth in the Supply Agreement; provided [...\*\*\*...]; and (b) the Parties will negotiate in good faith to [...\*\*\*...].
- 15.8. Effect of Termination by Novartis for Cause. In the event that Novartis would have the right to terminate this Agreement with respect to a Licensed Target Pair or Fc Target by Novartis pursuant to Sections 15.3.1.2 or 15.4, then, [...\*\*\*...], (a) if such Licensed Target is an Optioned Target, the Optioned Target will automatically be converted into a Global Target, and Novartis will be deemed granted all rights relating thereto as a Global Target, with Novartis' obligations to [...\*\*\*...], (b) if such Licensed Target is a Global Target or an Fc Target, any future Milestone Payments and future royalties applicable to Annual Net Sales of Global Licensed Products or Fc Licensed Products [...\*\*\*...] will be applicable in accordance with the terms of this Agreement, and (c) Novartis shall have the right to offset the full amount of any Losses it has suffered as a result of Xencor's breach against any such future Milestone Payments or future royalties.
- 15.9. <u>Effect of Expiration or Termination, Survival</u>. The following provisions will survive expiration or termination of this Agreement for any reason: all of Articles 1 (to the extent necessary to give effect to other surviving provisions), 11 and 13 and Sections 6.1.7, 7.1.3, 7.3 and 7.4, , 9.4, 9.6,

10.9 and 10.10, but in each case of the foregoing Sections only with respect to amounts accrued prior to the effective date of expiration or termination, 14.1, 14.2, 15.1 (second s entence, only upon expiration), 15.6 (to the extent applicable), 15.7 or 15.8, as applicable, 15.9, 16.2, 16.3, 16.4 (other than 16.4.4), 16.5, 16.6, 16.7, 16.8, 16.9, 16.10, 16.11, 16.16 and 16.17. Expiration or termination of this Agreement for any reason will not relieve the Parties of any liability or obligation which accrued hereunder prior to the effective date of such termination or expiration, nor preclude either Party from pursuing all rights and remedies it may have hereunder or at law or in equity, with respect to any breach of this Agreement. For the avoidance of doubt, termination of this Agreement will not affect any SDEA, which will continue to survive so long as any Licensed Products thereunder are being Commercialized. For clarity, the termination of this Agreement with respect to a Licensed Target Pair includes the termination of this Agreement with respect to all Licensed Antibodies and Licensed Products that Specifically Bind such Licensed Target Pair.

### 16. MISCELLANEOUS

- **16.1.** Effect of a Xencor Change of Control. In the event of a Change of Control of Xencor, in addition to any other rights of Novartis set forth in this Agreement, each Party shall have the right to initiate the following process by providing [...\*\*\*...] prior written notice of its desire to do so to the other Party: (a) [...\*\*\*...] with respect to all Optioned Licensed Products and indications [...\*\*\*...]; and (b) any unexercised Opt-In Rights shall likewise terminate immediately.
- **16.2. Assignment**. Except as provided in this Section 16.2, this Agreement may not be assigned or otherwise transferred, nor may any right or obligation hereunder be assigned or transferred, by either Party without the written consent of the other Party. Notwithstanding the foregoing, either Party may, without the other Party's written consent, assign this Agreement and its rights and obligations hereunder in whole or in part to an Affiliate or in whole to a party that acquires, by or otherwise in connection with, merger, sale of assets or otherwise, all or substantially all of the business of the assigning Party to which the subject matter of this Agreement relates. The assigning Party will remain responsible for the performance by its assignee of any obligation hereunder so assigned. An assignment to an Affiliate will terminate, and all rights so assigned will revert to the assigning Party, if and when such Affiliate ceases to be an Affiliate of the assigning Party. Any purported assignment in violation of this Section 16.2 will be void.
- **16.3. Governing Law**. The Agreement will be construed and the respective rights of the Parties determined in accordance with the substantive Laws of the State of New York, notwithstanding any provisions of New York Law or any other Law governing conflicts of laws to the contrary.

164	Arbitration
104	Armiranon

-137-

- **16.4.1.** Disputes. Except as otherwise expressly set forth in this Agreement, including Section 2.5.3, disputes of any nature arising under, relating to, or in connection with this Agreement ("**Disputes**") will be resolved pursuant to this Section 16.4.
- **16.4.2.** Dispute Escalation. In the event of a Dispute between the Parties, the Parties will first attempt to resolve such Dispute by negotiation and consultation between themselves or at the JSC. In the event that such Dispute is not resolved on an informal basis within [...\*\*\*...] from receipt of the written notice of a Dispute, any Party may, by written notice to the other, have such Dispute referred to the Executive Officers, who will attempt to resolve such Dispute by negotiation and consultation for a [...\*\*\*...] period following receipt of such written notice.
- **16.4.3.** Full Arbitration. Except as otherwise expressly set forth in this Agreement, in the event the Parties have not resolved such Dispute within [...\*\*\*...] of receipt of the written notice referring such Dispute to the Executive Officers, either Party may at any time after such [...\*\*\*...] period submit such Dispute to be finally settled by arbitration administered in accordance with the procedural rules of the American Arbitration Association (the "AAA") in effect at the time of submission, as modified by this Section 16.4. The arbitration will be governed by the Laws of the State of New York. The arbitration will be heard and determined by three arbitrators who are retired judges or attorneys with at least 20 years of relevant experience in the pharmaceutical and biotechnology industry, each of whom will be impartial and independent and will not have worked for or on behalf of either Party for at least five years. Each Party will appoint one arbitrator and the third arbitrator will be selected by the two Party-appointed arbitrators, or, failing agreement within [...\*\*\*...] following appointment of the second arbitrator, by the AAA. Such arbitration will take place in New York, New York. The arbitration award so given will, absent manifest error, be a final and binding determination of the Dispute, will be fully enforceable in any court of competent jurisdiction, and will not include any damages expressly prohibited by Section 13.5. Novartis will pay the fees, costs and expenses for the arbitrator it chooses, Xencor will pay the fees, costs and expenses for the arbitrator it chooses, and the Parties will share payment for the third arbitrator. Except in a proceeding to enforce the results of the arbitration or as otherwise required by Law or securities exchange, neither Party nor any arbitrator may disclose the existence, content or results of any arbitration hereunder without the prior written consent of both Parties.

### **16.4.4.** Expedited Arbitration.

- **16.4.4.1.** If a Party exercises its rights under this Agreement to refer a dispute to expedited arbitration (an "**Expedited Dispute**"), then the Parties will follow the expedited dispute resolution process in this Section 16.4.4 (and not the dispute resolution process in Section 16.4.3 of this Agreement) ("**Expedited Arbitration**"). The Parties agree and acknowledge that any good faith dispute under Expedited Arbitration will not be deemed to be a material breach of this Agreement.
- **16.4.4.2.** The Expedited Dispute will be submitted to fast-track, binding arbitration in accordance with the following:

- (a) Arbitration will be conducted in New York, New York under the rules of the AAA for the resolution of commercial disputes in the most expedited manner permitted by such rules. The Parties will appoint a single arbitrator to be selected by mutual agreement. If the Parties are unable to agree on an arbitrator, the Parties will request that the AAA select the arbitrator. The arbitrator will be a professional in business or licensing experienced in the valuation of biopharmaceutical products with at least 10 years of experience in the pharmaceutical and life sciences industries, including the conduct of research, development and commercialization collaborations. The cost of the arbitration will be borne equally by the Parties. Except in a proceeding to enforce the results of the arbitration or as otherwise required by Laws, neither Novartis nor Xencor nor any arbitrator may disclose the existence, content or results of any arbitration hereunder without the prior written agreement of Novartis and Xencor.
- (b) Within [...\*\*\*...] after such matter is referred to arbitration, each Party will provide the arbitrator with a proposal and written memorandum in support of its position regarding the Expedited Dispute, as well as any documentary evidence it wishes to provide in support thereof (each a "**Brief**") and the arbitrator will provide each Party's Brief to the other Party after it receives it from both Parties.
- (c) Within [...\*\*\*...] after a Party submits its Brief, the other Party will have the right to respond thereto. The response and any material in support thereof will be provided to the arbitrator and the other Party.
- (d) The arbitrator will have the right to meet with the Parties as necessary to inform the arbitrator's determination and to perform independent research and analysis. Within [...\*\*\*...] of the receipt by the arbitrator of both Parties' responses (or expiration of the [...\*\*\*...] period if any Party fails to submit a response), the arbitrator will deliver his/her decision regarding the Expedited Dispute in writing, <u>provided</u> that the arbitrator will select one of the resolutions proposed by the Parties.
- **16.4.5.** Injunctive Relief. Notwithstanding the dispute resolution procedures set forth in this Section 16.4, in the event of an actual or threatened breach of this Agreement, the aggrieved Party may seek provisional equitable relief (including restraining orders, specific performance or other injunctive relief), without first submitting to any dispute resolution procedures hereunder.
- **16.4.6.** Tolling. The Parties agree that all applicable statutes of limitation and time-based defenses (such as estoppel and laches), as well as all time periods in which a Party must exercise rights or perform obligation hereunder, will be tolled once the dispute resolution procedures set forth in this Section 16.4 have been initiated and for so long as they are pending, and the Parties will cooperate in taking all actions reasonably necessary to achieve such a result. In addition, during the pendency of any Dispute under this Agreement initiated before the end of

any applicable cure period, including under Section 15.3, (a) this Agreement will remain in full force and effect, (b) the provisions of this Agreement relating to termination for material breach with respect to such Dispute will not be effective, (c) the time periods for cure under Section 15.3 as to any termination notice given prior to the initiation of arbitration will be tolled, (d) any time periods to exercise rights or perform obligations will be tolled, and (e) neither Party will issue a notice of termination pursuant to this Agreement based on the subject matter of the arbitration, until the arbitral tribunal has confirmed the material breach and the existence of the facts claimed by a Party to be the basis for the asserted material breach, provided that if such breach can be cured by (i) the payment of money, the defaulting Party will have an additional [...\*\*\*...] within its receipt of the arbitral tribunal's decision to pay such amount or (ii) the taking of specific remedial actions, the defaulting Party will have a reasonably necessary period to diligently undertake and complete such remedial actions within such reasonably necessary period or any specific timeframe established by such arbitral tribunal's decision before any such notice of termination can be issued. Further, with respect to any time periods that have run during the pendency of the Dispute, the applicable Party will have a reasonable period of time or any specific timeframe established by such arbitral tribunal's decision to exercise any rights or perform any obligations affected by the running of such time periods.

- 16.5. Entire Agreement, Amendments. This Agreement, together with the SDEA, contains the entire understanding of the Parties with respect to the subject matter hereof, and supersedes all previous arrangements with respect to the subject matter hereof, whether written or oral, including, effective as of the Effective Date, that Confidentiality Agreement between the Parties dated as of May 29, 2015 and that Non-Disclosure Agreement between the Parties dated as of July 24, 2015, as amended (provided that all information disclosed or exchanged under such agreement will be treated as Confidential Information hereunder). This Agreement may be amended, or any term hereof modified, only by a written instrument duly-executed by authorized representatives of both Parties hereto. The Exhibits and Schedules attached hereto may be amended, or any term hereof modified, only by a written instrument duly-executed by authorized representatives of both Parties hereto.
- 16.6. <u>Severability</u>. If any provision hereof should be held invalid, illegal or unenforceable in any respect in any jurisdiction, the Parties hereto will substitute, by mutual consent, valid provisions for such invalid, illegal or unenforceable provisions, which valid provisions in their economic effect are sufficiently similar to the invalid, illegal or unenforceable provisions that it can be reasonably assumed that the Parties would have entered into this Agreement with such valid provisions. In case such valid provisions cannot be agreed upon, the invalid, illegal or unenforceable of one or several provisions of this Agreement will not affect the validity of this Agreement as a whole, unless the invalid, illegal or unenforceable provisions are of such essential importance to this Agreement that it is to be reasonably assumed that the Parties would not have entered into this Agreement without the invalid, illegal or unenforceable provisions.
- 16.7. **Headings**. The captions to the Sections hereof are not a part of this Agreement, but are merely for convenience to assist in locating and reading the several Sections hereof.
- 16.8. **Waiver of Rule of Construction**. Each Party has had the opportunity to consult with counsel in connection with the review, drafting and negotiation of this Agreement. Accordingly, the

rule of construction that any ambiguity in this Agreement will be construed against the drafting Party will not apply.

- **Interpretation**. Except where the context expressly requires otherwise, (a) the use of any gender herein will be deemed to encompass references to either or both genders, and the use of the singular will be deemed to include the plural (and vice versa), (b) the words "include", "includes" and "including" will be deemed to be followed by the phrase "without limitation" and will not be interpreted to limit the provision to which it relates, (c) the word "shall" will be construed to have the same meaning and effect as the word "will, (d) any definition of or reference to any agreement, instrument or other document herein will be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein), (e) any reference herein to any Person will be construed to include the Person's successors and assigns, (f) the words "herein", "hereof" and "hereunder", and words of similar import, will be construed to refer to this Agreement in each of their entirety, as the context requires, and not to any particular provision hereof, (g) all references herein to Sections, Exhibits or Schedules will be construed to refer to Sections, Exhibits or Schedules of this Agreement, and references to this Agreement include all Exhibits and Schedules hereto. (h) the word "notice" means notice in writing (whether or not specifically stated) and will include notices, consents, approvals and other written communications contemplated under this Agreement, (i) provisions that require that a Party, the Parties or any committee hereunder "agree," "consent" or "approve" or the like will require that such agreement, consent or approval be specific and in writing, whether by written agreement, letter, approved minutes or otherwise (but excluding e-mail and instant messaging), (j) references to any specific law, rule or regulation, or article, section or other division thereof, will be deemed to include the then-current amendments thereto or any replacement or successor law, rule or regulation thereof, (k) the term "or" will be interpreted in the inclusive sense commonly associated with the term "and/or", and (l) the "x" found between two Targets shall be understood to mean "and" and identifies the two Targets that a given Bispecific Antibody Specifically Binds.
- 16.10. **No Implied Waivers, Rights Cumulative.** No failure on the part of Xencor or Novartis to exercise, and no delay in exercising, any right, power, remedy or privilege under this Agreement, or provided by statute or at Law or in equity or otherwise, will impair, prejudice or constitute a waiver of any such right, power, remedy or privilege or be construed as a waiver of any breach of this Agreement or as an acquiescence therein, nor will any single or partial exercise of any such right, power, remedy or privilege preclude any other or further exercise thereof or the exercise of any other right, power, remedy or privilege.
- 16.11. **Notices**. All notices which are required or permitted hereunder will be in writing and sufficient if delivered personally, sent by nationally-recognized overnight courier or sent by registered or certified mail, postage prepaid, return receipt requested, addressed as follows:

If to Xencor, to: Xencor, Inc.

111 West Lemon Avenue, 2<sup>nd</sup> floor Monrovia, California 91016 Attention: Chief Executive Officer With a copy to: Xencor, Inc.

111 West Lemon Avenue, 2<sup>nd</sup> floor Monrovia, California 91016

Attention: General Counsel

If to Novartis, to: Novartis Institutes for BioMedical Research, Inc.

250 Massachusetts Avenue Cambridge, MA 02139 Attention: General Counsel

or to such other address as the Party to whom notice is to be given may have furnished to the other Party in writing in accordance herewith. Any such notice will be deemed to have been given (a) when delivered if personally delivered on a Business Day (or if delivered or sent on a non-Business Day, then on the next Business Day), (b) on the Business Day of receipt if sent by overnight courier, or (c) on the Business Day of receipt if sent by mail.

- 16.12. **Compliance with Export Regulations**. Neither Party will export any technology licensed to it by the other Party under this Agreement except in compliance with U.S. export Laws.
- 16.13. **Force Majeure**. Neither Party will be held liable to the other Party nor be deemed to have defaulted under or breached this Agreement for failure or delay in performing any obligation under this Agreement to the extent that such failure or delay is caused by or results from causes beyond the reasonable control of the affected Party, potentially including embargoes, war, acts of war (whether war be declared or not), insurrections, riots, civil commotions, strikes, fire, earthquakes, floods, or other acts of God. The affected Party will notify the other Party of such force majeure circumstances as soon as reasonably practical, and will promptly undertake all reasonable efforts necessary to cure such force majeure circumstances and resume performance of its obligations hereunder and will keep the other Party reasonably informed regarding the status of such circumstances and any efforts related to the cure thereof, and the implications for the resumption of performance of such Party's obligations.
- 16.14. <u>Independent Parties</u>. It is expressly agreed that Xencor and Novartis will be independent contractors and that the relationship between Xencor and Novartis will not constitute a partnership, joint venture or agency. Xencor will not have the authority to make any statements, representations or commitments of any kind, or to take any action, which will be binding on Novartis, without the prior written consent of Novartis, and Novartis will not have the authority to make any statements, representations or commitments of any kind, or to take any action, which will be binding on Xencor, without the prior written consent of Xencor.
- 16.15. <u>Counterparts</u>. The Agreement may be executed in two counterparts, including by facsimile or PDF signature pages, each of which will be deemed an original, but both of which together will constitute one and the same instrument.

16.16. **Performance by Affiliates**. Each Party acknowledges and accepts that the other Party may exercise its rights and perform its obligations (including granting or continuing licenses and other rights) under this Agreement either directly or through one or more of its Affiliates. A Party's Affiliates

will have the benefit of all rights (including all licenses and other rights) of such Party under this Agreement, but not be subject to such Party's obligations, unless expressly provided herein or to the extent that an Affiliate is exercising a right hereunder, or in the case of a permitted assignment, in accordance with Section 16.2. Accordingly, in this Agreement "Novartis" will be interpreted to mean "Novartis or its Affiliates" and "Xencor" will be interpreted to mean "Xencor or its Affiliates" where necessary to give each Party's Affiliates the benefit of the rights provided to such Party in this Agreement and the ability to perform its obligations (including granting or continuing licenses and other rights) under this Agreement, provided however, that in any event each Party will remain responsible for the acts and omissions, including financial liabilities, of its Affiliates and such Party hereby expressly waives any requirement that the other Party exhaust any right, power or remedy, or proceed against such Affiliates, for any obligation or performance hereunder prior to proceeding directly against the Party associated with such Affiliate.

16.17. Binding Effect, No Third Party Beneficiaries. As of the Effective Date, this Agreement will be binding upon and inure to the benefit of the Parties and their respective permitted successors and permitted assigns. Except as expressly set forth in this Agreement, no Person other than the Parties and their respective Affiliates and permitted assignees hereunder will be deemed an intended beneficiary hereunder or have any right to enforce any obligation of this Agreement.

### THE REMAINDER OF THIS PAGE HAS BEEN LEFT INTENTIONALLY BLANK

-144-

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the Effective Date.

NOVARTIS INSTITUTES FOR BIOMEDICAL XENCOR, INC. RESEARCH, INC.

BY: <u>/s/ Christian Klee</u>
BY: <u>/s/ Bassil I. Dahiyat</u>

NAME: Christian Klee NAME: Bassil I. Dahiyat

TITLE: Vice President and Chief Financial Officer TITLE: President and Chief Executive Officer

Exhibit and Schedule List:

Exhibit A-1 Fc Patents

Exhibit A-2 Xencor Patents

Exhibit A-3 Xencor In-Licenses

Exhibit B Initial JSC Representatives

Exhibit C Research Plan

Exhibit D Form of Invoice

Exhibit E Example Royalty Calculation

Schedule 1.1.766 Excluded Target Pairs

Schedule 10.6.1.2 Opt-Out Milestone Payments and Royalties

# EXHIBIT A-1

[...\*\*\*...]

## EXHIBIT A-2

[...\*\*\*...]

### EXHIBIT A-3

### Xencor In-Licenses

- 1. [...\*\*\*...]
- 2. [...\*\*\*...]
- 3. [...\*\*\*...]
- 4. [...\*\*\*...]
- 5.

### EXHIBIT B

## Initial JSC Representatives

## Xencor:

- 1. [...\*\*\*...]
- 2. [...\*\*\*...]
- 3. [...\*\*\*...]

## Novartis:

- 1. [...\*\*\*...]
- 2. [...\*\*\*...]
- 3. [...\*\*\*...]

### **EXHIBIT C**

### Research Plan

[...\*\*\*...]

#### EXHIBIT D

### Form of Invoice

[...\*\*\*...]

## EXHIBIT E

## Royalty Calculation Examples

<u>Section 10.6.1</u>
Regional Licensed Products
[***]
Section 10.6.2
Global Licensed Products
[***]
Section 10.6.3
Optioned Licensed Product
[***]

 ${\bf ***} {\bf Confidential\ Treatment\ Requested}$ 

## **SCHEDULE 1.1.76**

## **Excluded Target Pairs**

[...\*\*\*...]

## SCHEDULE 10.6.1.2

## **Opt-Out Royalties**

Royalties Upon Exercise of Xencor's Opt-Out Right						
USD million unless other noted						
Exercise of Xencor Opt Out Right Prior to [***]			Exercise of Xencor Opt Out Right After [***]			
	Regional	Optioned		Regional	Optioned	
	Licensed	Licensed		Licensed	Licensed	
	Products	Products		Products	Products	
Royalties (U.S. annual Net Sales per Licensed Product)			Royalties (U.S. annual Net Sales per Licensed Product)			
Portion up to and including \$[***]	[***]%	[***]%	Portion up to and including \$[***]	[***]%	[***]%	
Portion greater than \$[***] up to and including \$[***]	[***]%	[***]%	Portion greater than \$[***] up to and including \$[***]	[***]%	[***]%	
Portion greater than \$[***]	[***]%	[***]%	Portion greater than \$[***]	[***]%	[***]%	

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

- I, Bassil I. Dahiyat, Ph.D., certify that:
  - 1. I have reviewed this Quarterly Report on Form 10-Q of Xencor, Inc.;
  - 2.Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
  - 3.Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report;
  - 4.The Company's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in exchange act rules 13a-15(f) and 15d-15(f)) for the Company and have:
    - a)Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company is made known to us particularly during the period in which this report is being prepared;
    - b)Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
    - c)Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
    - d)Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter (the Company's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting; and
  - 5.The Company's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):
    - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; and
    - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

/s/ BASSIL I. DAHIYAT
Bassil I. Dahiyat, Ph.D.
President & Chief Executive Officer

Date: August 2, 2016

# CERTIFICATION OF THE CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

- I, John J. Kuch, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Xencor, Inc.;
- 2.Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3.Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report;
- 4.The Company's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in exchange act rules 13a-15(f) and 15d-15(F) for the Company and have:
  - a)Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company is made known to us, particularly during the period in which this report is being prepared;
  - b)Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c)Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d)Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter (the Company's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5.The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

/s/ JOHN J. KUCH

John J. Kuch

Vice President, Finance (Principal Financial Officer)

Date: August 2, 2016

#### CERTIFICATION

Pursuant to the requirement set forth in Rule 13a-14(b) of the Securities Exchange Act of 1934, as amended, (the "Exchange Act") and Section 1350 of Chapter 63 of Title 18 of the United States Code (18 U.S.C. §1350), Bassil I. Dahiyat, Chief Executive Officer of Xencor, Inc. (the "Company"), and John J. Kuch, Vice President, Finance of the Company, each hereby certifies that, to the best of his or her knowledge:

- 1. The Company's Quarterly Report on Form 10-Q for the period ended June 30, 2016, to which this Certification is attached as Exhibit 32.1 (the "Periodic Report"), fully complies with the requirements of Section 13(a) or Section 15(d) of the Exchange Act; and
- 2. The information contained in the Periodic Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: August 2, 2016

IN WITNESS WHEREOF, the undersigned have set their hands hereto as of the day of August, 2016.

/s/ BASSIL I. DAHIYAT

Bassil I. Dahiyat
Chief Executive Officer

/s/ JOHN J. KUCH
John J. Kuch
Vice President, Finance

This certification accompanies the Form 10-Q to which it relates, is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of Xencor, Inc. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before or after the date of the Form 10-Q), irrespective of any general incorporation language contained in such filing.